Court File No. CV-14-506652-00CP

### ONTARIO SUPERIOR COURT OF JUSTICE

)

)

THE HONOURABLE JUSTICE BELOBABA Mondoy, THE 13 DAY OF AUGUST, 2018

BETWEEN:

- + 1

ans cour

## SHERIDAN CHEVROLET CADILLAC LTD., PICKERING AUTO MALL LTD., FADY SAMAHA, and JORDAN RAMSAY

Plaintiffs

- and -

## **TTEKT** CORPORATION, JTEKT AUTOMOTIVE NORTH AMERICA, INC., KOYO CORPORATION OF U.S.A., KOYO CANADA INC., NSK LTD., NSK AMERICAS, INC., NSK CANADA INC., TRW AUTOMOTIVE HOLDINGS CORP., TRW AUTOMOTIVE INC., TRW DEUTSCHLAND HOLDING GMBH, MITSUBISHI ELECTRIC CORPORATION, MITSUBISHI ELECTRIC AUTOMOTIVE AMERICA, INC., MITSUBISHI ELECTRIC SALES CANADA INC., MITSUBA CORPORATION, AMERICAN MITSUBA CORPORATION, SHOWA CORPORATION, AMERICAN SHOWA, INC. and SHOWA CANADA INC.

Defendants

Proceeding under the Class Proceedings Act, 1992, S.O. 1992, c. 6

ORDER - Electric Powered Steering Assemblies -- Mitsubishi Electric Notice Approval and Consent Certification -

THIS MOTION made by the Plaintiffs for an Order approving the abbreviated, publication, and long-form notices of settlement approval hearings and the method of dissemination of said notices, and certifying this proceeding as a class proceeding for settlement purposes as against Mitsubishi Electric Corporation, Mitsubishi Electric Automotive America, Inc., and Mitsubishi Sales Canada, Inc. (collectively, the "Settling Defendants") was read this day at Osgoode Hall, 130 Queen Street West, Toronto, Ontario.

ON READING the materials filed, including the settlement agreement with the Settling Defendants dated as of March 2, 2018 attached to this Order as Schedule "A" (the "Settlement Agreement"), and on reading the submissions of counsel for the Plaintiffs and Counsel for the Settling Defendants, the Non-Settling Defendants taking no position;

AND WHEREAS a parallel class proceeding relating to the pricing of Electric Powered Steering Assemblies was commenced under Court File No. CV-16-549385-00CP (the "Second Ontario Action") and is being case managed with the Proceeding;

AND ON BEING ADVISED that the Plaintiffs and the Settling Defendants consent to this Order;

- 1. **THIS COURT ORDERS** that for the purposes of this Order, except to the extent that they are modified in this Order, the definitions set out in the Settlement Agreement apply to and are incorporated into this Order.
- 2. THIS COURT ORDERS that the abbreviated, publication, and long-form notices of settlement approval hearing are hereby approved substantially in the forms attached respectively hereto as Schedules "B" to "D".
- 3. THIS COURT ORDERS that the plan of dissemination for the abbreviated, publication, and long-form notices of settlement approval hearing (the "Plan of Dissemination") is hereby approved in the form attached hereto as Schedule "E" and that the notices of settlement approval hearing shall be disseminated in accordance with the Plan of Dissemination.
- 4. **THIS COURT ORDERS** that the Proceeding is certified as a class proceeding as against the Settling Defendants for settlement purposes only.

× .

1

## 5. THIS COURT ORDERS that the "Settlement Class" is certified as follows:

í

1

All Persons in Canada who, during the Class Period, (a) purchased, directly or indirectly, Electric Powered Steering Assemblies; and/or (b) purchased or leased, directly or indirectly, a new or used Automotive Vehicle; and/or (c) purchased for import into Canada, a new or used Automotive Vehicle. Excluded Persons are excluded from the Settlement Class.

- 6. THIS COURT ORDERS that Sheridan Chevrolet Cadillac Ltd., Pickering Auto Mall Ltd., Fady Samaha, and Jordan Ramsay are appointed as the representative plaintiffs for the Settlement Class.
- 7. THIS COURT ORDERS that the following issue is common to the Settlement Class:

Did the Settling Defendants conspire to fix, raise, maintain, or stabilize the prices of Electric Powered Steering Assemblies in Canada and elsewhere during the Class Period? If so, what damages, if any, did Settlement Class Members suffer?

8. THIS COURT ORDERS that this Order, including but not limited to the certification of this Action as against the Settling Defendants for settlement purposes and the definition of Settlement Class and Common Issue, and any reasons given by the Court in connection with this Order, is without prejudice to the rights and defences of the Non-Settling Defendants in connection with the ongoing Proceeding or the Second Ontario Action and, without restricting the generality of the foregoing, may not be relied on by any Person to establish jurisdiction, the criteria for certification (including class definition) or the existence or elements of the causes of action asserted in the Proceeding or the Second Ontario Action, as against the Non-Settling Defendants. Nothing in this paragraph shall affect the efficacy of the opt-out process provided for in this Order.

- 9. THIS COURT ORDERS that putative members of the Settlement Class seeking to opt out of the Proceeding must opt out of the Proceeding and the Second Ontario Action on a collective basis. For greater certainty, putative members of the Settlement Class cannot opt out of the Proceeding, but not the Second Ontario Action.
- 10. **THIS COURT ORDERS** that putative members of the Settlement Class can opt out of the Proceeding by sending a written request to opt out to Class Counsel, postmarked on or before the Opt-Out Deadline. The written election to opt out must be signed by the Person or the Person's designee and must include the following information:
  - (a) the Person's full name, current address, and telephone number;
  - (b) if the Person seeking to opt out is a corporation, the name of the corporation and the position of the Person submitting the request to opt out on behalf of the corporation;
  - (c) a statement to the effect that the Person wishes to be excluded from the Proceeding and the Second Ontario Action; and
  - (d) the reason(s) for opting out.

ť

L

- 11. **THIS COURT ORDERS** that where the postmark is not visible or legible, the request to opt out shall be deemed to have been postmarked four (4) business days prior to the date that it is received by Class Counsel.
- 12. **THIS COURT ORDERS** that any putative member of the Settlement Class who validly opts out of the Proceeding shall have no further right to participate in the Proceeding or the

Second Ontario Action or to share in the distribution of any funds received as a result of a judgment or settlement in the Proceeding or the Second Ontario Action.

- 13. **THIS COURT ORDERS** that no further right to opt out of the Proceeding or Second Ontario Action will be provided.
- 14. THIS COURT ORDERS that, within thirty (30) days of the Opt-Out Deadline, Class Counsel shall provide to the Defendants in the Proceeding and Second Ontario Action a report containing the names of each Person who has validly and timely opted out of the Proceeding and a summary of the information delivered by such Persons pursuant to paragraph 10 above.

ENTERED AT / INSCRIT À TORONTO ON / BOOK NO: LE / DANS LE REGISTRE NO.:

. Real J.

The Honourable Justice Belobaba

AUG 2 0 2018

PER/PAR: 🚺

10.1

Schedule "A"

λ.

1

## CANADIAN ELECTRIC POWERED STEERING ASSEMBLIES CLASS ACTIONS NATIONAL SETTLEMENT AGREEMENT

Made as of March 2, 2018

Between

# SHERIDAN CHEVROLET CADILLAC LTD., THE PICKERING AUTO MALL LTD., FADY SAMAHA, and JORDAN RAMSAY

(the "Plaintiffs")

and

# MITSUBISHI ELECTRIC CORPORATION, MITSUBISHI ELECTRIC AUTOMOTIVE AMERICA, INC. and MITSUBISHI SALES CANADA, INC.

(the "Settling Defendants")

.

# CANADIAN ELECTRIC POWERED STEERING ASSEMBLIES CLASS ACTIONS NATIONAL SETTLEMENT AGREEMENT TABLE OF CONTENTS

#### RECITALS ......1 2.1 2.2 2.3 Motion Seeking Approval of the Settlement ......9 2.4 3.1 3.2 Intervention in the U.S. Litigation ......11 3.3 Extent of Cooperation ......11 4.1 4.2 5.1 Right of Termination......23 6.1 6.2 If Settlement Agreement is Terminated ......24

| 6.3       | Allocation of Settlement Amount Following Termination | 25 |
|-----------|---|----|
| 6.4       | Survival of Provisions After Termination              | 26 |
| SECTION 7 | - RELEASES AND DISMISSALS                             |    |
| 7.1       | Release of Releasees                                  | 26 |
| 7.2       | Release by Releasees                                  | 26 |
| 7.3       | Covenant Not to Sue                                   | 27 |
| 7.4       | No Further Claims                                     | 27 |
| 7.5       | Dismissal of the Proceeding                           | 27 |
| 7.6       | Dismissal of Other Actions                            | 27 |
| 7.7       | Material Term   | 28 |
| SECTION 8 | - BAR ORDER AND OTHER CLAIMS                          |    |

| 8.1   | Bar Order  |                      |  |  |
|---|--|----------------------|--|--|
| 8.2   | Claims Against Other Entities Reserved   | 31                   |  |  |
| 8.3   | Material Term  |                      |  |  |
| SECTION 9 – EFFECT OF SETTLEMENT  |  |                      |  |  |
| 9.1   | No Admission of Liability  |                      |  |  |
| 9.2   | Agreement Not Evidence   |                      |  |  |
| 9.3   | No Further Litigation  |                      |  |  |
| SECTION   | <b>10 – CERTIFICATION FOR SETTLEMENT ONLY</b>  |                      |  |  |
| SECTION   | 11 - NOTICE TO SETTLEMENT CLASS  |                      |  |  |
| 11.   | Notices Required   |                      |  |  |
| 11.   | Form and Distribution of Notices   |                      |  |  |
| SECTION   | 12 – ADMINISTRATION AND IMPLEMENTATION   |                      |  |  |
| 12.   | Mechanics of Administration  |                      |  |  |
| 12.   | Information and Assistance   |                      |  |  |
| SECTION   | 13 – DISTRIBUTION OF THE SETTLEMENT AMOUNT<br>INTEREST   |                      |  |  |
| 13.   | Distribution Protocol  | 35                   |  |  |
| 13.   | No Responsibility for Administration or Fees   |                      |  |  |
| SECTION 14 – CLASS COUNSEL FEES, DISBURSEMENTS AND ADMINISTRATION<br>EXPENSES |  |                      |  |  |
|   | EXPENSES   |                      |  |  |
| 14.   |  |                      |  |  |
| 14.<br>14.  | Responsibility for Fees, Disbursements and Taxes   |                      |  |  |
|   | Responsibility for Fees, Disbursements and Taxes<br>Responsibility for Costs of Notices and Translation  | 36<br>36             |  |  |
| 14.:<br>14.:  | Responsibility for Fees, Disbursements and Taxes<br>Responsibility for Costs of Notices and Translation  | 36<br>36<br>36       |  |  |
| 14.:<br>14.:  | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li>15 – MISCELLANEOUS</li> </ul>   | 36<br>36<br>36<br>37 |  |  |
| 14.:<br>14.:<br>SECTION   | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li>15 – MISCELLANEOUS</li> <li>Motions for Directions</li> </ul>   |                      |  |  |
| 14.<br>14.<br>SECTION<br>15.  | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li>15 – MISCELLANEOUS</li> <li>Motions for Directions</li> <li>Headings, etc.</li> </ul>   |                      |  |  |
| 14.<br>14.<br>SECTION<br>15.<br>15.   | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li>15 - MISCELLANEOUS</li> <li>Motions for Directions</li> <li>Headings, etc.</li> <li>Computation of Time</li> </ul>  |                      |  |  |
| 14.<br>14.<br>SECTION<br>15.<br>15.   | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li>15 – MISCELLANEOUS</li></ul>  |                      |  |  |
| 14.<br>14.<br>SECTION<br>15.<br>15.<br>15.                                    | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li>15 – MISCELLANEOUS</li></ul>  |                      |  |  |
| 14.<br>14.<br>SECTION<br>15.<br>15.<br>15.<br>15.                             | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li><b>15 – MISCELLANEOUS</b></li> <li>Motions for Directions</li> <li>Headings, etc</li> <li>Computation of Time</li> <li>Ongoing Jurisdiction</li> <li>Governing Law</li> <li>Entire Agreement</li> </ul>   |                      |  |  |
| 14.:<br>14.:<br>SECTION<br>15.:<br>15.:<br>15.:<br>15.:<br>15.:               | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li><b>15 – MISCELLANEOUS</b></li> <li>Motions for Directions</li> <li>Headings, etc</li> <li>Computation of Time</li> <li>Ongoing Jurisdiction</li> <li>Governing Law</li> <li>Entire Agreement</li> <li>Amendments</li> </ul>                         |                      |  |  |
| 14.<br>14.<br>SECTION<br>15.<br>15.<br>15.<br>15.<br>15.                      | <ul> <li>Responsibility for Fees, Disbursements and Taxes</li> <li>Responsibility for Costs of Notices and Translation</li> <li>Court Approval for Class Counsel Fees and Disbursements</li> <li><b>15 – MISCELLANEOUS</b></li> <li>Motions for Directions</li> <li>Headings, etc</li> <li>Computation of Time</li> <li>Ongoing Jurisdiction</li> <li>Governing Law</li> <li>Entire Agreement</li> <li>Amendments</li> <li>Binding Effect</li> </ul> |                      |  |  |

1 I

| 15.11 | Language              | 39 |
|-------|-----------------------|----|
| 15.12 | Transaction           | 39 |
| 15.13 | Recitals              | 39 |
| 15.14 | Schedules             | 40 |
| 15.15 | Acknowledgements      | 40 |
|       | Authorized Signatures |    |
|       | Notice                |    |
| 15.18 | Date of Execution     | 41 |

) L

## CANADIAN ELECTRIC POWERED STEERING ASSEMBLIES CLASS ACTIONS NATIONAL SETTLEMENT AGREEMENT

## RECITALS

ĩ

A. WHEREAS the Proceeding was commenced by the Plaintiffs in Ontario and the Plaintiffs claim class-wide damages allegedly caused as a result of the conduct alleged therein;

B. WHEREAS the Proceeding alleges that some or all of the Releasees participated in an unlawful conspiracy with other manufacturers of Electric Powered Steering Assemblies to rig bids for, and to raise, fix, maintain or stabilize the prices of Electric Powered Steering Assemblies sold in Canada and elsewhere as early as January 1, 2005 until at least October 1, 2012, contrary to Part VI of the *Competition Act*, RSC 1985, c C-34 and the common law;

C. WHEREAS the Settling Defendants and Releasees do not admit, through the execution of this Settlement Agreement or otherwise, any allegation of unlawful conduct alleged in the Proceeding, or in any Other Actions, and otherwise deny all liability and assert that they have complete defences in respect of the merits of the Proceeding and any Other Actions or otherwise;

D. WHEREAS the Plaintiffs, Class Counsel and the Settling Defendants agree that neither this Settlement Agreement nor any statement made in the negotiation thereof shall be deemed or construed to be an admission by or evidence against the Releasees or evidence of the truth of any of the Plaintiffs' allegations against the Releasees, which allegations are expressly denied by the Settling Defendants;

E. WHEREAS the Settling Defendants are entering into this Settlement Agreement in order to achieve a final and nation-wide resolution of all claims asserted or which could have been asserted against the Releasees by the Plaintiffs and the Settlement Class in the Proceeding and any Other Actions, and to avoid further expense, inconvenience and the distraction of burdensome and protracted litigation;

F. WHEREAS the Settling Defendants do not hereby attorn to the jurisdiction of the Court or any other court or tribunal in respect of any civil, criminal or administrative process except to the extent they have previously done so in the Proceeding and as is expressly provided in this Settlement Agreement with respect to the Proceeding; G. WHEREAS Counsel for the Settling Defendants and Class Counsel have engaged in arm'slength settlement discussions and negotiations, resulting in this Settlement Agreement relating to Canada;

H. WHEREAS as a result of these settlement discussions and negotiations, the Settling Defendants and the Plaintiffs have entered into this Settlement Agreement, which embodies all of the terms and conditions of the settlement between the Settling Defendants and the Plaintiffs, both individually and on behalf of the class they seek to represent, subject to approval of the Court;

I. WHEREAS Class Counsel, on their own behalf and on behalf of the Plaintiffs and the proposed Settlement Class, have reviewed and fully understand the terms of this Settlement Agreement and, based on their analyses of the facts and law applicable to the Plaintiffs' claims, having regard to the burdens and expense in prosecuting the Proceeding, including the risks and uncertainties associated with trials and appeals, and having regard to the value of the Settlement Agreement, have concluded that this Settlement Agreement is fair, reasonable and in the best interests of the Plaintiffs and the class they seek to represent;

J. WHEREAS the Parties therefore wish to and hereby finally resolve on a national basis, without admission of liability, the Proceeding and any Other Actions as against the Releasees;

K. WHEREAS the Parties consent to certification of the Proceeding as a class proceeding and to the Settlement Class and a Common Issue in respect of the Proceeding solely for the purposes of implementing this Settlement Agreement and contingent on approvals by the Court as provided for in this Settlement Agreement, on the express understanding that such certification shall not derogate from the respective rights of the Parties in the event that this Settlement Agreement is not approved, is terminated or otherwise fails to take effect for any reason; and

L. WHEREAS the Plaintiffs assert that they are adequate class representatives for the class they seek to represent and will seek to be appointed representative plaintiffs in the Proceeding;

NOW THEREFORE, in consideration of the covenants, agreements and releases set forth herein and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, it is agreed by the Parties that the Proceeding be settled and dismissed with prejudice as to the Settling Defendants only, without costs as to the Plaintiffs, the class they seek to represent, or the Settling Defendants, subject to the approval of the Court, on the following terms and conditions:

#### **SECTION 1 – DEFINITIONS**

For the purposes of this Settlement Agreement only, including the recitals and schedules hereto:

(1) *Administration Expenses* means all fees, disbursements, expenses, costs, taxes and any other amounts incurred or payable by the Plaintiffs, Class Counsel or otherwise for the approval, implementation and operation of this Settlement Agreement, including the costs of notices, but excluding Class Counsel Fees and Class Counsel Disbursements.

(2) *Automotive Vehicle* means all automobiles, passenger cars, sport utility vehicles, vans, trucks, buses and, without limitation, any other type of vehicle containing Electric Powered Steering Assemblies.

(3) *Certification Date* means the later of the date on which an order granting certification of the Proceeding against one or more Non-Settling Defendants is issued by the Court and the time to appeal such certification has expired without any appeal being taken, or if an appeal is taken the date of the final disposition of such appeal.

(4) *Claims Administrator* means the firm proposed by Class Counsel and appointed by the Court to administer the Settlement Amount in accordance with the provisions of this Settlement Agreement and the Distribution Protocol, and any employees of such firm.

(5) Class Counsel means Siskinds LLP and Sotos LLP.

(6) *Class Counsel Disbursements* include the disbursements and applicable taxes incurred by Class Counsel in the prosecution of the Proceeding or the Second Ontario Action, as well as any adverse costs awards issued against the Plaintiffs in the Proceeding or the Second Ontario Action.

(7) *Class Counsel Fees* means the fees of Class Counsel, and any applicable taxes or charges thereon, including any amounts payable as a result of the Settlement Agreement by Class Counsel or the Settlement Class Members to any other body or Person.

(8) *Class Period* means January 1, 2005 to the date of the order certifying the Proceeding as against the Settling Defendants for settlement purposes.

- 3 -

1

(9) *Common Issue* means: Did the Settling Defendants conspire to fix, raise, maintain, or stabilize the prices of Electric Powered Steering Assemblies in Canada and elsewhere during the Class Period? If so, what damages, if any, did Settlement Class Members suffer?

(10) Counsel for the Settling Defendants means Torys LLP.

(11) Court means the Ontario Superior Court of Justice.

(12) *Date of Execution* means the date on the cover page as of which the Parties have executed this Settlement Agreement.

(13) **Defendants** means the entities named as defendants in the Proceeding or the Second Ontario Action and any Persons added as defendants in the Proceeding or the Second Ontario Action in the future. For greater certainty, Defendants includes the Settling Defendants and the Settled Defendants.

(14) *Distribution Protocol* means the plan for distributing the Settlement Amount and accrued interest, in whole or in part, as approved by the Court.

(15) *Effective Date* means the date when a Final Order has been received from the Court approving this Settlement Agreement.

(16) *Electric Powered Steering Assemblies* means a device in an Automotive Vehicle that links the steering wheel to the tires, and includes the column, intermediate shaft, and electric power steering electric control unit, among other parts, but does not include the steering wheel or tires.

(17) *Excluded Person* means each Defendant, the directors and officers of each Defendant, the subsidiaries or affiliates of each Defendant, the entities in which each Defendant or any of that Defendant's subsidiaries or affiliates have a controlling interest and the legal representatives, heirs, successors and assigns of each of the foregoing, and those Persons who validly and timely opted out of the Proceeding in accordance with the order of the Court.

(18) *Final Order* means a final order, judgment or equivalent decree entered by the Court approving this Settlement Agreement in accordance with its terms, once the time to appeal such order has expired without any appeal being taken, if an appeal lies, or if the order is appealed, once there has been affirmation of the order upon a final disposition of all appeals.

t

(19) *Non-Settling Defendant* means any Defendant that is not a Settling Defendant or a Settled Defendant.

(20) *Opt-Out Deadline* means the date which is sixty (60) days after the date in the notice described in Section 11.1(1) is first published.

(21) *Other Actions* means actions or proceedings, excluding the Proceeding and Second Ontario Action, relating to Released Claims commenced by a Settlement Class Member either before or after the Effective Date.

(22) *Party and Parties* means the Settling Defendants, the Plaintiffs, and, where necessary, the Settlement Class Members.

(23) *Person* means an individual, corporation, partnership, limited partnership, limited liability company, association, joint stock company, estate, legal representative, trust, trustee, executor, beneficiary, unincorporated association, government or any political subdivision or agency thereof, and any other business or legal entity and their heirs, predecessors, successors, representatives, or assignees.

(24) *Plaintiffs* means Sheridan Chevrolet Cadillac Ltd., The Pickering Auto Mall Ltd., Fady Samaha and Jordan Ramsay.

(25) *Proceeding* means the action commenced in the Court bearing Court File No. CV-14-506652-00CP and includes any action subsequently consolidated into such action, including the Second Ontario Action.

(26) *Proportionate Liability* means the proportion of any judgment that, had the Settling Defendants not settled, the Court would have apportioned to the Releasees.

(27) **Released Claims** means any and all manner of claims, demands, actions, suits, causes of action, whether class, individual or otherwise in nature, whether personal or subrogated, damages of any kind (including compensatory, punitive or other damages) whenever incurred, liabilities of any nature whatsoever, including interest, costs, expenses, class administration expenses (including Administration Expenses), penalties, and lawyers' fees (including Class Counsel Fees and Class Counsel Disbursements), known or unknown, suspected or unsuspected, actual or

contingent, and liquidated or unliquidated, in law, under statute or in equity, that any of the Releasors ever had, now have or hereafter can, shall or may have, relating in any way to any conduct related to, arising from, or described in the Proceeding or the Second Ontario Action prior to the date hereof on account of, arising out of, resulting from, or related to in any respect the purchase, sale, pricing, discounting, manufacturing, marketing, offering, or distributing of Electric Powered Steering Assemblies or relating to any conduct alleged (or which was previously or could have been alleged) in the Proceeding or the Second Ontario Action including, without limitation, any such claims which have been asserted or could have been asserted, directly or indirectly, whether in Canada or elsewhere, in respect of the purchase, sale, pricing, discounting, manufacturing, marketing, offering or distributing of Electric Powered Steering Assemblies or the purchase or lease of new Automotive Vehicles, including, without limitation, any claims for consequential, subsequent or follow-on harm that arises after the date hereof in respect of any agreement, combination or conduct that occurred prior to the date hereof. However, the Released Claims do not include: (i) claims based on negligence, personal injury, bailment, failure to deliver lost goods, damaged or delayed goods, product defects, or breach of product warranty, or breach of contract claims or similar claims between the Parties that relate to Electric Powered Steering Assemblies but do not relate to alleged anti-competitive conduct; (ii) claims brought (whether before or after the Effective Date) outside of Canada relating to purchases of Electric Powered Steering Assemblies outside of Canada; (iii) claims brought (whether before or after the Effective Date) under laws other than those of Canada relating to purchases of Electric Powered Steering Assemblies outside of Canada; or (iv) claims concerning any automotive part other than Electric Powered Steering Assemblies, where such claims do not concern Electric Powered Steering Assemblies.

(28) **Releasees** means, jointly and severally, individually and collectively, the Settling Defendants and all of their present and former direct and indirect parents, owners, subsidiaries, divisions, affiliates, associates (as defined in the *Canada Business Corporations Act*, RSC 1985, c C-44), partners, joint ventures, franchisees, dealers, insurers, and all other Persons, partnerships or corporations with whom any of the former have been, or are now, affiliated, and all of their respective past, present and future officers, directors, employees, agents, mandataries, shareholders, attorneys, trustees, servants and representatives, members, managers and the predecessors, successors, purchasers, heirs, executors, administrators and assigns of each of the

foregoing, excluding always the Non-Settling Defendants and the Non-Settling Defendants' related entities.

(29) **Releasors** means, jointly and severally, individually and collectively, the Plaintiffs and the Settlement Class Members and all of their present and former direct and indirect parents, owners, subsidiaries, divisions, affiliates, associates (as defined in the *Canada Business Corporations Act*, RSC 1985, c C-44), partners, insurers, and all other Persons, partnerships or corporations with whom any of the former have been, or are now, affiliated, and all of their respective past, present and future officers, directors, employees, shareholders, attorneys, trustees, servants and representatives, members, managers and the predecessors, successors, purchasers, heirs, executors, administrators and assigns of each of the foregoing.

(30) *Second Ontario Action* means the action commenced in the Court bearing Court File No. CV-16-549385-00CP.

(31) Settled Defendants means any Defendant that executes its own settlement agreement in the Proceeding or the Second Ontario Action and whose settlement agreement becomes effective in accordance with its terms, whether or not such settlement agreement is in existence at the Date of Execution.

(32) Settlement Agreement means this agreement, including the recitals and schedules.

(33) Settlement Amount means CDN\$150,000.

(34) *Settlement Class* means all Persons in Canada who, during the Class Period, (i) purchased, directly or indirectly, Electric Powered Steering Assemblies; (ii) purchased or leased, directly or indirectly, a new or used Automotive Vehicle; and/or (iii) purchased for import into Canada, a new or used Automotive Vehicle. Excluded Persons are excluded from the Settlement Class.

(35) Settlement Class Member means a member of the Settlement Class.

(36) *Settling Defendants* means Mitsubishi Electric Corporation, Mitsubishi Electric Automotive America, Inc. and Mitsubishi Electric Sales Canada Inc.

(37) *Trust Account* means a guaranteed investment vehicle, liquid money market account or equivalent security with a rating equivalent to or better than that of a Canadian Schedule I bank (a

bank listed in Schedule I of the *Bank Act*, S.C. 1991, c. 46) held at a Canadian financial institution under the control of Siskinds LLP or the Claims Administrator, once appointed, for the benefit of the Settlement Class Members or the Settling Defendants, as provided for in this Settlement Agreement.

(38) U.S. Litigation means the consolidated class action proceedings, in which a plaintiff alleges that certain auto parts manufacturers rigged bids for, and/or conspired to raise, fix, maintain or stabilize the prices of Electric Powered Steering Assemblies, in which the Settling Defendants are named as parties, currently pending in the United States District Court for the Eastern District of Michigan, Southern Division, including the actions under the captions *In re Automotive Parts Antitrust Litigation*, Electronic Powered Steering Assemblies Cases, Case No. 2:13-cv-01901 (MOB), *In re Automotive Parts Antitrust Litigation*, Electronic Powered Steering Assemblies Cases, Case No. 2:13-cv-01902 (MOB), *In re Automotive Parts Antitrust Litigation*, Electronic Powered Steering Assemblies Cases, Case No. 2:13-cv-01903 (MOB), and includes all class actions transferred by the Judicial Panel for Multidistrict Litigation for coordination, all class proceedings alleging price-fixing of Electric Powered Steering Assemblies, to the extent that the Settling Defendants are named as parties.

(39) **U.S.** Settlement Agreements includes any settlement reached with the Settling Defendants in the U.S. Litigation.

### SECTION 2 – SETTLEMENT APPROVAL

#### 2.1 Best Efforts

(1) The Parties shall use their best efforts to implement this settlement and to secure the prompt, complete and final dismissal with prejudice of the Proceeding as against the Settling Defendants.

#### 2.2 Motion Seeking Approval of Notice and Certification

(1) The Plaintiffs shall file a motion before the Court, as soon as practicable after the Date of Execution, for an order approving the notices described in Section 11.1(1) and certifying the Proceeding as a class proceeding as against the Settling Defendants (for settlement purposes only).

(2) The order approving the notices described in Section 11.1(1) and certifying the Proceeding for settlement purposes shall be substantially in the form attached as Schedule A.

## 2.3 Motion Seeking Approval of the Settlement

,

(1) The Plaintiffs shall make best efforts to file a motion before the Court for an order approving this Settlement Agreement as soon as practicable after:

(a) the order referred to in Section 2.2(1) has been granted; and

(b) the notices described in Section 11.1(1) have been published.

(2) The order approving this Settlement Agreement shall be substantially in the form attached as Schedule B.

(3) This Settlement Agreement shall only become final on the Effective Date.

## 2.4 Pre-Motion Confidentiality

(1) Until the motion required by Section 2.2 is brought, the Parties shall keep all of the terms of the Settlement Agreement confidential and shall not disclose them without the prior consent of Counsel for the Settling Defendants and Class Counsel, as the case may be, except as required for the purposes of financial reporting, the preparation of financial records (including tax returns and financial statements), as necessary to give effect to its terms, or as otherwise required by law.

### **SECTION 3 – SETTLEMENT BENEFITS**

### 3.1 Payment of Settlement Amount

(1) Within thirty (30) days of the Date of Execution, the Settling Defendants shall pay the Settlement Amount to Siskinds LLP, for deposit into the Trust Account.

(2) Payment of the Settlement Amount shall be made by wire transfer. At least thirty (30) days prior to the Settlement Amount becoming due, Siskinds LLP will provide, in writing, the following information necessary to complete the wire transfers: name of bank, address of bank, ABA number, SWIFT number, name of beneficiary, beneficiary's bank account number, beneficiary's address, and bank contact details.

(3) The Settlement Amount and other consideration to be provided in accordance with the terms of this Settlement Agreement shall be provided in full satisfaction of the Released Claims against the Releasees.

(4) The Settlement Amount shall be all-inclusive of all amounts, including, without limitation, interest, costs, Class Counsel Fees and Class Counsel Disbursements.

(5) The Releasees shall have no obligation to pay any amount in addition to the Settlement Amount, for any reason, pursuant to or in furtherance of this Settlement Agreement, the Proceeding, the Second Ontario Action, or any Other Actions.

(6) Once a Claims Administrator has been appointed, Siskinds LLP shall transfer control of the Trust Account to the Claims Administrator.

(7) Siskinds LLP and the Claims Administrator shall maintain the Trust Account as provided for in this Settlement Agreement.

(8) Siskinds LLP and the Claims Administrator shall not pay out all or any part of the monies in the Trust Account, except in accordance with this Settlement Agreement, or in accordance with an order of the Court obtained after notice to the Parties.

#### **3.2** Taxes and Interest

(1) Except as hereinafter provided, all interest earned on the Settlement Amount in the Trust Account shall accrue to the benefit of the Settlement Class and shall become and remain part of the Trust Account.

(2) Subject to Section 3.2(3), all taxes payable on any interest which accrues on the Settlement Amount in the Trust Account or otherwise in relation to the Settlement Amount shall be paid from the Trust Account. Siskinds LLP or the Claims Administrator, as appropriate, shall be solely responsible to fulfill all tax reporting and payment requirements arising from the Settlement Amount in the Trust Account, including any obligation to report taxable income and make tax payments. All taxes (including interest and penalties) due with respect to the income earned by the Settlement Amount shall be paid from the Trust Account. (3) The Settling Defendants shall have no responsibility to make any filings relating to the Trust Account and will have no responsibility to pay tax on any income earned on the Settlement Amount or pay any taxes on the monies in the Trust Account, unless this Settlement Agreement is not approved, is terminated, or otherwise fails to take effect for any reason, in which case the interest earned on the Settlement Amount in the Trust Account or otherwise shall be paid to the Settling Defendants who, in such case, shall be responsible for the payment of all taxes on such interest not previously paid by Siskinds LLP or the Claims Administrator.

## 3.3 Intervention in the U.S. Litigation

Т

(1) The Settling Defendants and other Releasees shall not oppose any application that may be brought by or on behalf of the Plaintiffs to intervene in the U.S. Litigation in order to gain access to discovery documents and other documents and information subject to a protective order that are relevant to the Proceeding and is not otherwise inconsistent with the terms of this Settlement Agreement, including Section 4.1(14). However it is understood and agreed that neither the Settling Defendants nor the other Releasees have any obligation to bring or otherwise participate in such an application.

#### **SECTION 4 – COOPERATION**

### 4.1 Extent of Cooperation

(1) Prior to December 31, 2017, Counsel for the Settling Defendants met with Class Counsel in Canada or the United States, to provide an oral evidentiary proffer which included information originating with the Settling Defendants that was not covered by privilege relating to the allegations in the Proceeding and the Second Ontario Action. Notwithstanding any other provision of this Settlement Agreement, and for greater certainty, it is agreed that all statements made and information provided by Counsel for the Settling Defendants are privileged, will be kept strictly confidential, may not be directly or indirectly disclosed to any other Person, unless disclosure is ordered by the Court. Further, absent a Court order, Class Counsel will not attribute any factual information obtained from the proffer to the Settling Defendants and/or Counsel for the Settling Defendants. Notwithstanding the foregoing, Class Counsel may: (i) use information obtained from the proffer in the prosecution of the Proceeding and the Second Ontario Action, including for the purpose of developing an allocation plan relating to any settlement or judgment proceeds, except the prosecution of any claims against Releasees: and (ii) may rely on such information to certify that, to the best of Class Counsel's knowledge, information and belief, such information has evidentiary support or will likely have evidentiary support after reasonable opportunity for further investigation or discovery, but, absent a Court Order, the Plaintiffs shall not introduce any information from a proffer into the record or subpoena any Counsel for the Settling Defendants related to a proffer.

(2) After the Effective Date, at the request of Class Counsel, the Settling Defendants shall provide to Class Counsel, within ninety (90) days of receiving the request, or at a time mutually agreed upon by the Parties, the following:

- (a) transactional sales data produced by the Settling Defendants pursuant to the U.S.
   Settlement Agreements. The transactional sales data shall be delivered in Microsoft
   Excel or such other format as may be agreed upon by Counsel for the Settling
   Defendants and Class Counsel, and shall be delivered as a separate production from
   the documents to be delivered pursuant to Sections 4.1(2)(c)-(e) or identified by
   bates number as part of the production of documents to be delivered pursuant to
- (b) reasonable assistance to Class Counsel in understanding the transactional sales data produced by the Settling Defendants, through Counsel for the Settling Defendants, including a reasonable number of written and/or telephonic communications with Class Counsel and/or the Plaintiffs' experts and between technical personnel;
- (c) electronic copies of any documents (as defined in Rule 30.01 of the Ontario *Rules* of *Civil Procedure*) regarding Electric Powered Steering Assemblies produced by the Settling Defendants in the U.S. Litigation, including any documents produced or made available by the Settling Defendants pursuant to the U.S. Settlement Agreements, including at any future date, and any pre-existing translations of those documents and any pre-existing and non-privileged electronic coding. In addition, where the documents previously produced in the U.S. Litigation contain bates stamps on their face, a field will be produced containing the corresponding bates stamps of the first page of each document;

- (d) electronic copies of transcripts and video recordings of all depositions or other testimony of current or former employees, officers or directors of the Releasees, including all exhibits thereto and any pre-existing translations of those transcripts, taken in the U.S. Litigation, within ten (10) business days of said transcripts becoming available, but only to the extent permitted under the U.S. Protective Order (as defined in Section 4.2(4) below); and
- (e) electronic copies of any documents (as defined in Rule 30.01 of the Ontario Rules of Civil Procedure), including any pre-existing translations of those documents, produced by the Settling Defendants to the Canadian Competition Bureau and the United States Department of Justice, to the extent such productions are not duplicative.

(3) After the Certification Date, at the request of Class Counsel acting in good faith and as reasonably necessary for the ongoing prosecution of the Proceeding or the Second Ontario Action, the Settling Defendants shall make reasonable efforts to provide to Class Counsel, within ninety (90) days of receiving the request, or at a time mutually agreed upon by the Parties, the following:

(a) transactional cost data for the global production and sale of Electric Powered Steering Assemblies between January 1, 1998 and, at least, December 31, 2013, to the extent such data relates to Electric Powered Steering Assemblies known or expected to be included in Automotive Vehicles that were sold in Canada. Class Counsel may request, and the Settling Defendants shall make reasonable efforts to provide, additional transactional cost data for the period between January 1, 2014 and the date of the order certifying the Proceeding for settlement purposes, if Class Counsel determine that such information is reasonably necessary for the ongoing prosecution of the Proceeding and/or the Second Ontario Action. The transactional cost data shall be delivered in Microsoft Excel or such other format as may be agreed upon by Counsel for the Settling Defendants and Class Counsel, and shall be delivered as a separate production from the documents to be delivered pursuant to Section 4.1(3)(d) or identified by bates number as part of the production of documents to be delivered pursuant to Section 4.1(3);

- sales data relating to the Settling Defendants' global sales of Electric Powered Steering Assemblies, to the extent such data relates to Electric Powered Steering Assemblies known or expected to be included in Automotive Vehicles that were sold in Canada. The transactional sales data shall be delivered in Microsoft Excel or such other format as may be agreed upon by Counsel for the Settling Defendants and Class Counsel;
- (c) reasonable assistance to Class Counsel in understanding the transactional cost and/or sales data produced by the Settling Defendants, through Counsel for the Settling Defendants, including a reasonable number of written and/or telephonic communications with Class Counsel and/or the Plaintiffs' experts and between technical personnel; and
- (d) to the extent not already produced, any documents (as defined in Rule 30.01 of the Ontario *Rules of Civil Procedure*), including any pre-existing translations of those documents, that are relevant to the allegations in the Proceeding or the Second Ontario Action.

(4) The obligation to produce documents pursuant to Sections 4.1(2) and (3) shall be a continuing obligation to the extent additional documents are identified by the Settling Defendants following the initial productions pursuant to this Settlement Agreement or are produced in the U.S. Litigation.

(5) The Settling Defendants shall not object to the Plaintiffs' participation in any evidentiary proffers and/or interviews of the Settling Defendants' representatives that occur in the U.S. Litigation pursuant to the U.S. Settlement Agreements. The Settling Defendants shall, where possible, provide notice to Class Counsel thirty (30) days before any evidentiary proffer or interview of representatives of the Settling Defendants that occurs in the U.S. Litigation pursuant to the U.S. Settlement Agreements.

(6) It is understood that the evidentiary proffer described in Section 4.1(1) and the evidentiary proffers and/or interviews of witnesses described in Section 4.1(5) might take place before the Effective Date. In such event:

(b)

- (a) any documents or information provided in the course of those evidentiary proffers and/or interviews shall be subject to the terms and protections of this Settlement Agreement; and
- (b) in the event that this Settlement Agreement is not approved, is terminated, or otherwise fails to take effect for any reason, the documents and information provided during the evidentiary proffers and/or interviews shall not be used by the Plaintiffs or Class Counsel, whether directly or indirectly, in any way for any reason, including, without limitation, against the Settling Defendants as an admission or evidence of any violation of any statute or law, or of any liability or wrongdoing by the Settling Defendants or of the truth of any claims or allegations in the Proceeding or the Second Ontario Action, and such information shall not be discoverable by any Person or treated as evidence of any kind, unless otherwise ordered by a Court. In order to give effect to this agreement, Class Counsel agrees to return all copies of any documents received during, and destroy all copies of any notes taken during (or subsequent reports provided about), these evidentiary proffers and/or interviews and to provide written confirmation to the Settling Defendants of having done so.

(7) In the event that any employee interviews provided for in the U.S. Settlement Agreements: (i) have already occurred; (ii) occur within six (6) months of the Effective Date, but Class Counsel is unable to attend or is not given reasonable advance notice of the time and location of the interviews; or (iii) do not occur within six (6) months of the Effective Date, the Settling Defendants shall, at the request of Class Counsel, upon at least sixty (60) days' notice, and subject to any legal restrictions, make reasonable efforts to make available at a mutually convenient time, up to five (5) current or former officers, directors or employees of the Settling Defendants who have knowledge about the allegations in the Proceeding and the Second Ontario Action to provide information regarding the allegations raised in the Proceeding and the Second Ontario Action in a personal interview with Class Counsel and/or experts retained by Class Counsel. Class Counsel shall use reasonable efforts to limit this requirement to a single current employee. Such personal interviews shall take place in a North American location and shall not exceed one (1) business day or seven (7) hours for each individual. Costs incurred by, and the expenses of, the employees of the Settling Defendants in relation to such interviews and costs of an interpreter or otherwise related to foreign language translation in connection with interviews shall be the responsibility of the Settling Defendants. If an employee refuses to provide information, or otherwise cooperate, the Settling Defendants shall use all reasonable efforts to make him/her available for an interview with Class Counsel and/or experts retained by Class Counsel. The failure of a current or former employee to agree to make him or herself available, or to otherwise cooperate with the Plaintiffs,

shall not constitute a violation of this Settlement Agreement.

Subject to the rules of evidence, any court order with respect to confidentiality and the (8) other provisions of this Settlement Agreement, the Settling Defendants agree to use reasonable efforts to produce at trial or otherwise in the Proceeding or Second Ontario Action (including through affidavit evidence): (i) a current representative qualified to establish for admission into evidence the Settling Defendants' transactional sales data and cost information provided pursuant to Sections 4.1(2)(a) and (3)(a); (ii) a representative qualified to establish for admission into evidence any of the Settling Defendants' documents provided as cooperation pursuant to Section 4.1(2) and (3) of this Settlement Agreement that is reasonable and necessary for the prosecution of the Proceeding and the Second Ontario Action (with Class Counsel using its best efforts to authenticate documents for use at trial or otherwise without use of a live witness); and (iii) representatives qualified to establish for admission into evidence information provided in cooperation pursuant to Section 4 of this Settlement Agreement, provided that Class Counsel shall use all reasonable efforts to limit this requirement to a single witness. To the extent reasonably possible, a single witness will be used both to authenticate documents and provide the information at trial or otherwise contemplated by this Section. The failure of a specific officer, director or employee to agree to make him or herself available, or to otherwise cooperate with the Plaintiffs, shall not constitute a violation of this Settlement Agreement. The Settling Defendants shall be responsible for all associated costs incurred by such representatives in connection with fulfilling the Settling Defendants' obligations under this Section.

(9) It is understood that the Settling Defendants are likely unable to make available for interviews, examinations or trial testimony, or any other court proceedings, the five (5) individuals referenced in Attachment A (filed under seal) of the plea agreement between Mitsubishi Electric Corporation and the United States of America (Case No. 2:13-cr-20710-PJK, E.D. Mich., ECF No. 9) who were not granted immunity from prosecution by the United States under that plea agreement, and may be unable to make available for interviews, examinations or trial testimony,

or any other court proceedings any individual who is no longer an officer, director, or employee of the Settling Defendants, and that this shall not constitute a violation of this Settlement Agreement.

(10) Nothing in this Settlement Agreement shall be construed to require the Settling Defendants to perform any act, including the transmittal or disclosure of any information, which would violate the law of this or any jurisdiction.

(11) Nothing in this Settlement Agreement shall require, or shall be construed to require, the Settling Defendants or any representative or employee of the Settling Defendants to disclose or produce any documents or information prepared by or for Counsel for the Settling Defendants, or that is not within the possession, custody or control of the Settling Defendants, or to disclose or produce any documents or information in breach of any order, regulatory directive, rule or law of this or any jurisdiction, or subject to solicitor-client privilege, litigation privilege, joint defence privilege or any other privilege, doctrine, or law, or to disclose or produce any information or documents they obtained on a privileged or co-operative basis from any party to any action or proceeding who is not a Releasee. The Settling Defendants are not required to create a privilege log. However, if a relevant privilege log was created in the context of the U.S. Litigation, Counsel for the Settling Defendants have created a relevant privilege log, or there is some other pre-existing document containing identifying information regarding the withheld documents, the Settling Defendants will provide Class Counsel with a copy of such log or document.

(12) If any documents protected by any privilege and/or any privacy law or other rule or law of this or any applicable jurisdiction are accidentally or inadvertently disclosed or produced, such documents shall be promptly returned to the Settling Defendants and the documents and the information contained therein shall not be disclosed or used directly or indirectly, except with the express written permission of the Settling Defendants, and the production of such documents shall in no way be construed to have waived in any manner any privilege, doctrine, law, or protection attached to such documents.

(13) The Settling Defendants' obligations to cooperate as particularized in this Section shall not be affected by the release provisions contained in Section 7 of this Settlement Agreement. Unless this Settlement Agreement is not approved, is terminated or otherwise fails to take effect for any reason, the Settling Defendants' obligations to cooperate shall cease at the date of final judgment in the Proceeding and the Second Ontario Action against all Defendants.

(14) Subject to Sections 4.1(15) and (16), the provisions set forth in this Section 4.1 are the exclusive means by which the Plaintiffs, Class Counsel and Settlement Class Members may obtain discovery or information or documents from the Releasees. The Plaintiffs, Class Counsel and Settlement Class Members agree that they shall not pursue any other means of discovery against, or seek to compel the evidence of, the Releasees, whether in Canada or elsewhere and whether under the rules or laws of this or any other Canadian or foreign jurisdiction.

(15) The Plaintiffs may exercise any rights they have to seek to obtain discovery in the Proceeding and the Second Ontario Action as against an officer, director and/or employee of the Settling Defendants put forward to participate in employee interviews, or provide testimony at trial or otherwise pursuant to Sections 4.1(7) or (8), if the current or former officer, director or employee of the Settling Defendants fails to cooperate in accordance with those Sections and the provisions of this Settlement Agreement, and/or any individuals who the Settling Defendants are unable to make available for interviews, examinations or trial testimony, or any other court proceedings, pursuant to Section 4.1(9).

(16) In the event that the Settling Defendants materially breach this Section 4.1, the Plaintiffs may move before the Court to enforce the terms of this Settlement Agreement, seek an order setting aside Section 4.1(14) and allowing the Plaintiffs to obtain discovery or information from the Settling Defendants as if the Settling Defendants remained parties to the Proceeding, or seek such other remedy that is available at law.

(17) A material factor influencing the Settling Defendants' decision to execute this Settlement Agreement is their desire to limit the burden and expense of this litigation. Accordingly, Class Counsel agree to exercise good faith in seeking cooperation from the Settling Defendants, agree not to seek information that is unnecessary, cumulative or duplicative and agree otherwise to avoid imposing undue or unreasonable burdens or expense on the Settling Defendants.

(18) The scope of the Settling Defendants' cooperation under this Settlement Agreement shall be limited to the allegations asserted in the Proceeding and the Second Ontario Action as presently filed.

(19) The Settling Defendants make no representation regarding and shall bear no liability with respect to the accuracy of any of the documents or information described in this Section 4.1, or that they have, can or will produce a complete set of any of the documents or information described in this Section 4.1, and the failure to do so shall not constitute a breach or violation of this Settlement Agreement.

#### 4.2 Limits on Use of Documents and Information

It is understood and agreed that all documents and information made available or provided (1)by the Settling Defendants and/or Counsel for the Settling Defendants to the Plaintiffs and Class Counsel under this Settlement Agreement shall be used only in connection with the prosecution of the claims in the Proceeding and the Second Ontario Action, and shall not be used directly or indirectly for any other purpose, except to the extent that the documents or information are or become publicly available. The Plaintiffs and Class Counsel agree they will not disclose the documents and information provided by the Settling Defendants and/or Counsel for the Settling Defendants beyond what is reasonably necessary for the prosecution of the Proceeding and the Second Ontario Action or as otherwise required by law, except to the extent that the documents or information are or become publicly available. Subject to the foregoing, Class Counsel shall take reasonable precautions to ensure and maintain the confidentiality of such documents and information, and of any work product of Class Counsel that discloses such documents and information. Notwithstanding the foregoing, Class Counsel may disclose such information and documents to Camp Fiorante Matthews Mogerman LLP, to the extent that Camp Fiorante Matthews Mogerman LLP is assisting Class Counsel in the prosecution of the Proceeding and the Second Ontario Action and they agree to keep such information confidential and only use it for the purpose of providing such assistance.

(2) If the Plaintiffs intend to produce or file in the Proceeding or the Second Ontario Action any documents or other information provided by the Settling Defendants and/or Counsel for the Settling Defendants as cooperation under the Settlement Agreement (and such disclosure is not otherwise prohibited by the Settlement Agreement) which, at the time of being provided, were marked or designated by the Settling Defendants as "Confidential — Subject to Procedure Under Section 4.2(2) of the Settlement Agreement," and there is not already a confidentiality order issued in the Proceeding or the Second Ontario Action that applies to the documents and information provided as cooperation by the Settling Defendants, Class Counsel shall provide the Settling Defendants with an advance description of the documents or other information sought to be produced or filed at least thirty (30) days in advance of the proposed production or filing, in order that the Settling Defendants may move to obtain a sealing or confidentiality order or similar relief. If, within that thirty (30) day period, a Settling Defendant so moves, the Plaintiffs and Class Counsel shall not produce or file the confidential information or documents until the earlier of: (i) one-hundred and ninety (190) days after being advised of such intention; or (ii) the Settling Defendants' motion has been decided and all applicable appeal periods have expired. Notwithstanding the foregoing, the Plaintiffs and Class Counsel may, so as not to delay prosecution of the Proceeding or the Second Ontario Action, (i) file or seek to file such documents or information with the Court in sealed envelopes or other appropriate containers, segregated from the public record, endorsed with the title of the action and the following statement: "This envelope/box/container containing documents which are filed by [name of Party] and subject to a pending confidentiality motion is not to be opened nor the contents thereof to be displayed or revealed to any non-Court personnel except by order of the Court" and such records shall not form part of the public record in the Proceeding or the Second Ontario Action except upon order of the Court or by agreement of all Parties and/or the Settling Defendant whose confidential information is contained therein; and (ii) provide on an interim basis, documents or information to counsel for the Non-Settling Defendants provided that counsel for the Non-Settling Defendants agree and give assurances that the documents or information will remain with external counsel and will only be disclosed to independent expert(s) retained by a Party for the purposes of the Proceeding or the Second Ontario Action, as well as secretarial, clerical or other support personnel of such expert(s) to whom disclosure is reasonably necessary. An independent expert may not be an employee of a Plaintiff or Defendant in the Proceeding and the Second Ontario Action, or a competitor of the Settling Defendants. In addition, if a Settling Defendant intervenes for this purpose, the Plaintiffs and Class Counsel shall not oppose a motion to intervene made by the Settling Defendants for this purpose, provided that the form and content of the requested order is similar in substance to the order issued by the Ontario Court in Ontario Superior Court of Justice Court File No. CV-12-44673700CP, dated July 15, 2015.

(3) In the event that a Person requests disclosure of documents or information provided by the Settling Defendants and/or Counsel for the Settling Defendants as cooperation under this Settlement Agreement which, at the time of being provided, were marked or designated by the

Settling Defendants as "Confidential - Subject to Procedure Under Section 4.2(2) of the Settlement Agreement", whether or not the Person applies for an order requiring the Plaintiffs to disclose or produce any documents or other information, and there is not already a confidentiality order issued in the Proceeding or the Second Ontario Action that applies to the documents and information provided as cooperation by the Settling Defendants, Class Counsel shall provide notice to the Settling Defendants promptly upon becoming aware of it in order that the Settling Defendants may intervene to oppose such disclosure or production. In no circumstances shall the Plaintiffs or Class Counsel apply for or consent to such an application for disclosure or production. The Plaintiffs and Class Counsel shall not disclose the confidential information or documents until the Settling Defendants' motion has been decided and all applicable appeal periods have expired, except: (i) to the extent such information or documents are or become otherwise publicly available; (ii) as ordered to do so by a Court; and (iii) in the event that the Person making the request is a Non-Settling Defendant, so as not to delay prosecution of the Proceeding or the Second Ontario Action, Class Counsel may provide, on an interim basis, documents or information to counsel for the Non-Settling Defendant provided that counsel for the Non-Settling Defendant agree and give assurances that the documents or information will remain with external counsel and will only be disclosed to independent expert(s) retained by a Party for the purposes of the Proceeding or the Second Ontario Action, as well as secretarial, clerical or other support personnel of such expert(s) to whom disclosure is reasonably necessary. An independent expert may not be an employee of a Plaintiff or Defendant in the Proceeding and the Second Ontario Action or a competitor of the Settling Defendants.

(4) In addition, until such a time as a confidentiality order is in place in the Proceeding or the Second Ontario Action that applies to the documents and information provided as cooperation by the Settling Defendants, Class Counsel shall treat any documents received from the Settling Defendants and designated as Confidential or Highly Confidential in accordance with the provisions of the "Stipulation and Protective Order Governing the Production and Exchange of Confidential Information" issued in the U.S. Litigation on July 10, 2012 (the "U.S. Protective Order"). Once a confidentiality or protective order(s) is issued in the Proceeding or the Second Ontario Action, that order(s) shall govern any documents and information received from the Settling Defendants.

### **SECTION 5 – OPTING OUT**

## 5.1 Procedure

(1) Class Counsel will seek approvals from the Court of the following opt out process as part of the order certifying the Proceeding as a class proceeding for settlement purposes:

- Putative members of the Settlement Class seeking to opt out of the Proceeding must opt out of the Second Ontario Action on a collective basis. For greater certainty, putative members of the Settlement Class cannot opt out of the Proceeding, but not the Second Ontario Action;
- (b) Persons seeking to opt out of the Proceeding must do so by sending a written request to opt out to Class Counsel at an address to be identified in the notice described in Section 11.1(1), postmarked on or before the Opt-Out Deadline. The written election to opt out must be signed by the Person or the Person's designee and must include the following information:
  - (i) the Person's full name, current address and telephone number;
  - (ii) if the Person seeking to opt out is a corporation, the name of the corporation and the position of the Person submitting the request to opt out on behalf of the corporation;
  - (iii) a statement to the effect that the Person wishes to be excluded from the Proceeding and the Second Ontario Action; and
  - (iv) the reason(s) for opting out.
- (c) Where the postmark is not visible or legible, the request to opt out shall be deemed to have been postmarked four (4) business days prior to the date that it is received by Class Counsel.
- (d) Persons who opt out of the Proceeding shall have no further right to participate in the Proceeding or the Second Ontario Action or to share in the distribution of any funds received as a result of a judgment or settlement in the Proceeding or the Second Ontario Action.

t

- (e) No further right to opt out of the Proceeding or Second Ontario Action will be provided.
- (f) Within thirty (30) days of the Opt-Out Deadline, Class Counsel shall provide to the Defendants in the Proceeding and Second Ontario Action a report containing the names of each Person who has validly and timely opted out of the Proceeding and a summary of the information delivered by such Persons pursuant to Section 5.1(1)(b) above.

(2) The Settling Defendants reserve all of their legal rights and defences with respect to any potential Settlement Class Member who validly opts out from the Proceeding and the Second Ontario Action. The Plaintiffs through their respective Class Counsel expressly waive their right to opt out of the Proceeding and the Second Ontario Action.

### SECTION 6 – TERMINATION OF SETTLEMENT AGREEMENT

### 6.1 Right of Termination

(1) In the event that:

t

- (a) the Court declines to certify the Proceeding for the purposes of the Settlement
   Agreement;
- (b) the Court declines to dismiss the Proceeding against the Settling Defendants;
- (c) the Court declines to approve this Settlement Agreement or any material part hereof;
- (d) the Court approves this Settlement Agreement in a materially modified form;
- (e) the Court issues a settlement approval order that is materially inconsistent with the terms of the Settlement Agreement or not substantially in the form attached to this Settlement Agreement as Schedule B; or
- (f) any order approving this Settlement Agreement made by the Court does not become a Final Order;

the Plaintiffs and the Settling Defendants shall each have the right to terminate this Settlement Agreement by delivering a written notice pursuant to Section 15.17, within thirty (30) days following an event described above.

(2) In addition, if the Settlement Amount is not paid in accordance with Section 3.1(1), the Plaintiffs shall have the right to terminate this Settlement Agreement by delivering a written notice pursuant to Section 15.17.

(3) Except as provided for in Section 6.4, if the Settlement Agreement is terminated, the Settlement Agreement shall be null and void and have no further force or effect, and shall not be binding on the Parties, and shall not be used as evidence or otherwise in any litigation or in any other way for any reason.

(4) Any order, ruling or determination made or rejected by any Court with respect to:

- (a) Class Counsel Fees or Class Counsel Disbursements;
- (b) the opt-out process; or
- (c) the Distribution Protocol

shall not be deemed to be a material modification of all, or a part, of this Settlement Agreement and shall not provide any basis for the termination of this Settlement Agreement.

## 6.2 If Settlement Agreement is Terminated

(1) If this Settlement Agreement is not approved, is terminated in accordance with its terms or otherwise fails to take effect for any reason:

- (a) no motion to certify the Proceeding as a class proceeding on the basis of this Settlement Agreement, or to approve this Settlement Agreement, which has not been decided, shall proceed;
- (b) the Parties will cooperate in seeking to have any issued order certifying the Proceeding as a class proceeding on the basis of the Settlement Agreement or approving this Settlement Agreement set aside and declared null and void and of no force or effect, and any Party shall be estopped from asserting otherwise;

- (c) any prior certification of the Proceeding as a class proceeding on the basis of this Settlement Agreement, including the definitions of the Settlement Class and the Common Issue pursuant to this Settlement Agreement, shall be without prejudice to any position that any of the Parties or Releasees may later take on any issue in the Proceeding, the Second Ontario Action, or any Other Actions or other litigation; and
- within ten (10) days of such termination having occurred, Class Counsel shall (d) destroy all documents or other materials provided by the Settling Defendants and/or Counsel for the Settling Defendants under this Settlement Agreement or containing or reflecting information derived from such documents or other materials received from the Settling Defendants and/or Counsel for the Settling Defendants and, to the extent Class Counsel has disclosed any documents or information provided by the Settling Defendants and/or Counsel for the Settling Defendants to any other Person, shall recover and destroy such documents or information. Class Counsel shall provide Counsel for the Settling Defendants with a written certification by Class Counsel of such destruction. Nothing contained in this Section 6.2 shall be construed to require Class Counsel to destroy any of their work product. However, any documents or information provided by the Settling Defendants and/or Counsel for the Settling Defendants, or received from the Settling Defendants and/or Counsel for the Settling Defendants in connection with this Settlement Agreement, may not be disclosed to any Person in any manner or used, directly or indirectly, by Class Counsel or any other Person in any way for any reason, without the express prior written permission of the relevant Settling Defendants. Class Counsel shall take appropriate steps and precautions to ensure and maintain the confidentiality of such documents, information and any work product of Class Counsel derived from such documents or information.

### 6.3 Allocation of Settlement Amount Following Termination

If the Settlement Agreement is not approved, is terminated or otherwise fails to take effect for any reason, Siskinds LLP shall, within thirty (30) days of the written notice pursuant to Section 6.1(1), return to the Settling Defendants the amount they have paid to Siskinds LLP, plus all

accrued interest thereon, but less the Settling Defendants' proportional share of the costs of notices required by Section 11.1(1) and any translations required by Section 15.11.

#### 6.4 Survival of Provisions After Termination

(1) If this Settlement Agreement is not approved, is terminated or otherwise fails to take effect for any reason, the provisions of Sections 3.1(8), 3.2(3), 4.1(6)(b), 6.1(3), 6.2, 6.3, 6.4, 9.1, 9.2, 11.1(2) and 12.2(4), and the definitions and Schedules applicable thereto shall survive the termination and continue in full force and effect. The definitions and Schedules shall survive only for the limited purpose of the interpretation of Sections 3.1(8), 3.2(3), 4.1(6)(b), 6.1(3), 6.2, 6.3, 6.4, 9.1, 9.2, 11.1(2) and 12.2(4) within the meaning of this Settlement Agreement, but for no other purposes. All other provisions of this Settlement Agreement and all other obligations pursuant to this Settlement Agreement shall cease immediately.

### SECTION 7 - RELEASES AND DISMISSALS

#### 7.1 Release of Releasees

(1) Upon the Effective Date, subject to Section 7.3, and in consideration of payment of the Settlement Amount and for other valuable consideration set forth in the Settlement Agreement, the Releasors forever and absolutely release and forever discharge the Releasees from the Released Claims that any of them, whether directly, indirectly, derivatively, or in any other capacity, ever had, now have, or hereafter can, shall, or may have.

(2) The Plaintiffs and Settlement Class Members acknowledge that they may hereafter discover facts in addition to, or different from, those facts which they know or believe to be true regarding the subject matter of the Settlement Agreement, and it is their intention to release fully, finally and forever all Released Claims and, in furtherance of such intention, this release shall be and remain in effect notwithstanding the discovery or existence of different facts.

#### 7.2 Release by Releasees

(1) Upon the Effective Date, each Release forever and absolutely releases each of the other Release from any and all claims for contribution or indemnity with respect to the Released Claims.

#### 7.3 Covenant Not to Sue

(1) Upon the Effective Date, and notwithstanding Section 7.1, for any Settlement Class Members resident in any province or territory where the release of one tortfeasor is a release of all other tortfeasors, the Releasors do not release the Releasees but instead the Releasors covenant and undertake not to make any claim in any way or to threaten, commence, participate in or continue any proceeding in any jurisdiction against the Releasees in respect of or in relation to the Released Claims.

### 7.4 No Further Claims

(1) Upon the Effective Date, each Releasor shall not now or hereafter institute, continue, maintain, intervene in or assert, either directly or indirectly, whether in Canada or elsewhere, on their own behalf or on behalf of any class or any other Person, any proceeding, cause of action, claim or demand against any Releasee, or any other Person who may claim contribution or indemnity, or other claims over relief, from any Releasee, whether pursuant to the *Negligence Act*, RSO 1990, c. N. 1 or other legislation or at common law or equity in respect of any Released Claim, except for the continuation of the Proceeding and Second Ontario Action against the Non-Settling Defendants or named or unnamed co-conspirators that are not Releasees or, if the Proceeding or Second Ontario Action are not certified with respect to the Non-Settling Defendants, the continuation of the claims asserted in the Proceeding or Second Ontario Action on an individual basis or otherwise against any Non-Settling Defendant or named or unnamed co-conspirator to the low of the foregoing, the Releasee. For greater certainty and without limiting the generality of the foregoing, the Releasors shall not assert or pursue a Released Claim, against any Releasee under the laws of any foreign jurisdiction.

## 7.5 Dismissal of the Proceeding

(1) Upon the Effective Date, the Proceeding shall be dismissed with prejudice and without costs as against the Settling Defendants.

### 7.6 Dismissal of Other Actions

(1) Upon the Effective Date, each member of the Settlement Class shall be deemed to irrevocably consent to the dismissal, without costs, with prejudice and without reservation, of his,

her or its Other Actions against the Releasees, to the extent such Other Actions relate to Electric Powered Steering Assemblies.

(2) Upon the Effective Date, all Other Actions commenced by any Settlement Class Member, to the extent such Other Actions relate to Electric Powered Steering Assemblies, shall be dismissed as against the Releasees, without costs, with prejudice and without reservation.

#### 7.7 Material Term

(1) The releases, covenants, dismissals, and granting of consent contemplated in this Section shall be considered a material term of the Settlement Agreement and the failure of the Court to approve the releases, covenants, dismissals, and granting of consent contemplated herein shall give rise to a right of termination pursuant to Section 6.1 of the Settlement Agreement.

# **SECTION 8 – BAR ORDER AND OTHER CLAIMS**

#### 8.1 Bar Order

- (1) Class Counsel shall seek a bar order from the Court providing for the following:
  - (a) to the extent such claims are recognized at law, all claims for contribution, indemnity or other claims over, whether asserted, unasserted or asserted in a representative capacity, inclusive of interest, taxes and costs, relating to the Released Claims, which were or could have been brought in the Proceeding or any Other Actions, or otherwise, by any Non-Settling Defendant, any named or unnamed co-conspirator that is not a Releasee, any Settled Defendant or any other Person or party against a Releasee, or by a Releasee against any Non-Settling Defendant, any named or unnamed co-conspirator that is not a Release against any Non-Settling Defendant, or any other Person or party, are barred, prohibited and enjoined in accordance with the terms of this Section (unless such claim is made in respect of a claim by a Person who has validly opted out of the Proceeding);
  - (b) if the Court ultimately determines that a claim for contribution and indemnity or other claim over, whether in equity or in law, by statute or otherwise is a legally recognized claim:

- (i) the Plaintiffs and the Settlement Class Members shall not be entitled to claim or recover from the Non-Settling Defendants and/or named or unnamed co-conspirators and/or any other Person or party that is not a Release that portion of any damages (including punitive damages, if any), restitutionary award, disgorgement of profits, interest and costs (including investigative costs claimed pursuant to section 36 of the *Competition Act*) that corresponds to the Proportionate Liability of the Releasees proven at trial or otherwise;
- (ii) the Plaintiffs and the Settlement Class Members shall limit their claims against the Non-Settling Defendants and/or named or unnamed co-conspirators and/or any other Person or party that is not a Release to include only, and shall only seek to recover from the Non-Settling Defendants and/or named or unnamed co-conspirators and/or any other Person or party that is not a Releasee, those claims for damages (including punitive damages, if any), restitutionary award, disgorgement of profits, interest and costs (including investigative costs claimed pursuant to section 36 of the *Competition Act*) attributable to the aggregate of the several liability of the Non-Settling Defendants and/or named or unnamed coconspirators and/or any other Person or party that is not a Releasee to the Plaintiffs and Settlement Class Members, if any, and, for greater certainty, Settlement Class Members shall be entitled to claim and seek to recover on a joint and several basis as between the Non-Settling Defendants and/or named or unnamed co-conspirators and/or any other Person or party that is not a Releasee, if permitted by law; and
- (iii) the Court shall have full authority to determine the Proportionate Liability of the Releasees at the trial or other disposition of the Proceeding, whether or not the Releasees remain in the Proceeding or appear at the trial or other disposition, and the Proportionate Liability of the Releasees shall be determined as if the Releasees are

parties to the Proceeding and any determination by the Court in respect of the Proportionate Liability of the Releasees shall only apply in the Proceeding and shall not be binding on the Releasees in any other proceeding;

- (c) after the Proceeding against the Non-Settling Defendants has been certified and all appeals or times to appeal have been exhausted and on at least twenty (20) days' notice to Counsel for the Settling Defendants, a Non-Settling Defendant may, on motion to the Court seeking an Order for the following, which order shall be determined as if the Settling Defendants remained parties to the Proceeding:
  - documentary discovery and affidavit(s) of documents from Settling Defendant(s) in accordance with that Ontario Rules of Civil Procedure;
  - (ii) oral discovery of representative(s) of Settling Defendant(s), the transcript of which may be read in at trial;
  - (iii) leave to serve a request(s) to admit on Settling Defendant(s) in respect of factual matters; and/or
  - (iv) the production of a representative(s) of Settling Defendant(s) to testify at trial, with such witness to be subject to cross-examination by counsel for the Non-Settling Defendants.
- (d) the Settling Defendants retain all rights to oppose any motion brought pursuant to Section 8.1(1)(c), including any such motion brought at trial seeking an order requiring the Settling Defendants to produce a representative to testify at trial. Moreover, nothing herein restricts the Settling Defendants from seeking a protective order to maintain confidentiality and protection of proprietary information in respect of documents to be produced and/or for information obtained from discovery in accordance with Section 8.1(1)(c);
- (e) on any motion brought pursuant to Section 8.1(1)(c), the Court may make such orders as to costs and other terms as it considers appropriate;

- (f) to the extent that such an order is granted and discovery is provided to a Non-Settling Defendant, a copy of all discovery provided, whether oral or documentary in nature, shall be provided by the Settling Defendants to the Plaintiffs and Class Counsel within ten (10) days of such discovery being provided to a Non-Settling Defendant;
- (g) the Court will retain an ongoing supervisory role over the discovery process and the Settling Defendants will attorn to the jurisdiction of the Court for these purposes; and
- (h) a Non-Settling Defendant may effect service of the motion(s) referred to in Section
   8.1(1)(c) on a Settling Defendant by service on Counsel for the Settling Defendants.

# 8.2 Claims Against Other Entities Reserved

(1) Except as provided herein, this Settlement Agreement does not settle, compromise, release or limit in any way whatsoever any claim by the Releasors against any Person other than the Releasees.

# 8.3 Material Term

(1) The Parties acknowledge that the bar order and reservation of rights contemplated in this Section shall be considered a material term of the Settlement Agreement and the failure of the Court to approve the bar order and reservation of rights contemplated herein shall give rise to a right of termination pursuant to Section 6.1 of the Settlement Agreement.

#### **SECTION 9 - EFFECT OF SETTLEMENT**

# 9.1 No Admission of Liability

(1) The Plaintiffs and the Releasees expressly reserve all of their rights if the Settlement Agreement is not approved, is terminated, or otherwise fails to take effect for any reason. Further, whether or not the Settlement Agreement is finally approved, is terminated, or otherwise fails to take effect for any reason, this Settlement Agreement and anything contained herein, and any and all negotiations, documents, discussions and proceedings associated with this Settlement Agreement, and any action taken to carry out this Settlement Agreement, shall not be deemed, construed, or interpreted to be an admission of any violation of any statute or law, or of any wrongdoing or liability by the Releasees, or of the truth of any of the claims or allegations contained in the Proceeding, the Second Ontario Action, or any Other Actions, or any other pleading filed by the Plaintiffs.

#### 9.2 Agreement Not Evidence

(1) The Parties agree that, whether or not it is finally approved, is terminated, or otherwise fails to take effect for any reason, this Settlement Agreement and anything contained herein, and any and all negotiations, documents, discussions and proceedings associated with this Settlement Agreement, and any action taken to carry out this Settlement Agreement, shall not be referred to, offered as evidence or received in evidence in any pending or future civil, criminal or administrative action or proceeding, except in a proceeding to approve and/or enforce this Settlement Agreement, to defend against the assertion of Released Claims, as necessary in any insurance-related proceeding, or as otherwise required by law.

#### 9.3 No Further Litigation

No Class Counsel, nor anyone currently or hereafter employed by or a partner with Class (1)Counsel, may directly or indirectly participate or be involved in or in any way assist with respect to any claim made or action commenced by any Person against the Settling Defendants which relates to or arises from the Released Claims, except in relation to the continued prosecution of the Proceeding and the Second Ontario Action against any Non-Settling Defendants or unnamed coconspirators that are not Releasees or, if the Proceeding or the Second Ontario Action are not certified, the continuation of the claims asserted in the Proceeding or the Second Ontario Action on an individual basis or otherwise against any Non-Settling Defendant or unnamed co-conspirator that is not a Releasee. Moreover, these Persons may not divulge to anyone for any purpose any information obtained in the course of the Proceeding or the negotiation and preparation of this Settlement Agreement, except to the extent such information is or becomes otherwise publicly available or unless ordered to do so by a court, and subject to Section 4.2 of this Settlement Agreement, except that such information can be disclosed to Camp Fiorante Matthews Mogerman LLP, to the extent that Camp Fiorante Matthews Mogerman LLP is assisting Class Counsel in the prosecution of the Proceeding and the Second Ontario Action and they agree to keep such information confidential and only use it for the purpose of providing such assistance.

#### SECTION 10 – CERTIFICATION FOR SETTLEMENT ONLY

(1) The Parties agree that the Proceeding shall be certified as a class proceeding as against the Settling Defendants solely for purposes of settlement of the Proceeding and the approval of this Settlement Agreement by the Court.

(2) The Plaintiffs agree that, in the motion for certification of the Proceeding as a class proceeding for settlement purposes, the only common issue that they will seek to define is the Common Issue and the only class that they will assert is the Settlement Class.

(3) The Parties agree that the certification of the Proceeding as against the Settling Defendants for the purpose of implementing this Settlement Agreement, shall not derogate in any way from the rights of the Plaintiffs as against the Non-Settling Defendants, except as expressly set out in this Settlement Agreement.

# SECTION 11 – NOTICE TO SETTLEMENT CLASS

#### 11.1 Notices Required

(1) The proposed Settlement Class shall be given a single notice of: (i) the certification of the Proceeding as a class proceeding as against the Settling Defendants for settlement purposes; (ii) the right to opt out of the Proceeding; (iii) the hearing at which the Court will be asked to approve the Settlement Agreement; and (iv) if brought with the hearing to approve the Settlement Agreement, the hearing to approve Class Counsel Fees and Class Counsel Disbursements.

(2) If this Settlement Agreement is not approved, is terminated, or otherwise fails to take effect, the proposed Settlement Class shall be given notice of such event.

# 11.2 Form and Distribution of Notices

(1) The notices shall be in a form agreed upon by the Parties and approved by the Court or, if the Parties cannot agree on the form of the notices, the notices shall be in a form ordered by the Court.

(2) The notices shall be disseminated by a method agreed upon by the Parties and approved by the Court or, if the Parties cannot agree on a method for disseminating the notices, the notices shall be disseminated by a method ordered by the Court.

#### SECTION 12 – ADMINISTRATION AND IMPLEMENTATION

# 12.1 Mechanics of Administration

(1) Except to the extent provided for in this Settlement Agreement, the mechanics of the implementation and administration of this Settlement Agreement shall be determined by the Court on motions brought by Class Counsel.

# 12.2 Information and Assistance

(1) The Settling Defendants will make best efforts to provide to Class Counsel a list of the names and addresses (including any relevant email addresses) of those Persons in Canada who received delivery of Electric Powered Steering Assemblies from the Settling Defendants during the Class Period. The information shall be delivered in Microsoft Excel or such other format as may be agreed upon by Counsel for the Settling Defendants and Class Counsel.

(2) The information required by Section 12.2(1) shall be delivered to Class Counsel within thirty (30) days of the Date of Execution or at a time mutually agreed upon by the Parties.

- (3) Class Counsel may use the information provided under Section 12.2(1):
  - (a) to facilitate the dissemination of the notices required in Section 11.1;
  - (b) to advise Persons in Canada who purchased Electric Powered Steering Assemblies from the Settling Defendants during the Class Period of any subsequent settlement agreement reached in the Proceeding or the Second Ontario Action, any related approval hearings, and any other major steps in the Proceeding or the Second Ontario Action;
  - (c) to facilitate the claims administration process with respect to this Settlement Agreement and any other settlement agreement(s) achieved or court awards issued in the Proceeding or the Second Ontario Action; and
  - (d) as otherwise authorized in Section 4.

(4) All information provided by the Settling Defendants pursuant to Section 12.2(1) shall be dealt with in accordance with Section 4, except that Class Counsel may disclose all information provided by the Settling Defendants pursuant to Section 12.2(1) to any Court-appointed notice

provider and/or the Claims Administrator, to the extent reasonably necessary for the purposes enumerated in Section 12.2(3). Any Court-appointed notice provider and/or the Claims Administrator shall be bound by the same confidentiality obligations set out in Section 4. If this Settlement Agreement is terminated, all information provided by the Settling Defendants pursuant to Section 12.2(1) shall be dealt with in accordance with Section 6.2(1)(d) and no record of the information so provided shall be retained by Class Counsel in any form whatsoever.

(5) The Settling Defendants will make themselves reasonably available to respond to questions respecting the information provided pursuant to Section 12.2(1) from Class Counsel or any Court-appointed notice provider and/or the Claims Administrator. The Settling Defendants' obligations to make themselves reasonably available to respond to questions as particularized in this Section shall not be affected by the release provisions contained in Section 7 of this Settlement Agreement. Unless this Settlement Agreement is not approved, is terminated or otherwise fails to take effect for any reason, the Settling Defendants' obligations to cooperate pursuant to this Section 12.2 shall cease when the Proceeding and the Second Ontario Action are resolved as against all Defendants and all settlement funds or court awards have been distributed.

(6) The Settling Defendants shall bear no liability with respect to the completeness or accuracy of the information provided pursuant to this Section 12.2.

# SECTION 13 – DISTRIBUTION OF THE SETTLEMENT AMOUNT AND ACCRUED INTEREST

#### **13.1** Distribution Protocol

(1) At a time wholly within the discretion of Class Counsel, but on notice to the Settling Defendants, Class Counsel will bring a motion seeking an order from the Court approving the Distribution Protocol. The motion can be brought before the Effective Date, but the order approving the Distribution Protocol shall be conditional on the Effective Date occurring.

(2) The Distribution Protocol shall require Settlement Class Members seeking compensation to give credit for any compensation received through other proceedings or in private out-of-class settlements, unless by such proceedings or private out-of-class settlements the Settlement Class Member's claim was released in its entirety, in which case the Settlement Class Member shall be deemed ineligible for any further compensation.

#### 13.2 No Responsibility for Administration or Fees

(1) Except as otherwise provided for in this Settlement Agreement, the Settling Defendants shall not have any responsibility, financial obligations or liability whatsoever with respect to the administration of the Settlement Agreement or the investment, distribution or administration of monies in the Trust Account including, but not limited to Administration Expenses.

# SECTION 14 – CLASS COUNSEL FEES, DISBURSEMENTS AND ADMINISTRATION EXPENSES

# 14.1 Responsibility for Fees, Disbursements and Taxes

(1) The Settling Defendants shall not be liable for any Class Counsel Fees, Class Counsel Disbursements, or taxes of any of the lawyers, experts, advisors, agents, or representatives retained by Class Counsel, the Plaintiffs or the Settlement Class Members, or any lien of any Person on any payment to any Settlement Class Member from the Settlement Amount.

# 14.2 Responsibility for Costs of Notices and Translation

(1) Siskinds LLP shall pay the costs of the notices required by Section 11 and any costs of translation required by Section 15.11 from the Trust Account, as they become due. Subject to Section 6.3, the Releasees shall not have any responsibility for the costs of the notices or translation.

# 14.3 Court Approval for Class Counsel Fees and Disbursements

(1) Class Counsel may seek the Court's approval to pay Class Counsel Disbursements and Class Counsel Fees contemporaneous with seeking approval of this Settlement Agreement. Class Counsel Disbursements and Class Counsel Fees shall be reimbursed and paid solely out of the Trust Account after the Effective Date. Except as provided herein, Administration Expenses may only be paid out of the Trust Account after the Effective Date. No other Class Counsel Disbursements or Class Counsel Fees shall be paid from the Trust Account prior to the Effective Date.

#### SECTION 15 – MISCELLANEOUS

### 15.1 Motions for Directions

(1) Class Counsel or the Settling Defendants may apply to the Court as may be required for directions in respect of the interpretation, implementation and administration of this Settlement Agreement.

(2) All motions contemplated by this Settlement Agreement shall be on notice to the Parties, except for those applications concerned solely with the implementation and administration of the Distribution Protocol.

#### 15.2 Headings, etc.

(1) In this Settlement Agreement:

- (a) the division of the Settlement Agreement into sections and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Settlement Agreement; and
- (b) the terms "this Settlement Agreement," "hereof," "hereunder," "herein," and similar expressions refer to this Settlement Agreement and not to any particular section or other portion of this Settlement Agreement.

#### **15.3** Computation of Time

(1) In the computation of time in this Settlement Agreement, except where a contrary intention appears,

- (a) where there is a reference to a number of days between two events, the number of days shall be counted by excluding the day on which the first event happens and including the day on which the second event happens, including all calendar days; and
- (b) only in the case where the time for doing an act expires on a holiday as "holiday" is defined in the *Rules of Civil Procedure*, RRO 1990, Reg 194, the act may be done on the next day that is not a holiday.

#### 15.4 Ongoing Jurisdiction

(1) The Court shall exercise jurisdiction with respect to implementation, administration, interpretation and enforcement of the terms of this Settlement Agreement, and the Plaintiffs, Settlement Class Members and Settling Defendants attorn to the jurisdiction of the Court for such purposes.

#### 15.5 Governing Law

(1) This Settlement Agreement shall be governed by and construed and interpreted in accordance with the laws of the Province of Ontario and the laws of Canada applicable therein.

#### 15.6 Entire Agreement

(1) This Settlement Agreement constitutes the entire agreement among the Parties, and supersedes all prior and contemporaneous understandings, undertakings, negotiations, representations, promises, agreements, agreements in principle and memoranda of understanding in connection herewith. None of the Parties will be bound by any prior obligations, conditions or representations with respect to the subject matter of this Settlement Agreement, unless expressly incorporated herein.

#### 15.7 Amendments

(1) This Settlement Agreement may not be modified or amended except in writing and on consent of all Parties hereto, and any such modification or amendment must be approved by the Court.

#### 15.8 Binding Effect

(1) This Settlement Agreement shall be binding upon, and enure to the benefit of, the Plaintiffs, the Settlement Class Members, the Settling Defendants, the Releasors, the Releasees and all of their successors and assigns. Without limiting the generality of the foregoing, each and every covenant and agreement made herein by the Plaintiffs shall be binding upon all Releasors and each and every covenant and agreement made herein by the Settling Defendants shall be binding upon all Releasors and each and every covenant and agreement made herein by the Settling Defendants shall be binding upon all of the Releasees.

### 15.9 Counterparts

(1) This Settlement Agreement may be executed in counterparts, all of which taken together will be deemed to constitute one and the same agreement, and a facsimile or electronic signature shall be deemed an original signature for purposes of executing this Settlement Agreement.

#### 15.10 Negotiated Agreement

(1) This Settlement Agreement has been the subject of negotiations and discussions among the undersigned, each of which has been represented and advised by competent counsel, so that any statute, case law, or rule of interpretation or construction that would or might cause any provision to be construed against the drafter of this Settlement Agreement shall have no force and effect. The Parties further agree that the language contained in or not contained in previous drafts of this Settlement Agreement, or any agreement in principle, shall have no bearing upon the proper interpretation of this Settlement Agreement.

# 15.11 Language

(1) The Parties acknowledge that they have required and consented that this Settlement Agreement and all related documents be prepared in English; les parties reconnaissent avoir exigé que la présente convention et tous les documents connexes soient rédigés en anglais. Nevertheless, if required to by the Court, Class Counsel and/or a translation firm selected by Class Counsel shall prepare a French translation of the Settlement Agreement, the cost of which shall be paid from the Settlement Amount. In the event of any dispute as to the interpretation or application of this Settlement Agreement, only the English version shall govern.

#### 15.12 Transaction

(1) The present Settlement Agreement constitutes a transaction in accordance with Articles 2631 and following of the *Civil Code of Quebec*, and the Parties are hereby renouncing any errors of fact, of law and/or of calculation.

#### 15.13 Recitals

(1) The recitals to this Settlement Agreement are true and form part of the Settlement Agreement.

#### 15.14 Schedules

(1) The schedules annexed hereto form part of this Settlement Agreement.

#### 15.15 Acknowledgements

- (1) Each of the Parties hereby affirms and acknowledges that:
  - (a) he, she or a representative of the Party with the authority to bind the Party with respect to the matters set forth herein has read and understood the Settlement Agreement;
  - (b) the terms of this Settlement Agreement and the effects thereof have been fully explained to him, her or the Party's representative by his, her or its counsel;
  - he, she or the Party's representative fully understands each term of the Settlement
     Agreement and its effect; and
  - (d) no Party has relied upon any statement, representation or inducement (whether material, false, negligently made or otherwise) of any other Party, beyond the terms of the Settlement Agreement, with respect to the first Party's decision to execute this Settlement Agreement.

#### 15.16 Authorized Signatures

(1) Each of the undersigned represents that he or she is fully authorized to enter into the terms and conditions of, and to execute, this Settlement Agreement on behalf of the Parties identified above their respective signatures and their law firms.

#### 15.17 Notice

(1) Where this Settlement Agreement requires a Party to provide notice or any other communication or document to another, such notice, communication or document shall be provided by email, facsimile or letter by overnight delivery to the representatives for the Party to whom notice is being provided, as identified below:

# For the Plaintiffs and for Class Counsel in the Proceeding:

Charles M. Wright and Linda Visser SISKINDS LLP Barristers and Solicitors 680 Waterloo Street London, ON N6A 3V8 Tel: 519.672.2121 Fax: 519.672.6065 Email: charles.wright@siskinds.com linda.visser@siskinds.com

# For the Settling Defendants:

Linda Plumpton and James Gotowiec Torys LLP 79 Wellington Street West, 30th Floor Box 270, TD Centre Toronto, ON M5K 1N2 Tel: 416.865.0040 Fax: 416.865.7380 Email: lplumpton@torys.com jgotowiec@torys.com

David Sterns and Jean Marc Leclerc SOTOS LLP Barristers and Solicitors 180 Dundas Street West, Suite 1250 Toronto, ON M5G 1Z8 Tel: 416.977.0007 Fax: 416.977.0717 Email: dsterns@sotosllp.com jleclerc@sotosllp.com

#### 15.18 Date of Execution

The Parties have executed this Settlement Agreement as of the date on the cover page. (1)

FADY SAMAHA and JORDAN RAMSAY on their own behalf and on behalf of the Settlement Class, by their counsel

Name of Authorized Signatory:

Signature of Authorized Signatory:

Korry McGladdery Dent KMalladdery Dent

Class Counsel

SHERIDAN CHEVROLET CADILLAC LTD. and THE PICKERING AUTO MALL LTD., on their own behalf and on behalf of the Settlement Class, by their counsel

Name of Authorized Signatory:

Signature of Authorized Signatory:

FM Haddeny

Class Counsel

# MITSUBISHI ELECTRIC CORPORATION, MITSUBISHI ELECTRIC AUTOMOTIVE AMERICA, INC. and MITSUBISHI ELECTRIC SALES CANADA INC., by their counsel

.

Name of Authorized Signatory:

Linda Plumpton

Signature of Authorized Signatory:

Torys LLP Counsel for the Settling Defendants

#### **SCHEDULE "A"**

Court File No. CV-14-506652-00CP

#### ONTARIO SUPERIOR COURT OF JUSTICE

| THE HONOURABLE   | ) | , THE | DAY    |
|------------------|---|-------|--------|
| JUSTICE BELOBABA | ) | OF    | , 2018 |

BETWEEN:

# SHERIDAN CHEVROLET CADILLAC LTD., PICKERING AUTO MALL LTD., FADY SAMAHA, and JORDAN RAMSAY

Plaintiffs

- and -

# JTEKT CORPORATION, JTEKT AUTOMOTIVE NORTH AMERICA, INC., KOYO CORPORATION OF U.S.A., KOYO CANADA INC., NSK LTD., NSK AMERICAS, INC., NSK CANADA INC., TRW AUTOMOTIVE HOLDINGS CORP., TRW AUTOMOTIVE INC., TRW DEUTSCHLAND HOLDING GMBH, MITSUBISHI ELECTRIC CORPORATION, MITSUBISHI ELECTRIC AUTOMOTIVE AMERICA, INC., MITSUBISHI ELECTRIC SALES CANADA INC., MITSUBA CORPORATION, AMERICAN MITSUBA CORPORATION, SHOWA CORPORATION, AMERICAN SHOWA, INC. and SHOWA CANADA INC.

Defendants

Proceeding under the Class Proceedings Act, 1992, S.O. 1992, c. 6

ORDER - Electric Powered Steering Assemblies -- Mitsubishi Electric Notice Approval and Consent Certification -

THIS MOTION made by the Plaintiffs for an Order approving the abbreviated, publication and long-form notices of settlement approval hearings and the method of dissemination of said notices, and certifying this proceeding as a class proceeding for settlement purposes as against Mitsubishi Electric Corporation, Mitsubishi Electric Automotive America, Inc. and Mitsubishi Sales Canada, Inc. (collectively, the "Settling Defendants") was read this day at Osgoode Hall, 130 Queen Street West, Toronto, Ontario.

**ON READING** the materials filed, including the settlement agreement with the Settling Defendants dated as of  $\bullet$ , 2018 attached to this Order as Schedule "A" (the "Settlement Agreement"), and on reading the submissions of counsel for the Plaintiffs and Counsel for the Settling Defendants, the Non-Settling Defendants taking no position;

AND WHEREAS a parallel class proceeding relating to the pricing of Electric Powered Steering Assemblies was commenced under Court File No. CV-16-549385-00CP (the "Second Ontario Action") and is being case managed with the Proceeding;

AND ON BEING ADVISED that the Plaintiffs and the Settling Defendants consent to this Order;

- 1. **THIS COURT ORDERS** that for the purposes of this Order, except to the extent that they are modified in this Order, the definitions set out in the Settlement Agreement apply to and are incorporated into this Order.
- 2. THIS COURT ORDERS that the abbreviated, publication and long-form notices of settlement approval hearing are hereby approved substantially in the forms attached respectively hereto as Schedules "B" to "D".
- 3. THIS COURT ORDERS that the plan of dissemination for the abbreviated, publication and long-form notices of settlement approval hearing (the "Plan of Dissemination") is hereby approved in the form attached hereto as Schedule "E" and that the notices of settlement approval hearing shall be disseminated in accordance with the Plan of Dissemination.
- THIS COURT ORDERS that the Proceeding is certified as a class proceeding as against the Settling Defendants for settlement purposes only.

#### 5. **THIS COURT ORDERS** that the "Settlement Class" is certified as follows:

All Persons in Canada who, during the Class Period, (a) purchased, directly or indirectly, Electric Powered Steering Assemblies; and/or (b) purchased or leased, directly or indirectly, a new or used Automotive Vehicle; and/or (c) purchased for import into Canada, a new or used Automotive Vehicle. Excluded Persons are excluded from the Settlement Class.

 THIS COURT ORDERS that Sheridan Chevrolet Cadillac Ltd., Pickering Auto Mall Ltd., Fady Samaha, and Jordan Ramsay are appointed as the representative plaintiffs for the Settlement Class.

# 7. THIS COURT ORDERS that the following issue is common to the Settlement Class:

Did the Settling Defendants conspire to fix, raise, maintain, or stabilize the prices of Electric Powered Steering Assemblies in Canada and elsewhere during the Class Period? If so, what damages, if any, did Settlement Class Members suffer?

8. THIS COURT ORDERS that paragraphs 1 and 4-7 of this Order, including the certification of the Proceeding as against the Settling Defendants for settlement purposes and the definition of Settlement Class and Common Issue, and any reasons given by the Court in connection with paragraphs 1 and 4-7 of this Order, are without prejudice to the rights and defences of the Non-Settling Defendants in connection with the ongoing Proceeding or the Second Ontario Action and, without restricting the generality of the foregoing, may not be relied on by any Person to establish jurisdiction, the criteria for certification (including class definition) or the existence or elements of the causes of action asserted in the Proceeding or the Second Ontario Action, as against the Non-Settling Defendants.

- 9. THIS COURT ORDERS that putative members of the Settlement Class seeking to opt out of the Proceeding must opt out of the Second Ontario Action on a collective basis. For greater certainty, putative members of the Settlement Class cannot opt out of the Proceeding, but not the Second Ontario Action.
- 10. THIS COURT ORDERS that putative members of the Settlement Class can opt out of the Proceeding by sending a written request to opt out to Class Counsel, postmarked on or before the Opt-Out Deadline. The written election to opt out must be signed by the Person or the Person's designee and must include the following information:
  - (a) the Person's full name, current address and telephone number;
  - (b) if the Person seeking to opt out is a corporation, the name of the corporation and the position of the Person submitting the request to opt out on behalf of the corporation;
  - (c) a statement to the effect that the Person wishes to be excluded from the Proceeding and the Second Ontario Action; and
  - (d) the reason(s) for opting out.
- 11. **THIS COURT ORDERS** that where the postmark is not visible or legible, the request to opt out shall be deemed to have been postmarked four (4) business days prior to the date that it is received by Class Counsel.
- 12. **THIS COURT ORDERS** that any putative member of the Settlement Class who validly opts out of the Proceeding shall have no further right to participate in the Proceeding or the

Second Ontario Action or to share in the distribution of any funds received as a result of a judgment or settlement in the Proceeding or the Second Ontario Action.

- THIS COURT ORDERS that no further right to opt out of the Proceeding or Second Ontario Action will be provided.
- 14. THIS COURT ORDERS that, within thirty (30) days of the Opt-Out Deadline, Class Counsel shall provide to the Defendants in the Proceeding and Second Ontario Action a report containing the names of each Person who has validly and timely opted out of the Proceeding and a summary of the information delivered by such Persons pursuant to paragraph 10 above.

The Honourable Justice Belobaba

#### **SCHEDULE "B"**

Court File No. CV-14-506652-00CP

#### ONTARIO SUPERIOR COURT OF JUSTICE

| THE HONOURABLE   | ) | , THE | DAY    |
|------------------|---|-------|--------|
| JUSTICE BELOBABA | ) | OF    | , 2018 |

BETWEEN:

# SHERIDAN CHEVROLET CADILLAC LTD., PICKERING AUTO MALL LTD., FADY SAMAHA, and JORDAN RAMSAY

Plaintiffs

- and -

# JTEKT CORPORATION, JTEKT AUTOMOTIVE NORTH AMERICA, INC., KOYO CORPORATION OF U.S.A., KOYO CANADA INC., NSK LTD., NSK AMERICAS, INC., NSK CANADA INC., TRW AUTOMOTIVE HOLDINGS CORP., TRW AUTOMOTIVE INC., TRW DEUTSCHLAND HOLDING GMBH, MITSUBISHI ELECTRIC CORPORATION, MITSUBISHI ELECTRIC AUTOMOTIVE AMERICA, INC., MITSUBISHI ELECTRIC SALES CANADA INC., MITSUBA CORPORATION, AMERICAN MITSUBA CORPORATION, SHOWA CORPORATION, AMERICAN SHOWA, INC. and SHOWA CANADA INC.

Defendants

Proceeding under the Class Proceedings Act, 1992, S.O. 1992, c. 6

ORDER - Electric Powered Steering Assemblies -- Mitsubishi Electric Settlement Approval -

THIS MOTION made by the Plaintiffs for an Order approving the settlement agreement

entered into with Mitsubishi Electric Corporation, Mitsubishi Electric Automotive America, Inc.

and Mitsubishi Electric Sales Canada Inc. (collectively, the "Settling Defendants") and dismissing

this action as against the Settling Defendants, was heard this day at Osgoode Hall, 130 Queen

Street West, Toronto, Ontario.

AND ON READING the materials filed, including the settlement agreement dated  $\bullet$ , 2018, attached to this Order as Schedule "A" (the "Settlement Agreement"), and on hearing the submissions of counsel for the Plaintiffs and counsel for the Settling Defendants, the Non-Settling Defendants taking no position;

AND WHEREAS a parallel class proceeding relating to the pricing of Electric Powered Steering Assemblies was commenced under Court File No. CV-16-549385-00CP (the "Second Ontario Action") and is being case managed with the Proceeding;

AND ON BEING ADVISED that the deadline for objecting to the Settlement Agreement has passed and there have been • written objections to the Settlement Agreement;

AND ON BEING ADVISED that the deadline for opting out of the Proceeding and Second Ontario Action has passed, and • Persons validly exercised the right to opt out;

AND ON BEING ADVISED that the Plaintiffs and the Settling Defendants consent to this Order:

- 1. **THIS COURT ORDERS** that, in addition to the definitions used elsewhere in this Order, for the purposes of this Order, the definitions set out in the Settlement Agreement apply to and are incorporated into this Order.
- 2. **THIS COURT ORDERS** that in the event of a conflict between this Order and the Settlement Agreement, this Order shall prevail.
- 3. **THIS COURT ORDERS** that this Order, including the Settlement Agreement, is binding upon each member of the Settlement Class including those Persons who are minors or

- 4. **THIS COURT ORDERS** that the Settlement Agreement is fair, reasonable and in the best interests of the Settlement Class.
- 5. THIS COURT ORDERS that the Settlement Agreement is hereby approved pursuant to section 29 of the *Class Proceedings Act*, 1992 and shall be implemented and enforced in accordance with its terms.
- 6. **THIS COURT ORDERS** that, upon the Effective Date, each member of the Settlement Class shall be deemed to have consented to the dismissal as against the Releasees of any Other Actions he, she or it has commenced, without costs and with prejudice.
- 7. THIS COURT ORDERS that, upon the Effective Date, each Other Action commenced in Ontario by any member of the Settlement Class shall be and is hereby dismissed against the Releasees, without costs and with prejudice.
- 8. **THIS COURT ORDERS** that, upon the Effective Date, subject to paragraph 10, each Releasor has released and shall be conclusively deemed to have forever and absolutely released the Releasees from the Released Claims.
- 9. THIS COURT ORDERS that, upon the Effective Date, each Releasor shall not now or hereafter institute, continue, maintain, intervene in or assert, either directly or indirectly, whether in Canada or elsewhere, on their own behalf or on behalf of any class or any other Person, any proceeding, cause of action, claim or demand against any Releasee, or any other Person who may claim contribution or indemnity, or other claims over relief, from any Releasee, whether pursuant to the *Negligence Act*, RSO 1990, c. N. 1 or other

legislation or at common law or equity in respect of any Released Claim, except for the continuation of the Proceeding and the Second Ontario Action against the Non-Settling Defendants or named or unnamed co-conspirators that are not Releasees or, if the Proceeding or the Second Ontario Action are not certified with respect to the Non-Settling Defendants, the continuation of the claims asserted in the Proceeding or the Second Ontario Action on an individual basis or otherwise against any Non-Settling Defendant or named or unnamed co-conspirator that is not a Releasee.

- 10. THIS COURT ORDERS that the use of the terms "Releasors" and "Released Claims" in this Order does not constitute a release of claims by those members of the Settlement Class who are resident in any province or territory where the release of one tortfeasor is a release of all tortfeasors.
- 11. THIS COURT ORDERS that, upon the Effective Date, each member of the Settlement Class who is resident in any province or territory where the release of one tortfeasor is a release of all tortfeasors covenants and undertakes not to make any claim in any way nor to threaten, commence, participate in or continue any proceeding in any jurisdiction against the Releasees in respect of or in relation to the Released Claims.
- 12. THIS COURT ORDERS that all claims for contribution, indemnity or other claims over, whether asserted, unasserted or asserted in a representative capacity, inclusive of interest, taxes and costs, relating to the Released Claims, which were or could have been brought in the Proceeding, the Second Ontario Action, or any Other Actions, or otherwise, by any Non-Settling Defendant, any named or unnamed co-conspirator that is not a Releasee, any Settled Defendant or any other Person or party against a Releasee, or by a Releasee against any Non-Settling Defendant, any named or unnamed co-conspirator that is not a Releasee, any

any Settled Defendant or any other Person or party, are barred, prohibited and enjoined in accordance with the terms of this Order (unless such claim is made in respect of a claim by a Person who has validly opted out of the Proceeding).

- 13. **THIS COURT ORDERS** that if this Court ultimately determines that a claim for contribution and indemnity or other claim over, whether in equity or in law, by statute or otherwise is a legally recognized claim:
  - (a) the Plaintiffs and the Settlement Class Members shall not be entitled to claim or recover from the Non-Settling Defendants and/or named or unnamed coconspirators and/or any other Person or party that is not a Release that portion of any damages (including punitive damages, if any), restitutionary award, disgorgement of profits, interest and costs (including investigative costs claimed pursuant to section 36 of the *Competition Act*) that corresponds to the Proportionate Liability of the Releasees proven at trial or otherwise;
  - (b) the Plaintiffs and the Settlement Class Members shall limit their claims against the Non-Settling Defendants and/or named or unnamed co-conspirators and/or any other Person or party that is not a Releasee to include only, and shall only seek to recover from the Non-Settling Defendants and/or named or unnamed coconspirators and/or any other Person or party that is not a Releasee, those claims for damages (including punitive damages, if any), restitutionary award, disgorgement of profits, interest and costs (including investigative costs claimed pursuant to section 36 of the *Competition Act*) attributable to the aggregate of the several liability of the Non-Settling Defendants and/or named or unnamed coconspirators and/or any other Person or party that is not a Releasee to the Plaintiffs

and Settlement Class Members, if any, and, for greater certainty, the Settlement Class Members shall be entitled to claim and seek to recover on a joint and several basis as between the Non-Settling Defendants and/or named or unnamed coconspirators and/or any other Person or party that is not a Releasee, if permitted by law; and

- (c) this Court shall have full authority to determine the Proportionate Liability of the Releasees at the trial or other disposition of the Proceeding, whether or not the Releasees remain in the Proceeding or appear at the trial or other disposition, and the Proportionate Liability of the Releasees shall be determined as if the Releasees are parties to the Proceeding and any determination by this Court in respect of the Proportionate Liability of the Releasees shall only apply in the Proceeding and shall not be binding on the Releasees in any other proceeding.
- 14. THIS COURT ORDERS that nothing in this Order is intended to or shall limit, restrict or affect any arguments which the Non-Settling Defendants may make regarding the reduction of any assessment of damages (including punitive damages, if any), restitutionary award, disgorgement of profits, interest and costs (including investigative costs claimed pursuant to section 36 of the *Competition Act*) or judgment against them in favour of Settlement Class Members in the Proceeding or the rights of the Plaintiffs and the Settlement Class Members to oppose or resist any such arguments, except as provided for in this Order.
- 15. **THIS COURT ORDERS** that a Non-Settling Defendant may, on motion to this Court determined as if the Settling Defendants remained parties to the Proceeding, and on at least twenty (20) days' notice to Counsel for the Settling Defendants, and not to be brought

unless and until the Proceeding against the Non-Settling Defendants has been certified and all appeals or times to appeal have been exhausted, seek orders for the following:

- (a) documentary discovery and affidavit(s) of documents from Settling Defendant(s)
   in accordance with the *Rules of Civil Procedure*, RRO 1990, Reg. 194;
- (b) oral discovery of representative(s) of Settling Defendant(s), the transcript of which may be read in at trial;
- (c) leave to serve request(s) to admit on Settling Defendant(s) in respect of factual matters; and/or
- (d) the production of representative(s) of Settling Defendant(s) to testify at trial, with such witness(es) to be subject to cross-examination by counsel for the Non-Settling Defendants.
- 16. THIS COURT ORDERS that the Settling Defendants retain all rights to oppose such motion(s) brought under paragraph 15. Moreover, nothing herein restricts the Settling Defendants from seeking a protective order to maintain confidentiality and protection of proprietary information in respect of documents to be produced and/or for information obtained from discovery in accordance with paragraph 15. Notwithstanding any provision in this Order, on any motion brought pursuant to paragraph 15, the Court may make such orders as to costs and other terms as it considers appropriate.
- 17. THIS COURT ORDERS that a Non-Settling Defendant may effect service of the motion(s) referred to in paragraph 15 above by service on Counsel for the Settling Defendants.

- 18. THIS COURT ORDERS that for purposes of administration and enforcement of the Settlement Agreement and this Order, this Court will retain an ongoing supervisory role and the Settling Defendants attorn to the jurisdiction of this Court solely for the purpose of implementing, administering and enforcing the Settlement Agreement and this Order, and subject to the terms and conditions set out in the Settlement Agreement and this Order.
- 19. THIS COURT ORDERS that, except as provided herein, this Order does not affect any claims or causes of action that any members of the Settlement Class has or may have in the Proceeding or the Second Ontario Action against the Non-Settling Defendants or named or unnamed co-conspirators who are not Releasees.
- 20. **THIS COURT ORDERS** that no Release shall have any responsibility or liability whatsoever relating to the administration of the Settlement Agreement; to administration, investment, or distribution of the Trust Account; or to the Distribution Protocol.
- 21. THIS COURT ORDERS that the Settlement Amount shall be held in the Trust Account by Siskinds LLP for the benefit of Class Members and after the Effective Date the Settlement Amount may be used to pay Class Counsel Disbursements incurred for the benefit of the Settlement Class in the continued prosecution of the Proceeding or the Second Ontario Action against the Non-Settling Defendants. This paragraph shall not be interpreted as affecting the rights of the Plaintiffs or the Settlement Class to claim such Disbursements in the context of a future costs award in their favour against the Non-Settling Defendants, or the rights of the Non-Settling Defendants to oppose and resist any such claim.

- 22. **THIS COURT ORDERS** that in the event that some of the Settlement Amount remains in the Trust Account after payment of Class Counsel Disbursements, Class Counsel Fees and Administrative Expenses, Class Counsel shall seek direction from this Court regarding the distribution of the remaining funds.
- 23. THIS COURT ORDERS that, in the event that the Settlement Agreement is terminated in accordance with its terms, this Order shall be declared null and void on subsequent motion made on notice.
- 24. **THIS COURT ORDERS** that the Proceeding be and is hereby dismissed against the Settling Defendants, without costs and with prejudice.
- 25. THIS COURT ORDERS that the approval of the Settlement Agreement and any reasons given by the Court in relation thereto, except any reasons given in connection with paragraphs 12-17 of this Order, are without prejudice to the rights and defences of the Non-Settling Defendants in connection with the ongoing Proceeding and the Second Ontario Action and, without restricting the generality of the foregoing, may not be relied on by any person to establish jurisdiction, the criteria for certification (including class definition) or the existence or elements of the causes of action asserted in the Proceeding or the Second Ontario Action as against the Non-Settling Defendants.

The Honourable Justice Belobaba

# PROPOSED SETTLEMENTS IN CANADIAN AUTO PARTS PRICE-FIXING CLASS ACTIONS

# BETWEEN 1998 AND 2018, DID YOU OR YOUR COMPANY:

- 1. Purchase and/or lease a new or used automotive vehicle in Canada or for import into Canada; and/or
- 2. Purchase any of the following automotive parts: alternators, electric powered steering assemblies, electronic control units, electronic throttle bodies, fuel injection systems, high intensity discharge ballasts, ignition coils, inverters, occupant safety systems, oxygen sensors, spark plugs, starters, switches, valve timing control devices and/or windshield wiper systems?

If so, you might be affected by settlements reached in the class actions.

| SETTLEMENTS ACHIE<br>Settlements have been reached with   |  |  |                            |
|---|--|--|----------------------------|
| • Aisan Industry Co. Ltd., Franklin Pr<br>in the following class actions:   | recision Industry, Inc., Aisar   | n Corporation of America and Hyundar   | n Industrial Co., Ltd.,    |
| PART<br>Electronic Throttle Bodies<br>Fuel Injection Systems  | RELEVANT PERIOD<br>2000 to 2017<br>2000 to 2017  | AMOUNT<br>USD\$440,000<br>USD\$ 40,000<br>TOTAL: USD\$480,000  |                            |
| • Robert Bosch GmbH, Robert Bosch<br>in the following class actions:  | I INC., RODERT BOSCH LLU AN  | nd RBKB Bosch Electrical Drives Co.,   | L <b>IG</b> ,              |
| PART<br>Fuel Injection Systems<br>Oxygen Sensors<br>Spark Plugs<br>Starters<br>Windshield Wiper Systems   | <b>RELEVANT PERIOD</b><br>2000 to 2017<br>2000 to 2018<br>2000 to 2018<br>2000 to 2017<br>2000 to 2018                                       | AMOUNT<br>CDN\$ 375,000<br>USD\$1,747,750<br>USD\$1,747,750<br>CDN\$ 250,000<br>CDN\$ 150,000<br>TOTAL: CDN\$ 775,000 + USD\$3   | 9,495,000                  |
| Mitsubishi Electric Corporation, Mi in the following class actions:   | tsubishi Electric Automotiv  | e America, Inc. and Mistubishi Sales (   | Canada, Inc.               |
| PART<br>Alternators<br>Electric Powered Steering Assemblic<br>Electronic Control Units<br>Electronic Throttle Bodies<br>Fuel Injection Systems<br>High Intensity Discharge Ballasts<br>Ignition Coils<br>Inverters<br>Occupant Safety Systems<br>Starters<br>Valve Timing Control Devices | 1999 to 2016<br>2000 to 2017<br>2000 to 2017<br>1998 to 2018<br>2000 to 2017<br>2000 to 2017<br>2000 to 2017<br>2000 to 2017<br>2000 to 2017 | AMOUNT<br>CDN\$2,200,000<br>CDN\$150,000<br>CDN\$150,000<br>CDN\$50,000<br>CDN\$400,000<br>CDN\$150,000<br>CDN\$150,000<br>CDN\$150,000<br>CDN\$150,000<br>CDN\$150,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$150,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$2,000,000<br>CDN\$ |                            |
| Omron Corporation, Omron Electror<br>in the following class action:   | ics Co. Ltd., Omron Autom  | otive Electronics, Inc. and Omron Au   | iomotive Technologies Inc. |
| PART<br>Switches  | RELEVANT PERIOD  | AMOUNT<br>USD\$390,000   |                            |

# PROPOSED SETTLEMENTS IN CANADIAN AUTO PARTS PRICE-FIXING CLASS ACTIONS

The settlements are a compromise of disputed claims and are not an admission of liability, wrongdoing or fault by any of the parties. The settlements require court approval in Ontario, British Columbia and/or Quebec.

At this stage, the settlement funds (less approved fees and expenses) are being held in interest-bearing trust accounts for the benefit of settlement class members. At a later date, the courts will be asked to approve the distribution of the settlement funds.

Settlement class members have the following options:

- 1. Tell the court what you think about the proposed settlements in writing or speak to the court at the hearings;
- Opt out of (or exclude yourself from) the Electric Powered Steering Assemblies, High Intensity Discharge Ballasts, Oxygen Sensors, Spark Plugs, Switches and/or Windshield Wiper Systems actions (the deadline to opt out of the other actions has already passed); or
- 3. Do nothing.

See the long-form notice online at <u>www.siskinds.com/autoparts</u> for relevant deadlines.

# CLASS COUNSEL

The law firms of Siskinds LLP, Sotos LLP, Camp Fiorante Matthews Mogerman, Siskinds Desmeules s.e.n.c.r.l. and Klein Lawyers LLP (Oxygen Sensors, Spark Plugs and Switches actions only) represent members of these class actions.

# QUESTIONS?

For more information (including relevant deadlines): Visit: www.siskinds.com/autoparts; Email: autoparts@sotosllp.com; or Call: 1.888.977.9806 \*

# AUTO PARTS PRICE-FIXING CLASS ACTIONS

Did you purchase or lease an automotive vehicle in Canada and/or for import into Canada between July 1998 and 2018? And/or did you purchase any of the following automotive parts: alternators, electric powered steering assemblies, electronic control units, electronic throttle bodies, fuel injection systems, high intensity discharge ballasts, ignition coils, inverters, occupant safety systems, oxygen sensors, spark plugs, starters, switches, valve timing control devices and/or windshield wiper systems?

If so, you might be affected by recent settlements in related class actions. The settlements are not admissions of liability, wrongdoing or fault. The settlements require court approval in Ontario, British Columbia and/or Quebec.

At a later date, the courts will be asked to approve the distribution of the settlement funds.

If you would like to remove yourself from the electric powered steering assemblies, high intensity discharge ballasts, oxygen sensors, spark plugs, switches or windshield wiper systems actions, you must act by XX/XX/XXXX. The time to remove yourself from the other actions has expired.

The law firms of Siskinds LLP, Sotos LLP, Camp Fiorante Matthews Mogerman, Siskinds Desmeules s.e.n.c.r.l. and Klein Laywers LLP (oxygen sensors, spark plugs and switches actions only) represent members of these class actions.

For more information visit www.siskinds.com/autoparts email autoparts@sotosllp.com or call 1.888.977.9806

# NOTICE OF CERTIFICATION FOR SETTLEMENT PURPOSES AND SETTLEMENT APPROVAL HEARING IN CANADIAN AUTO PARTS PRICE-FIXING CLASS ACTIONS

#### If you bought or leased a new or used Automotive Vehicle<sup>1</sup>, or certain automotive parts, since July 1998, you should read this notice carefully. It may affect your legal rights.

#### A. WHAT IS A CLASS ACTION?

A class action is a lawsuit filed by one person on behalf of a large group of people.

# B. WHAT ARE THESE CLASS ACTIONS ABOUT?

Class actions have been started in Canada claiming that many companies participated in conspiracies to fix the prices of automotive parts sold in Canada or to manufacturers for installation in automotive vehicles sold in Canada.

This notice is about class actions relating to the following automotive parts (the "Relevant Parts"):

| Part                                    | Description   | Class Period                           |
|---|---|--|
| Alternators                             | An Alternator is a device that charges an Automotive<br>Vehicle's battery and powers the electrical system of an<br>Automotive Vehicle when its engine is running.  | January 1, 2000 to<br>March 20, 2017   |
| Electric Powered<br>Steering Assemblies | An Electric Powered Steering Assembly is a device in an<br>Automotive Vehicle that links the steering wheel to the<br>tires, and includes the column, intermediate shaft, and<br>electric power steering electric control unit, among other<br>parts, but does not include the steering wheel or tires.   | January 1, 2005 to ●                   |
| Electronic Control<br>Units             | An Electronic Control Unit is a wide variety of different<br>types of electronic modules that control one or more of<br>the different electrical systems or sub-systems in an<br>Automotive Vehicle.  | January 1, 1999 to<br>November 2, 2016 |
| Electronic Throttle<br>Bodies           | An Electronic Throttle Body is a component of an<br>electronic throttle control system in an Automotive<br>Vehicle that controls the volume of air flowing into the<br>engine according to a signal from the engine control unit.   | January 1, 2000 to<br>March 20, 2017   |
| Fuel Injection Systems                  | A Fuel Injection System is a system that admits fuel or a<br>fuel/air mixture into engine cylinders of an Automotive<br>Vehicle. The term, "Fuel Injection Systems" as used<br>herein also includes any components of such systems<br>including, without limitation: injectors; high pressure<br>pumps; rail assemblies; feed lines; fuel pumps and fuel<br>pump modules. | January 1, 2000 to<br>March 20, 2017   |

<sup>&</sup>lt;sup>1</sup> In the Mitsubishi Electric Settlement Agreements, Automotive Vehicle is defined as: all automobiles, passenger cars, sport utility vehicles, vans, trucks, buses, and without limitation any other type of vehicle that contains a Relevant Part. In the Aisan, Bosch and Omron Settlement Agreements, Automotive Vehicle is defined as: all passenger cars, sport utility vehicles, vans, and light trucks (up to 10,000 lbs).

| Part                                 | Description  | Class Period                           |
|--------------------------------------|--|--|
| High Intensity<br>Discharge Ballasts | A High Intensity Discharge Ballast is an electrical device<br>that limits the amount of electrical current flowing to a<br>high intensity discharge headlamp installed in an<br>Automotive Vehicle, which would otherwise rise to<br>destructive levels due to a high intensity discharge<br>headlamp's negative resistance. | July 1, 1998 to ●                      |
| Ignition Coils                       | An Ignition Coil is an induction coil that transforms the<br>low voltage of an Automotive Vehicle's battery into the<br>necessary voltage required to create an electric spark in<br>the spark plugs to ignite the fuel.   | January 1, 2000 to<br>March 20, 2017   |
| Inverters                            | An Inverter is a device that provides power to an<br>Automotive Vehicle's motor by converting direct current<br>electricity from an Automotive Vehicle's battery to<br>alternating current electricity.  | January 1, 2000 to<br>March 20, 2017   |
| Occupant Safety<br>Systems           | An Occupant Safety System means any one of the following: seatbelts, steering wheels, airbags and safety electronic systems which control airbag deployment in an Automotive Vehicle.  | January 1, 2003 to<br>December 4, 2014 |
| Oxygen Sensors                       | An Oxygen Sensor means a device that detects the<br>amount of oxygen in the exhaust gases of Automotive<br>Vehicles and sends signals to the engine management<br>computer or "engine control unit". Oxygen Sensors<br>include air fuel ratio sensors.   | January 1, 2000 to ●                   |
| Spark Plugs                          | A Spark Plug means Automotive Vehicle components<br>that use electric currents to ignite the compressed fuel/air<br>mixture in the combustion chamber of an internal<br>combustion engine.   | January 1, 2000 to ●                   |
| Starters                             | A Starter is a device that powers an Automotive Vehicle's battery, causing it to "turn over" and start when the driver turns the ignition switch.  | January 1, 2000 to<br>March 20, 2017   |
| Switches                             | Switches means the following switches used within an<br>Automotive Vehicle: steering wheel switches, turn<br>switches, wiper switches, combination switches, door<br>courtesy switches and power window switches.  | June 1, 2003 to ●                      |
| Valve Timing Control<br>Devices      | A Valve Timing Control Device is a device within the variable valve timing system of an Automotive Vehicle that controls the timing of the opening and closing of engine valves.   | January 1, 2000 to<br>March 20, 2017   |
| Windshield Wiper<br>Systems          | A Windshield Wiper System means a system that includes<br>windshield wipers and components such as the motor,<br>linkage arm and blade.  | January 1, 2000 to ●                   |

For more information about the particular class actions, see the relevant webpage online at <u>www.siskinds.com/autoparts</u> or <u>www.sotosllp.com/class-actions/current-cases/auto-parts/</u>.

The class actions were started in BC, Ontario and/or Quebec, but include Canadian residents in all provinces and territories who were affected by the alleged conspiracies. These class actions claim that the companies that sell the

J.

Relevant Parts were involved in conspiracies to illegally increase the prices of these products. These class actions ask the Court to require these companies to return any extra money they may have received due to the alleged conspiracies.

#### C. WHO IS AFFECTED BY THE CLASS ACTIONS?

The class actions were certified as class proceedings as against the settling defendants for the purposes of implementing the settlement agreements.

You are affected by the class actions and/or are a "member" of the settlement class if you are a person in Canada who, during the relevant class period (as set out in the chart above):

- purchased or leased a new or used Automotive Vehicle in Canada;
- purchased a new or used Automotive Vehicle for import into Canada; or
- purchased a Relevant Part in Canada.

# D. WHAT SETTLEMENTS HAVE BEEN REACHED IN THE CLASS ACTIONS?

A settlement is when a defendant agrees to pay money to the members of the class action in exchange for being released from the case, without admitting liability for any of the claims.

Settlements have been achieved with:

- Aisan Industry Co. Ltd., Franklin Precision Industry, Inc., Aisan Corporation of America and Hyundam Industrial Co., Ltd. ("Aisan");
- Robert Bosch GmbH, Robert Bosch Inc., Robert Bosch LLC and RBKB Bosch Electrical Drives Co., Ltd. ("Bosch");
- Mitsubishi Electric Corporation, Mitsubishi Electric Automotive America, Inc. and Mitsubishi Sales Canada, Inc. ("Mitsubishi Electric"); and
- Omron Corporation, Omron Electronics Co. Ltd., Omron Automotive Electronics, Inc. and Omron Automotive Technologies Inc. ("Omron").

These defendants have agreed to pay the amounts set out below in exchange for a full release of the claims against them relating to the pricing of the Relevant Parts.

| Aisan Settlements:               |                |
|----------------------------------|----------------|
| Electronic Throttle Bodies       | USD\$440,000   |
| Fuel Injection Systems           | USD\$40,000    |
| Bosch Settlements:               |                |
| Fuel Injection Systems           | CAD\$375,000   |
| Oxygen Sensors                   | USD\$1,747,750 |
| Spark Plugs                      | USD\$1,747,750 |
| Starters                         | CAD\$250,000   |
| Windshield Wiper Systems         | CAD\$150,000   |
| Omron Settlement:                |                |
| Switches                         | USD\$390,000   |
| Mitsubishi Electric Settlements: |                |
| Alternators                      | CAD\$2,200,000 |

| Electric Powered Steering Assemblies | CAD\$150,000                                |
|--------------------------------------|---|
| Electronic Control Units             | CAD\$150,000                                |
| Electronic Throttle Bodies           | CAD\$50,000                                 |
| Fuel Injection Systems               | CAD\$400,000                                |
| High Intensity Discharge Ballasts    | CAD\$150,000                                |
| Ignition Coils                       | CAD\$2,500,000                              |
| Inverters                            | CAD\$150,000                                |
| Occupant Safety Systems              | CAD\$150,000                                |
| Starters                             | CAD\$2,000,000                              |
| Valve Timing Control Devices         | CAD\$600,000                                |
| Occupant Safety Systems Starters     | CAD\$150,00<br>CAD\$150,00<br>CAD\$2,000,00 |

The Aisan, Bosch, Mitsubishi Electric and Omron defendants have also agreed to provide cooperation to the plaintiffs in pursuing the applicable class actions against the remaining defendants. The Aisan, Bosch, Mitsubishi Electric and Omron defendants do not admit any liability, wrongdoing or fault.

The settlements are subject to the approval of the Ontario, British Columbia and/or Quebec Courts. The Courts will hold hearings to decide whether to approve these settlements, where applicable, in the British Columbia Court in the City of Vancouver on  $\bullet$ , 2018 at  $\bullet$ :00 a.m., in the Quebec Court in the City of Quebec on  $\bullet$ , 2018 at  $\bullet$ :00 a.m., and in the Ontario Court in the City of Toronto on  $\bullet$ , 2018 at  $\bullet$ :00 a.m. The Courts will decide whether the settlements are fair, reasonable, and in the best interests of settlement class members.

# E. WHAT DO I NEED TO DO?

If you want to be a member of these class actions, you do not need to do anything. However, there are three steps that you should take to protect your legal rights:

- 1. You should keep records of any purchases or leases of all Automotive Vehicles or automotive parts from January 1995 onward.<sup>2</sup> Records include invoices, receipts and bank or loan statements.
- 2. Automotive dealerships should keep records of any sales or leases of Automotive Vehicles or automotive parts from January 1995 onward.
- 3. You should register online at <u>www.siskinds.com/autoparts</u> to receive updates about these class actions and the other auto parts price-fixing class actions.

If you want to tell the Court what you think about the proposed settlements or speak to the Court at the hearings listed above, you must send your written submissions to Class Counsel by email at <u>autoparts@sotosllp.com</u> no later than  $\bullet$ . The written submissions must state the nature of any comments or objections, and whether you intend to appear at the settlement approval hearing.

You may (but do not need to) attend the hearings. If you wish to attend the hearings, please contact Class Counsel for additional details.

#### F. WHAT HAPPENS TO THE MONEY PAID UNDER THE SETTLEMENTS?

At this stage, the settlement funds (less approved fees and expenses) are being held in interest-bearing trust accounts for the benefit of settlement class members. At a later date, the courts will be asked to approve the distribution of the settlement funds.

#### G. WHAT IF I DON'T WANT TO BE IN THE CLASS ACTIONS?

Members of the alternators, electronic control units, electronic throttle bodies, fuel injection systems, ignition coils, inverters, occupant safety systems, starters and valve timing control devices class actions were provided an opportunity to exclude themselves from the class actions ("opt out") and advised that no further right to opt out

<sup>&</sup>lt;sup>2</sup> The earliest class period in the Canadian auto parts class actions commences in January 1995.

would be provided. Members of the electric powered steering assemblies, high intensity discharge ballasts, oxygen sensors, spark plugs, switches and windshield wiper systems settlement classes can opt out.

You can opt out by sending a signed letter to Class Counsel, with the following information:

- your full name, current address and telephone number;
- if you are writing on behalf of a company, the name of the company and your position at the company;
- a statement saying that you (or the company) want to opt out of the class actions. You must identify which class actions you (or the company) want to opt out of; and
- your reason for opting out.

Requests to opt out of the proceedings must be post-marked by ●. Residents of Quebec must also send their opt out letter to the Clerk of the Superior Court of Quebec, by ●.

If you exclude yourself or opt out:

- you will not be eligible to participate in the ongoing class action, and
- you will not receive any money from the class action, but
- you will be able to start or continue your own case against the defendants regarding the claims at issue in the class action.

If you do nothing, and so do not exclude yourself or opt out:

- you will be eligible to participate in the ongoing class action, and
- you may receive money from the class action, but
- you will <u>not</u> be able to start or continue your own case against the defendants regarding the claims at issue in the class action.

This is your only chance to exclude yourself or opt out of the electric powered steering assemblies, high intensity discharge ballasts, oxygen sensors, spark plugs, switches and windshield wiper systems class actions. No further right to opt out will be provided.

Parallel class proceedings were commenced in Ontario against other defendants relating to the pricing of electric powered steering assemblies and switches. The right to opt out applies in those actions as well and no further right to opt out will be provided in respect of those actions. Information on those actions and the companies named as defendants is available on Class Counsel's websites at: <u>www.siskinds.com/autoparts</u>.

# H. WHO ARE THE LAWYERS WORKING ON THESE CLASS ACTIONS AND HOW ARE THEY PAID?

The law firms of Siskinds LLP and Sotos LLP represent members of these class actions in Ontario, and in provinces other than British Columbia or Quebec, as well as corporations of more than 50 employees in Quebec.

Siskinds LLP can be reached at:

Telephone (toll free): 1-800-461-6166 ext. 1315 Email: <u>autopartsclassaction@siskinds.com</u> Mail: 680 Waterloo Street, London, ON N6A 3V8 Attention: Charles Wright

Sotos LLP can be reached at:

Telephone (toll free): 1-888-977-9806 Email: <u>autoparts@sotosllp.com</u> Mail: 180 Dundas Street West, Suite 1200, Toronto, ON M5G 1Z8 Attention: Jean-Marc Leclerc The law firm of Camp Fiorante Matthews Mogerman represents members of the alternators, fuel injection systems, ignition coils, occupant safety systems, starters and windshield wiper systems class actions in British Columbia. They can be reached at:

Telephone: 1-800-689-2322 Email: <u>aslevin@cfmlawyers.ca</u> Mail: #400 - 856 Homer Street, Vancouver, BC V6B 2W5 Attention: David Jones

The law firm of Klein Lawyers LLP represents members of the oxygen sensors, spark plugs and switches class actions in British Columbia. They can be reached at:

Telephone: 604-874-7171 Email: <u>dtanjuatco@callkleinlawyers.com</u> Mail: 1385 West 8<sup>th</sup> Avenue, #400, Vancouver, BC V6H 3V9 Attention: Angela Bespflug

The law firm of Siskinds Desmeules s.e.n.c.r.l. represents individuals and corporations of 50 or less employees who are members of the alternators, electronic control units, electronic throttle bodies, fuel injection systems, ignition coils, inverters, occupant safety systems, starters, valve timing control devices and windshield wiper systems class actions in Quebec. They can be reached at:

Telephone: 418-694-2009 Email: <u>recours@siskindsdesmeules.com</u> Mail: Les promenades du Vieux-Quebec, 43 rue De Buade, bureau 320, Quebec City, QC G1R 4A2 Attention: Barbara Ann Cain

As an individual, you do not have to pay the lawyers working on these class actions any money. The lawyers will be paid from the money collected in these class actions. The Courts will be asked to decide how much the lawyers will be paid. The lawyers will collectively be asking that the Courts approve legal fees of up to 25% of the Aisan, Bosch, Mitsubishi Electric and Omron settlement funds, plus disbursements and applicable taxes. Any approved legal fees will be paid out of the settlement funds. Class Counsel reserve the right to ask the Courts to allow Class Counsel to use the Aisan, Bosch, Mitsubishi Electric and Omron settlement funds to pay for any future adverse cost awards or future disbursements.

### I. WHERE CAN I ASK MORE QUESTIONS?

For more information, please visit <u>www.siskinds.com/autoparts</u>. If you have questions that are not answered online, please contact Class Counsel at the numbers listed above.

To receive future notices and updates regarding the auto parts class actions and any future settlements, register online at <u>www.siskinds.com/autoparts</u>.

#### J. INTERPRETATION

L

This notice contains a summary of some of the terms of the Aisan, Bosch, Mitsubishi Electric and Omron settlement agreements. If there is a conflict between the provisions of this notice and the settlement agreements, the terms of the settlement agreements shall prevail.

ı

## METHOD OF DISSEMINATION OF NOTICE OF HEARING OF AISAN, BOSCH, MITSUBISHI ELECTRIC AND OMRON SETTLEMENTS IN CANADIAN AUTO PARTS PRICE-FIXING CLASS ACTIONS

The Notice of Certification for Settlement Purposes and Settlement Approval Hearing shall be distributed in the following manner:

## Abbreviated Notice:

- 1. sent by direct mail, in English and/or French, as is appropriate, to:
  - (a) the Canadian corporate headquarters of the automotive original equipment manufacturers identified in Schedule "A";
  - (b) the automotive dealerships located in Canada and identified in Schedule "B";
  - (c) car rental companies located in Canada and identified in Schedule "C";
  - (d) taxi companies located in Canada and identified in Schedule "D";
  - (e) car sharing companies located in Canada and identified in Schedule "E";
  - (f) anyone who has inquired with class counsel regarding the auto parts class actions, to the extent that class counsel has their name and address information; and
  - (g) the direct purchaser customers of the settled and settling defendants, to the extent that such information has been provided to class counsel pursuant to the terms of the settled and settling defendants' respective settlements.

## **Publication Notice**

- 2. published once in the following newspapers, no larger than 1/8 newsprint page, in either
  English or French, as is appropriate for each newspaper, subject to each having reasonable
  publication deadlines and costs:
  - (a) The Globe and Mail, national edition;
  - (b) Le Journal de Montreal;
  - (c) The Gazette (Montreal);
  - (d) Le Soleil (Québec City);
  - (e) The Vancouver Sun;
  - (f) The Regina Leader Post;
  - (g) The StarPhoenix (Saskatoon); and

(h) Winnipeg Free Press.

#### Abbreviated and Long-Form Notice

- 3. sent to the following industry associations, in English and/or French, as is appropriate for each association, requesting voluntary distribution to their membership and/or that a copy of the notice or information about the actions be posted on their website:
  - (a) Automobile Protection Association;
  - (b) Alberta Motor Vehicle Industry Council (AMVIC);
  - (c) Motor Vehicle Sales Authority of British Columbia;
  - (d) Ontario Motor Vehicle Industry Council (OMVIC);
  - (e) Canadian Automobile Association (CAA);
  - (f) Alberta Motor Association (AMA);
  - (g) British Columbia Automobile Association (BCAA);
  - (h) CAA Saskatchewan;
  - (i) CAA Manitoba;
  - (j) CAA South Central Ontario;
  - (k) CAA Niagara;
  - (l) CAA North & East Ontario;
  - (m) CAA Quebec;
  - (n) CAA Atlantic;
  - (o) Automobile Journalists Association of Canada;
  - (p) Consumer Interest Alliance Inc.;
  - (q) Consumers' Association of Canada;
  - (r) Consumer Council of Canada;
  - (s) Union des consommateurs;
  - (t) Option Consommateurs;
  - (u) Protégez-Vous;

L

- (v) Canadian Automotive Dealers Association;
- (w) Motor Dealers' Association of Alberta;
- (x) Trillium Automobile Dealers Association;
- (y) La Corporation des Concessionnaires d'Automobiles du Québec;
- (z) Manitoba Motor Dealer Association;
- (aa) New Brunswick Automotive Dealers Association;
- (bb) Nova Scotia Automotive Dealers Association;
- (cc) Prince Edward Island Automotive Dealers Association;
- (dd) Newfoundland & Labrador Automotive Dealers Association.

#### Long-Form Notice

- 4. posted in English and French on class counsel's respective websites; and
- 5. sent by direct mail, in English and/or French, as is appropriate, to any person (or their counsel) known by class counsel as having commenced a similar or related action in Canada.

|  | Head Office for Major Auto N        | Manufacturers in C | anada    |         |   |
|--|-------------------------------------|--------------------|----------|---------|---|
|  |                                     |                    |          | Postal  |   |
| Company                                | Address                             | City               | Province | Code    | Notes   |
| BMW Canada                             | 50 Ultimate Drive                   | Richmond Hill      | ON       | L4S 0C8 | Also Mini   |
| Chrysler Canada Inc.                   | One Riverside Drive West            | Windsor            | ON       | N9A 5K3 | Also Plymouth, Dodge  |
| Ford Motor Company of Canada Limited   | The Canadian Road, PO Box 2000      | Oakville           | ON       | L6K 0C8 |   |
| General Motors of Canada Limited       | 1908 Colonel Sam Drive              | Oshawa             | ON       | L1H 8P7 | Also Pontiac, Buick, Cadillac,<br>Chevrolet, Oldsmobile, GMC,<br>Lincoln) |
| Honda Canada                           | 180 Honda Blvd.                     | Markham            | ON       | L6C 0H9 | Also Acura  |
| Hyundai Auto Canada Corp.              | 75 Frontenac Drive                  | Markham            | ON       | L3R 6H2 |   |
| Isuzu Commercial Truck of Canada, Inc. | 6205-B Airport Road, Suite 410      | Mississauga        | ON       | L4V 1E3 | Isuzu passenger vehicles not<br>sold in Canada since 2002                 |
| Jaguar Canada                          | 75 Courtney Park Drive West, Unit 3 | Mississauga        | ON       | L5W 0E3 | Also Land Rover   |
| Kia Canada Inc.                        | 180 Foster Crescent                 | Mississauga        | ON       | LSR 4J5 |   |
| Mazda Canada Inc.                      | 55 Vogell Road                      | Richmond Hill      | ON       | L4B 3K5 |   |
| Mercedes-Benz Canada Inc.              | 98 Vanderhoof Ave                   | Toronto            | ON       | M4G 4C9 |   |
| Mitsubishi Motor Sales of Canada       | 4141 Dixie Road, PO Box 41009       | Mississauga        | ON       | L4W 5C9 |   |
| Nissan Canada Inc.                     | 2 Hunter Valley Road                | Orillia            | ON       | L3V 6H2 | Also Infiniti   |
| Porsche Cars Canada, Ltd.              | 5925 Airport Road, Suite 420        | Mississauga        | ON       | L4V 1W1 |   |
| Subaru Canada, Inc.                    | 560 Suffolk Court                   | Mississauga        | ON       | L5R 4J7 |   |
| Suzuki Canada Inc.                     | 100 East Beaver Creek Road          | Richmond Hill      | ON       | L4B 1J6 |   |
| Toyota Canada Inc.                     | 1 Toyota Place                      | Toronto            | ON       | M1H 1H9 | Also Scion  |
| Volkswagen Group Canada                | 777 Bayly St. West                  | Ajax               | ON       | L1S 7G7 | Also Audi   |
| Volvo Cars of Canada Corp.             | 9130 Leslie Street, Suite 101       | Richmond Hill      | ON       | L4B 0B9 |   |

5

۳. ۲

|                               | New Car D                | ealers in Ca | anada with Annual | Sales Over \$50 | M           |                |                |
|-------------------------------|--------------------------|--------------|-------------------|-----------------|-------------|----------------|----------------|
|                               |                          | Suite        |                   |                 |             |                |                |
| Company Name                  | Address                  | Number       | City              | Province        | Postal Code | Phone Number   | Fax Number     |
| 2-20 Mazda                    | 2101 Boul Du Traversier  |              | Pincourt          | QC              | J7W0K9      | (514) 453-7220 | (514) 453-6596 |
| 440 Chevrolet Buick Gmc       |                          |              |                   |                 |             |                |                |
| Ltee                          | 3670 Sud Laval A-440 O   |              | Laval             | QC              | Н7Т2Н6      | (514) 333-8333 | Not Available  |
| Acura                         | 2250 Queen St E          |              | Brampton          | ON              | L6S5X9      | (416) 640-5740 | (905) 458-7109 |
| Acura 2000                    | 2250 Queen St E          |              | Brampton          | ON              | L6S5X9      | (905) 458-7100 | (905) 458-7109 |
| Acura Of North Toronto        | 7064 Yonge St            |              | Thornhill         | ON              | L4J1V7      | (905) 882-9660 | (905) 882-9661 |
| Addison Chevrolet             | 1220 Eglinton Ave E      |              | Mississauga       | ON              | L4W2M7      | (905) 238-2886 | (905) 238-3598 |
| Agincourt Autohaus Inc        | 3450 Sheppard Av E       |              | Toronto           | ON              | M1T 3K4     | (416) 291-6456 | Not Available  |
| Agincourt Mazda               | 5500 Finch Ave E         |              | Scarborough       | ON              | M1S0C7      | (416) 283-7100 | (416) 940-8850 |
| Agincourt Nissan              | 1871 Mccowan Rd          |              | Scarborough       | ON              | M1S4L4      | (416) 291-1188 | (416) 291-3388 |
|                               |                          |              |                   |                 |             |                |                |
| Airdrie Chrysler Dodge Jeep   | 139 East Lake Cres NE    |              | Airdrie           | AB              | T4A2H7      | (403) 948-2600 | (403) 948-3234 |
|                               |                          |              |                   |                 |             |                |                |
| Al Palladini's Pine Tree Ford | 100 Auto Park Cir        |              | Woodbridge        | ON              | L4L9T5      | (416) 798-4777 | Not Available  |
| Albi Le Geant                 | 3550 Av De La Gare       |              | Mascouche         | QC              | J7K3C1      | (514) 722-5555 | Not Available  |
| Albi Mazda                    | 3550 Av De La Gare       |              | Mascouche         | QC              | J7K3C1      | (450) 474-7000 | Not Available  |
| Alta Nissan                   | 7625 Martin Grove Rd     | В            | Woodbridge        | ON              | L4L2C5      | (905) 851-1279 | (905) 851-9778 |
| Applewood Chevrolet           |                          |              |                   |                 |             |                |                |
| Cadillac                      | 3000 Woodchester Dr      |              | Mississauga       | ON              | L5L2R4      | (905) 828-2221 | Not Available  |
| Arbour Volkswagen             | 2475 Boul Chomedey       |              | Laval             | QC              | H7T2R2      | (450) 681-8110 | (450) 681-7671 |
| Audi Of Mississauga           | 787 Bancroft Dr          |              | Mississauga       | ON              | L5V2Y6      | (905) 593-6040 | Not Available  |
|                               |                          |              |                   |                 |             |                |                |
| Audi Prestige                 | 5905 Rte Transcanadienne |              | Saint-Laurent     | QC              | H4T1A1      | (514) 364-7777 | (514) 748-7958 |
|                               | 15505 Yellowhead Trail   |              |                   |                 | Ţ           |                |                |
| Auto Canada Inc               | NW                       | 200          | Edmonton          | AB              | T5V1E5      | (780) 732-3135 | (780) 447-0651 |
| Auto Clearing Chrysler        |                          |              |                   |                 |             | 1              |                |
| Dodge                         | 331 Circle Dr W          |              | Saskatoon         | SK _            | S7L5S8      | (306) 244-2186 | Not Available  |
| Auto Gallery Subaru           | 609 Winnipeg St          |              | Regina            | SK              | S4R8P2      | (306) 525-6700 | (306) 543-9496 |
|                               |                          |              |                   |                 |             |                |                |
| Auto Gouverneur Nissan Inc    | 305 Boul Saint-Martin E  |              | Laval             | QC              | H7M1Z1      | (450) 668-1650 | (450) 668-6019 |
| Auto Haus Volkswagen          | 660 Pembina Hwy          |              | Winnipeg          | MB              | R3M2M5      | (204) 284-7520 | (204) 284-0390 |

£,

|                               | New Car D                 | ealers in Ca | anada with Annual S | ales Over \$50 | M                        |                |                |
|-------------------------------|---------------------------|--------------|---------------------|----------------|--------------------------|----------------|----------------|
| · · · · · · · · ·             |                           | Suite        |                     |                | 1                        |                |                |
| Company Name                  | Address                   | Number       | City                | Province       | Postal Code              | Phone Number   | Fax Number     |
| Automobile Lauzon Inc         | 2435 Boul Chomedey        |              | Laval               | QC             | H7T2W5                   | (450) 688-1120 | (450) 688-9091 |
| Automobile Toyota             |                           | -            |                     |                |                          |                |                |
| Charlesbourg                  | 16070 Boul Henri-Bourassa |              | Quebec              | QC             | G1G3Z8                   | (418) 623-9843 | (418) 623-3578 |
| Automobiles Honda De Laval    | 700 Boul Saint-Martin E   |              | Laval               | QC             | H7M4M8                   | (450) 629-7603 | (450) 629-0334 |
|                               | 1917 Boul Sir-Wilfrid-    |              |                     |                |                          |                | ·              |
| Automobiles Niquet Inc        | Laurier                   |              | Saint-Bruno         | QC             | J3V4P6                   | (450) 653-7553 | (450) 653-1653 |
| Automobiles Niquet            | 1905 Boul Sir-Wilfrid-    |              |                     |                |                          |                |                |
| Volkswagen                    | Laurier                   |              | Saint-Bruno         | QC             | J3V0G8                   | (450) 653-1553 | (450) 653-1653 |
| Automobiles Popular Inc       | 5441 Rue Saint-Hubert     |              | Montreal            | QC             | H2J2Y4                   | (514) 274-5471 | (514) 270-2381 |
| Automobiles Val Estrie Inc    | 4141 Rue King O           |              | Sherbrooke          | QC             | J1L1P5                   | (819) 563-4466 | (819) 563-6494 |
| Bank Street Hyundai           | 2788 Bank St              |              | Gloucester          | ON             | K1T1N3                   | (613) 739-7530 | (613) 739-8952 |
| Baril Ford Inc                | 6875 Boul Laurier O       |              | Saint-Hyacinthe     | QC             | J2S9A5                   | (450) 774-2000 | (450) 773-8886 |
| Baril Ford Lincoln            | 6875 Boul Laurier Ouest   |              | Saint-Hyacinthe     | QC             | J2S9A5                   | (450) 773-7070 | Not Available  |
| Barnes Wheaton Surrey Ltd     | 6280 120 St               |              | Surrey              | вс             | V3X1Y7                   | (604) 594-2277 | (604) 501-7106 |
| Barrie Ford                   | 55 Mapleview Dr W         |              | Barrie              | ON             | L4N9H7                   | (705) 737-2310 | (705) 737-0447 |
| Barrie Honda                  | 80 Mapleview Dr W         |              | Barrie              | ON             | L4N9H6                   | (705) 733-2100 | (705) 733-3448 |
| Barry Cullen Chevrolet Cad    |                           |              |                     |                |                          |                |                |
| Ltd                           | 905 Woodlawn Road West    |              | Guelph              | ON             | N1K1B7                   | (519) 824-0210 | (519) 824-9837 |
| Bay-King Chrysler Dodge       |                           |              |                     |                |                          |                |                |
| Jeep                          | 55 Rymal Rd E             |              | Hamilton            | ON             | L9B1B9                   | (905) 383-7700 | Not Available  |
| Bayview Chrysler Dodge Ltd    | 255 Indian Rd S           |              | Sarnia              | ON             | N7T3W5                   | (519) 337-7561 | Not Available  |
| Beauport Mazda                | 585 Rue Clemenceau        |              | Quebec              | QC             | G1C7Z9                   | (418) 667-3131 | Not Available  |
| Beaupre Capitale Chrysler     |                           |              |                     |                | ························ |                |                |
| Inc                           | 225 Rue Du Marais         |              | Quebec              | QC             | G1M3C8                   | (418) 687-2604 | (418) 681-6936 |
| Bel-Air Lexus Toyota Inc      | 435 Mcarthur Ave          |              | Ottawa              | ON             | K1K1G4                   | (613) 741-3270 | Not Available  |
| Belanger Ford Lincoln Ctr Ltd | 204 Michael St            |              | Chelmsford          | ON             | POM1L0                   | (705) 855-4504 | (705) 855-5754 |

э

|                             | New Car I              | Dealers in Ca | anada with Annual | Sales Over \$50 | M           |                |                |
|-----------------------------|------------------------|---------------|-------------------|-----------------|-------------|----------------|----------------|
|                             |                        | Suite         | Î                 |                 |             |                |                |
| Company Name                | Address                | Number        | City              | Province        | Postal Code | Phone Number   | Fax Number     |
| Big 4 Motors Ltd            | 7330 Macleod Trail SE  |               | Calgary           | AB              | T2H0L9      | (403) 252-6671 | (403) 259-0077 |
| Birchcliff Ford Lincoln Inc | 2150 Lawrence Ave E    |               | Scarborough       | ON              | M1R3A7      | (416) 752-9180 | (416) 752-9298 |
| Birchwood Honda On Regent   | 1401 Regent Ave W      |               | Winnipeg          | МВ              | R2C3B2      | (204) 661-6644 | (204) 654-8518 |
| Bmw Autohaus                | 480 Steeles Ave W      |               | Thornhill         | ON              | L4J6X6      | (905) 886-3380 | (905) 886-3381 |
| Bmw Canbec Automobiles      |                        |               |                   |                 |             |                |                |
| Inc                         | 4090 Rue Jean-Talon O  |               | Montreal          | QC              | H4P1V5      | (514) 731-7871 | (514) 731-1072 |
| Bmw Durham                  | 100 Achilles Rd        |               | Ajax              | ON              | L1Z0C5      | (905) 619-5522 | Not Available  |
| Bmw Gallery                 | 650 Crowfoot Cres NW   |               | Calgary           | AB              | T3G4S3      | (403) 275-6492 | (403) 275-6431 |
| Bmw Laval                   | 2450 Boul Chomedey     |               | Laval             | QC              | H7T2X3      | (450) 682-1212 | (450) 682-1123 |
| Bmw Of Missisauga           | 4505 Dixie Rd          |               | Mississauga       | ON              | L4W5K3      | (905) 625-7533 | (905) 625-8230 |
| Bmw Store Ltd               | 2040 Burrard St        |               | Vancouver         | BC              | V6J3H5      | (604) 736-7381 | (604) 736-6738 |
| Bmw Toronto                 | 11 Sunlight Park Rd    |               | Toronto           | ON              | M4M1B5      | (416) 623-4269 | Not Available  |
| Borssard Mazda              | 8450 Boul Taschereau   |               | Brossard          | QC              | J4X1C2      | (450) 466-0999 | (450) 466-0099 |
| Boulevard Toyota            | 120 Rue Du Marais      |               | Quebec            | QC              | G1M3G2      | (418) 683-6565 | (418) 681-8209 |
| Boyer Chevrolet Buick Gmc   | 715 Kingston Rd        |               | Pickering         | ON              | L1V1A9      | (416) 798-7989 | Not Available  |
| Brampton Chrysler Dodge     | 190 Canam Cres         |               | Brampton          | ON              | L7A1A9      | (905) 459-1810 | (905) 459-2516 |
| Jeep<br>Brasso Nissan Ltd   | 195 Glendeer Cir SE    |               | Calgary           | AB              | T2H2S8      | (403) 253-5555 | (403) 258-3460 |
| Brett Chevrolet Cadillac    | 183 Rothesay Ave       |               | Saint John        | NB              | E2J2B4      | (506) 634-5555 | (506) 634-5574 |
| Brian Finch Buick Gmc       | 640 Wonderland Rd N    |               | London            | ON              | N6C5Y7      | (519) 649-7779 | Not Available  |
| Bhan Finen Bulek Gine       |                        |               | London            |                 |             | (,             |                |
| Bridge City Chrysler Dodge  | 3216 1 Ave S           |               | Lethbridge        | AB              | T1J4H2      | (403) 328-3325 | (403) 320-0742 |
| Brock Ford Sales Inc        | 4500 Drummond Rd       |               | Niagara Falls     | ON              | L2E6C7      | (905) 357-5410 | (905) 354-0063 |
| Brossard Cadillac Inc       | 2555 Boul Matte        | 100           | Brossard          | QC              | J4Y2P4      | (450) 619-6669 | Not Available  |
| Brossard Honda              | 8905 Boul Taschereau   |               | Brossard          | QC              | J4Y1A4      | (450) 445-7161 | (450) 678-2406 |
| Burrard Acura               | 2430 Burrard St        |               | Vancouver         | BC              | V6J5L3      | (604) 736-8890 | Not Available  |
| Cadillac Store              | 580 Bayfield St        |               | Barrie            | ON              | L4M5A2      | (705) 722-6348 | Not Available  |
|                             | 34 Heritage Meadows Rd | 1             |                   |                 |             |                |                |
| Calgary Bmw                 | SE                     | 1             | Calgary           | AB              | T2H3C1      | (403) 253-0338 | (403) 253-6638 |

|                             | New Car I                | Dealers in Ca | anada with Annual Sa | ales Over \$50 | DM          |                |                |
|-----------------------------|--------------------------|---------------|----------------------|----------------|-------------|----------------|----------------|
|                             |                          | Suite         | T                    |                |             |                | <u> </u>       |
| Company Name                | Address                  | Number        | City                 | Province       | Postal Code | Phone Number   | Fax Number     |
| Cambrian Ford Sales Ltd     | 1615 Kingsway Rd         |               | Sudbury              | ON             | P3A4S9      | (705) 560-3673 | (705) 560-4662 |
| Campbell Ford               | 1500 Carling Ave         |               | Ottawa               | ON             | K1Z0A3      | (613) 725-3611 | (613) 725-1893 |
| Campbell Parts              | 1500 Carling Ave         |               | Ottawa               | ON             | K1Z0A3      | (613) 725-3647 | (613) 725-1893 |
|                             | 370 Canyon Meadows Dr    |               |                      |                |             |                |                |
| Canyon Creek Toyota         | SE                       |               | Calgary              | AB             | T2J7C6      | (403) 278-6066 | (403) 278-6223 |
| Capital Auto Ctr            | 1311 101 St SW           |               | Edmonton             | AB             | T6X1A1      | (780) 437-5832 | Not Available  |
| Capital Chrysler Jeep Dodge | 1311 101 St SW           |               | Edmonton             | AB             | T6X1A1      | (780) 435-4711 | Not Available  |
| Capital Ford Lincoln Inc    | 1201 N Pasqua St         |               | Regina               | SK             | S4X4P7      | (306) 543-5410 | (306) 775-1716 |
| Capital Gmc Buick Cad Ltd   | 4020 Rochdale Blvd       | ·····         | Regina               | SK             | S4X4P7      | (306) 523-2600 | Not Available  |
| Carter Chev Cad Buick Gmc   | 800 Automall Drive       |               | North Vancouver      | BC             | V7P3R8      | (604) 987-5231 | (604) 980-6588 |
| Centennial Chrysler Dodge   |                          |               |                      |                |             | -              |                |
| Jeep                        | 395 Dunlop St W          |               | Barrie               | ON             | L4N1C3      | (705) 726-0393 | (705) 739-7034 |
| Centennial Plymouth         |                          |               |                      |                |             |                |                |
| Chrysler                    | 395 Dunlop St W          |               | Barrie               | ON             | L4N1C3      | (416) 777-9944 | (705) 739-7034 |
| Centre Auto Collision Ltd   | 354 Richmond St E        |               | Toronto              |                | M5A1P7      | (416) 364-1116 | (416) 364-8532 |
| Centre Honda Downtown       | 354 Richmond St E        |               | Toronto              | ON             | M5A1P7      | (416) 364-1116 | (416) 364-8532 |
| Chambly Toyota Inc          | 7875 Ch De Chambly       |               | Saint-Hubert         | QC             | J3Y5K2      | (450) 658-4334 | (450) 678-1325 |
| Champlain Dodge Chrysler    |                          |               |                      |                |             |                |                |
| Ltee                        | 3350 Rue Wellington      |               | Verdun               | QC             | H4G1T5      | (514) 761-4801 | (514) 761-4404 |
| Chartrand Ford Laval        | 1610 Boul Saint-Martin E |               | Laval                | QC             | H7G4W6      | (450) 669-6110 | (450) 669-8196 |
| Chesswood Group Ltd         | 4077 Chesswood Dr        |               | North York           | ON             | M3J2R8      | (416) 386-3099 | (416) 386-3085 |
| Chevrolet Buick Gmc De      |                          |               |                      |                |             |                |                |
| Lasalle                     | 8000 Newman Blvd         |               | Lasalle              | QC             | H8N1X9      | (514) 595-5666 | (514) 595-5674 |
| Chomedey Toyota Inc         | 3665 Boul Saint-Martin O | -             | Laval                | QC             | H7T1A7      | (450) 687-2634 | Not Available  |
| Christin Automobile Inc     | 12011 Rue Sherbrooke Est |               | Pte-Aux-Trembles     | QC             | H1B1C6      | (514) 640-1050 | (514) 645-0212 |

|                                  | New Car I                | Dealers in Ca | anada with Annual Sa | les Over \$50 | M           |                |                |
|----------------------------------|--------------------------|---------------|----------------------|---------------|-------------|----------------|----------------|
|                                  |                          | Suite         |                      |               |             |                |                |
| Company Name                     | Address                  | Number        | City                 | Province      | Postal Code | Phone Number   | Fax Number     |
|                                  | 6300 Boul Henri-Bourassa |               |                      |               |             |                |                |
| Circuit Ford Lincoln             | E                        |               | Montreal-Nord        | QC            | H1G5W9      | (514) 325-4700 | (514) 325-9872 |
| City Buick Cadillac              | 1900 Victoria Park Ave   |               | North York           | ON            | M1R1T6      | (416) 491-2000 | Not Available  |
| City Ford Sales Ltd              | 14750 St Albert Trail NW |               | Edmonton             | AB            | T6V1H5      | (780) 454-2000 | (780) 447-2912 |
| Civic Honda Ddo Svc              | 3650 Boul Des Sources    |               | Dollard-Ormeaux      | QC            | H9B1Z9      | (514) 683-5533 | Not Available  |
| Clark Chevrolet-Cadillac         | 358 King St              | 101           | Fredericton          | NB            | E3B1E3      | (506) 452-1010 | (506) 458-5657 |
| Classic Honda                    | 30 Van Kirk Dr           |               | Brampton             | ON            | L7A2Y4      | (905) 454-1434 | Not Available  |
| Cmp Automotive Ltd               | 1313 36 St NE            | i             | Calgary              | AB            | T2A6P9      | (403) 207-2455 | Not Available  |
| Cmp Chev Cad Buick Gmc           | 1313 36th Street NE      |               | Calgary              | AB            | T2A6P9      | (403) 207-1000 | Not Available  |
| Coastal Ford Sales Ltd           | 5750 Lougheed Hwy        |               | Burnaby              | BC            | V5B2Z9      | (604) 294-6525 | Not Available  |
| Colonial Honda                   | 2657 Robie St            |               | Halifax              | NS            | B3K4N9      | (902) 453-1940 | (902) 454-5590 |
| Colony Ford Lincoln Sales Inc    | 300 Queen St E           |               | Brampton             | ON            | L6V1C2      | (905) 451-4094 | (905) 451-0004 |
| Competition Chevrolet Ltd        | 40 Boulder Blvd          |               | Stony Plain          | AB            | T7Z1V7      | (780) 963-6121 | Not Available  |
| Cooksville Dodge Chrysler<br>Inc | 290 Dundas St E          |               | Mississauga          | ON            | L5A1W9      | (905) 279-3031 | (905) 279-9686 |
| Coquitlam Chrysler Dodge<br>Jeep | 2960 Christmas Way       |               | Coquitlam            | ВС            | V3C4E6      | (604) 464-6611 | Not Available  |
| Country Hills Toyota             | 20 Freeport Landng NE    |               | Calgary              | AB            | T3J5H6      | (403) 290-1111 | (403) 290-1110 |
| Courtesy Chrysler Dodge          |                          |               |                      |               |             |                |                |
| Jeep                             | 125 Glendeer Cir SE      |               | Calgary              | AB            | T2H2S8      | (403) 255-8111 | Not Available  |
| Courtesy Ford Lincoln Sales      | 684 Wharncliffe Rd S     |               | London               | ON            | N6J2N4      | (519) 680-1200 | (519) 680-3222 |
| Cowell Motors Volkswagen         | 13611 Smallwood Pl       |               | Richmond             | BC            | V6V1W8      | (604) 273-3922 | (604) 273-0758 |
| Crestview Chrysler Dodge<br>Jeep | 601 Albert St            |               | Regina               | SK            | S4R2P4      | (306) 584-4195 | (306) 359-0077 |
| Crosstown Chevrolet Ltd          | 280 Falconbridge Road    | 1             | Sudbury              | ON            | P3A5K3      | (705) 566-4804 | (705) 566-9723 |
| Crosstown Chrysler               | 15520 123 Ave NW         |               | Edmonton             | AB            | T5V1K8      | (780) 488-2627 | (780) 488-0958 |

.

-

|                                   | New Car D                | ealers in Ca | anada with Annual Sa | les Over \$50 | )M          |                |                |
|-----------------------------------|--------------------------|--------------|----------------------|---------------|-------------|----------------|----------------|
|                                   |                          | Suite        |                      |               |             |                |                |
| Company Name                      | Address                  | Number       | City                 | Province      | Postal Code | Phone Number   | Fax Number     |
| Crosstown Chrysler Jeep           |                          |              |                      |               |             |                |                |
| Dodge                             | 15520 123 Ave NW         |              | Edmonton             | AB            | T5V1K8      | (780) 488-4881 | Not Available  |
| Dale Downie Nissan                | 1111 Oxford St E         |              | London               | ON            | N5Y3L7      | (519) 451-4560 | (519) 451-2824 |
| Dams Ford Lincoln Sales Ltd       | 14530 104 Ave            |              | Surrey               | ВС            | V3R1L9      | (604) 588-9921 | Not Available  |
| Dams Ford Lincoln Sales Ltd       | 19330 Langley Bypass     |              | Surrey               | BC            | V3S7R2      | (604) 532-9921 | Not Available  |
| Dan Kane Chevrolet Cadillac       | 500 Division Road        |              | Windsor              | ON            | N9A6M9      | (519) 969-6000 | (519) 969-2363 |
| Dartmouth Chrysler Jeep<br>Dodge  | 61 Athorpe Dr            |              | Dartmouth            | NS            | B2W1K9      | (902) 469-9050 | (902) 463-9908 |
| David Morris Fine Cars Ltd        | 17407 111 Ave NW         |              | Edmonton             | AB            | T5S0A1      | (780) 484-9000 | (780) 484-8827 |
| Davis Buick Gmc                   | 525 Mayor Magrath Dr S   |              | Lethbridge           | AB            | T1J4E3      | (403) 329-6755 | Not Available  |
| Davis Buick Gmc (Medicine<br>Hat) | 1450 Trans Canada Way SE |              | Medicine Hat         | АВ            | T1B4M2      | (403) 527-1115 | Not Available  |
| Davis Gmc Buick Ltd               | 115 Wt Hill Boulevard S  |              | Lethbridge           | AB            | T1J4T6      | (403) 329-4444 | (403) 329-0015 |
| Dearborn Ford                     | 2555 Trans Canada Hwy E  |              | Kamloops             | ВС            | V2C4A9      | (250) 372-7101 | Not Available  |
| Denny Andrews Ford                | 18208 Stony Plain Rd NW  |              | Edmonton             | AB            | T5S1A7      | (780) 489-9999 | Not Available  |
| Derrick Dodge Ltd                 | 6211 104 St NW           |              | Edmonton             | AB            | T6H2K8      | (780) 435-3611 | Not Available  |
|                                   | 3400 Boul Des Sources    |              | Dollard-Ormeaux      | QC            | H9B1Z9      | (514) 685-3310 | (514) 685-2841 |
| Deschamps Chev Buick Cad          |                          |              |                      |               |             |                |                |
| Gmc                               | 333 Boul Armand-Frappier |              | Sainte-Julie         | QC            | J3E0C7      | (450) 649-9333 | (450) 649-9125 |
| Desjardins Auto Collection        | 175 Rue Du Marais        |              | Quebec               | QC            | G1M3C8      | (418) 683-4451 | (418) 683-0199 |
| Desmeules Dodge Chrysler<br>Jeep  | 3500 Sud Laval A-440 O   |              | Laval                | QC            | H7T2H6      | (450) 682-1700 | Not Available  |

.

|                                  | New Car               | Dealers in Ca | anada with Annual | Sales Over \$50 | M           |                |                |
|----------------------------------|-----------------------|---------------|-------------------|-----------------|-------------|----------------|----------------|
| <u> </u>                         |                       | Suite         |                   |                 |             |                |                |
| Company Name                     | Address               | Number        | City              | Province        | Postal Code | Phone Number   | Fax Number     |
| Dilawri Chrysler Jeep Dodge      | 370 West Hunt Club Rd |               | Nepean            | ON              | K2E1A5      | (613) 523-9959 | (613) 523-1392 |
| Dilawri Mitsubishi               | 1750 6 Ave            |               | Regina            | SK              | S4R0X3      | (306) 525-2333 | (306) 546-1725 |
| Discovery Ford Sales Ltd         | 850 Brant St          |               | Burlington        | ON              | L7R2J5      | (905) 634-6399 | (905) 632-0914 |
| Discovery Ford Sls               |                       |               |                   |                 |             |                |                |
| Burlington                       | 850 Brant St          |               | Burlington        | ON              | L7R2J5      | (905) 632-8696 | (905) 632-0914 |
| Dixie Ford Sales Ltd             | 5495 Dixie Rd         |               | Mississauga       | ON              | L4W1E6      | (905) 629-1300 | (905) 629-2081 |
| Dixie Plymouth Chrysler Jeep     | 8050 Dixie Rd         |               | Brampton          | ON              | L6T4W6      | (905) 452-1000 | (905) 452-8983 |
| Dodge City Auto                  | 2200 8th St E         |               | Saskatoon         | SK              | S7H0V3      | (306) 374-2120 | (306) 374-9899 |
| Dodge City-Royal Garage          | 709 Topsail Rd        |               | Mount Pearl       | NL              | A1N3N4      | (709) 748-2110 | (709) 748-2108 |
| Don Valley North Lexus           | 3120 Steeles Ave E    |               | Markham           | ON              | L3R1G9      | (905) 479-8555 | Not Available  |
| Don Valley North Toyota          | 3300 Steeles Ave E    |               | Markham           | ON              | L3R1G9      | (905) 475-0722 | Not Available  |
| Don Valley Volkswag0n Ltd        | 1695 Eglinton Ave E   | 3             | North York        | ON              | M4A1J6      | (416) 751-3131 | (416) 751-3051 |
| Don Wheaton Gm                   | 10727 82 Ave NW       |               | Edmonton          | AB              | T6E2B1      | (780) 439-0071 | (780) 431-5501 |
| Donnelly Automotive Group        | 2496 Bank St          |               | Ottawa            | ON              | K1V0W8      | (613) 737-5000 | Not Available  |
| Donway Ford Lincoln Sales<br>Ltd | 1975 Eglinton Ave E   |               | Scarborough       | ON              | M1L2N1      | (416) 751-2200 | (416) 751-9301 |
| Douglas Ford Lincoln Sales       |                       |               |                   |                 |             |                |                |
| Ltd                              | 379 Bayfield St       |               | Barrie            | ON              | L4M3C5      | (705) 728-5558 | Not Available  |
| Dow Honda                        | 835 Carling Ave       |               | Ottawa            | ON              | K1S2E7      | (613) 237-2777 | (613) 237-0229 |
| Downey's Ford Lincoln            | 35 Consumers Dr       |               | Saint John        | NB              | E2J4Z7      | (506) 632-6000 | Not Available  |
| Downsview Chrysler Dodge         | 199 Rimrock Rd        |               | North York        | ON              | мзјзсб      | (416) 635-1660 | (416) 635-1797 |
| Jeep<br>Duchesne Auto Ltee       | 450 Boul De Quen      | <u> </u>      | Alma              | QC              | G8B5P5      | (418) 662-3431 | (418) 662-3294 |
| Duck Chev Buick Cadillac         |                       |               |                   |                 |             |                |                |
| Gmc                              | 400 SE Marine Drive   |               | Vancouver         | BC              | V5X4X2      | (604) 324-7222 | (604) 324-4414 |
| Dueck Downtown Chev<br>Buick Gmc | 888 Terminal Avenue   |               | Vancouver         | ВС              | V6A2M5      | (604) 675-7900 | Not Available  |

|                               | New Car D                   | Dealers in Ca | anada with Annual | Sales Over \$50 | M           | , <b></b>      | -              |
|-------------------------------|-----------------------------|---------------|-------------------|-----------------|-------------|----------------|----------------|
|                               |                             | Suite         |                   |                 |             |                |                |
| Company Name                  | Address                     | Number        | City              | Province        | Postal Code | Phone Number   | Fax Number     |
| Dueck Richmond Chev Buick     |                             |               |                   |                 |             |                |                |
| Cad                           | 12100 Featherstone Way      |               | Richmond          | BC              | V6W1K9      | (604) 273-1311 | (604) 273-9038 |
| Dunlop Ford Sales Ltd         | 1510 Mayor Magrath Dr S     |               | Lethbridge        | AB              | T1K2R3      | (403) 328-8861 | (403) 329-9156 |
| Dupre Chevrolet-Cadillac Inc  | 18 Boul Sir-Wilfrid-Laurier |               | Beloeil           | QC              | J3G4E8      | (450) 467-2804 | (450) 467-0850 |
| Eagle Ridge Chev Buick Gmc    |                             |               |                   |                 |             |                |                |
| Ltd                           | 2595 Barnet Highway         |               | Coquitlam         | BC              | V3E1K9      | (604) 464-3941 | Not Available  |
| East-Court Ford Lincoln Sales | 4700 Sheppard Ave E         |               | Scarborough       | ON              | M1S3V6      | (416) 292-1171 | (416) 292-3342 |
| Eastside Chevrolet Buick      |                             |               |                   |                 |             |                |                |
| Gmc                           | 8435 Woodbine Ave           |               | Markham           | ON              | L3R2P4      | (905) 475-7373 | Not Available  |
| Eastside Dodge Chrysler       |                             |               |                   |                 |             |                |                |
| Jeep                          | 815 36 St NE                |               | Calgary           | AB              | T2A4W3      | (403) 273-4313 | (403) 235-1267 |
| Edmonton Bmw                  | 7450 Roper Rd NW            |               | Edmonton          | AB              | T6B3L9      | (780) 490-5269 | Not Available  |
| Edmonton Scion                | 10220170 St NW              |               | Edmonton          | AB              | T5S1N9      | (780) 437-2466 | Not Available  |
| Ens Scion                     | 631 Brant Ct                |               | Saskatoon         | SK              | S7J5L3      | (306) 500-8838 | (306) 664-6965 |
| Ens Toyota                    | 285 Venture Cres            |               | Saskatoon         | SK              | S7K6N8      | (306) 653-5611 | Not Available  |
| Erin Dodge Chrysler           | 2365 Motorway Blvd          |               | Mississauga       | ON              | L5L2M4      | (905) 828-2004 | Not Available  |
| Erin Park Lexus               | 2360 Motorway Blvd          |               | Mississauga       | ON              | L5L1X3      | (905) 828-7727 | (905) 828-9282 |
| Erin Park Toyota              | 2411 Motorway Blvd          |               | Mississauga       | ON              | L5L3R2      | (905) 828-7711 | (905) 828-0617 |
| Estrie Toyota                 | 6 Rue Irwin                 |               | Granby            | QC              | J2J2P1      | (450) 378-8404 | (450) 378-1478 |
| Fairview Chrysler Dodge Ltd   | 2377 Fairview St            |               | Burlington        | ON              | L7R2E3      | (905) 632-4335 | (905) 333-4454 |
| Falls Chev Cad Buick Gmc      | 5888 Thorold Stone Road     |               | Niagara Falls     | ON              | L2J1A1      | (905) 358-6173 | Not Available  |
| Fast Lane                     | 1438 Youville Dr            |               | Orleans           | ON              | K1C2X8      | (613) 830-8806 | (613) 841-1011 |
| Finch Chev Cadillac Buick     |                             |               |                   |                 |             |                |                |
| Gmc                           | 640 Wonderland Rd North     |               | London            | ON              | N6H3E5      | (519) 657-9411 | Not Available  |
| Forbes Motors Inc             | 165 Weber Street South      |               | Waterloo          | ON              | N2J2B1      | (519) 742-4463 | (519) 742-4107 |
| Ford Lincoln Laval 440        | 2705 Boul Chomedey          |               | Laval             | QC              | H7P0C2      | (450) 666-3673 | Not Available  |

\*

|                              | New Car D                  | ealers in Ca | anada with Annual | Sales Over \$50 | M           |                |                |
|------------------------------|----------------------------|--------------|-------------------|-----------------|-------------|----------------|----------------|
|                              |                            | Suite        |                   |                 |             |                |                |
| Company Name                 | Address                    | Number       | City              | Province        | Postal Code | Phone Number   | Fax Number     |
| Formula Honda                | 2240 Markham Rd            |              | Scarborough       | ON              | M1B2W4      | (416) 754-4555 | Not Available  |
| Fort Motors Ltd              | 11104 Alaska Rd            |              | Fort St. John     | BC              | V1J5T5      | (250) 785-6661 | (250) 785-2283 |
|                              | 7000 Boul Louis-H-         |              |                   |                 |             |                |                |
| Fortier Auto Ltee            | Lafontaine                 |              | Anjou             | QC              | H1M2X3      | (514) 353-9821 | (514) 353-4556 |
| Fraser Ford Sales Ltd        | 815 King St W              |              | Oshawa            | ON              | L1J2L4      | (905) 576-1800 | Not Available  |
| Freedom Ford Sales Ltd       | 7505 75 St NW              |              | Edmonton          | AB              | T6C4H8      | (780) 462-7575 | Not Available  |
| Frontier Honda               | 8704 100 St                |              | Morinville        | AB              | T8R1K6      | (780) 939-3666 | (780) 939-5306 |
| Frost Chevrolet              | 150 Bovaird Dr W           |              | Brampton          | ON              | L7A0H3      | (905) 459-0126 | (905) 459-8442 |
| Gabriel Ford                 | 7100 Rue Saint-Jacques     |              | Montreal          | QC              | H4B1V2      | (514) 489-3077 | (514) 489-8710 |
| Gabriel Lexus                | 3303 Cote De Liesse        |              | Montreal          | QC              | H4N3C2      | (514) 747-7777 | Not Available  |
| Gananoque Chev Buick Gmc     |                            |              |                   |                 |             |                |                |
| Cad                          | 439 King Street East       |              | Gananoque         | ON              | K7G1G9      | (613) 382-3366 | Not Available  |
| Garage Hyundai Victoriaville | 433 Boul Des Bois-Francs N |              | Victoriaville     | QC              | G6P1H1      | (819) 758-5775 | Not Available  |
| Gauthier Dodge Chrysler      |                            |              |                   |                 |             |                |                |
| Jeep                         | 1375 Regent Ave W          |              | Winnipeg          | MB              | R2C3B2      | (204) 661-8999 | (204) 663-9261 |
| General Motors Of Canada     |                            |              |                   |                 |             |                |                |
| Ltd                          | 5000 Aut Transcanadienne   |              | Pointe-Claire     | QC              | H9R1B6      | (514) 630-6209 | (514) 630-7395 |
|                              | 201 Highway 17 E &         |              |                   |                 |             |                |                |
| George Stockfish Ford Sales  | Highway 11                 |              | North Bay         | ON              | P1B8J5      | (705) 476-1506 | (705) 476-0323 |
| Georgian Buick Gmc           | 65 Barrie View Dr          |              | Barrie            | ON              | L4N8V4      | (705) 730-1310 | Not Available  |
| Georgian Chevrolet Buick     |                            |              |                   |                 |             |                |                |
| Gmc                          | 65 Barrie View Drive       |              | Barrie            | ON              | L4N8V4      | (705) 733-2222 | Not Available  |
| Gold Key Langley Vw          |                            |              |                   |                 |             |                |                |
| Canada's                     | 19545 Langley Bypass       |              | Surrey            | BC              | V3S6K1      | (604) 534-7431 | Not Available  |
| Granby Chevrolet Cad Buick   |                            |              |                   |                 |             |                |                |
| Gmc                          | 595 Rue Boivin             |              | Granby            | QC              | J2G2M1      | (450) 372-4242 | Not Available  |
| Gravel Chev Buick Cad Gmc    |                            |              |                   |                 |             |                |                |
| Ltee                         | 1007 Boul Rene Levesque    |              | Verdun            | QC              | H3E0H1      | (514) 769-5353 | Not Available  |
| Gravel Chevrolet             | 1000 Boul Rene-Levesque    |              | Verdun            | QC              | H3E0H1      | (514) 861-2000 | (514) 769-5239 |

|                             | New Car D                 | ealers in Ca | anada with Annual S | Sales Over \$50 | )M          |                | -              |
|-----------------------------|---------------------------|--------------|---------------------|-----------------|-------------|----------------|----------------|
|                             |                           | Suite        |                     |                 |             |                |                |
| Company Name                | Address                   | Number       | City                | Province        | Postal Code | Phone Number   | Fax Number     |
| Groupe Lauzon               | 2435 Boul Chomedey        |              | Laval               | QC              | H7T2W5      | (514) 990-1192 | (450) 688-9091 |
| Gsl Chevrolet Cadillac Ltd  | 1720 Bow Trail SW         |              | Calgary             | AB              | T3C2E4      | (403) 265-7690 | Not Available  |
| Gus Brown Buick Gmc         | 1201 Dundas Street East   |              | Whitby              | ON              | L1N2K6      | (905) 668-5846 | (905) 668-3211 |
| Gus Revenberg Chev Buick    | 10150 Tecumseh Road       |              |                     |                 |             |                |                |
| Gmc                         | East                      |              | Windsor             | ON              | N8R1A2      | (519) 979-2800 | (519) 979-3093 |
| Gyro Mazda                  | 134 Laird Dr              |              | East York           | ON              | M4G3V7      | (416) 421-5730 | (416) 421-0822 |
| H J Pfaff Motors Inc Audi   | 9088 Jane St              |              | Concord             | ON              | L4K2M9      | (905) 907-2834 | (905) 907-0904 |
| Halifax Chrysler Dodge Jeep | 12 Lakelands Blvd         |              | Halifax             | NS              | B3S1S8      | (902) 455-0566 | Not Available  |
| Halton Honda                | 4100 Harvester Rd         |              | Burlington          | ON              | L7L0C1      | (905) 632-5371 | Not Available  |
| Hamel Chevrolet Buick Gmc   |                           |              |                     |                 |             |                |                |
| Ltee                        | 9455 Boulevard Lacordaire |              | Saint-Leonard       | QC              | H1R3E8      | (514) 327-3540 | (514) 327-1403 |
| Hansen Ford Lincoln Sales   |                           |              |                     |                 |             |                |                |
| Inc                         | 10101 112 St              |              | Grande Prairie      | AB              | T8V8H6      | (780) 538-1774 | (780) 538-4326 |
| Heffner Lexus               | 3131 King St E            |              | Kitchener           | ON              | N2A1B1      | (519) 895-9130 | (519) 748-9616 |
| Heffner Motors Ltd          | 3121 King St E            |              | Kitchener           | ON              | N2A1B1      | (519) 748-9669 | (519) 748-9904 |
| Heffner Toyota Svc          | 3121 King St E            |              | Kitchener           | ON              | N2A1B1      | (519) 748-9667 | Not Available  |
| Heninger Toyota             | 3640 Macleod Trail SE     |              | Calgary             | AB              | T2G2P9      | (403) 243-6118 | Not Available  |
| Highbury Ford Sales Ltd     | 1365 Dundas St            |              | London              | ON              | N5W3B5      | (519) 455-1800 | (519) 455-9867 |
| Highland Chev Buick Gmc     |                           |              |                     |                 |             |                |                |
| Cad                         | 15783 Yonge Street        |              | Aurora              | ON              | L4G6S9      | (905) 727-9444 | Not Available  |
| Highland Chevrolet          | 15783 Yonge St            |              | Aurora              | ON              | L4G1P4      | (416) 798-7699 | (905) 727-6099 |
|                             | 3700 Rue Sainte-Catherine |              |                     |                 |             |                |                |
| Hochelaga Chevrolet Buick   | E                         |              | Montreal            | QC              | H1W2E8      | (514) 524-4471 | (514) 525-7198 |
| Hogan Chevrolet Buick Gmc   |                           |              |                     |                 |             |                |                |
| Ltd                         | 5000 Sheppard Ave E       |              | Scarborough         | ON              | M1S4L9      | (416) 291-5054 | Not Available  |
| Holiday Ford Sales Ltd      | 1555 Lansdowne St W       |              | Peterborough        | ON              | K9J7M3      | (705) 742-5432 | (705) 742-7722 |
| Honda Canada Inc            | 13240 Worster Crt         |              | Richmond            | BC              | V6V2B8      | (604) 278-7121 | Not Available  |
| Honda De Blainville         | 700 Boul Du Cure-Labelle  |              | Blainville          | QC              | J7C2J6      | (450) 435-1122 | Not Available  |
| Honda De Terrebonne         | 2850 Ch Gascon            |              | Terrebonne          | QC              | J6X4H6      | (514) 331-0555 | Not Available  |

|                            | New Car I              | Dealers in Ca | anada with Annual Sa | les Over \$50 | M           |                |                |
|----------------------------|------------------------|---------------|----------------------|---------------|-------------|----------------|----------------|
|                            |                        | Suite         |                      |               |             |                |                |
| Company Name               | Address                | Number        | City                 | Province      | Postal Code | Phone Number   | Fax Number     |
| Honda Ile Perrot           | 40 Boul Don-Quichotte  |               | L'Ile-Perrot         | QC            | J7V6N5      | (514) 453-8416 | (514) 453-8047 |
| Honda Ready Import Ltd     | 260 Dundas St E        |               | Mississauga          | ON            | L5A1W9      | (905) 896-3500 | (905) 896-0759 |
| Houle Toyota               | 12305 Rue Sherbrooke E |               | Pte-Aux-Trembles     | QC            | H1B1C8      | (514) 640-5010 | (514) 640-0300 |
| Humberview Buick Gmc       | 1650 The Queensway     |               | Etobicoke            | ON            | M8Z1X1      | (416) 259-3030 | (416) 259-8995 |
| Hunt Club Nissan           | 275 West Hunt Club Rd  |               | Nepean               | ON            | K2E1A6      | (613) 521-6262 | (613) 521-8241 |
| Infiniti 417               | 1599 Star Top Rd       |               | Gloucester           | ON            | K185P5      | (613) 749-9417 | Not Available  |
| Islington Chrysler Dodge   |                        |               |                      |               |             |                |                |
| Jeep                       | 5476 Dundas St W       |               | Etobicoke            | ON            | M9B1B6      | (416) 239-3541 | (416) 239-2743 |
| Jack Carter Chev Cad Buick |                        |               |                      |               |             |                |                |
| Gmc                        | 11555 29th Street SE   |               | Calgary              | AB            | T2ZON4      | (403) 258-6300 | Not Available  |
| Jack Carter Chevrolet      |                        |               |                      |               |             |                |                |
| Cadillac                   | 11555 29 Street SE     |               | Calgary              | AB            | T2ZON4      | (403) 258-6349 | Not Available  |
| Jack Mcgee Chevrolet-      |                        |               |                      |               |             |                |                |
| Cadillac                   | 1053 Clonsilla Avenue  |               | Peterborough         | ON            | K9J5Y2      | (705) 741-9000 | (705) 743-1898 |
| Jacques Olivier Ford Inc   | 4405 Ch De Chambly     |               | Saint-Hubert         | QC            | J3Y3M7      | (450) 445-3673 | Not Available  |
| Jerome D Taylor Chevrolet  |                        |               |                      |               |             |                |                |
| Cad                        | 2440 Princess Street   |               | Kingston             | ON            | K7M3G4      | (613) 549-1311 | (613) 548-7665 |
| Jerome Taylor Auto Mall    | 2440 Princess St       |               | Kingston             | ON            | K7M3G4      | (613) 545-3400 | (613) 548-7665 |
| Jim Gauthier Cad Buick Gmc |                        |               |                      |               |             |                |                |
| Ltd                        | 2400 Mcphillips Street |               | Winnipeg             | MB            | R2V4J6      | (204) 633-8833 | Not Available  |
| Jim Gauthier Chevrolet Ltd | 1400 Mcphillips Street |               | Winnipeg             | MB            | R2V4G6      | (204) 697-1400 | (204) 697-2878 |
| Jim Pattison Chrysler Jeep | 15377 Guildford Dr     |               | Surrey               | BC            | V3R0H9      | (604) 581-2231 | (604) 581-0301 |
| Jim Pattison Scion         | 1290 Burrard St        |               | Vancouver            | BC            | V6Z1Z4      | (604) 681-8824 | Not Available  |
| Jim Pattison Scion-        |                        |               | 1                    |               |             |                |                |
| Downtown                   | 1229 Hornby St         |               | Vancouver            | BC            | V6Z1W4      | (604) 568-8836 | Not Available  |
| Jim Pattison Scion-Surrey  | 15389 Guildford Dr     |               | Surrey               | BC            | V3R0H9      | (604) 495-4150 | Not Available  |
| Jim Tubman Motors          | 1770 Bank Street       | -             | Ottawa               | ON            | K1V7Y6      | (613) 733-4050 | Not Available  |
| Jim Wilson Chev Pntc Buick |                        | 1             |                      |               |             |                |                |
| Gmc                        | 20 Mulcahy Crt         |               | Orillia              | ON            | L3V7W7      | (705) 329-2000 | Not Available  |

|                             | New Car Dealers in Canada with Annual Sales Over \$50M |        |                |          |             |                |                                       |  |  |  |  |
|-----------------------------|--|--------|----------------|----------|-------------|----------------|---------------------------------------|--|--|--|--|
|                             |  | Suite  |                | <u>,</u> |             | T              |                                       |  |  |  |  |
| Company Name                | Address  | Number | City           | Province | Postal Code | Phone Number   | Fax Number                            |  |  |  |  |
| John Bear Buick Chev Gmc    |  |        |                |          |             |                | · · · · · · · · · · · · · · · · · · · |  |  |  |  |
| Cad                         | 1200 Upper James Street                                |        | Hamilton       | ON       | L9C3B1      | (905) 575-9400 | (905) 575-5233                        |  |  |  |  |
| John Bear Buick Gmc Ltd     | 333 Lake Street  |        | St Catharines  | ON       | L2N7T3      | (905) 934-2571 | Not Available                         |  |  |  |  |
| John Scotti Jaguar          | 4305 Boul Metropolitain E                              |        | Saint-Leonard  | QC       | H1R1Z4      | (514) 725-2221 | (514) 725-1543                        |  |  |  |  |
| Johnston Chrysler Dodge     |  |        |                |          | _           | 1              |                                       |  |  |  |  |
| Jeep                        | 1350 Upper James St                                    |        | Hamilton       | ON       | L9C3B4      | (905) 388-5502 | (905) 575-1102                        |  |  |  |  |
| Kamloops Ford Lincoln       | 940 Halston Ave  |        | Kamloops       | BC       | V2B2B8      | (250) 376-7266 | Not Available                         |  |  |  |  |
| Kanata Ford                 | 8000 Campeau Dr  |        | Kanata         | ON       | K2T1C2      | (613) 591-9000 | (613) 591-9191                        |  |  |  |  |
| Kanata Honda                | 2500 Palladium Dr                                      | 800    | Kanata         | ON       | K2V1E2      | (613) 591-3311 | Not Available                         |  |  |  |  |
| Kelowna Infiniti Nissan     | 2570 Enterprise Way                                    |        | Kelowna        | BC       | V1X7X5      | (250) 712-0404 | (250) 712-0400                        |  |  |  |  |
| Ken Sargent Gmc Buick Ltd   | 12308 100 St   |        | Grande Prairie | AB       | T8V4H7      | (780) 532-8865 | Not Available                         |  |  |  |  |
| Ken Shaw Lexus              | 2336 St Clair Ave W                                    |        | Toronto        |          | M6N1K8      | (416) 766-0055 | Not Available                         |  |  |  |  |
| Ken Shaw Lexus Toyota       | 2336 St Clair Ave W                                    |        | Toronto        |          | M6N1K8      | (416) 766-1155 | (416) 766-6112                        |  |  |  |  |
| Kennedy Ford Sales Ltd      | 280 South Service Rd W                                 |        | Oakville       |          | L6K3X5      | (905) 845-1646 | Not Available                         |  |  |  |  |
| Kentwood Ford Sales Inc     | 13344 97 St NW   |        | Edmonton       | AB       | T5E4C9      | (780) 476-8600 | (780) 475-5522                        |  |  |  |  |
| Keystone Ford Sales         | 1300 Regent Ave W                                      |        | Winnipeg       | MB       | R2C3A8      | (204) 661-9555 | (204) 661-4347                        |  |  |  |  |
| Kia Canada Inc              | 180 Foster Cres  |        | Mississauga    | ON       | L5R4J5      | Not Available  | Not Available                         |  |  |  |  |
| Kia St Constant Inc         | 48 Rue Saint-Pierre                                    |        | Saint-Constant | QC       | J5A1B9      | (450) 632-0700 | Not Available                         |  |  |  |  |
|                             |  |        |                |          |             |                |                                       |  |  |  |  |
| Kia Trois-Rivieres          | 3790 Boul Gene-HKruger                                 |        | Trois-Rivieres | QC       | G9A4M3      | (819) 373-7300 | Not Available                         |  |  |  |  |
| Kings County Chrysler Ltd   | 5128 A A Macdonald Hwy                                 | AA     | Montague       | PE       | COA1RO      | (902) 892-9281 | Not Available                         |  |  |  |  |
| Kingston Dodge Chrysler     |  |        |                |          |             |                |                                       |  |  |  |  |
| Jeep                        | 1429 Princess St                                       |        | Kingston       | ON       | K7M3E9      | (613) 549-8900 | (613) 549-4606                        |  |  |  |  |
| Kingsway Toyota             | 12820 97 St NW   |        | Edmonton       | AB       | T5E4C3      | (780) 478-8300 | (780) 478-1025                        |  |  |  |  |
| Kipp Scott Gmc Cadillac     |  |        |                |          |             |                |                                       |  |  |  |  |
| Buick                       | 6841 50th Avenue                                       |        | Red Deer       | AB       | T4N4E2      | (403) 343-6633 | Not Available                         |  |  |  |  |
| Koch Ford Lincoln Sales Ltd | 5121 Gateway Blvd NW                                   |        | Edmonton       | AB       | T6H5W5      | (780) 434-8411 | (780) 435-5169                        |  |  |  |  |

|                                   | New Car D                    | ealers in Ca | anada with Annual S | ales Over \$50 | M              |                |                |
|-----------------------------------|------------------------------|--------------|---------------------|----------------|----------------|----------------|----------------|
|                                   |                              | Suite        |                     |                |                |                |                |
| Company Name                      | Address                      | Number       | City                | Province       | Postal Code    | Phone Number   | Fax Number     |
| Laird Wheaton Chev Buick          |                              |              |                     |                |                |                |                |
| Gmc                               | 2590 Bowen Road              | -            | Nanaimo             | BC             | V9T3L3         | (250) 758-1514 | Not Available  |
| Lakeland Ford Sales 2009 Ltd      | 3434 2nd Ave W               | 1            | Prince Albert       | SK             | S6V5G2         | (306) 764-3325 | (306) 764-3290 |
| Lakeshore Honda                   | 3526 Lake Shore Blvd W       |              | Etobicoke           | ON             | M8W1 <u>N6</u> | (416) 252-0066 | (416) 252-7477 |
| Lakewood Chevrolet Ltd            | 9150 34 Ave NW               |              | Edmonton            | AB             | T6E5P2         | (780) 462-5959 | (780) 461-5332 |
| Laking Toyota Inc                 | 695 Kingsway                 |              | Sudbury             | ON             | P3B2E4         | (705) 674-7534 | (705) 674-6176 |
| Lallier Ste-Foy                   | 2000 Rue Cyrille-Duquet      |              | Quebec              | QC             | G1N2E8         | (418) 687-2525 | (418) 687-3005 |
| Land Rover Metro West             | 3526 Lake Shore Blvd W       |              | Etobicoke           | ON             | M8W1N6         | (416) 252-1010 | (416) 252-8905 |
| Landau Ford Lincoln Sales         | 555 Empress St               |              | Winnipeg            | MB             | R3G3H1         | (204) 772-2411 | (204) 772-5198 |
| Landmark Mazda                    | 13145 97 St NW               |              | Edmonton            | AB             | T5E4C4         | (780) 476-6221 | Not Available  |
| Langley Chrysler Dodge Jeep       | 19418 Langley Bypass         |              | Surrey              | BC             | V3S7R2         | (604) 534-5355 | (604) 534-2800 |
| Langley Toyotatown                | 20622 Langley Bypass         |              | Langley             | BC             | V3A6K8         | (604) 530-3156 | (604) 530-1471 |
| Laquerre Chev Buick Gmc Inc       | 124 Rue Olivier              |              | Laurier-Station     | QC             | G0S1N0         | (418) 728-4233 | (418) 728-4307 |
| Le Relais Chev Cad Buick<br>Gmc   | 9411 Avenue Papineau         |              | Montreal            | QC             | H2M2G5         | (514) 384-6380 | (514) 384-5795 |
| Ledingham Gm                      | 200 Pth 12 N                 |              | Steinbach           | MB             | R5G1T6         | (204) 326-3451 | (204) 326-6486 |
| Leggat Chevrolet Cad Buick<br>Gmc | 4100 Harvester Rd            |              | Burlington          | ON             | L7L0C1         | (905) 632-4141 | Not Available  |
| Leveille Toyota                   | 1369 Montee Masson           |              | Terrebonne          | QC             | J6W6A6         | (450) 471-4117 | (450) 471-3996 |
| Lexus Canada                      | 1 Toyota Pl                  |              | Scarborough         | ON             | M1H1H9         | (416) 438-6535 | Not Available  |
| Lexus Of Calgary Svc              | 22 Heritage Meadows Rd<br>SE |              | Calgary             | АВ             | T2H3C1         | (403) 225-3987 | Not Available  |
| Lexus On The Park                 | 1075 Leslie St               |              | North York          | ON             | M3C2J7         | (416) 391-1600 | Not Available  |
| Lombardi Honda                    | 4356 Boul Metropolitain E    |              | Saint-Leonard       | QC             | H1S1A2         | (514) 728-2222 | (514) 728-1717 |
| Lombardi Honda                    | 560 Wharncliffe Rd S         | <del> </del> | London              |                | N6J2N4         | (519) 649-0889 | (519) 649-1811 |

|                             | New Car D                      | ealers in C | anada with Annual S | ales Over \$50 | M           | · · · · ·      |                |
|-----------------------------|--------------------------------|-------------|---------------------|----------------|-------------|----------------|----------------|
|                             |                                | Suite       |                     |                |             |                |                |
| Company Name                | Address                        | Number      | City                | Province       | Postal Code | Phone Number   | Fax Number     |
| Longue-Pointe Chrysler Ltee | 6200 Boul Metropolitain E      |             | Saint-Leonard       | QC             | H1S1A9      | (514) 256-5092 | (514) 259-6437 |
| Longueuil Honda             | 3551 Ch De Chambly             |             | Longueuil           | QC             | J4L4E2      | (450) 679-4710 | (450) 679-2759 |
| Lounsbury Automotive Ltd    | 2155 West Main Street          |             | Moncton             | NB             | E1C9P2      | (506) 857-4300 | (506) 857-4308 |
| Lussier Chevrolet Buick Gmc | 3000 Rue Dessaulles            |             | Saint-Hyacinthe     | QC             | J2S2V8      | (450) 778-1112 | (450) 778-1422 |
| Mac Donald Buick Gmc        | 111 Baig Blvd                  |             | Moncton             | NB             | E1E1C9      | (506) 853-6205 | (506) 853-6225 |
| Macmaster Chev Cad Buick    |                                |             |                     |                |             |                |                |
| Gmc                         | 1470 Dundas St                 |             | London              | ON             | N5W3B9      | (519) 455-6200 | Not Available  |
| Maison Chrysler Fiat        | 15070 Boul Henri-Bourassa      |             | Quebec              | QC             | G1G3Z4      | (418) 622-4700 | (418) 622-8467 |
| Maisonneuve Chevrolet       | 3700 Rue Sainte-Catherine<br>E |             | Montreal            | QC             | H1W2E8      | (514) 527-8733 | (514) 525-7198 |
| Maple Ridge Chryslr Jeep    |                                |             |                     |                |             |                |                |
| Dodge                       | 11911 West St                  |             | Maple Ridge         | BC             | V2X3M6      | (604) 465-8931 | Not Available  |
| Maple Toyota                | 88 Auto Vaughan Dr             |             | Maple               | ON             | L6A4A1      | (905) 417-9100 | (905) 417-4999 |
| Maranello Motors Ltd        | 55 Auto Park Cir               |             | Woodbridge          | ON             | L4L8R1      | (905) 663-8011 | Not Available  |
| Maranello Motors Ltd-Bmw    | 55 Auto Park Cir               |             | Woodbridge          | ON             | L4L8R1      | (416) 213-5699 | (905) 850-2741 |
| Marino's Fine Cars          | 3526 Lake Shore Blvd W         |             | Etobicoke           | ON             | M8W1N6      | (416) 252-2277 | Not Available  |
| Markham Honda               | 8220 Kennedy Rd                |             | Markham             | ON             | L3R5X3      | (905) 477-2451 | (905) 477-3063 |
| Markville Ford Lincoln Svt  | 8210 Kennedy Rd                |             | Markham             | ON             | L3R5X3      | (905) 474-1350 | Not Available  |
| Marlin Chevrolet Buick Gmc  |                                |             |                     |                |             |                |                |
| Inc                         | 2145 Rue Frank-Carrel          |             | Quebec              | QC             | G1N2G2      | (418) 688-1212 | (418) 688-1281 |
| Marvin Starr Pontiac        | 3132 Eglinton Ave E            |             | Scarborough         | ON             | M1J2H1      | (416) 264-3451 | (416) 264-4513 |
| Marvin Starr Pontiac Buick  |                                |             |                     |                |             |                |                |
| Cad                         | 3132 Eglinton Ave E            |             | Scarborough         | ON             | M1J2H1      | (416) 264-2311 | (416) 264-4513 |
| Mayfield Toyota Scion       | 10220 170 St NW                |             | Edmonton            | AB             | T5S1N9      | (780) 420-1111 | (780) 486-9014 |
| Mazda De Blainville         | 20900 Ch De La Cote N          |             | Mirabel             | QC             | J7J0E5      | (514) 875-3938 | (450) 433-7914 |

|                           | New Car I               | Dealers in Ca | anada with Annual S | ales Over \$50 | M           |                |                |
|---------------------------|-------------------------|---------------|---------------------|----------------|-------------|----------------|----------------|
|                           |                         | Suite         |                     |                |             |                |                |
| Company Name              | Address                 | Number        | City                | Province       | Postal Code | Phone Number   | Fax Number     |
| Mazda De Laval            | 2385 Boul Chomedey      |               | Laval               | QC             | H7T2W5      | (450) 688-4787 | (450) 688-8500 |
| Mazda Gabriel             | 5333 Rue Saint-Jacques  |               | Montreal            | QC             | H4A2C9      | (514) 484-7777 | (514) 484-6343 |
| Mc Curdy Chevrolet Buick  | 174 Trenton-Frankford   |               |                     |                |             |                |                |
| Gmc                       | Roadwy 3                |               | Trenton             | ON             | K8V5R6      | (613) 392-1245 | (613) 392-4689 |
| Mcnaught Cadillac Buick   |                         |               |                     |                |             |                |                |
| Gmc                       | 1717 Waverley St        | 1000          | Winnipeg            | MB             | R3T6A9      | (204) 786-3811 | (204) 775-4077 |
| Meadowvale Ford Sales &   |                         |               |                     |                |             |                |                |
| Svc                       | 2230 Battleford Rd      |               | Mississauga         | ON             | L5N3K6      | (905) 542-3673 | (905) 542-1787 |
| Mendes Toyota             | 1181 Bank St            |               | Ottawa              | ON             | K1S3X7      | (613) 523-8666 | Not Available  |
| Mercedes-Benz Burlington  | 441 North Service Rd    |               | Burlington          | ON             | L7P0A3      | (905) 632-4222 | Not Available  |
| Mercedes-Benz Canada Inc  | 849 Eglinton Ave E      |               | East York           | ON             | M4G2L5      | (416) 847-7400 | (416) 847-7473 |
| Mercedes-Benz Canada Inc  | 98 Vanderhoof Ave       |               | East York           | ON             | M4G4C9      | (416) 425-3550 | (416) 423-5027 |
| Mercedes-Benz Canada Inc  | 4844 Boul Taschereau    |               | Greenfield Park     | QC             | J4V2J2      | (450) 672-2720 | Not Available  |
| Mercedes-Benz Canada Inc  | 1325 Boundary Rd        |               | Vancouver           | BC             | V5K4T9      | (604) 639-1451 | Not Available  |
| Mercedes-Benz Canada Inc  | 1395 Broadway W         |               | Vancouver           | BC             | V6H1G9      | (604) 736-7411 | (604) 736-0313 |
| Mercedes-Benz Downtown    | 761 Dundas St E         |               | Toronto             | ON             | M5A4N5      | (416) 947-9000 | (416) 947-9085 |
| Mercedes-Benz Laval       | 3131 Nord Laval A-440 O |               | Laval               | QC             | H7P5P2      | (514) 856-2929 | Not Available  |
| Mercedes-Benz Maple       | 9300 Jane St            |               | Maple               | ON             | 1.6A0C5     | (905) 585-9300 | (905) 585-9301 |
| Merlin Ford Lincoln       | 3750 Idylwyld Dr N      |               | Saskatoon           | SK             | \$7L6G3     | (306) 931-6611 | Not Available  |
| Metro Chrysler Dodge Jeep | 1047 Richmond Rd        |               | Ottawa              | ON             | K2B6R1      | (613) 596-1006 | Not Available  |
| Metro Lexus Toyota        | 625 Frances Ave         |               | Victoria            | BC             | V8Z1A2      | (250) 386-3516 | Not Available  |
| Metro Mitsubishi          | 230 Wyse Rd             |               | Dartmouth           | NS             | B3A1M9      | (902) 463-6792 | (902) 465-7261 |
| Metro Motors Ltd          | 2505 Lougheed Hwy       |               | Port Coquitlam      | BC             | V3B1B2      | (604) 464-0271 | Not Available  |

•

.

|                            | New Car D                | ealers in C | anada with Annual Sa | ales Over \$50 | M           |                |                |
|----------------------------|--------------------------|-------------|----------------------|----------------|-------------|----------------|----------------|
|                            |                          | Suite       |                      |                |             |                |                |
| Company Name               | Address                  | Number      | City                 | Province       | Postal Code | Phone Number   | Fax Number     |
| Metro-Ford Sales Ltd       | 1111 9 Ave SW            |             | Calgary              | AB             | T2P1L3      | (403) 263-4530 | (403) 233-8195 |
| Mgm Ford Lincoln Sales     | 3010 50 Ave              |             | Red Deer             | AB             | T4R1M5      | (403) 346-6621 | Not Available  |
| Michael Boyer Pont Buick   |                          |             |                      |                |             | - <u> </u>     | 1              |
| Gmc                        | 715 Kingston Road        |             | Pickering            | ON             | L1V1A9      | (905) 831-2693 | Not Available  |
| Midtown Honda              | 3400 Dufferin St         |             | North York           | ON             | M6A2V1      | (416) 789-4101 | (416) 789-0641 |
| Mills Motors Buick Gmc Ltd | 240 Bond Street East     |             | Oshawa               | ON             | L1H7L3      | (905) 436-1500 | Not Available  |
| Mills Nissan Ltd           | 1275 101 St SW           |             | Edmonton             | AB             | T6X1A1      | (780) 463-5700 | Not Available  |
| Mini Crowfoot              | 650 Crowfoot Cres NW     |             | Calgary              | AB             | T3G4S3      | (403) 275-6464 | (403) 275-6431 |
| Mini Markham               | 8111 Kennedy Rd          |             | Markham              | ON             | L3R5M2      | (905) 513-8900 | (905) 513-0500 |
| Moncton Chrysler           |                          |             |                      |                |             |                |                |
| Superstore                 | 1365 Mountain Rd         |             | Moncton              | NB             | E1C2T9      | (506) 857-3939 | (506) 857-3902 |
| Montmorency Ford Ltd       | 7225 Boul Taschereau     | ·           | Brossard             | QC             | J4Y1A1      | (450) 678-9940 | Not Available  |
| Morrey Mazda               | 818 Automall Dr          |             | North Vancouver      | BC             | V7P3R8      | (604) 984-9299 | Not Available  |
| Murray Chev Cad Medicine   |                          |             |                      |                |             |                |                |
| Hat                        | 1270 Trans Canada Way SE |             | Medicine Hat         | AB             | T1B1J5      | (403) 527-1141 | (403) 526-8865 |
| Murray Chev Cad Pb Gmc     | 15 Chester Road          |             | Moose Jaw            | SK             | S6J1N3      | (306) 693-4605 | Not Available  |
| Murray Chevrolet Cadillac  | 30355 Automall Drive     |             | Abbotsford           | ВС             | V2T5M1      | (604) 857-5200 | (604) 857-5243 |
| Murray Chevrolet Cadillac  | 1500 Richmond Ave        |             | Brandon              | МВ             | R7A7E3      | (204) 728-0130 | (204) 726-8344 |
| Murray Chevrolet Ltd       | 1700 Waverley Street     |             | Winnipeg             | МВ             | R3T5V7      | (204) 261-6200 | (204) 261-1645 |
| Murray Motors Yarmouth     | · · · ·                  |             |                      |                |             |                |                |
| Ltd                        | 45 Starrs Road           |             | Yarmouth             | NS             | B5A2T2      | (902) 742-7191 | Not Available  |
| Myers Cadillac Chev Buick  |                          |             |                      |                |             |                | İ              |
| Gmc                        | 1200 Baseline Road       |             | Ottawa               | ON             | K2C0A6      | (613) 225-1260 | (613) 225-2231 |
| Myers Chevrolet Cadillac   | 1701 Woodroffe Ave       |             | Nepean               | ON             | K2G1W2      | (613) 228-7348 | Not Available  |
| Myers Kanata Inc           | 2500 Palladium Dr        | 200         | Kanata               | ON             | K2V1E2      | (613) 596-9221 | Not Available  |
| Newmarket Toyota           | 1171 Davis Dr            |             | Newmarket            | ON             | L3Y8R1      | (905) 953-2890 | (905) 953-2895 |
|                            | 1917 Boul Sir-Wilfrid-   |             |                      |                |             |                |                |
| Niquet Audi                | Laurier                  |             | Saint-Bruno          | QC             | J3V4P6      | (450) 653-2834 | (450) 653-1653 |

|                              | New Car I               | Dealers in Ca | anada with Annual | Sales Over \$50 | M           |                |                |
|------------------------------|-------------------------|---------------|-------------------|-----------------|-------------|----------------|----------------|
|                              |                         | Suite         |                   |                 |             |                |                |
| Company Name                 | Address                 | Number        | City              | Province        | Postal Code | Phone Number   | Fax Number     |
| Nissan-Cite                  | 3500 Rue Jean-Talon O   |               | Montreal          | QC              | H3R2E8      | (514) 739-3175 | (514) 735-1250 |
| North York Chrysler Jeep     |                         |               |                   |                 |             |                |                |
| Dodge                        | 7200 Yonge St           |               | North York        | ON              | M2M3V7      | (416) 221-5581 | (416) 221-4725 |
| Northgate Chevrolet Buick    |                         |               |                   |                 |             |                |                |
| Gmc                          | 13215 97th Street       |               | Edmonton          | AB              | T5E4C7      | (780) 476-3371 | (780) 476-3381 |
| Northstar Hyundai            | 14803 137th Ave         |               | Edmonton          | AB              | T5L4P3      | (780) 478-7669 | (780) 473-2437 |
| Northwest Toyota             | 2 Maritime Ontario Blvd |               | Brampton          | ON              | L6S0C2      | (905) 791-3500 | Not Available  |
| Number 7 Sales               | 555 Highway 7           |               | Woodbridge        | ON              | L4L1T5      | (905) 851-1130 | Not Available  |
| Nurse Chevrolet Cadillac     | 1520 Dundas St E        |               | Whitby            | ON              | L1N2K7      | (905) 430-2350 | (905) 668-7809 |
| Nurse Chevrolet Cadillac Ltd | 1530 Dundas Street East |               | Whitby            | ON              | L1N2K7      | (905) 686-2112 | Not Available  |
| O'Regan's South Shore        |                         |               | <i></i>           |                 |             | <u> </u>       |                |
| Toyota                       | 219 North St            |               | Bridgewater       | NS              | B4V2V7      | (902) 543-7168 | Not Available  |
| Oakland Ford Lincoln Sales   | 570 Trafalgar Rd        |               | Oakville          | ON              | L6J3J2      | (905) 844-3273 | (905) 844-4472 |
| Oakville Volkswagen          | 1355 North Service Rd W |               | Oakville          | ON              | L6M2W2      | (905) 844-3285 | (905) 844-2719 |
| Okotoks Chevrolet Buick      |                         |               |                   |                 |             |                |                |
| Gmc                          | 101 Northgate Drive     |               | Okotoks           | AB              | Т150Н9      | (403) 938-7874 | Not Available  |
| Olivier Ford                 | 4405 Ch De Chambly      | 1             | Saint-Hubert      | QC              | J3Y3M7      | (450) 616-8402 | Not Available  |
| Ontario Motor Sales Ltd      | 140 Bond Street West    |               | Oshawa            | ON              | L1H7L8      | (905) 725-6501 | Not Available  |
| Openroad Scion               | 13251 Smallwood Pl      |               | Richmond          | BC              | V6V1W8      | (604) 697-2466 | Not Available  |
| Openroad Toyota              | 13251 Smallwood Pl      |               | Richmond          | BC              | V6V1W8      | (604) 273-3766 | (604) 273-2892 |
| Orchard Ford Sales Ltd       | 2741 Highway 97 N       |               | Kelowna           | BC              | V1X4J8      | (250) 860-1000 | (250) 861-1870 |
| Otto's Subaru                | 225 Richmond Rd         | 225           | Ottawa            | ON              | K1Z6W7      | (613) 728-5813 | Not Available  |
| Owasco Audi                  | 2100 Champlain Ave      |               | Whitby            | ON              | L1N6A7      | (905) 579-0088 | Not Available  |
| Pacific Mazda                | 1060 Yates St           |               | Victoria          | BC              | V8V3M6      | (250) 385-1451 | Not Available  |
| Park Avenue Audi             | 9800 Boul Taschereau    |               | Brossard          | QC              | J4X0A4      | (450) 445-4811 | Not Available  |
| Park Avenue Toyota           | 8855 Boul Taschereau    |               | Brossard          | QC              | J4Y1A4      | (450) 445-0577 | Not Available  |
| Park Lane Chevrolet Cad Ltd  | 1290 London Road        |               | Sarnia            | ON              | N7S1P5      | (519) 541-8883 | Not Available  |

~

| - mm                       | New Car D                | ealers in C | anada with Annual Sa | les Over \$50 | M           |                |                |
|----------------------------|--------------------------|-------------|----------------------|---------------|-------------|----------------|----------------|
|                            |                          | Suite       |                      |               |             |                | 1              |
| Company Name               | Address                  | Number      | City                 | Province      | Postal Code | Phone Number   | Fax Number     |
| Parkside Ford Sales Ltd    | 2000 Main St             |             | Winnipeg             | MB            | R2V2B8      | (204) 339-2000 | (204) 339-5572 |
|                            |                          |             |                      |               |             |                |                |
| Parkway Buick Inc          | 9595 Rte Transcanadienne |             | Saint-Laurent        | QC            | H4S1T6      | (514) 333-7070 | (514) 333-8208 |
| Parkway Honda              | 1681 Eglinton Ave E      |             | North York           | ON            | M4A1J6      | (416) 752-6666 | (416) 752-2438 |
| Parkway The Ford           |                          |             |                      |               |             |                |                |
| Superstore                 | 455 King St N            |             | Waterloo             | ON            | N2J2Z5      | (519) 579-1100 | (519) 884-4129 |
| Pathway Hyundai            | 1375 Youville Dr         |             | Orleans              | ON            | K1C4R1      | (613) 837-4222 | (613) 837-7178 |
| Paul Albert Chev Buick Cad |                          |             |                      |               |             |                |                |
| Gmc                        | 870 Boulevard Talbot     |             | Chicoutimi           | QC            | G7H4B4      | (418) 696-4444 | Not Available  |
| Paul Sadlon Motors Inc     | 550 Bayfield Street      |             | Barrie               | ON            | L4M5A2      | (705) 726-1811 | Not Available  |
|                            |                          |             |                      |               |             |                |                |
| Pembina Dodge Chrysler Ltd | 300 Pembina Hwy          |             | Winnipeg             | МВ            | R3L2E2      | (204) 284-6650 | (204) 284-2325 |
| Peninsula Ford Lincoln     | 202392 Sunset Strip      |             | Owen Sound           | ON            | N4K5N7      | (519) 376-3252 | (519) 376-9559 |
| Performance Cars           | 371 Ontario St           |             | St Catharines        | ON            | L2R5L3      | (905) 323-6770 | (905) 684-6044 |
| Pie IX Dodge Chrysler      | 9350 Boul Pie-IX         |             | Montreal             | QC            | H1Z4E9      | (514) 327-9000 | Not Available  |
| Pike Wheaton Chevrolet     | 3110 Gaetz Avenue        |             | Red Deer             | AB            | T4R1M6      | (403) 347-3301 | Not Available  |
| Pinnacle Chrysler Jeep     |                          |             |                      |               |             |                |                |
| Dodge                      | 2300 Tecumseh Rd W       |             | Windsor              | ON            | N9B1W1      | (519) 256-2303 | Not Available  |
| Plaza Chev Cadillac Inc    | 3650 Boul Des Sources    |             | Dollard-Ormeaux      | QC            | H9B1Z9      | (514) 332-1673 | (514) 332-1205 |
| Portland Street Honda      | 36 Baker Dr              |             | Dartmouth            | NS            | B2W6K1      | (902) 435-3330 | (902) 435-9696 |
| Queenston Chevrolet        | 282 Centennial Pky N     |             | Hamilton             | ON            | L8E2X4      | (905) 560-2020 | (905) 560-3800 |
|                            |                          |             |                      |               |             |                |                |
|                            | 1306 The Queensway       |             | Etobicoke            | ON            | M8Z1S4      | (416) 259-7656 | (416) 259-1320 |
| Ray Cullen Chev Buick Gmc  |                          |             |                      |               |             |                |                |
| Ltd                        | 730 Wharncliffe Rd S     |             | London               | ON            | N6J2N4      | (519) 686-7282 | Not Available  |
| Red Deer Toyota            | 413 Lantern St           |             | Red Deer County      | AB            | T4E0A5      | (403) 343-3736 | Not Available  |
| Regina Motor Products      |                          |             |                      |               |             |                |                |
| (1970)                     | Albert St S @ Hwy 1 & 6  |             | Regina               | SK            | S4P3A8      | (306) 585-1770 | (306) 584-2339 |
| Regional Commercial Credit | 1310 Regent Ave W        |             | Winnipeg             | МВ            | R2C3A8      | (204) 669-5626 | (204) 661-8123 |
| Relais Toyota Inc          | 2059 King St W           |             | Sherbrooke           | QC            | J1J2E9      | (819) 823-7919 | (819) 564-1286 |

.....

|                              | New Car [                | Dealers in Ca | anada with Annual S | ales Over \$50 | M           |                |                |
|------------------------------|--------------------------|---------------|---------------------|----------------|-------------|----------------|----------------|
|                              |                          | Suite         |                     |                |             |                |                |
| Company Name                 | Address                  | Number        | City                | Province       | Postal Code | Phone Number   | Fax Number     |
| Richport Ford Lincoln        | 13580 Smallwood Pl       |               | Richmond            | BC             | V6V2C1      | (604) 273-7331 | (604) 273-4574 |
| Roadsport Honda              | 940 Ellesmere Rd         |               | Scarborough         | ON             | M1P2W8      | (416) 291-9501 | (416) 291-5091 |
| Ron Hodgson Chev Buick       |                          |               |                     |                |             |                |                |
| Gmc Ltd                      | 5 Galarneau Place        |               | St. Albert          | AB             | T8N2Y3      | (780) 458-7100 | (780) 459-4450 |
| Rond Point Dodge &           |                          |               |                     |                |             |                |                |
| Chrysler Lt                  | 5055 Boul Cousineau      |               | Saint-Hubert        | QC             | J3Y3K7      | (450) 656-4110 | (450) 656-1284 |
| Roy Foss Motors Ltd          | 7200 Yonge Street        | 1             | Thornhill           | ON             | L4J1V8      | (905) 886-2002 | Not Available  |
| Roy Foss Vaughan             | 2 Auto Park Cir          |               | Woodbridge          | ON             | L4L8R1      | (416) 798-4875 | Not Available  |
| Royal Chevrolet-Cadillac Inc | 1 Morona Park Dr         |               | Orangeville         | ON             | L9W0E1      | (519) 941-0420 | Not Available  |
| Royfoss Motors               | 7200 Yonge St            | -             | Thornhill           | ON             | L4J1V8      | (905) 884-4481 | (905) 709-7640 |
| Scarboro Town Chrysler       |                          | İ             |                     |                |             |                |                |
| Dodge                        | 4960 Sheppard E          |               | Scarborough         | ON             | M1B1W4      | (416) 751-7757 | Not Available  |
| Scarborotown Dodge           |                          |               |                     |                |             |                |                |
| Chrysler                     | 4960 Sheppard Ave E      |               | Scarborough         | ON             | M1S4A7      | (416) 298-7600 | (416) 298-9331 |
| Scarborough Nissan Ltd       | 1941 Eglinton Ave E      |               | Scarborough         | ON             | M1L2M4      | (416) 751-3511 | Not Available  |
| Scarborough Toyota Scion     | 1897 Eglinton Ave E      |               | Scarborough         | ON             | M1L2L6      | (416) 751-1530 | Not Available  |
| Scherer Chevrolet Buick Gmc  | 1225 Courtland Avenue E  |               | Kitchener           | ON             | N2C2N8      | (519) 893-8888 | (519) 893-8384 |
| Scion Calgary                | 20 Freeport Dr NE        |               | Calgary             | AB             | T3J5E3      | (403) 277-2466 | Not Available  |
| Scion Town                   | 1065 Wharncliffe Rd S    |               | London              | ON             | N6L1J9      | (519) 680-1400 | Not Available  |
| Shaganappi Motors (1976)     |                          |               |                     |                |             |                |                |
| Ltd                          | 4720 Crowchild Trail N W |               | Calgary             | AB             | T3A2N2      | (403) 288-0444 | Not Available  |
| Shanahan Ford Sales          | 18001 Leslie St          |               | Newmarket           | ON             | L3Y9A4      | (416) 798-4858 | (905) 853-3004 |
| Shaw Gmc Chevrolet Buick     |                          |               |                     |                |             | 1              |                |
| Ltd                          | 4620 Blackfoot Trail SE  |               | Calgary             | AB             | T2G4G2      | (403) 243-4200 | Not Available  |
| Sherbrooke Honda             | 2615 Rue King O          |               | Sherbrooke          | QC             | J1J2H3      | (819) 566-5322 | (819) 566-8868 |
| Sherwood Buick Gmc           | 10 Automall Road         |               | Sherwood Park       | AB             | T8H2N1      | (780) 464-5123 | (780) 467-5851 |
| Sherwood Chevrolet           | 550 Brand Rd             |               | Saskatoon           | SK             | S7J5J3      | (306) 374-6330 | Not Available  |
| Sherwood Ford                | 2540 Broadmoor Blvd      |               | Sherwood Park       | AB             | T8H1B4      | (780) 449-3673 | Not Available  |

а

| New Car Dealers in Canada with Annual Sales Over \$50M |                        |        |                 |          |             |                |                |  |  |  |
|--|------------------------|--------|-----------------|----------|-------------|----------------|----------------|--|--|--|
| · · ···  |                        | Suite  |                 | 1        |             | 1              | T              |  |  |  |
| Company Name   | Address                | Number | City            | Province | Postal Code | Phone Number   | Fax Number     |  |  |  |
| Sherwood Nissan  | 10 Broadway Blvd       |        | Sherwood Park   | AB       | T8H2A2      | (780) 449-5775 | (780) 449-5799 |  |  |  |
| Sherwood Park Toyota                                   | 31 Automall Rd         |        | Sherwood Park   | AB       | T8H0C7      | (780) 410-2455 | (780) 410-2456 |  |  |  |
| Sherwood Pk Dodge Chryslr                              |                        |        |                 |          |             |                |                |  |  |  |
| Jeep   | 230 Provincial Ave     |        | Sherwood Park   | AB       | T8H0E1      | (780) 410-4100 | (780) 410-4116 |  |  |  |
| Silverhill Acura                                       | 5811 3 St SE           |        | Calgary         | AB       | T2H1K1      | (403) 252-7725 | (403) 253-0077 |  |  |  |
| Sisley Motors Inc                                      | 88 Steeles Ave W       |        | Thornhill       | ON       | L4J1A1      | (416) 223-3131 | (905) 695-8880 |  |  |  |
| South Fort Chevrolet Ltd                               | 10109 89 Ave           |        | Ft Saskatchewan | AB       | T8L3V5      | (780) 467-3400 | Not Available  |  |  |  |
| Southbank Dodge Chrysler                               | 1255 Johnston Rd       |        | Ottawa          | ON       | K1V8Z1      | (613) 731-1974 | (613) 731-7274 |  |  |  |
| Southgate Buick Gmc Ltd                                | 9751 34th Avenue       |        | Edmonton        | AB       | T6E5X9      | (780) 435-4000 | (780) 435-7662 |  |  |  |
| Southgate Volkswagen                                   | 1223 101 St SW         |        | Edmonton        | AB       | T6X1A1      | (780) 435-4821 | Not Available  |  |  |  |
| Southside Dodge Chrysler                               |                        |        |                 |          |             | -              |                |  |  |  |
| Jeep   | 2804 50 Ave            |        | Red Deer        | AB       | T4R1M4      | (403) 346-5577 | (403) 346-6466 |  |  |  |
| Southtown Chrysler Jeep                                |                        |        |                 |          |             |                |                |  |  |  |
| Dodge  | 4404 66 St NW          |        | Edmonton        | AB       | T6K4E7      | (780) 490-3200 | (780) 490-3220 |  |  |  |
| Spinelli Infiniti                                      | 345 Boul Brunswick     |        | Pointe-Claire   | QC       | H9R4S1      | (514) 697-5222 | Not Available  |  |  |  |
| Spinelli Lexus Toyota                                  | 561 Boul Saint-Joseph  |        | Lachine         | QC       | H8S2K9      | (514) 482-5423 | Not Available  |  |  |  |
| Spinelli Nissan  | 345 Boul Brunswick     |        | Pointe-Claire   | QC       | H9R4S1      | (514) 697-9141 | (514) 697-9404 |  |  |  |
| Spinelli Toyota Pointe Claire                          | 12 Auto Plaza Ave      |        | Pointe-Claire   | QC       | H9R3H9      | (514) 694-1510 | (514) 694-6262 |  |  |  |
| St James Volkswagen Audi                               | 670 Century St         |        | Winnipeg        | МВ       | R3H0A1      | (204) 788-1100 | (204) 788-1109 |  |  |  |
| St-Jerome Chevrolet Buick                              |                        |        |                 |          |             |                |                |  |  |  |
| Gmc  | 265 Rue John-F-Kennedy |        | Saint-Jerome    | QC       | J7Y4B5      | (450) 438-1203 | Not Available  |  |  |  |
| Stampede Toyota  | 2508 24 Ave NE         |        | Calgary         | AB       | T1Y6R8      | (403) 291-2111 | Not Available  |  |  |  |
|  | 2777 Boul Du Versant-  |        |                 |          |             |                |                |  |  |  |
| Ste-Foy Toyota   | Nord                   |        | Quebec          | QC       | G1V1A4      | (418) 658-1340 | Not Available  |  |  |  |
| Ste-Marie Automobiles Ltee                             | 540 Rue Notre-Dame     |        | Saint-Remi      | QC       | JOL2LO      | (450) 454-4614 | Not Available  |  |  |  |
| Steele Chevrolet Buick Gmc<br>Cad                      | 636 Portland Street    |        | Dartmouth       | NS       | B2W2M3      | (902) 434-4100 | Not Available  |  |  |  |

.

.

|                                       | New Car                | Dealers in Ca | anada with Annual S | ales Over \$50 | M           |                |                |
|---------------------------------------|------------------------|---------------|---------------------|----------------|-------------|----------------|----------------|
| · · · · · · · · · · · · · · · · · · · |                        | Suite         |                     |                | ·           |                |                |
| Company Name                          | Address                | Number        | City                | Province       | Postal Code | Phone Number   | Fax Number     |
| Steele Dodge Chryslr Jeep             |                        | · · · · ·     |                     |                |             |                |                |
| Plym                                  | 6050 Livingstone St    |               | Halifax             | NS             | B3K2C6      | (902) 454-7341 | (902) 454-5490 |
| Steele Ford Lincoln                   | 3773 Windsor St        |               | Halifax             | NS             | B3K0A2      | (902) 453-1130 | (902) 453-4628 |
| Suburban Motors Ltd                   | 3377 Douglas St        | 1             | Victoria            | BC             | V8Z3L4      | (250) 475-2255 | (250) 475-2256 |
| Summit Ford Sales Ltd                 | 12 Carrier Dr          |               | Etobicoke           | ON             | M9V2C1      | (416) 741-6221 | (416) 741-2874 |
| Surgenor Chev Buick Gmc               | ( <u> </u>             |               |                     |                |             |                |                |
| Cad                                   | 939 St Laurent Blvd    |               | Ottawa              | ON             | К1КЗВ1      | (613) 741-0741 | (613) 741-1313 |
| Surrey Honda                          | 15291 Fraser Hwy       |               | Surrey              | BC             | V3R3P3      | (604) 583-7421 | (604) 583-0281 |
| Taylor Lexus Toyota                   | 655 Broad St           |               | Regina              | SK             | S4R1X5      | (306) 347-2400 | Not Available  |
| Team Ford Sales Ltd                   | 3304 91 St NW          |               | Edmonton            | AB             | T6N1C1      | (780) 462-8300 | (780) 461-0055 |
| Terra Nova Motors Ltd                 | 595 Kenmount Road      |               | St. John's          | NL             | A1B3P9      | (709) 364-4130 | (709) 364-7705 |
|                                       |                        |               |                     |                |             |                |                |
| Terrace Ford Lincoln Sales            | 900 Walker's Line      |               | Burlington          | ON             | L7N2G2      | (905) 632-6252 | (905) 632-1876 |
| Terrebonne Ford Inc                   | 2730 Chemin Gascon     |               | Terrebonne          | QC             | J6X4H6      | (450) 654-3322 | Not Available  |
| Thibault Chev Cad Buick               |                        |               |                     |                |             |                |                |
| Gmc                                   | 3839 Rue King Ouest    |               | Sherbrooke          | QC             | J1L1W7      | (819) 563-7878 | Not Available  |
| Thorncrest Sherway Ford               |                        |               |                     |                |             |                |                |
| Sales                                 | 1575 The Queensway     |               | Etobicoke           | ON             | M8Z1T8      | (416) 521-7000 | (416) 521-7070 |
|                                       |                        |               |                     |                |             |                |                |
| Tony Graham Scion Toyota              | 1855 Merivale Rd       |               | Nepean              | ON             | K2G1E3      | (613) 225-1212 | (613) 225-8603 |
|                                       |                        |               |                     |                |             |                |                |
| Toronto Dodge Jeep Chrysler           | 321 Front St E         |               | Toronto             | ON             | M5A1G3      | (416) 368-7000 | (416) 368-6454 |
| Toronto Honda                         | 2300 Danforth Ave      |               | East York           | ON             | M4C1K6      | (416) 423-2300 | (416) 423-4579 |
| Tower Chrysler Dodge Jeep             |                        |               |                     |                |             |                |                |
| Ram                                   | 10901 Macleod Trail SW |               | Calgary             | AB             | T2J4L3      | (403) 278-2020 | Not Available  |
| Toyota Canada Inc                     | 1 Toyota Pl            |               | Scarborough         | ON             | M1H1H9      | (416) 438-6320 | (416) 431-1867 |
| Toyota Credit Canada                  | 130 Thornhill Dr       |               | Dartmouth           | NS             | B3B1S3      | (902) 468-3473 | Not Available  |
| Toyota Drummondville                  | 10 Rue Cormier         |               | Drummondville       | QC             | J2COL4      | (819) 477-1777 | Not Available  |
|                                       | 3333 Ch De La Cote-De- |               |                     |                |             |                |                |
| Toyota Gabriel                        | Liesse                 |               | Saint-Laurent       | QC             | H4N3C2      | (514) 748-7777 | (514) 747-3267 |
| Toyota Heffner                        | 3121 King St E         |               | Kitchener           | ON             | N2A1B1      | (519) 748-9668 | Not Available  |

~

÷

|                            | New Car I               | Dealers in Ca | anada with Annual | Sales Over \$50 | )M          | . <u></u>                   |                |
|----------------------------|-------------------------|---------------|-------------------|-----------------|-------------|-----------------------------|----------------|
|                            |                         | Suite         |                   |                 |             | <u> </u>                    | 1              |
| Company Name               | Address                 | Number        | City              | Province        | Postal Code | Phone Number                | Fax Number     |
| Toyota Plaza Ltd           | 73 Kenmount Rd          |               | St. John's        | NL              | A1B3P8      | (709) 753-4051              | Not Available  |
| Toyotatown London          | 1065 Wharncliffe Rd S   |               | London            | ON              | N6L1J9      | (519) 680-1800              | (519) 686-0153 |
| Triple Seven Dodge         | 700 Broad St            |               | Regina            | SK              | S4R8H7      | (306) 522-2222              | (306) 525-4400 |
| Trois-Rivieres Honda       | 3115 Boul Saint-Jean    |               | Trois-Rivieres    | QC              | G9B2M3      | (819) 377-7500              | (819) 377-4775 |
| True North                 | 1370 Seymour St         |               | North Bay         | ON              | P1B9V6      | (705) 995-9820              | Not Available  |
| Tusket Sales & Svc Ltd     | 4143 Gavel Rd           |               | Tusket            | NS              | B0W3M0      | (902) 648-2600              | (902) 648-3268 |
| Universal Ford Lincoln     | 2800 Barlow Trail NE    |               | Calgary           | AB              | T1Y1A2      | (403) 291-2800              | Not Available  |
| Vandusen Chevrolet Buick   |                         |               |                   |                 |             |                             |                |
| Gmc                        | 425 Bayly Street W      |               | Ajax              | ON              | L1S6M3      | (905) 427-2500              | (905) 427-5496 |
| Vaughan Chrysler           | 1 Auto Park Cir         |               | Woodbridge        | ON              | L4L8R1      | (905) 850-7879              | (905) 850-8401 |
| Vickar Community Chevrolet |                         |               |                   |                 |             |                             |                |
| Ltd                        | 964 Regent Avenue West  |               | Winnipeg          | МВ              | R2C3A8      | (204) 661-8391              | (204) 661-3976 |
| Villa Toyota Scion         | 1235 Boul Saint-Joseph  |               | Gatineau          | QC              | J8Z3J6      | (819) 776-0077              | Not Available  |
|                            |                         |               |                   |                 |             |                             |                |
| Volkswagen Group Canada    | 777 Bayly St W          |               | Ajax              | ON              | L1S7G7      | (905) 428-6700              | (905) 428-5898 |
| Volvo Laval                | 2350 Boul Chomedey      |               | Laval             | QC              | H7T2W3      | (450) 682-3336              | Not Available  |
| Volvo Metro West           | 5450 Dundas St W        |               | Etobicoke         | ON              | M9B1B4      | (416) 231-1984              | (416) 231-0638 |
| Waterloo Ford Lincoln      | 11420 107 Ave NW        |               | Edmonton          | AB              | T5H0Y5      | (780) 423-4330              | Not Available  |
| Watrous Mainline Motor     |                         |               |                   |                 |             |                             |                |
| Prods                      | 208 1st Ave E           |               | Watrous           | SK              | SOK4TO      | (306) 946-3336              | (306) 946-2229 |
| Waverley Chrysler Dodge    |                         |               |                   |                 |             |                             |                |
| Jeep                       | 1700 Waverley St        | А             | Winnipeg          | MB              | R3T5V7      | (204) 661-5337              | Not Available  |
|                            |                         |               |                   |                 |             |                             |                |
| Wellington Motors Chrysler | 935 Woodlawn Rd W       |               | Guelph            | ON              | N1K1B7      | (519) 651-2422              | Not Available  |
| Wendell Motor Sales Ltd    | 549 Fairway Rd S        |               | Kitchener         | ON              | N2C1X4      | (519) 893-1501              | (519) 893-1579 |
| Westend Hyundai            | 3526 Lake Shore Blvd W  |               | Etobicoke         | ON              | M8W1N6      | (416) 259-8401              | (416) 252-7477 |
|                            | 18325 Stony Plain Road  |               |                   |                 |             |                             |                |
| Western Gmc Buick Ltd      | NW                      |               | Edmonton          | AB              | T5S1C6      | (780) 48 <del>6</del> -3333 | Not Available  |
| Westgate Chevrolet Ltd     | 10145 178 St NW         |               | Edmonton          | AB              | T5S1E4      | (780) 483-3320              | (780) 483-8068 |
| Westgate Honda             | 680 Oxford St W         |               | London            | ON              | N6H1T9      | (519) 472-4890              | (519) 472-9165 |
| Wheaton Chevrolet Ltd      | 260 Albert Street North |               | Regina            | SK              | S4R3C1      | (306) 543-1555              | (306) 949-8802 |

~

| · · · · · · · · · · · · · · · · · · · | New Car              | Dealers in Ca   | anada with Annual S | ales Over \$50 | M           |                        | -              |
|---------------------------------------|----------------------|-----------------|---------------------|----------------|-------------|------------------------|----------------|
| Company Name                          | Address              | Suite<br>Number | City                | Province       | Postal Code | Phone Number           | Fax Number     |
| Wheaton Gmc Buick Cadillac            |                      |                 |                     |                |             |                        |                |
| Ltd                                   | 2102 Millar Avenue   |                 | Saskatoon           | SK             | S7K6P4      | (306) 244-8131         | Not Available  |
| Wheaton Honda                         | 9688 34 Ave NW       |                 | Edmonton            | AB             | T6E6S9      | (780) 463-7888         | (780) 463-2031 |
| Whitby Toyota                         | 1025 Dundas St W     |                 | Whitby              | ON             | L1P1Z1      | (905) 668-4792         | Not Available  |
| Whitby-Oshawa Honda                   | 300 Thickson Rd S    |                 | Whitby              | ON             | L1N9Z1      | (905) 666- <u>1772</u> | (905) 666-4571 |
| Whiteoak Ford Lincoln Sales           | 3285 Mavis Rd        |                 | Mississauga         | ON             | L5C1T7      | (905) 270-8210         | Not Available  |
| Wietzes Motors                        | 7080 Dufferin St     |                 | Concord             | ON             | L4K0A1      | (905) 761-5133         | Not Available  |
| Williamson Buick Gmc                  | 259 Toronto St S     |                 | Uxbridge            | ON             | L9P1R1      | (905) 852-3357         | (905) 852-5332 |
| Williamson Chevrolet                  | 259 Toronto St S     |                 | Uxbridge            | ON             | L9P1R1      | (905) 852-3332         | (905) 852-5332 |
| Williamson Uxbridge                   | 259 Toronto Street S |                 | Uxbridge            | ON             | L9P1S3      | (905) 852-3331         | (905) 852-5332 |
| Williamson Uxbridge Gmc               | 259 Toronto St S     |                 | Uxbridge            | ON             | L9P1R1      | (905) 852-3337         | (905) 852-3336 |
| Willowdale Nissan Infiniti Ltd        | 7200 Yonge St        |                 | Thornhill           | ON             | L4J1V8      | (905) 881-4200         | Not Available  |
| Willowdale Nissan Ltd                 | 7200 Yonge St        |                 | Thornhill           | ON             | L4J1V8      | (416) 226-1116         | Not Available  |
| Wilson-Niblett Motors Ltd             | 10675 Yonge Street   |                 | Richmond Hill       | ON             | L4C3E1      | (905) 884-0991         | (905) 884-0033 |
| Winnipeg Dodge Chrysler               |                      |                 |                     |                |             |                        |                |
| Jeep                                  | 3965 Portage Ave     | 90              | Winnipeg            | MB             | R3K2H3      | (204) 772-2600         | Not Available  |
| Wood Wheaton Chev Cad                 |                      | 1               |                     |                |             |                        |                |
| Buick                                 | 2879 Highway 16 West |                 | Prince George       | BC             | V2N0A3      | (250) 564-4466         | (250) 562-0327 |
| Woodchester Infiniti                  | 2560 Motorway Blvd   |                 | Mississauga         | ON_            | L5L1X3      | (905) 820-4545         | (905) 828-1847 |
| Woodchester Nissan Inc                | 2560 Motorway Blvd   |                 | Mississauga         | ON             | L5L1X3      | (905) 828-7001         | (905) 828-1847 |
| Yonge Ford Lincoln                    | 7120 Yonge St        |                 | Thornhill           | ON             | L4J1V8      | (905) 731-8366         | (905) 764-4118 |
| Zender Ford Sales Ltd                 | 99 Golden Spike Rd   |                 | Spruce Grove        | AB             | T7X2Y3      | (780) 962-3000         | (780) 962-6943 |
| Zimmer Wheatong Gmc<br>Buick          | 685 Notre Dame Dr    |                 | Kamloops            | BC             | V2C5N7      | (250) 571-5930         | Not Available  |

--

|                                |  | Car Renta | al Agencies in Cana | da       |             |                | · · · · · · · · · · · · · · · · · · · |
|--------------------------------|--|-----------|---------------------|----------|-------------|----------------|---------------------------------------|
|                                |  | Suite     |                     |          |             |                |                                       |
| Company Name                   | Address                                    | Number    | City                | Province | Postal Code | Phone Number   | Fax Number                            |
| 4 A's Car & Truck Rental       | 808 Britannia Rd W                         | 111       | Mississauga         | ON       | L5V0A7      | (905) 813-8100 | (905) 813-8100                        |
| A Dollar Rent A Car            | 1380 Sargent Ave                           |           | Winnipeg            | MB       | R3E0G5      | (204) 949-3770 | (204) 775-5640                        |
| A Value Car & Truck Rentals    | 1600 Bayly St                              |           | Pickering           | ON       | L1W3N2      | (905) 421-9777 | Not Available                         |
| A Zoom Car & Truck Rentals Inc | 5835 Dixie Rd                              | 3         | Mississauga         | ON       | L4W4V7      | (905) 670-7368 | Not Available                         |
| ABC Car & Truck Rentals        | 2040 Dundas St                             |           | London              | ON       | N5V1R2      | (519) 659-2177 | (519) 659-7207                        |
| ABC Car & Truck Rentals        | 2001 Ontario St E                          |           | Stratford           | ON       | N5A6S5      | (519) 273-3116 | (519) 273-6876                        |
| ABC Rent-A-Car                 | 255 Broadway W                             |           | Vancouver           | BC       | V5Y1P5      | (604) 873-6622 | (604) 879-7447                        |
| Advantage Car & Truck Rentals  | 1270 Matheson Blvd E                       |           | Mississauga         | ON       | L4W1R2      | (905) 238-3432 | Not Available                         |
| Advantage Car & Truck Rentals  | 1380 Matheson Blvd E                       | 2         | Mississauga         | ON       | L4W4P8      | (905) 678-2719 | (905) 238-9655                        |
| Advantage Car Rentals Ltd      | 1335 Dalhousie Dr                          | 1         | Kamloops            | BC       | V2C5P6      | (250) 377-3003 | (250) 374-9171                        |
| Advantage Rent-A-Car           | 4730 Sheppard Ave E                        |           | Scarborough         | ON       | M1S3V6      | (416) 298-9911 | (416) 298-5769                        |
| Affinity Car & Truck Rental    | 1110 Dupont St                             |           | Toronto             | ON       | M6H2A2      | (416) 588-3936 | (416) 588-4987                        |
| Affinity Car Luxury Car Rntls  | 7115 Yonge St                              |           | Thornhill           | ON       | L3T2A8      | (905) 678-7368 | Not Available                         |
| Affinity Car Rental            | 1110 Dupont St                             |           | Toronto             | ON       | M6H2A2      | (416) 724-0724 | (416) 588-4987                        |
| Affordable Auto & Atv Rntls    | 9904 121 Ave                               |           | Grande Prairie      | AB       | T8V4J1      | (780) 539-5073 | (780) 532-6454                        |
| Alamo Rent-A-Car               | 2000 Airport Rd NE                         |           | Calgary             | AB       | T2E6W5      | (403) 543-3985 | (403) 543-3990                        |
| Alamo Rent-A-Car               | 975 Romeo Vachon                           |           | Dorval              | QC       | H4Y1H1      | (514) 633-1222 | Not Available                         |
| Alamo Rent-A-Car               | 926 King St E                              |           | Kitchener           | ON       | N2G2M8      | (519) 742-4377 | Not Available                         |
| Alamo Rent-A-Car               | 936 King St W                              |           | Kitchener           | ON       | N2G1G4      | (519) 742-2713 | (519) 742-0502                        |
| Alamo Rent-A-Car               | 1206 Rue Stanley                           |           | Montreal            | QC       | H3B2S7      | (514) 875-9988 | (514) 633-1059                        |
| Alamo Rent-A-Car               | 405 11 Ave                                 |           | Nisku               | AB       | T9E7N2      | (780) 955-8280 | Not Available                         |
|                                | 1003 Airport Parkway                       |           |                     |          |             |                |                                       |
| Alamo Rent-A-Car               | Pvt  |           | Ottawa              | ON       | K1V2C1      | (613) 737-7264 | Not Available                         |
|                                | Vancouver Internatl<br>Airport, 3211 Grant |           |                     |          |             |                |                                       |
| Alamo Rent-A-Car               | Mcconachie Way                             |           | Richmond            | вс       | V7B0A4      | (604) 231-1400 | (604) 821-2970                        |

~

|                          |  | Car Renta | al Agencies in Can | ada      |             |                |                |
|--------------------------|--|-----------|--------------------|----------|-------------|----------------|----------------|
|                          |  | Suite     |                    |          |             |                |                |
| Company Name             | Address                                      | Number    | City               | Province | Postal Code | Phone Number   | Fax Number     |
| Alamo Rent-A-Car         | 200 Wellington St W                          | 1200      | Toronto            | ON       | M5J1E6      | (416) 364-6748 | Not Available  |
| Alamo Rent-A-Car         | 930 Yonge St                                 |           | Toronto            | ON       | M4W2J2      | (416) 426-2964 | Not Available  |
| Alamo Rent-A-Car         | 2000 Wellington Ave                          |           | Winnipeg           | MB       | R3H1C1      | (204) 925-3448 | (204) 772-6510 |
| All West Rv Rentals      | 4053 208 St                                  |           | Langley            | BC       | V3A2H3      | (604) 514-2668 | Not Available  |
| A-Plus Rent-A-Car Co Ltd | 13340 Smallwood Pl                           | 102       | Richmond           | BC       | V6V1W8      | (604) 278-2112 | (604) 278-2115 |
| Autorentalcanada.Com     | 5700 Rue D'Iberville                         | В         | Montreal           | QC       | H2G2B3      | (514) 990-8217 | Not Available  |
| Avis Rent A Car          | 425 Dunlop St W                              |           | Barrie             | ON       | L4N1C3      | (705) 726-6527 | Not Available  |
| Avis Rent A Car          | 160 Charing Cross St                         |           | Brantford          | ON       | N3R2J4      | (519) 756-8857 | Not Available  |
| Avis Rent A Car          | 777 Av Aviation                              |           | Dieppe             | NB       | E1A7Z5      | (506) 855-7212 | Not Available  |
|                          | 575 Boul Albert-De                           |           |                    |          |             |                |                |
| Avis Rent A Car          | Niverville                                   |           | Dorval             | QC       | H4Y1J3      | (514) 636-1920 | (514) 636-3338 |
|                          | 975 Boul Romeo-Vachon                        |           |                    |          |             |                |                |
| Avis Rent A Car          | N  | CT6       | Dorval             | QC       | H4Y1H2      | (514) 636-1902 | (514) 636-3338 |
| Avis Rent A Car          | 8th Ave & Service Rd                         |           | Edmonton           | AB       |             | (780) 890-7596 | (780) 890-7004 |
| Avis Rent A Car          | 1 Bell Blvd                                  | <u>_</u>  | Enfield            | NS       | B2T1K2      | (902) 429-0963 | Not Available  |
| Avis Rent A Car          | 5481 Dundas St W                             |           | Etobicoke          | ON       | M9B1B5      | (416) 233-2847 | Not Available  |
| Avis Rent A Car          | 626 Dixon Rd                                 |           | Etobicoke          | ON       | M9W1J1      | (416) 245-2527 | Not Available  |
| Avis Rent A Car          | 5 Portage Ave                                |           | Fort Smith         | NT       | X0E0P0      | (867) 872-2211 | (867) 872-5001 |
| Avis Rent A Car          | 551 King St                                  |           | Fredericton        | NB       | E3B1E7      | (506) 454-2847 | (506) 450-4324 |
| Avis Rent A Car          | 5 Churchill St                               |           | Goose Bay          | NF       | AOP 1EO     | (709) 896-5923 | Not Available  |
| Avis Rent A Car          | 42 Gordon St                                 |           | Guelph             | ON       | N1H4H3      | (519) 836-1338 | Not Available  |
| Avis Rent A Car          | 414 Connaught Dr                             |           | Jasper             | AB       | TOE1E0      | (780) 852-3970 | (780) 852-9663 |
| Avis Rent A Car          | 38 Edgewater St                              |           | Kanata             | ON       | K2L1V8      | (613) 831-3380 | Not Available  |
| Avis Rent A Car          | 5533 Airport Way                             | 12        | Kelowna            | BC       | V1V1S1      | (250) 554-3888 | Not Available  |
| Avis Rent A Car          | 1114 Len Birchall Way                        |           | Kingston           | ON       | K7M9A1      | (613) 389-2228 | Not Available  |
| Avis Rent A Car          | 1412 Princess St                             | 9         | Kingston           | ON       | K7M3E5      | (613) 531-3311 | (613) 530-3311 |
| Avis Rent A Car          | 696 Fairway Rd S                             |           | Kitchener          | ON       | N2C1X3      | (519) 893-1110 | (519) 896-1202 |
|                          | Aeroport Montreal-<br>Trudeau, 975 Boulevard |           |                    |          |             |                |                |
| Avis Rent A Car          | Romeo Vachon                                 |           | Montreal           | QC       | H4Y 1H1     | (514) 636-1902 | Not Available  |

|                 |                         | Car Renta | al Agencies in Cana | ada      |             |                |                |
|-----------------|-------------------------|-----------|---------------------|----------|-------------|----------------|----------------|
|                 |                         | Suite     |                     |          |             | [              |                |
| Company Name    | Address                 | Number    | City                | Province | Postal Code | Phone Number   | Fax Number     |
| Avis Rent A Car | 205 York St             | 1         | London              | ON       | N6A1B1      | (519) 672-2847 | Not Available  |
| Avis Rent A Car | 290 Horton St           |           | London              | ON       | N6B1L4      | (519) 672-2847 | Not Available  |
| Avis Rent A Car | 8500 Warden Ave         |           | Markham             | ON       | L6G1A5      | (905) 305-0004 | Not Available  |
| Avis Rent A Car | 305 Steeles Ave E       |           | Milton              | ON       | L9T1Y2      | (905) 875-0161 | Not Available  |
| Avis Rent A Car | 5990 Airport            |           | Mississauga         | ON       | L5P 1A2     | (905) 672-5318 | Not Available  |
| Avis Rent A Car | 6400 Millcreek Dr       |           | Mississauga         | ON       | L5N3E7      | (905) 813-7265 | Not Available  |
| Avis Rent A Car | 1225 Rue Metcalfe       |           | Montreal            | QC       | H3B2V5      | (514) 866-7906 | Not Available  |
| Avis Rent A Car | 1640 Electra Blvd       | 131       | North Saanich       | BC       | V8L5V4      | (250) 656-6033 | Not Available  |
| Avis Rent A Car | 78 Bond St W            |           | Oshawa              | ON       | L1G1A5      | (905) 666-2085 | Not Available  |
|                 | 1003 Airport Parkway    |           |                     |          |             |                |                |
| Avis Rent A Car | Pvt                     |           | Ottawa              | ON       | K1V9B4      | (613) 739-3334 | Not Available  |
| Avis Rent A Car | 180 Paul Benoit Dr      |           | Ottawa              | ON       | K1V2E5      | (613) 521-7541 | Not Available  |
| Avis Rent A Car | 851 Industrial Ave      |           | Ottawa              | ON       | K1G4L3      | (613) 288-0935 | Not Available  |
| Avis Rent A Car | 37 Lanark Rd            |           | Perth               | ON       | K7H2R9      | (613) 267-6003 | Not Available  |
|                 | 1100 Boul Rene          |           |                     |          |             |                |                |
| Avis Rent A Car | Levesque E              |           | Quebec              | QC       | G1R 5V2     | (418) 523-1075 | Not Available  |
| Avis Rent A Car | 3125 Boul Hochelaga     |           | Quebec              | QC       | G1W2P9      | (418) 652-1847 | Not Available  |
| Avis Rent A Car | 510 Rue Principale      |           | Quebec              | QC       | G2G2T9      | (418) 872-2861 | (418) 872-1594 |
| Avis Rent A Car | 7175 Boul Wilfrid-Hamel |           | Quebec              | QC       | G2G1B6      | (418) 872-0409 | (418) 877-2151 |
|                 | Aeroport - 505 Rue      |           |                     |          |             |                |                |
| Avis Rent A Car | Principale              |           | Quebec              | QC       | G2G 0J4     | (418) 872-2861 | Not Available  |
| Avis Rent A Car | 5201 Regina Ave         |           | Regina              | SK       | S4W1B3      | (306) 751-5460 | (306) 751-5466 |
| Avis Rent A Car | 560 Stewart St          |           | Renfrew             | ON       | K7V1Y6      | (613) 432-1112 | Not Available  |
| Avis Rent A Car | 3820 Mcdonald Rd        |           | Richmond            | BC       | V7B1L8      | (604) 606-2847 | Not Available  |
|                 | 4911 Grant Mcconachie   |           |                     |          |             |                |                |
| Avis Rent A Car | Way                     |           | Richmond            | BC       | V7B1M8      | (604) 606-2855 | Not Available  |
| Avis Rent A Car | 5533 Airport Rd S       | 12        | Richmond            |          | V7B1B5      | (604) 491-9500 | Not Available  |
| Avis Rent A Car | 4 Water St              |           | Saint John          | NB       | E2L2A5      | (506) 634-7750 | Not Available  |
| Avis Rent A Car | 2301 Avenue C N         | 114       | Saskatoon           |          | S7L5Z5      | (306) 652-3434 | Not Available  |
| Avis Rent A Car | 2625 Airport Dr         |           | Saskatoon           | SK       | S7L7L1      | (306) 373-1161 | Not Available  |

| Car Rental Agencies in Canada |                          |         |                  |          |             |                |                |  |  |
|-------------------------------|--------------------------|---------|------------------|----------|-------------|----------------|----------------|--|--|
|                               |                          | Suite   |                  |          |             |                |                |  |  |
| Company Name                  | Address                  | Number  | City             | Province | Postal Code | Phone Number   | Fax Number     |  |  |
| Avis Rent A Car               | 573 Second Line          |         | Sault Ste. Marie | ON       | P6B4K2      | (705) 253-5489 | Not Available  |  |  |
|                               |                          |         |                  |          |             |                |                |  |  |
|                               | Sault Ste Marie Airport, |         |                  |          |             |                |                |  |  |
| Avis Rent A Car               | 475 Airport Road, RR#1   |         | Sault Ste. Marie | ON       | P6A 5K6     | (705) 779-2644 | Not Available  |  |  |
| Avis Rent A Car               | 136 Airport Rd           |         | St. John's       | NL       | A1A4Y3      | (709) 722-6620 | (709) 722-6623 |  |  |
| Avis Rent A Car               | 10076 King George Blvd   |         | Surrey           | BC       | V3T2W4      | (604) 930-2847 | (604) 584-2847 |  |  |
| Avis Rent A Car               | 280 Airport Road         |         | Sydney           | NS       | B1M 1B8     | (902) 563-2847 | Not Available  |  |  |
| Avis Rent A Car               | 1465 Walsh St W          |         | Thunder Bay      | ON       | P7E4X6      | (807) 577-2847 | (807) 577-6559 |  |  |
| Avis Rent A Car               | 161 Bay St               | 140     | Toronto          | ON       | M5J2S1      | (416) 777-2847 | Not Available  |  |  |
| Avis Rent A Car               | 3320 Dufferin            |         | Toronto          | ON       | M6A 2T6     | (416) 256-9060 | Not Available  |  |  |
| Avis Rent A Car               | 80 Bloor St E            |         | Toronto          | ON       | M4W3G8      | (416) 964-2051 | Not Available  |  |  |
| Avis Rent A Car               | 757 Hornby St            | -       | Vancouver        | BC       | V6Z1S2      | (604) 606-2869 | (604) 606-2877 |  |  |
| Avis Rent A Car               | 8326 St George St        |         | Vancouver        | BC       | V5X3S7      | (604) 606-2886 | Not Available  |  |  |
| Avis Rent A Car               | 1001 Douglas St          | 2G      | Victoria         | BC       | V8W2C5      | (250) 386-8468 | (250) 386-7726 |  |  |
| Avis Rent A Car               | 160 Weber St S           |         | Waterloo         | ON       | N2J2A8      | (519) 886-5846 | (519) 742-1012 |  |  |
| Avis Rent A Car               | 4315 Northlands Blvd     |         | Whistler         | BC       | VON1B4      | (604) 932-1236 | (604) 932-3026 |  |  |
|                               | 4215 Seventh             |         |                  |          |             |                |                |  |  |
| Avis Rent A Car               | Concession Rd            |         | Windsor          | ON       | N9A6J3      | (519) 966-2782 | (519) 966-9859 |  |  |
| Avis Rent A Car               | 234 York Ave             |         | Winnipeg         | MB       | R3C0N5      | (204) 989-7521 | (204) 956-4329 |  |  |
| Bargain Car & Truck Rentals   | 554 Bathurst St          |         | London           | ON       | N6B3P1      | (519) 673-1290 | (519) 679-1408 |  |  |
| Bristol Auto & Truck Rentals  | 9205 Yonge St            |         | Richmond Hill    | ON       | L4C6Z2      | (905) 881-3303 | (905) 881-3310 |  |  |
| Bristol Car & Truck Rentals   | 190 Bovaird Dr W         | 1       | Brampton         | ON       | L7A1A2      | (905) 453-8383 | Not Available  |  |  |
| Bristol Rentals Inc           | 22 Kennedy Rd S          |         | Brampton         | ON       | L6W3E2      | (905) 874-6639 | Not Available  |  |  |
| Budget Car & Truck Rental     | 30824 South Fraser Way   |         | Abbotsford       | вс       | V2T6L4      | (604) 852-0038 | Not Available  |  |  |
| Budget Rent A Car             | 147 Westney Rd S         |         | Ajax             |          | L1S2C9      | (905) 619-9440 | Not Available  |  |  |
| Budget Rent A Car             | 1525 Av Du Pont S        | <u></u> | Alma             | QC       | G8B6N2      | (418) 668-5032 | Not Available  |  |  |
|                               | Bathurst Airport, 2929   |         |                  |          |             |                | <u> </u>       |  |  |
| Budget Rent A Car             | Route 180                |         | Bathurst         | NB       | E2A7B9      | (506) 549-6868 | Not Available  |  |  |

.

-

|  | Car Rental Agencies in Canada  |        |                             |          |                  |                                  |                                |  |  |
|--|--|--------|-----------------------------|----------|------------------|----------------------------------|--------------------------------|--|--|
|  |  | Suite  | 1                           |          | T                |                                  |                                |  |  |
| Company Name                           | Address  | Number | City                        | Province | Postal Code      | Phone Number                     | Fax Number                     |  |  |
| Budget Rent A Car                      | 73 Kennedy Rd S  |        | Brampton                    | ON       | L6W3E4           | (905) 451-2931                   | Not Available                  |  |  |
| Budget Rent A Car                      | 215 6th St   |        | Brandon                     | MB       | R7A3N3           | (204) 725-3550                   | Not Available                  |  |  |
| Budget Rent A Car                      | 2685 Gilmore Ave   |        | Burnaby                     | BC       | V5C4T7           | (604) 668-7000                   | Not Available                  |  |  |
| Budget Rent A Car                      | 15180 Bannister Rd SE  | 24     | Calgary                     | AB       | T2X1Z5           | (403) 226-1550                   | (403) 284-5570                 |  |  |
| Budget Rent A Car                      | 2000 Airport Rd NE   | 24     | Calgary                     | AB       | T2E6W5           | (403) 226-1550                   | (403) 221-1709                 |  |  |
| Budget Rent A Car                      | 3328 26 St NE  | 24     | Calgary                     | AB       | T1Y4T7           | (403) 226-1550                   | (403) 299-6803                 |  |  |
| Budget Rent A Car                      | 5210 Macleod Trail SW  | 24     | Calgary                     | AB       | т2ној2           | (403) 226-1550                   | (403) 299-6813                 |  |  |
| Budget Rent A Car                      | 2602 South Island<br>Highway   |        | Campbell River              | вс       | V9W1C7           | (250) 923-4283                   | Not Available                  |  |  |
| Budget Rent A Car                      | 22 Roseberry St  |        | Campbellton                 | NB       | E3N2G2           | (506) 780-0088                   | Not Available                  |  |  |
| Budget Rent A Car                      | 215 University Ave   |        | Charlottetown               | PE       | C1A4L6           | (902) 566-5525                   | (902) 368-2165                 |  |  |
| Budget Rent A Car<br>Budget Rent A Car | Charlottetown Airport,<br>250 Maple Hills Avenue<br>1910 Boul Saint-Paul |        | Charlottetown<br>Chicoutimi | PE<br>QC | C1C1N2<br>G7K1C9 | (902) 566-5525<br>(418) 545-4444 | Not Available<br>Not Available |  |  |
| Budget Rent A Car                      | 1555 Comox Rd  |        | Courtenay                   | BC       | V9N3P7           | (250) 338-7717                   | (250) 338-6611                 |  |  |
| Budget Rent A Car                      | 1460 Ryan Road   |        | Lazo                        | BC       | VOR2KO           | (250) 338-7717                   | Not Available                  |  |  |
| Budget Rent A Car                      | 19370 Airport Access Rd  | 3      | Cranbrook                   | BC       | V1C7E4           | (250) 426-5425                   | Not Available                  |  |  |
| Budget Rent A Car                      | 607 Cranbrook St N   |        | Cranbrook                   | BC       | V1C3R8           | (250) 489-4371                   | Not Available                  |  |  |
| Budget Rent A Car                      | 575 Boul Albert-De<br>Niverville   |        | Dorval                      | QC       | Н4Ү1ЈЗ           | (514) 633-1416                   | Not Available                  |  |  |
| Budget Rent A Car                      | 975 Boul Romeo-Vachon<br>N   |        | Dorval                      | QC       | H4Y1H2           | (514) 636-0052                   | (514) 636-5448                 |  |  |
| Budget Rent A Car                      | 5430 Trans-Canada Hwy  |        | Duncan                      | BC       | V9L6W4           | (250) 748-3221                   | Not Available                  |  |  |
| Budget Rent A Car                      | 10016 106 St NW  |        | Edmonton                    | AB       | T5J1G1           | (780) 448-2009                   | Not Available                  |  |  |
| Budget Rent A Car                      | 11106 151 St NW  |        | Edmonton                    | AB       | T5M1X2           | (780) 442-1540                   | Not Available                  |  |  |
| Budget Rent A Car                      | 8112 Roper Rd NW   |        | Edmonton                    | АВ       | T6E6T8           | (780) 448-2000                   | Not Available                  |  |  |

.

.

|                   |  | Car Renta | I Agencies in Cana | da       | <b>_</b> ,  |                 |                |
|-------------------|--|-----------|--------------------|----------|-------------|-----------------|----------------|
|                   |  | Suite     | 1                  |          |             |                 |                |
| Company Name      | Address  | Number    | City               | Province | Postal Code | Phone Number    | Fax Number     |
|                   | 8804 Yellowhead Trail  |           |                    |          |             |                 |                |
| Budget Rent A Car | NW   |           | Edmonton           | AB       | T5B1G1      | (780) 448-2009  | Not Available  |
| Budget Rent A Car | 2 Bell Blvd  |           | Enfield            | NS       | B2T1K2      | (902) 492-7553  | Not Available  |
| Budget Rent A Car | 424 Gregoire Dr  |           | Fort Mcmurray      | AB       | T9H3R2      | (780) 743-8215  | Not Available  |
| Budget Rent A Car | 1285 Hanwell Rd  |           | Fredericton        | NB       | E3C1A6      | (506) 452-1107  | Not Available  |
| Budget Rent A Car | 130 Boul Greber  |           | Gatineau           | QC       | J8T6H5      | (819) 243-9999  | Not Available  |
| Budget Rent A Car | 576 Barnes Dr  |           | Goffs              | NS       | B2T1K3      | (902) 492-7546  | Not Available  |
| Budget Rent A Car | 11044 100 St   | 101       | Grande Prairie     | AB       | T8V2N1      | (780) 538-3526  | (780) 538-1029 |
| Budget Rent A Car | 42 Gordon St   |           | Guelph             | ON       | N1H4H3      | (519) 821-6040  | (519) 821-8591 |
| Budget Rent A Car | 362 Lacewood Dr  | 14        | Halifax            | NS       | B3S1M7      | (902) 450-7368  | Not Available  |
| Budget Rent A Car | 324 Parkdale Ave N   |           | Hamilton           | ON       | L8H5X9      | (905) 581-0746  | Not Available  |
| Budget Rent A Car | Airport Rd   |           | High Level         | AB       | T0H1Z0      | (780) 926-3114  | (780) 926-4020 |
| Budget Rent A Car | 3025 Airport Rd  |           | Kamloops           | BC       | V2B7X1      | (250) 374-7368  | Not Available  |
| Budget Rent A Car | 820 Notre Dame Dr  |           | Kamloops           | BC       | V2C6L5      | (250) 374-7368  | Not Available  |
| Budget Rent A Car | 2702 Hwy 97 N  |           | Kelowna            | BC       | V1X4J7      | (250) 491-7368  | Not Available  |
|                   | Kelowna Airport, 14-   |           |                    |          |             |                 |                |
| Budget Rent A Car | 5533 Kelowna Intl Apt  |           | Kelowna            | BC       | V1V1S1      | (250) 491-7368  | Not Available  |
| Budget Rent A Car | 856 Queen St   |           | Kincardine         | ON       | N2Z2Y2      | (519) 396-7000  | Not Available  |
| Budget Rent A Car | 1412 Princess St   |           | Kingston           | ON       | K7M3E5      | (613) 546-3231  | Not Available  |
| Budget Rent A Car | 7000 Ch De L'Aeroport  |           | La Baie            | QC       | G7B3N8      | (418) 677-2720  | (418) 677-1374 |
| Budget Rent A Car | 10040 101 Ave  |           | Lac La Biche       | AB       | T0A2C0      | (780) 623-4175  | Not Available  |
| Dudget Dept & Con | Aeroport Montreal-<br>Trudeau, 975 Boulevard<br>Romeo Vachon |           | Dorval             | QC       | H4Y1H1      | (514) 636-0052  | Not Available  |
| Budget Rent A Car |  |           | Lasalle            |          | H8N1X8      | (514) 368-2847  | Not Available  |
| Budget Rent A Car | 7794 Boul Newman   |           | Lasalle            |          |             | (314) 508-2847  |                |
|                   | 1655 Boul Daniel   |           |                    |          | H7V4C2      | (450) 221 0171  | Not Available  |
| Budget Rent A Car | Johnson  |           | Laval              | QC       |             | (450) 231-0171  |                |
|                   | 1655 Boul Daniel-  |           |                    |          | H7V4C2      | (450) 978-4412  | Not Available  |
| Budget Rent A Car | Johnson  |           | Laval              | QC       |             | 1(450) 978-4412 |                |

|                   | Car Rental Agencies in Canada                  |                 |                 |          |             |                |                |  |  |  |
|-------------------|--|-----------------|-----------------|----------|-------------|----------------|----------------|--|--|--|
| Company Name      | Address  | Suite<br>Number | City            | Province | Postal Code | Phone Number   | Fax Number     |  |  |  |
| Budget Rent A Car | 5103 25 St                                     |                 | Lloydminster    | AB       | T9V3G2      | (780) 875-1785 | (780) 875-5595 |  |  |  |
| Budget Rent A Car | 1750 Crumlin                                   |                 | London          | ON       | N5V3B6      | (519) 451-2700 | Not Available  |  |  |  |
| Budget Rent A Car | 20803 Lougheed Hwy                             |                 | Maple Ridge     | BC       | V2X2R2      | (604) 476-9888 | (604) 476-9886 |  |  |  |
| Budget Rent A Car | 7604 Woodbine Ave                              |                 | Markham         | ON       | L3R2N2      | (905) 477-0460 | (905) 477-5288 |  |  |  |
| Budget Rent A Car | 95 Durick St                                   |                 | Miramichi       | NB       | E1V0G1      | (506) 624-9020 | Not Available  |  |  |  |
| Budget Rent A Car | 5905 Campus Rd                                 | _               | Mississauga     | ON       | L4V1P9      | (905) 676-1500 | (905) 676-0391 |  |  |  |
|                   | Pearson International<br>Airport, 5980 Airport |                 |                 |          |             |                |                |  |  |  |
| Budget Rent A Car | Road   |                 | Mississauga     | ON       | L4V1P9      | (905) 676-0311 | (905) 676-0391 |  |  |  |
| Budget Rent A Car | 1965 Main St                                   |                 | Moncton         | NB       | E1E1J1      | (506) 857-3993 | (506) 853-8103 |  |  |  |
| Budget Rent A Car | 1335 Route 117                                 |                 | Mont-Tremblant  | QC       | J8E2X9      | (819) 425-1222 | Not Available  |  |  |  |
| Budget Rent A Car | 933 Topsail Rd                                 |                 | Mount Pearl     | NL       | A1N3K1      | (709) 747-1234 | (709) 747-0716 |  |  |  |
| Budget Rent A Car | 2501 Kenworth Rd                               |                 | Nanaimo         | BC       | V9T3M4      | (250) 758-2270 | (250) 729-2428 |  |  |  |
| Budget Rent A Car | 33 Terminal Ave                                |                 | Nanaimo         | BC       | V9R5C4      | (250) 760-7368 | Not Available  |  |  |  |
| Budget Rent A Car | 6450 Island N Hwy                              |                 | Nanaimo         | BC       | V9T1X2      | (250) 760-7368 | Not Available  |  |  |  |
| Budget Rent A Car | 611 Sixth St                                   |                 | New Westminster | вс       | V3L3C1      | (604) 524-4593 | Not Available  |  |  |  |
| Budget Rent A Car | 3320 Dufferin St                               |                 | North York      | ON       | M6A2T6      | (416) 787-1224 | (416) 785-6317 |  |  |  |
| Budget Rent A Car | 1003 Airport Parkway<br>Pvt                    |                 | Ottawa          | ON       | K1V2C1      | (613) 521-0608 | Not Available  |  |  |  |
| Budget Rent A Car | 1200 Baseline Rd                               |                 | Ottawa          | ON       | K2C0A6      | (613) 225-8006 | Not Available  |  |  |  |
| Budget Rent A Car | 851 Industrial Ave                             |                 | Ottawa          | ON       | K1G4L3      | (613) 739-4231 | Not Available  |  |  |  |
| Budget Rent A Car | 188 Westminster Ave W                          |                 | Penticton       | BC       | V2A1J8      | (250) 493-0212 | Not Available  |  |  |  |
| Budget Rent A Car | 2504 Skaha Lake Rđ                             | 106             | Penticton       | BC       | V2A6G1      | (250) 493-0212 | Not Available  |  |  |  |
| Budget Rent A Car | 3000 Airport Rd                                | 9               | Penticton       | BC       | V2A8X1      | (250) 493-0212 | (250) 492-9109 |  |  |  |
| Budget Rent A Car | 19669 Blaney Dr                                |                 | Pitt Meadows    | BC       | V3Y2V5      | (604) 460-0773 | Not Available  |  |  |  |
| Budget Rent A Car | 750 Av Saint-Louis                             |                 | Plessisville    | QC       | G6L2M1      | (819) 362-8816 | Not Available  |  |  |  |
| Budget Rent A Car | 3500 3rd Ave                                   |                 | Port Alberni    | BC       | V9Y4E5      | (250) 724-4511 | (250) 724-9737 |  |  |  |
| Budget Rent A Car | 2484 Lougheed Hwy                              |                 | Port Coquitlam  | BC       | V3B1A9      | (604) 927-7501 | Not Available  |  |  |  |
| Budget Rent A Car | 955 1st Ave                                    |                 | Prince George   | BC       | V2L2Y4      | (250) 563-9669 | (250) 614-7214 |  |  |  |

•

ч.

|                   |                        | Car Renta | al Agencies in Canac | la       |             |                |                |
|-------------------|------------------------|-----------|----------------------|----------|-------------|----------------|----------------|
|                   |                        | Suite     |                      |          |             |                |                |
| Company Name      | Address                | Number    | City                 | Province | Postal Code | Phone Number   | Fax Number     |
| Budget Rent A Car | 505 Rue Principale     |           | Quebec               | QC       | G2G0J4      | (418) 872-8413 | Not Available  |
| Budget Rent A Car | 5214 50 Ave            |           | Red Deer             | AB       | T4N4B5      | (403) 346-7858 | Not Available  |
|                   | Regina Airport, 5200   |           |                      |          |             |                |                |
| Budget Rent A Car | Regina Avenue          | _         | Regina               | SK       | S4P 2M4     | (306) 992-7983 | Not Available  |
| Budget Rent A Car | 3691 No 3 Rd           |           | Richmond             | BC       | V6X2B8      | (604) 656-8888 | Not Available  |
| Budget Rent A Car | 3840 Mcdonald Rd       |           | Richmond             | BC       | V7B1L8      | (604) 668-7144 | Not Available  |
| <u> </u>          | Vancouver Internatl    |           |                      |          | 1           |                |                |
|                   | Airport, 4440 Cowlie   |           |                      |          |             |                |                |
| Budget Rent A Car | Cres                   |           | Richmond             | BC       | V7B1B8      | (604) 668-7000 | Not Available  |
| Budget Rent A Car | 10619 Yonge St         |           | Richmond Hill        | ON       | L4C3C5      | (905) 884-1165 | Not Available  |
| Budget Rent A Car | 1177 Boul Marcotte     |           | Roberval             | QC       | G8H2B5      | (418) 275-5517 | Not Available  |
| Budget Rent A Car | 390 Boul Rideau        |           | Rouyn-Noranda        | QC       | J9X1P4      | (819) 797-0266 | (819) 797-3997 |
| Budget Rent A Car | 529 Rothesay Ave       |           | Saint John           | NB       | E2J2C6      | (506) 633-3434 | Not Available  |
| Budget Rent A Car |                        |           | Sandspit             | BC       | V0T1T0      | (250) 637-5747 | Not Available  |
| Budget Rent A Car | 234 1st Ave S          |           | Saskatoon            | SK       | S7K1K3      | (306) 244-7925 | (306) 653-3003 |
| Budget Rent A Car | 602 1st Ave N          |           | Saskatoon            | SK       | S7K1X8      | (306) 500-1259 | Not Available  |
| Budget Rent A Car | 801 45 St W            |           | Saskatoon            | SK       | S7L5X2      | (306) 664-0674 | Not Available  |
| Budget Rent A Car | 573 Second Line E      | -         | Sault Ste. Marie     | ON       | P6B4K2      | (705) 942-1144 | (705) 942-4985 |
| Budget Rent A Car | 503 Centennial Rd N    | 1         | Scarborough          | ON       | M1C2A5      | (416) 287-5141 | (416) 282-3399 |
| Budget Rent A Car | 3265 Rue King O        |           | Sherbrooke           | QC       | J1L1C8      | (819) 200-2907 | Not Available  |
| Budget Rent A Car | 434 St Paul St         |           | St Catharines        | ON       | L2R3N3      | (905) 682-1427 | Not Available  |
| Budget Rent A Car | 22 Muir Dr             | В         | St. Albert           | AB       | T8N1G3      | (780) 460-8445 | Not Available  |
|                   | St John's Airport, 80  |           |                      |          |             |                |                |
|                   | Craig Dobbins Way      |           |                      |          |             |                |                |
| Budget Rent A Car | Terminal               |           | St. John's           | NF       | A1A5T2      | (709) 747-1234 | Not Available  |
| Budget Rent A Car | 10305 King George Blvd |           | Surrey               | BC       | V3T2W6      | (604) 656-8888 | (604) 586-3803 |
| Budget Rent A Car | 2570 King George Blvd  |           | Surrey               | BC       | V4P1H5      | (604) 535-5424 | (604) 531-2472 |
| Budget Rent A Car | 1266 Rue Notre-Dame E  |           | Thetford Mines       | QC       | G6G2V5      | (418) 335-5527 | (418) 338-8120 |

|                                       |                         | Car Renta | al Agencies in Cana | nda      |             |                |                |
|---------------------------------------|-------------------------|-----------|---------------------|----------|-------------|----------------|----------------|
| · · · · · · · · · · · · · · · · · · · |                         | Suite     |                     |          |             |                | 1              |
| Company Name                          | Address                 | Number    | City                | Province | Postal Code | Phone Number   | Fax Number     |
| Budget Rent A Car                     | 100 Princess St         |           | Thunder Bay         | ON       | P7E6S2      | (807) 473-5040 | (807) 473-5161 |
| Budget Rent A Car                     | 230 Waterloo St S       |           | Thunder Bay         | ON       | P7E2C3      | (807) 622-3366 | (807) 622-0211 |
| Budget Rent A Car                     | 4599 Airport Rd         |           | Timmins             | ON       | P4N 7C3     | (705) 264-2300 | Not Available  |
| Budget Rent A Car                     | 271 Front St E          |           | Toronto             | ON       | M5A1G1      | (416) 363-0985 | Not Available  |
| Budget Rent A Car                     | 2746 Yonge St           |           | Toronto             | ON       | M4N2J2      | (416) 487-1511 | Not Available  |
| Budget Rent A Car                     | 333 College St          |           | Toronto             | ON       | M5T1S2      | (416) 963-9496 | (416) 413-4846 |
| Budget Rent A Car                     | 2931 Highway Dr         |           | Trail               | BC       | V1R2T2      | (250) 364-2476 | Not Available  |
| Budget Rent A Car                     | 2222 Boul Des Recollets |           | Trois-Rivieres      | QC       | G8Z3X5      | (819) 373-1515 | Not Available  |
| Budget Rent A Car                     | 50 4e Ave               |           | Val-D'Or            | QC       | J9P1G1      | (819) 874-2823 | Not Available  |
|                                       | Aeroport De Val D'Or,   |           |                     |          |             |                |                |
| Budget Rent A Car                     | 125 7th Street          |           | Val-D'Or            | QC       | J9P4N9      | (819) 874-7094 | Not Available  |
| Budget Rent A Car                     | 1404 Marine Dr SW       |           | Vancouver           | BC       | V6P5Z9      | (604) 261-3343 | (604) 264-8171 |
| Budget Rent A Car                     | 1412 Marine Dr SW       |           | Vancouver           | BC       | V6P5Z9      | (604) 656-8888 | Not Available  |
| Budget Rent A Car                     | 416 Georgia St W        |           | Vancouver           | BC       | V6B1Z3      | (604) 668-7228 | (604) 668-7291 |
| Budget Rent A Car                     | 99 Pender St W          |           | Vancouver           | BC       | V6B1R3      | (604) 683-7299 | (604) 683-3541 |
| Budget Rent A Car                     | 4419 27 St              |           | Vernon              | BC       | V1T4Y5      | (250) 545-6162 | (250) 545-0071 |
| Budget Rent A Car                     | 2224 Douglas St         |           | Victoria            | BC       | V8T4L5      | (250) 953-5353 | (250) 382-2560 |
| Budget Rent A Car                     | 2634 Douglas St         |           | Victoria            | BC       | V8T4M1      | (250) 953-5300 | Not Available  |
| Budget Rent A Car                     | 2916 Jacklin Rd         |           | Victoria            | BC       | V9B3Y5      | (250) 391-2150 | Not Available  |
| Budget Rent A Car                     | 3657 Harriet Rd         |           | Victoria            | BC       | V8Z3T1      | (250) 953-5300 | Not Available  |
|                                       | Victoria International  |           |                     |          |             |                |                |
|                                       | Airport, 132-1640       |           |                     |          |             |                |                |
| Budget Rent A Car                     | Electra Boulevard       |           | Victoria            | BC       | V8L5V4      | (250) 953-5300 | Not Available  |
|                                       | 434 Rue Notre Dame      |           |                     |          |             |                |                |
| Budget Rent A Car                     | Ouest                   |           | Victoriaville       | QC       | G6P1S8      | (819) 809-2202 | Not Available  |
| Budget Rent A Car                     | 160 Weber St S          |           | Waterloo            | ON       | N2J2A8      | (519) 749-7912 | Not Available  |
| Budget Rent A Car                     | 2575 Ross Rd            |           | West Kelowna        | BC       | V1Z1M2      | (250) 491-7368 | Not Available  |
| Budget Rent A Car                     | 3200 County Rd 42       |           | Windsor             | ON       | N9A6J3      | (519) 969-5611 | (519) 969-6566 |
| Budget Rent A Car                     | 201 Winnipeg Airport    |           | Winnipeg            | MB       | R3H1C1      | (204) 989-8510 | (204) 786-2962 |
| Budget Rent A Car                     | 2140 Mcphillips St      | 5         | Winnipeg            | MB       | R2V3C8      | (204) 953-1150 | Not Available  |

÷.

\_

|                              |                            | Car Renta | al Agencies in Cana | da       |             |                |                |
|------------------------------|----------------------------|-----------|---------------------|----------|-------------|----------------|----------------|
|                              |                            | Suite     |                     |          |             |                |                |
| Company Name                 | Address                    | Number    | City                | Province | Postal Code | Phone Number   | Fax Number     |
| Budget Rent A Car            | 20 Airport Road            |           | Yellowknife         | NT       | X1A3T2      | (867) 920-9200 | (867) 920-9201 |
| Budget Rent-A-Car            | 3170 Kempt Rd              |           | Halifax             | NS       | B3K4X1      | (902) 873-3509 | Not Available  |
| Canada Car Rental Ltd        | 1277 Caledonia Rd          |           | North York          | ON       | M6A2X7      | (416) 780-0902 | (416) 780-0898 |
| Capital Car & Truck Rentals  | 227 Kenmount Rd            | -         | St. John's          | NL       | A1B3P9      | (709) 726-7111 | (709) 726-8811 |
| Car Rental Place             | 3219 Dufferin St           |           | North York          | ON       | M6A2T2      | (416) 787-0209 | (416) 787-9240 |
| Clearway Car & Truck Rentals | 31808s South Fraser<br>Way |           | Abbotsford          | вс       | V2T1V3      | (604) 850-6522 | (604) 864-4045 |
| Clearway Rentals Inc         | 31808 South Fraser Way     |           | Abbotsford          | вс       | V2T1V3      | (604) 850-2570 | Not Available  |
| Cmp Classic Leasing & Rental | 1313 36 St NE              |           | Calgary             | AB       | T2A6P9      | (403) 207-1043 | (403) 207-1018 |
| Discount Car & Truck Rentals | 8 Valleyview Dr            |           | Bancroft            | ON       | KOL1CO      | (613) 332-3028 | (613) 332-5404 |
| Discount Car & Truck Rentals | 15 George St               |           | Barrie              | ON       | L4N2G5      | (705) 722-8900 | (705) 722-0184 |
| Discount Car & Truck Rentals | 252 North Front St         |           | Belleville          | ON       | К8РЗС4      | (613) 969-1645 | (613) 969-1647 |
| Discount Car & Truck Rentals | 157 East Dr                | D         | Brampton            | ON       | L6T1B5      | (905) 799-1233 | Not Available  |
| Discount Car & Truck Rentals | 310 Queen St E             |           | Brampton            | ON       | L6W2B5      | (905) 454-0400 | Not Available  |
| Discount Car & Truck Rentals | 786 Colborne St            |           | Brantford           | ON       | N3S3S1      | (519) 759-1316 | Not Available  |
| Discount Car & Truck Rentals | 40 Hopewell Way NE         |           | Calgary             | AB       | T3J5H7      | (403) 299-1203 | Not Available  |
| Discount Car & Truck Rentals | 1250 Bishop St N           |           | Cambridge           | ON       | N3H2K9      | (519) 622-9004 | Not Available  |
| Discount Car & Truck Rentals | 185 Brackley Point Rd      |           | Charlottetown       | PE       | C1A6Z2      | (902) 566-3213 | (902) 368-8076 |
| Discount Car & Truck Rentals | 135 Av Ilsley              |           | Dartmouth           | NS       | B3B1T1      | (902) 468-7171 | Not Available  |

••

|                              |                     | Car Renta | al Agencies in Can | ada      |             |                |                |
|------------------------------|---------------------|-----------|--------------------|----------|-------------|----------------|----------------|
|                              |                     | Suite     |                    |          |             |                | Ĩ              |
| Company Name                 | Address             | Number    | City               | Province | Postal Code | Phone Number   | Fax Number     |
| Discount Car & Truck Rentals | 12503 125 St NW     |           | Edmonton           | AB       | T5L0T4      | (780) 452-3430 | Not Available  |
| Discount Car & Truck Rentals | 340 Evans Ave       |           | Etobicoke          | ON       | M8Z1K5      | (416) 249-5554 | Not Available  |
| Discount Car & Truck Rentals | 354 Guelph St       | 23        | Georgetown         | ON       | L7G4B5      | (905) 873-1828 | (905) 873-1079 |
| Discount Car & Truck Rentals | 73 Gordon St        |           | Guelph             | ON       | N1H4H5      | (519) 837-2030 | (519) 837-3911 |
| Discount Car & Truck Rentals | 2710 Agricola St    |           | Halifax            | NS       | B3K4E1      | (902) 453-5335 | Not Available  |
| Discount Car & Truck Rentals | 1588 Upper James St |           | Hamilton           | ON       | L9B1K3      | (905) 575-9963 | (905) 575-9814 |
| Discount Car & Truck Rentals | 174 Main St E       |           | Hamilton           | ON       | L8N1G9      | (905) 523-7777 | Not Available  |
| Discount Car & Truck Rentals | 1081 King St E      |           | Kitchener          | ON       | N2G2N1      | (519) 743-6844 | (519) 743-8826 |
| Discount Car & Truck Rentals | 1616 Matheson Blvd  | 3         | Mississauga        | ON       | L4W1R9      | (905) 238-1084 | Not Available  |
| Discount Car & Truck Rentals | 206 Dundas St E     |           | Mississauga        | ON       | L5A1W4      | (905) 279-2966 | Not Available  |
| Discount Car & Truck Rentals | 1543 Mountain Rd    |           | Moncton            | NB       | E1G1A3      | (506) 857-2309 | Not Available  |
| Discount Car & Truck Rentals | 2250 Rue Guy        |           | Montreal           | QC       | НЗН2МЗ      | (514) 932-3313 | (514) 932-1533 |
| Discount Car & Truck Rentals | 709 Topsail Rd      |           | Mount Pearl        | NL       | A1N3N4      | (709) 748-2105 | (709) 748-2188 |
| Discount Car & Truck Rentals | 710 Arrow Rd        |           | North York         | ON       | М9М2М1      | (416) 744-7942 | Not Available  |
| Discount Car & Truck Rentals | 720 Arrow Rd        |           | North York         | ON       | M9M2M1      | (416) 443-2277 | Not Available  |

|                              |                                   | Car Renta | al Agencies in Canada | a        |             |                |                |
|------------------------------|-----------------------------------|-----------|-----------------------|----------|-------------|----------------|----------------|
|                              |                                   | Suite     |                       |          |             |                |                |
| Company Name                 | Address                           | Number    | City                  | Province | Postal Code | Phone Number   | Fax Number     |
| Discount Car & Truck Rentals | Don Mills & Eglinton              |           | North York            | ON       | МЗВ2МЗ      | (416) 443-2622 | Not Available  |
| Discount Car & Truck Rentals | 101 Simcoe St N                   |           | Oshawa                | ON       | L1G4S4      | (905) 571-0571 | Not Available  |
| Discount Car & Truck Rentals | 938 St Laurent Blvd               |           | Ottawa                | ON       | К1КЗВЗ      | (613) 747-3784 | Not Available  |
| Discount Car & Truck Rentals | 1271 Pembroke St W                |           | Pembroke              | ON       | K8A5R3      | (613) 735-7860 | (613) 735-7873 |
| Discount Car & Truck Rentals | 1585 Lansdowne St W               |           | Peterborough          | ON       | K9K1R2      | (705) 749-6116 | (705) 749-6525 |
| Discount Car & Truck Rentals | 3673 Boul Saint-Jean-<br>Baptiste |           | Pte-Aux-Trembles      | QC       | H1B4B2      | (514) 640-6670 | (514) 640-6782 |
| Discount Car & Truck Rentals | 10620 Yonge St                    |           | Richmond Hill         | ON       | L4C3C8      | (905) 737-1954 | Not Available  |
| Discount Car & Truck Rentals | 335 Brand Rd                      | 1         | Saskatoon             | SK       | S7J5L4      | (306) 242-6556 | Not Available  |
| Discount Car & Truck Rentals | 1975 Eglinton Ave E               |           | Scarborough           | ON       | M1L2N1      | (416) 288-4577 | Not Available  |
| Discount Car & Truck Rentals | 306 Lake St                       |           | St Catharines         | ON       | L2N4H2      | (905) 687-9979 | (905) 687-9958 |
| Discount Car & Truck Rentals | 1190 Kingsway                     |           | Sudbury               | ON_      | P3B2E8      | (705) 673-3039 | Not Available  |
| Discount Car & Truck Rentals | 7089 Yonge St                     |           | Thornhill             | ON       | L3T2A7      | (416) 222-2227 | Not Available  |
| Discount Car & Truck Rentals | 1351 Dufferin St                  |           | Toronto               | ON       | M6H4C7      | (416) 533-9435 | (416) 840-0129 |
| Discount Car & Truck Rentals | 2336 St Clair Ave W               |           | Toronto               | ON       | M6N1K8      | (416) 443-2622 | Not Available  |
| Discount Car & Truck Rentals | 2440 Dundas St W                  |           | Toronto               | ON       | M6P1W9      | (416) 533-9433 | Not Available  |

-

~

|                              |                                | Car Renta | al Agencies in Cana | ada      |             |                |                |
|------------------------------|--------------------------------|-----------|---------------------|----------|-------------|----------------|----------------|
|                              |                                | Suite     |                     |          |             | · ·            |                |
| Company Name                 | Address                        | Number    | City                | Province | Postal Code | Phone Number   | Fax Number     |
| Discount Car & Truck Rentals | 133 Weber St N                 |           | Waterloo            | ON       | N2J3G9      | (519) 888-6261 | Not Available  |
| Discount Car & Truck Rentals | 195 Union St E                 |           | Waterloo            | ON       | N2J1C4      | (519) 888-6261 | Not Available  |
| Discount Car & Truck Rentals | 1110 Dundas St E               |           | Whitby              | ON       | L1N2K2      | (905) 666-1900 | (905) 666-8877 |
| Discount Car & Truck Rentals | 2565 Dougall Ave               |           | Windsor             | ON       | N8X1T5      | (519) 972-0966 | Not Available  |
| Discount Car & Truck Rentals | 383 Mcmillan Ave               |           | Winnipeg            | МВ       | R3LON3      | (204) 287-8002 | Not Available  |
| Discount Car & Truck Rentals | 8099 Weston Rd                 |           | Woodbridge          | ON       | L4L0C1      | (905) 856-3470 | Not Available  |
| Discount Location D'Autos    | 9500 Boul Henri-<br>Bourassa O |           | Saint-Laurent       | QC       | H4S1N8      | (514) 333-0070 | Not Available  |
| Dixon's Car & Truck Rental   | 2392 Princess St               |           | Kingston            | ON       | K7M3G4      | (613) 542-2222 | Not Available  |
| Dollar Rent A Car            | 2000 Airport Rd NE             |           | Calgary             | AB       | T2E6W5      | (403) 291-4129 | Not Available  |
| Dollar Rent A Car            | 12505 Cote-De-Liesse<br>Rd     |           | Dorval              | QC       | H9P1B7      | (514) 990-0078 | Not Available  |
| Dollar Rent A Car            | 2005 55e Ave                   |           | Dorval              | QC       | H9P2Y6      | (514) 631-8950 | Not Available  |
| Dollar Rent A Car            | 1 Bell Blvd                    |           | Enfield             | NS       | B2T1K2      | (902) 873-2075 | (902) 873-3531 |
| Dollar Rent A Car            | 5553 Dundas St W               |           | Etobicoke           | ON       | M9B1B8      | (416) 236-7888 | (416) 743-8057 |
| Dollar Rent A Car            | 1181 Hollis St                 |           | Halifax             | NS       | B3H2P6      | (902) 453-6925 | (902) 477-6164 |
| Dollar Rent A Car            | 8004 Sparrow Cres              |           | Leduc               | AB       | T9E7G1      | (780) 986-3327 | Not Available  |
| Dollar Rent A Car            | 20214 Lougheed Hwy             |           | Maple Ridge         | BC       | V2X2P7      | (604) 460-1508 | Not Available  |
| Dollar Rent A Car            | 4470 Highway 7                 |           | Markham             | ON       | L3R1M3      | (905) 475-5280 | Not Available  |
| Dollar Rent A Car            | 1369 Dundas St E               |           | Mississauga         | ON       | L4Y2C7      | (905) 281-3226 | (905) 281-3317 |
| Dollar Rent A Car            | 1630 Matheson Blvd             |           | Mississauga         | ON       | L4W1Y4      | (905) 212-9991 | Not Available  |
| Dollar Rent A Car            | 3660 Hurontario St             |           | Mississauga         | ON       | L5B3C4      | (905) 897-7808 | (905) 897-7502 |
| Dollar Rent A Car            | 6050 Indian Line               |           | Mississauga         | ON       | L4V1G5      | (905) 612-1881 | (905) 612-1893 |
| Dollar Rent A Car            | 251 Grant Dr                   |           | New Liskeard        | ON       | POJ1PO      | (705) 647-8989 | Not Available  |
| Dollar Rent A Car            | 2055 16th Ave E                | 2         | Owen Sound          | ON       | N4K5R7      | (519) 376-0047 | (519) 376-8901 |

|                               |                       | Car Renta | I Agencies in Can | ada      |             |                |                |
|-------------------------------|-----------------------|-----------|-------------------|----------|-------------|----------------|----------------|
|                               |                       | Suite     |                   |          |             |                |                |
| Company Name                  | Address               | Number    | City              | Province | Postal Code | Phone Number   | Fax Number     |
| Dollar Rent A Car             | 1975 Broad St         |           | Regina            | SK       | S4P1Y2      | (306) 565-0500 | Not Available  |
| Dollar Rent A Car             | 5201 Regina Ave       | İ         | Regina            | SK       | S4W1B3      | (306) 352-1000 | (306) 525-1389 |
| Dollar Rent A Car             | 9429 Canora Rd        |           | Sidney            | BC       | V8L1P2      | (250) 656-8804 | Not Available  |
| Dollar Rent A Car             | 4 Craig Dobbin's Way  |           | St. John's        | NL       | A1A4Y3      | (709) 722-7300 | (709) 576-0953 |
| Dollar Rent A Car             | 7319 King George Blvd |           | Surrey            | вс       | V3W5A8      | (604) 635-3333 | (604) 635-3540 |
| Dollar Rent A Car             | 6005 50 Ave           |           | Taber             | AB       | T1G1W7      | (403) 223-0502 | Not Available  |
| Dollar Rent A Car             | 1245 Danforth Ave     |           | Toronto           | ON       | M4J5B5      | (416) 778-5446 | Not Available  |
| Dollar Rent A Car             | 220 Yonge St          |           | Toronto           | ON       | M5B2H1      | (647) 795-9140 | Not Available  |
| Dollar Rent A Car             | 330 Front St W        |           | Toronto           | ON       | M5V3B7      | (416) 598-4249 | Not Available  |
| Dollar Rent A Car             | 65 Front St W         |           | Toronto           | ON       | M5J1E6      | (416) 364-4952 | Not Available  |
| Dollar Rent A Car             | 3211 Grant St         |           | Vancouver         | BC       | V5K3J1      | (604) 606-1655 | Not Available  |
| Dollar Rent A Car             | 413 Seymour St        |           | Vancouver         | BC       | V6B3H5      | (604) 689-5303 | (604) 606-1698 |
| Dollar Rent A Car             | 1750 Sargent Avenue   |           | Winnipeg          | MB       | R3H0C7      | (204) 949-3779 | Not Available  |
| Driving Force                 | 9415 100 Ave          |           | Fort St. John     | BC       | V1J6W1      | (250) 787-0634 | (250) 787-1909 |
| Driving Force Nunavut         | 2018 40 W             | В         | Iqaluit           | NU       | ХОАОНО      | (867) 979-2088 | (867) 979-4326 |
| Driving Force Vehicle Rentals | 2332 23 St NE         |           | Calgary           | AB       | T2E8N3      | (403) 296-0770 | Not Available  |
| Enterprise Rent A Car         | 291 Speers Rd         |           | Oakville          | ON       | L6K2G1      | (905) 845-6161 | Not Available  |
| Enterprise Rent-A-Car         | 2015 Abbotsford Way   | 1         | Abbotsford        | BC       | V2S6Y5      | (604) 859-7368 | (604) 557-1847 |
| Enterprise Rent-A-Car         | 220 East Lake Blvd NE | 2         | Airdrie           | AB       | T4A2G2      | (403) 216-3346 | Not Available  |
| Enterprise Rent-A-Car         | 355 Bayly St W        |           | Ajax              | ON       | L1S1P1      | (905) 686-5587 | (905) 686-1365 |
| Enterprise Rent-A-Car         | 304 Dunlop St W       | 42        | Barrie            | ON       | L4N7L2      | (705) 728-1212 | (705) 728-5949 |
| Enterprise Rent-A-Car         | 301 Coleman St        |           | Belleville        | ON       | К8РЗН9      | (613) 967-2004 | (613) 967-3776 |
| Enterprise Rent-A-Car         | 2074 Steeles Ave E    | 18        | Brampton          | ON       | L6T4Z9      | (905) 799-3322 | Not Available  |
| Enterprise Rent-A-Car         | 320 Queen St E        |           | Brampton          | ON       | L6V1C2      | (905) 456-8288 | Not Available  |
| Enterprise Rent-A-Car         | 1500 Richmond Ave     |           | Brandon           | MB       | R7A7E3      | (204) 727-7732 | (204) 727-8391 |
| Enterprise Rent-A-Car         | 759 1st St            |           | Brandon           | МВ       | R7A2X5      | (204) 725-1300 | Not Available  |
| Enterprise Rent-A-Car         | 14 Easton Rd          |           | Brantford         | ON       | N3P1J5      | (519) 759-3100 | (519) 759-3033 |
| Enterprise Rent-A-Car         | 4756 Lougheed Hwy     |           | Burnaby           | BC       | V5C4A8      | (604) 291-8871 | Not Available  |
| Enterprise Rent-A-Car         | 1036 9 Ave SW         |           | Calgary           | AB       | T2P1L9      | (403) 212-5232 | Not Available  |

~

÷

|                       | Car Rental Agencies in Canada                      |        |             |          |             |                |                |  |  |  |
|-----------------------|--|--------|-------------|----------|-------------|----------------|----------------|--|--|--|
|                       |  | Suite  |             |          |             |                |                |  |  |  |
| Company Name          | Address  | Number | City        | Province | Postal Code | Phone Number   | Fax Number     |  |  |  |
| Enterprise Rent-A-Car | 11488 24 St SE                                     | 500    | Calgary     | AB       | T2Z4C9      | (403) 720-1916 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 2000 Airport Rd NE                                 |        | Calgary     | AB       | T2E6W5      | (403) 233-8021 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 2520 23 St NE                                      | 5      | Calgary     | AB       | T2E8L2      | (403) 735-6080 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 4115 Macleod Trail SW                              |        | Calgary     | AB       | T2G2R6      | (403) 243-2100 | (403) 243-4173 |  |  |  |
| Enterprise Rent-A-Car | 5821 6 St SE                                       |        | Calgary     | AB       | T2H1M4      | (403) 216-3490 | (403) 216-3498 |  |  |  |
| Enterprise Rent-A-Car | 626 58 Ave SE                                      |        | Calgary     | AB       | Т2НОР8      | (403) 253-1853 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 64 Crowfoot Cir NW                                 | 157    | Calgary     | AB       | T3G2T3      | (403) 241-7053 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 7820 Macleod Trail SE                              |        | Calgary     | AB       | T2H0L9      | (403) 252-0222 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 499 Hespeler Rd                                    | 6      | Cambridge   | ON       | N1R6J2      | (519) 622-8292 | (519) 622-9982 |  |  |  |
| Enterprise Rent-A-Car | 1250 Knight Rd                                     | 106    | Comox       | BC       | V9M4H2      | (250) 339-3313 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 8600 Keele St                                      | 65     | Concord     | ON       | L4K4H8      | (905) 761-0105 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 560 Pitt St  |        | Cornwall    | ON       | K6J3R6      | (613) 933-7711 | (613) 933-6024 |  |  |  |
| Enterprise Rent-A-Car | 597 Portland St                                    |        | Dartmouth   | NS       | B2W2M5      | (902) 462-6825 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 600 Windmill Rd                                    | 2      | Dartmouth   | NS       | B3B1B5      | (902) 469-1689 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 439 Rue Champlain                                  | 1      | Dieppe      | NB       | E1A1P2      | (506) 855-4660 | (506) 855-4685 |  |  |  |
| Enterprise Rent-A-Car | 10645 101 St NW                                    |        | Edmonton    | AB       | T5H2S2      | (780) 429-4883 | (780) 429-0057 |  |  |  |
| Enterprise Rent-A-Car | 10903 156 St                                       |        | Edmonton    | AB       | T5P1S7      | (780) 489-7577 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 4517 Gateway Blvd NW                               |        | Edmonton    | AB       | Т6Н5СЗ      | (780) 415-4350 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 7615 Argyll Rd NW                                  |        | Edmonton    | AB       | T6C4A7      | (780) 495-1205 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 9943 Jasper Ave NW                                 |        | Edmonton    | AB       | T5J2X4      | (780) 422-6097 | Not Available  |  |  |  |
|                       | Edmonton Enternational<br>Airport, 1, 1000 Airport |        |             |          |             |                |                |  |  |  |
| Enterprise Rent-A-Car | Road   |        | Leduc       | AB       | T9E8B7      | (780) 980-2338 | Not Available  |  |  |  |
| Enterprise Rent-A-Car | 5475 Dundas St W                                   |        | Etobicoke   | ON       | M9B1B5      | (416) 237-0557 | (416) 237-1672 |  |  |  |
| Enterprise Rent-A-Car | 562 Kipling Ave                                    | 4      | Etobicoke   | ON       | M8Z5E8      | (416) 252-0123 | (416) 252-1222 |  |  |  |
| Enterprise Rent-A-Car | 1270 Hanwell Rd                                    |        | Fredericton | NB       | E3C1A7      | (506) 459-4100 | (506) 459-4105 |  |  |  |
| Enterprise Rent-A-Car | 2621 Skead Rd                                      | 1      | Garson      | ON       | P3L1V4      | (705) 693-9993 | (705) 693-2999 |  |  |  |
| Enterprise Rent-A-Car | 1235 Donald St                                     | 8      | Gloucester  | ON       | K1J8W3      | (613) 742-6433 | Not Available  |  |  |  |

|                       |                         | Car Renta | al Agencies in Cana | da       |             |                |                |
|-----------------------|-------------------------|-----------|---------------------|----------|-------------|----------------|----------------|
|                       |                         | Suite     |                     |          |             |                |                |
| Company Name          | Address                 | Number    | City                | Province | Postal Code | Phone Number   | Fax Number     |
| Enterprise Rent-A-Car | 209 Aerotech Dr         | 1         | Goffs               | NS       | B2T1K3      | (902) 873-3300 | Not Available  |
|                       |                         |           |                     |          |             |                |                |
| Enterprise Rent-A-Car | 843 A.E.H. Daigle Blvd. |           | Grand Falls         | NB       | E3Z3C7      | (506) 473-6945 | Not Available  |
| Enterprise Rent-A-Car | 12712 100 St            |           | Grande Prairie      | AB       | T8V4H8      | (780) 538-8444 | (780) 538-8445 |
| Enterprise Rent-A-Car | 56 Gordon St            |           | Guelph              | ON       | N1H4H3      | (519) 767-5337 | Not Available  |
| Enterprise Rent-A-Car | 3387 Kempt Rd           |           | Halifax             | NS       | B3K4X5      | (902) 453-6929 | Not Available  |
| Enterprise Rent-A-Car | 5201 Duke St            | 110       | Halifax             | NS       | B3J1N9      | (902) 446-4388 | Not Available  |
| Enterprise Rent-A-Car | 100 Victoria St W       |           | Kamloops            | BC       | V2C1A4      | (250) 374-8288 | Not Available  |
| Enterprise Rent-A-Car | 3025 Airport Rd         |           | Kamloops            | BC       | V2B7W9      | (250) 376-2883 | Not Available  |
| Enterprise Rent-A-Car | 5533 Airport Way        | 11        | Kelowna             | BC       | V1V1S1      | (250) 491-9611 | (250) 491-9615 |
| Enterprise Rent-A-Car | 22 Government Rd E      |           | Kirkland Lake       | ON       | P2N1A3      | (705) 567-3277 | Not Available  |
| Enterprise Rent-A-Car | 509 Wilson Ave          | 15        | Kitchener           | ON       | N2C2M4      | (519) 893-6800 | Not Available  |
| Enterprise Rent-A-Car | 19898 Langley Bypass    |           | Langley             | BC       | V3A4Y1      | (604) 532-8969 | (604) 532-8912 |
|                       |                         |           |                     |          |             |                |                |
| Enterprise Rent-A-Car | 1555 Boul Cure-Labelle  |           | Laval               | QC       | H7V2W5      | (450) 688-0414 | (450) 688-4421 |
| Enterprise Rent-A-Car | 1106 3 Ave S            |           | Lethbridge          | AB       | Т1Ј0Ј6      | (403) 328-3517 | Not Available  |
| Enterprise Rent-A-Car | 1111 Oxford St E        |           | London              | ON       | N5Y3L7      | (519) 455-9585 | Not Available  |
| Enterprise Rent-A-Car | 288 Horton St E         |           | London              | ON       | N6B1L4      | (519) 439-8507 | Not Available  |
| Enterprise Rent-A-Car | 500 Oxford St W         |           | London              | ON       | N6H1T5      | (519) 472-6676 | (519) 472-6977 |
| Enterprise Rent-A-Car | 750 Wharncliffe Rd S    |           | London              | ON       | N6J2N4      | (519) 681-8500 | (519) 681-8416 |
| Enterprise Rent-A-Car | 7 Heritage Rd           |           | Markham             | ON       | L3P1M3      | (905) 471-2335 | Not Available  |
| Enterprise Rent-A-Car | 7390 Woodbine Ave       | 200       | Markham             | ON       | L3R1A5      | (905) 477-1688 | Not Available  |
|                       | 1071 Trans Canada Way   |           |                     |          |             |                |                |
| Enterprise Rent-A-Car | SE                      | 1         | Medicine Hat        | AB       | T1B1H9      | (403) 526-8064 | (403) 528-4665 |
| Enterprise Rent-A-Car | 88 Ontario St N         | 2         | Milton              | ON       | L9T2T1      | (905) 875-4515 | Not Available  |
| Enterprise Rent-A-Car | 1475 Aerowood Dr        | 2         | Mississauga         | ON       | L4W1C2      | (905) 238-0775 | Not Available  |
| Enterprise Rent-A-Car | 1490 Dundas St E        | В         | Mississauga         | ON       | L4X1L4      | (905) 897-5036 | Not Available  |
| Enterprise Rent-A-Car | 3111 Woodchester Dr     |           | Mississauga         | ON       | L5L1J2      | (905) 828-2377 | Not Available  |
| Enterprise Rent-A-Car | 6465 Millcreek Dr       | 15        | Mississauga         | ON       | L5N5R3      | (905) 812-8445 | Not Available  |
| Enterprise Rent-A-Car | 777 Dundas St W         | B1        | Mississauga         | ON       | L5C4P6      | (905) 281-0869 | Not Available  |
| Enterprise Rent-A-Car | 1903 Main St            |           | Moncton             | NB       | E1E4V8      | (506) 857-9000 | Not Available  |

|                       |                       | Car Renta | al Agencies in Canada | a        |             |                |                |
|-----------------------|-----------------------|-----------|-----------------------|----------|-------------|----------------|----------------|
|                       |                       | Suite     |                       |          | 1           |                | _ <u></u>      |
| Company Name          | Address               | Number    | City                  | Province | Postal Code | Phone Number   | Fax Number     |
| Enterprise Rent-A-Car | 33 Av Mozart E        |           | Montreal              | QC       | H2S1B1      | (514) 273-4284 | Not Available  |
| Enterprise Rent-A-Car | 1154 Topsail Rd       |           | Mount Pearl           | NL       | A1N5E8      | (709) 700-0805 | Not Available  |
| Enterprise Rent-A-Car | 835 Topsail Rd        | 3         | Mount Pearl           | NL       | A1N3J6      | (709) 739-6570 | (709) 739-6580 |
| Enterprise Rent-A-Car | 623 County Rd 41      |           | Napanee               | ON       | K7R3L2      | (613) 354-2262 | Not Available  |
| Enterprise Rent-A-Car | 2194 Robertson Rd     | 11        | Nepean                | ON       | K2H9J5      | (613) 596-2255 | (613) 596-2185 |
| Enterprise Rent-A-Car | 9241 Commercial St    |           | New Minas             | NS       | B4N3G1      | (902) 679-4919 | Not Available  |
| Enterprise Rent-A-Car | 1321 Third Ave        |           | New Westminster       | вс       | V3M1R3      | (604) 523-1377 | Not Available  |
| Enterprise Rent-A-Car | 17065 Yonge St        |           | Newmarket             | ON       | L3Y4V7      | (905) 967-1062 | Not Available  |
| Enterprise Rent-A-Car | 17725 Yonge St N      | 16        | Newmarket             | ON       | L3Y7C1      | (905) 895-9722 | Not Available  |
| Enterprise Rent-A-Car | 1881 Cassells St      | C348      | North Bay             | ON       | P1B4C8      | (705) 498-6224 | Not Available  |
| Enterprise Rent-A-Car | 282 Airport Rd        |           | North Bay             | ON       | P1B8W6      | (705) 498-6224 | Not Available  |
| Enterprise Rent-A-Car | 751 Marine Dr         |           | North Vancouver       | BC       | V7M1H4      | (604) 988-5878 | (604) 988-5819 |
| Enterprise Rent-A-Car | 2835 Dufferin St      |           | North York            | ON       | M6B3S4      | (416) 787-2636 | Not Available  |
| Enterprise Rent-A-Car | 4700 Dufferin St      | 4         | North York            | ON       | M3H5S7      | (416) 633-2211 | (416) 633-1779 |
|                       | 2071 South Service Rd |           |                       |          |             |                |                |
| Enterprise Rent-A-Car | W                     |           | Oakville              | ON       | L6L5W2      | (905) 338-8747 | (905) 465-1408 |
| Enterprise Rent-A-Car | 547 Trafalgar Rd      |           | Oakville              | ON       | L6J3J1      | (905) 339-0852 | (905) 339-2836 |
| Enterprise Rent-A-Car | 184 King St W         |           | Oshawa                | ON       | L1J2J2      | (905) 571-1571 | (905) 571-0527 |
| Enterprise Rent-A-Car | 423 Bloor St W        | 1         | Oshawa                | ON       | L1J5Y5      | (905) 728-3370 | Not Available  |
| Enterprise Rent-A-Car | 1615 Laperriere Ave   |           | Ottawa                | ON       | K1Z8S7      | (613) 729-4009 | (613) 729-2648 |
| Enterprise Rent-A-Car | 1788 Rue Bank         |           | Ottawa                | ON       | K1V7Y6      | (613) 247-0220 | (613) 247-7664 |
| Enterprise Rent-A-Car | 226 Queen St          |           | Ottawa                | ON       | K1P5E4      | (613) 236-3366 | Not Available  |
| Enterprise Rent-A-Car | 297 Leckie Pvt        |           | Ottawa                | ON       | K1V1S3      | (613) 247-0796 | (613) 247-1465 |
| Enterprise Rent-A-Car | 1322 Pembroke St W    |           | Pembroke              | ON       | K8A7A2      | (613) 635-7368 | (613) 635-7139 |
| Enterprise Rent-A-Car | 975 Clonsilla Ave     |           | Peterborough          | ON       | K9J5Y2      | (705) 748-2707 | (705) 748-2811 |
| Enterprise Rent-A-Car | 715 Kingston Rd       |           | Pickering             | ON       | L1V1A9      | (905) 839-3284 | Not Available  |
| Enterprise Rent-A-Car | 1050 Patricia Blvd    |           | Prince George         | BC       | V2L3V8      | (250) 614-8366 | (250) 614-8324 |
| Enterprise Rent-A-Car | 3422 50 Ave           |           | Red Deer              | AB       | T4N3Y4      | (403) 347-5145 | Not Available  |
| Enterprise Rent-A-Car | 4770 Riverside Dr     | 100       | Red Deer              | AB       | T4N2N7      | (403) 348-2550 | Not Available  |
| Enterprise Rent-A-Car | 1631 Victoria Ave     |           | Regina                | SK       | S4P0P8      | (306) 352-3222 | (306) 352-2787 |

•

|                       |                       | Car Renta | al Agencies in Canac | la       |             |                |                |
|-----------------------|-----------------------|-----------|----------------------|----------|-------------|----------------|----------------|
|                       |                       | Suite     |                      |          |             |                |                |
| Company Name          | Address               | Number    | City                 | Province | Postal Code | Phone Number   | Fax Number     |
| Enterprise Rent-A-Car | 200 Albert St         |           | Regina               | SK       | S4R2N4      | (306) 352-4222 | (306) 352-8482 |
| Enterprise Rent-A-Car | 5201 Regina Ave       |           | Regina               | SK       | S4W1B3      | (306) 359-3535 | (306) 359-3537 |
| Enterprise Rent-A-Car | 11051 Bridgeport Rd   | 128       | Richmond             | BC       | V6X1T3      | (604) 278-1223 | (604) 278-8076 |
| Enterprise Rent-A-Car | 9051 Beckwith Rd      |           | Richmond             | BC       | V6X1V7      | (604) 303-1117 | Not Available  |
| Enterprise Rent-A-Car | 10737 Yonge St        | 17        | Richmond Hill        | ON       | L4C9M9      | (905) 508-6981 | (905) 508-8520 |
| Enterprise Rent-A-Car | 60 Industrial Rd      |           | Richmond Hill        | ON       | L4C2Y1      | (905) 508-6981 | Not Available  |
| Enterprise Rent-A-Car | 132 Station St        |           | Saint John           | NB       | E2L3H6      | (506) 693-8688 | (506) 693-6504 |
| Enterprise Rent-A-Car | 782 Confederation St  |           | Sarnia               | ON       | N7T2E3      | (519) 337-7000 | Not Available  |
| Enterprise Rent-A-Car | 1206 Central Ave      |           | Saskatoon            | SK       | S7N2H2      | (306) 244-6900 | Not Available  |
| Enterprise Rent-A-Car | 2130 Airport Dr       |           | Saskatoon            | SK       | S7L6M6      | (306) 652-9119 | (306) 244-1813 |
| Enterprise Rent-A-Car | 2625 Airport Dr       | 17        | Saskatoon            | SK       | \$7L7L1     | (306) 664-4454 | (306) 665-1602 |
| Enterprise Rent-A-Car | 475 Airport Rd        |           | Sault Ste. Marie     | ON       | P6A5K6      | (705) 779-2727 | Not Available  |
| Enterprise Rent-A-Car | 97 Great Northern Rd  |           | Sault Ste. Marie     | ON       | P6B4Y9      | (705) 254-3227 | (705) 254-5504 |
| Enterprise Rent-A-Car | 2005 Eglinton Ave E   |           | Scarborough          | ON       | M1L2M9      | (416) 750-3712 | (416) 750-4191 |
| Enterprise Rent-A-Car | 4865 Sheppard Ave E   | 2         | Scarborough          | ON       | M1S3V8      | (416) 299-0096 | (416) 299-3288 |
| Enterprise Rent-A-Car | 404 Ontario St        | 8         | St Catharines        | ON       | L2R5L9      | (905) 685-1500 | (905) 685-8578 |
| Enterprise Rent-A-Car | 449 Queen St W        |           | St Marys             | ON       | N4X1B7      | (519) 284-0723 | (519) 284-3160 |
| Enterprise Rent-A-Car | 229 Kenmount Rd       | 101       | St. John's           | NL       | A1B3P9      | (709) 722-9480 | Not Available  |
| Enterprise Rent-A-Car | 80 Airport Rd         |           | St. John's           | NL       | A1A4Y3      | (709) 738-3900 | Not Available  |
| Enterprise Rent-A-Car | 80 Craig Dobbin's Way | l.        | St. John's           | NL       | A1A5T2      | (709) 738-3900 | (709) 738-3300 |
| Enterprise Rent-A-Car | 1715 Regent St        |           | Sudbury              | ON       | P3E3Z7      | (705) 522-5888 | (705) 522-1128 |
| Enterprise Rent-A-Car | 900 Kingsway          |           | Sudbury              | ON       | P3B2E5      | (705) 675-3227 | Not Available  |
| Enterprise Rent-A-Car | 446 Grand Lake Rd     |           | Sydney               | NS       | B1P5S8      | (902) 539-6700 | (902) 539-8368 |
| Enterprise Rent-A-Car | 248 Steeles Ave W     | 5         | Thornhill            | ON       | L4J1A1      | (905) 882-0400 | (905) 882-5529 |
| Enterprise Rent-A-Car | 100 Princess St       |           | Thunder Bay          | ON       | P7E6S2      | (807) 473-5222 | (807) 473-5225 |
| Enterprise Rent-A-Car | 1100 Arthur St W      |           | Thunder Bay          | ON       | P7E6P9      | (807) 622-1222 | (807) 623-0432 |
| Enterprise Rent-A-Car | 727 Memorial Ave      |           | Thunder Bay          | ON       | P7B3Z7      | (807) 344-2800 | (807) 344-6257 |
| Enterprise Rent-A-Car | 1092 Riverside Dr     |           | Timmins              | ON       | P4R1A1      | (705) 264-3227 | (705) 264-3282 |
| Enterprise Rent-A-Car | 700 Bay St            | 19        | Toronto              | ON       | M5G1Z6      | (416) 599-1375 | Not Available  |
| Enterprise Rent-A-Car | 1250 Granville St     |           | Vancouver            | BC       | V6Z1M4      | (604) 688-5500 | Not Available  |

|                        | Car Rental Agencies in Canada |        |             |          |             |                |                |  |  |  |
|------------------------|-------------------------------|--------|-------------|----------|-------------|----------------|----------------|--|--|--|
|                        |                               | Suite  |             |          | <u> </u>    |                |                |  |  |  |
| Company Name           | Address                       | Number | City        | Province | Postal Code | Phone Number   | Fax Number     |  |  |  |
| Enterprise Rent-A-Car  | 3510 Fraser St                |        | Vancouver   | BC       | V5V4C6      | (604) 872-7368 | (604) 872-7319 |  |  |  |
| Enterprise Rent-A-Car  | 970 Marine Dr SW              | 10     | Vancouver   | BC       | V6P5Z2      | (604) 266-0883 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 2507 Government St            |        | Victoria    | BC       | V8T4P6      | (250) 475-6900 | (250) 382-6900 |  |  |  |
| Enterprise Rent-A-Car  | 854 Murray St                 |        | Wallaceburg | ON       | N8A1W4      | (519) 627-2210 | (519) 351-1847 |  |  |  |
| Enterprise Rent-A-Car  | 379 Weber St N                |        | Waterloo    | ON       | N2J3J2      | (519) 884-6001 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 1230 Dundas St E              |        | Whitby      | ON       | L1N2K5      | (905) 430-4535 | (905) 430-4521 |  |  |  |
| Enterprise Rent-A-Car  | 1014 Tecumseh Rd E            |        | Windsor     | ON       | N8X2S8      | (519) 966-8500 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 9250 Tecumseh Rd E            |        | Windsor     | ON       | N8R1A2      | (519) 735-0755 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 1015 King Edward St           |        | Winnipeg    | MB       | R3HOR1      | (204) 779-9999 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 1392 Regent Ave W             | A      | Winnipeg    | MB       | R2C3A8      | (204) 668-0808 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 1700 Waverley St              |        | Winnipeg    | MB       | R3M2M7      | (204) 488-2292 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 2000 Wellington Ave           | 249    | Winnipeg    | MB       | R3H1C2      | (204) 779-2422 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 2195 Pembina Hwy              | 4      | Winnipeg    | MB       | R3T5S9      | (204) 269-5500 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 283 Ellice Ave                |        | Winnipeg    | MB       | R3B1X6      | (204) 925-3525 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | C Portage Ave                 | 35C    | Winnipeg    | MB       | R3K2H7      | (204) 831-4244 | Not Available  |  |  |  |
| Enterprise Rent-A-Car  | 1000 Rowntree Dairy Rd        | 5      | Woodbridge  | ON       | L4L5X3      | (289) 236-1462 | Not Available  |  |  |  |
| G N Auto Rentals       | 5730 Marine Dr                |        | Burnaby     | BC       | V5J3G8      | (604) 437-4434 | (604) 437-9990 |  |  |  |
| Gillam Rentals         | 123 Airport Rd                |        | Gillam      | MB       | ROBOLO      | (204) 652-2109 | (204) 652-2980 |  |  |  |
| Gta Car Truck Rentals  | 55 Stafford Dr                | F      | Brampton    | ON       | L6W1L3      | (905) 792-2277 | Not Available  |  |  |  |
| Hertz Location D'Autos | 555 Rue Arthur-Fecteau        |        | Dorval      | QC       | H4Y1J7      | (450) 476-3385 | Not Available  |  |  |  |
| Hertz Location D'Autos | 975 Boul Romeo-Vachon         |        | Dorval      | QC       | H4Y1H1      | (514) 636-9530 | (514) 636-9503 |  |  |  |
| Hertz Location D'Autos | 623 Boul Saint-Joseph         |        | Gatineau    | QC       | J8Y4A6      | (819) 770-7500 | (819) 770-7502 |  |  |  |
|                        | 1450 Boul Marie               |        |             |          |             |                |                |  |  |  |
| Hertz Location D'Autos | Victorin                      |        | Longueuil   | oc       | J4G1A4      | (450) 928-3464 | (450) 678-1226 |  |  |  |
| Hertz Location D'Autos | 1475 Aylmer St                |        | Montreal    | QC       | H3A2C6      | (514) 842-8537 | (514) 842-4820 |  |  |  |
| Hertz Location D'Autos | 2825 Boul Laurier             |        | Quebec      | QC       | G1V2L9      | (418) 658-6795 | (418) 658-8003 |  |  |  |
| Hertz Location D'Autos | 44 Cote Du Palais             |        | Quebec      | QC       | G1R4H8      | (418) 694-1224 | Not Available  |  |  |  |
| Hertz Location D'Autos | 505 Rue Principale            |        | Quebec      | QC       | G2G0J4      | (418) 871-1571 | (418) 871-6139 |  |  |  |

|                  | Car Rental Agencies in Canada |        |                  |          |             |                |                        |  |  |  |
|------------------|-------------------------------|--------|------------------|----------|-------------|----------------|------------------------|--|--|--|
|                  |                               | Suite  |                  |          |             |                |                        |  |  |  |
| Company Name     | Address                       | Number | City             | Province | Postal Code | Phone Number   | Fax Number             |  |  |  |
| Hertz Rent A Car | 117 Kennedy Rd S              | 215    | Brampton         | ON       | L6W3G3      | (905) 451-1550 | (905) 451-8656         |  |  |  |
| Hertz Rent A Car | 4881 Fountain St              | 33     | Breslau          | ON       | NOB1M0      | (519) 648-3404 | (519) 648-3040         |  |  |  |
| Hertz Rent A Car | 15180 Auto Row Hwy 3          |        | Bridgewater      | NS       | B4V2W3      | (902) 543-0353 | Not Available          |  |  |  |
| Hertz Rent A Car | 3360 Harvester Rd             |        | Burlington       | ON       | L7N3M8      | (905) 333-8844 | Not Available          |  |  |  |
| Hertz Rent A Car | 45730 Hocking Ave             |        | Chilliwack       | BC       | V2P1B3      | (604) 701-3383 | Not Available          |  |  |  |
| Hertz Rent A Car | 777 Aviation Ave              |        | Dieppe           | NB       | E1A7Z5      | (506) 858-8525 | (506) 854-7333         |  |  |  |
|                  | 435 Airport Perimeter         | _      |                  |          |             |                |                        |  |  |  |
| Hertz Rent A Car | Rd                            |        | Edmontn Intl Arp | AB       | T9E0V4      | (780) 890-4436 | Not Available          |  |  |  |
| Hertz Rent A Car | 2 Convair Dr                  |        | Etobicoke        | ON       | M9W7A1      | (416) 674-2020 | (416) 213-4433         |  |  |  |
| Hertz Rent A Car | 315 Macalpine Cres            |        | Fort Mcmurray    | AB       | T9H4Y4      | (780) 743-4047 | Not Available          |  |  |  |
| Hertz Rent A Car | 659 Queen St                  |        | Fredericton      | NB       | E3B1C3      | (506) 459-3444 | Not Available          |  |  |  |
| Hertz Rent A Car | 2600 Bank St                  |        | Gloucester       | ON       | K1T1M9      | (613) 521-3294 | (613) 521-4563         |  |  |  |
| Hertz Rent A Car | 10610 Airport Dr              | 103D   | Grande Prairie   | AB       | T8V7Z5      | (780) 539-3045 | (780) 402-9722         |  |  |  |
| Hertz Rent A Car | 2418 Barton St E              |        | Hamilton         | ON       | L8E2W7      | (905) 560-8522 | Not Available          |  |  |  |
| Hertz Rent A Car | 57 Glennie Ave                |        | Hamilton         | ON       | L8H5V6      | (905) 547-9936 | Not Available          |  |  |  |
| Hertz Rent A Car | 525 Legget Dr                 | 11     | Kanata           |          | K2K2W2      | (613) 591-3377 | (613) 591-1064         |  |  |  |
| Hertz Rent A Car | 5923 Production Way           |        | Langley          | BC       | V3A4N5      | (604) 532-9360 | Not Available          |  |  |  |
| Hertz Rent A Car | 3120 Steeles Ave E            |        | Markham          | ON       | L3R1G9      | (905) 475-0727 | Not Available          |  |  |  |
| Hertz Rent A Car | 3000 Woodchester Dr           |        | Mississauga      | ON       | L5L2R4      | (905) 828-1566 | Not Available          |  |  |  |
| Hertz Rent A Car | 470 Mountain Rd               |        | Moncton          | NB       | E1C2N3      | (506) 858-5519 | (506) 857-2814         |  |  |  |
| Hertz Rent A Car | 1916 Merivale Rd              |        | Nepean           | ON       | K2G1E8      | (613) 228-0330 | (613) 228-8917         |  |  |  |
| Hertz Rent A Car | 1640 Electra Blvd             | 130    | North Saanich    | BC       | V8L5V4      | (250) 657-0380 | Not Available          |  |  |  |
| Hertz Rent A Car | 3200 Dufferin St              | 16B    | North York       | ON       | M6A3B2      | (416) 256-7311 | Not Available          |  |  |  |
| Hertz Rent A Car | 3006 St Joseph Blvd           |        | Orleans          | ON       | K1E1E2      | (613) 841-0777 | Not Available          |  |  |  |
|                  | 1003 Airport Parkway          |        |                  |          | 1           |                |                        |  |  |  |
| Hertz Rent A Car | Pvt                           |        | Ottawa           | ON       | K1V2C1      | (613) 521-3332 | (613) <u>5</u> 21-2287 |  |  |  |
| Hertz Rent A Car | 1160 Newmarket St             |        | Ottawa           | ON       | K1B1A6      | (613) 745-1396 | (613) 474-1122         |  |  |  |
| Hertz Rent A Car | 30 York St                    |        | Ottawa           | ON       | K1N5S6      | (613) 241-7681 | (613) 241-6072         |  |  |  |
| Hertz Rent A Car | 881 Boul St Laurent           |        | Ottawa           | ON       | K1K3B1      | (613) 746-6683 | (613) 746-7049         |  |  |  |

-

| Car Rental Agencies in Canada |                       |        |                 |          |             |                |                |  |  |  |
|-------------------------------|-----------------------|--------|-----------------|----------|-------------|----------------|----------------|--|--|--|
|                               |                       | Suite  | · ·             |          |             |                |                |  |  |  |
| Company Name                  | Address               | Number | City            | Province | Postal Code | Phone Number   | Fax Number     |  |  |  |
|                               | Ottawa Airport, 1003  |        |                 |          |             |                |                |  |  |  |
| Hertz Rent A Car              | Airport Road          |        | Ottawa          | ON       | K1B 9B4     | (613) 521-3332 | Not Available  |  |  |  |
| Hertz Rent A Car              | 970 Brock Rd          |        | Pickering       | ON       | L1W2A1      | (905) 839-9844 | (905) 837-8940 |  |  |  |
| Hertz Rent A Car              | 4141 Airport          |        | Prince George   | BC       | V2L3H9      | (250) 963-0300 | Not Available  |  |  |  |
| Hertz Rent A Car              | 2755 Caribou St       |        | Regina          | SK       | S4W1A5      | (306) 791-9131 | (306) 585-1548 |  |  |  |
| Hertz Rent A Car              | 3846 Mcdonald Rd      |        | Richmond        | BC       | V7B1L8      | (604) 606-3700 | (604) 606-3799 |  |  |  |
|                               | 5140 Grant Mcconachie |        |                 |          |             |                |                |  |  |  |
| Hertz Rent A Car              | Way                   | 4      | Richmond        | BC       | V7A4X6      | (604) 276-0356 | Not Available  |  |  |  |
| Hertz Rent A Car              | 4180 Loch Lomond Rd   |        | Saint John      | NB       | E2N1L7      | (506) 634-6655 | (506) 634-5393 |  |  |  |
| Hertz Rent A Car              | 2507 Eglinton Ave E   |        | Scarborough     | ON       | M1K2R1      | (416) 261-4234 | Not Available  |  |  |  |
| Hertz Rent A Car              | 4032 Finch Ave E      |        | Scarborough     | ON       | M1S3T6      | (416) 396-2835 | Not Available  |  |  |  |
|                               |                       |        |                 |          |             |                |                |  |  |  |
| Hertz Rent A Car              | 80 Craig Dobbin's Way |        | St. John's      | NL       | A1A5T2      | (709) 726-0795 | Not Available  |  |  |  |
| Hertz Rent A Car              | 13409 72 Ave          | В      | Surrey          | BC       | V3W2N7      | (604) 592-5650 | Not Available  |  |  |  |
| Hertz Rent A Car              | 7545 Yonge St         |        | Thornhill       | ON       | L3T2C2      | (905) 707-0522 | (905) 707-5960 |  |  |  |
| Hertz Rent A Car              | 1480 Walsh St W       |        | Thunder Bay     | ON       | P7E6H6      | (807) 475-3334 | Not Available  |  |  |  |
|                               | Thunder Bay Airport,  |        |                 |          |             |                |                |  |  |  |
| Hertz Rent A Car              | 100 Princess Street   |        | Thunder Bay     | ON       | P7E6S2      | (807) 473-8111 | (807) 473-8290 |  |  |  |
| Hertz Rent A Car              | 65 Front St W         |        | Toronto         | ON       | M5J1E6      | (416) 364-2080 | (416) 364-2009 |  |  |  |
| Hertz Rent A Car              | 1038 Canada Pl        |        | Vancouver       | BC       | V6C0B9      | (604) 482-5544 | Not Available  |  |  |  |
| Hertz Rent A Car              | 1278 Granville St     |        | Vancouver       | BC       | V6Z3B2      | (604) 606-4711 | (604) 606-4728 |  |  |  |
| Hertz Rent A Car              | 396 Marine Dr SW      |        | Vancouver       | BC       | V5X2R6      | (604) 324-9425 | Not Available  |  |  |  |
|                               | Halifax International |        |                 |          |             |                |                |  |  |  |
| Hertz Rent A Car              | Airport, 1 Bell Blvd  |        | Halifax         | NS       | В2Т1КЗ      | (902) 873-3700 | Not Available  |  |  |  |
| Hertz Rent A Car              | 1255 Ouellette Ave    |        | Windsor         | ON       | N8X1J3      | (519) 560-3685 | Not Available  |  |  |  |
| Hertz Rent A Car              | 2000 Wellington Ave   | 249    | Winnipeg        | МВ       | R3H1C2      | (204) 925-6625 | (204) 774-2940 |  |  |  |
| Holiand Automotive            | 8525 Boul Decarie     |        | Mont-Royal      | QC       | H4P2J2      | (514) 737-3577 | (514) 737-0514 |  |  |  |
| bargain Rent A Car Ltd        | 1105 Granville St     |        | Vancouver       | BC       | V6Z1M1      | (604) 689-9666 | Not Available  |  |  |  |
| Location Viau Autos & Camions | 9055 Boul Pie-IX      | -      | Montreal        | QC       | H1Z3V6      | (514) 321-2700 | Not Available  |  |  |  |
|                               |                       |        |                 |          |             |                |                |  |  |  |
| o-Cost Rent-A-Car & Truck     | 1835 Marine Dr        |        | North Vancouver | вс       | V7P1V5      | (604) 986-1266 | (604) 986-8792 |  |  |  |

|                              |  | Car Renta | al Agencies in Canad | а        |             |                |                |
|------------------------------|--|-----------|----------------------|----------|-------------|----------------|----------------|
|                              |  | Suite     |                      |          |             |                |                |
| Company Name                 | Address  | Number    | City                 | Province | Postal Code | Phone Number   | Fax Number     |
| Luxury Auto Rental & Sales   | 186 Merritt St                                     |           | St Catharines        | ON       | L2T1J6      | (905) 680-9191 | (905) 680-9872 |
| Monahan Ford Rentals &       |  |           |                      | 1        |             |                |                |
| Leasing                      | Po Box 1869  |           | High Prairie         | AB       | T0G1E0      | (780) 523-5600 | (780) 523-5000 |
| Mr Rent Car                  | 8511 Capstan Way                                   | 110       | Richmond             | BC       | V6X1R3      | (604) 303-0063 | Not Available  |
| Mr Rent-A-Car                | 968 Kingsway                                       |           | Vancouver            | BC       | V5V3C4      | (604) 876-7777 | Not Available  |
| National Car & Truck Rentals | 5 Industrial Dr                                    |           | Campbellford         | ON       | KOL1LO      | (705) 653-9991 | Not Available  |
| National Car & Truck Rentals | 5116 Gateway Blvd NW                               |           | Edmonton             | AB       | T6H2H4      | (780) 436-8631 | (780) 437-2667 |
| National Car & Truck Rentals | Grande Prairie Airport,<br>10601 Airport Drive     |           | Grande Prairie       | AB       | T8V7Z5      | (780) 539-2435 | (780) 532-6454 |
| National Car Rental          | 30440 Liberator Ave                                |           | Abbotsford           | BC       | V2S1A1      | (604) 850-6534 | Not Available  |
| National Car Rental          | 55 Mountainash Rd                                  |           | Brampton             | ON       | L6R1W4      | (905) 799-7276 | Not Available  |
| National Car Rental          | 680 Stewart Blvd                                   |           | Brockville           | ON       | K6V5T4      | (613) 342-1177 | (613) 342-4283 |
| National Car Rental          | 2000 Airport Rd NE                                 | 4         | Calgary              | AB       | T2E6W5      | (403) 221-1690 | Not Available  |
| National Car Rental          | 2000 Jubilee                                       |           | Campbell River       | BC       | V9H1T5      | (250) 923-1234 | Not Available  |
| National Car Rental          | 1250 Knight Rd                                     | 106       | Comox                | BC       | V9M4H2      | (250) 339-0611 | Not Available  |
| National Car Rental          | 1520 Vincent Massey Dr                             |           | Cornwall             | ON       | K6H5R6      | (613) 933-2460 | (613) 932-3124 |
| National Car Rental          | 320 Old Island Hwy                                 |           | Courtenay            | BC       | V9N3P2      | (250) 334-0202 | (250) 334-0222 |
| National Car Rental          | 4563 36th St E                                     |           | Edmontn Intl Arp     | AB       | T9E0V4      | (780) 890-7700 | Not Available  |
| National Car Rental          | 9943 Jasper Ave NW                                 |           | Edmonton             | AB       | T5J2X4      | (780) 422-6097 | Not Available  |
|                              | Edmonton International<br>Airport, 1, 1000 Airport |           |                      |          |             |                |                |
| National Car Rental          | Road   |           | Leduc                | AB       | T9E8B7      | (780) 890-7232 | (780) 890-7009 |
| National Car Rental          | 280 Attwell Dr                                     |           | Etobicoke            | ON       | M9W5B2      | (416) 798-8802 | (416) 798-9753 |
| National Car Rental          | 4810 49th Ave                                      |           | Fort Nelson          | BC       | V0C1R0      | (250) 774-2679 | Not Available  |
| National Car Rental          | 257 Rd   |           | Fort St. John        | BC       | V1J4M6      | (250) 785-5590 | (250) 785-5592 |
| National Car Rental          | 5000 Air Terminal Dr                               | T108      | Garson               | ON       | P3L1V4      | (705) 693-5833 | Not Available  |
| National Car Rental          | 1003 Airport Pkwy                                  |           | Gloucester           | ON       | K1V9B3      | (613) 737-7023 | Not Available  |

|                     | <u> </u>              | Car Renta | al Agencies in Canad | la       |             |                |                |
|---------------------|-----------------------|-----------|----------------------|----------|-------------|----------------|----------------|
|                     |                       | Suite     |                      |          |             | T              |                |
| Company Name        | Address               | Number    | City                 | Province | Postal Code | Phone Number   | Fax Number     |
| National Car Rental | 915 11th Ave N        |           | Golden               | BC       | V0A1H0      | (250) 344-9899 | (250) 344-2009 |
| National Car Rental | 9904 121 Ave          |           | Grande Prairie       | AB       | T8V4J1      | (780) 532-3761 | (780) 532-6454 |
| National Car Rental | Halifax Intl Airport  |           | Halifax              | NS       | B2T1K3      | (902) 873-3505 | Not Available  |
| National Car Rental | 174 Victoria St W     |           | Kamloops             | BC       | V2C1A4      | (250) 374-5737 | (250) 374-3329 |
| National Car Rental | 3035 Airport Rd       |           | Kamloops             | BC       | V2B7X1      | (250) 376-4911 | (250) 374-3329 |
| National Car Rental | 5533 Airport Way      | 13        | Kelowna              | BC       | V1V1S1      | (250) 765-2800 | Not Available  |
| National Car Rental | 926 King St E         |           | Kitchener            | ON       | N2G2M8      | (519) 742-5815 | Not Available  |
| National Car Rental | 936 King St W         |           | Kitchener            | ON       | N2G1G4      | (519) 742-5814 | (519) 742-0502 |
| National Car Rental | 312 Enterprise Ave    | С         | Kitimat              | BC       | V8C2C9      | (250) 632-6112 | Not Available  |
| National Car Rental | 780 Lahakas Blvd S    |           | Kitimat              | BC       | V8C1T9      | (250) 632-6112 | Not Available  |
| National Car Rental | 1750 Crumlin          |           | London               | ON       | N5V3B6      | (519) 455-7413 | Not Available  |
| National Car Rental | 67 Peninsula Rd       |           | Marathon             | ON       | POT2E0      | (807) 229-0194 | (807) 229-3030 |
| National Car Rental | 5 Mcintosh Dr         |           | Markham              | ON       | L3R8C8      | (905) 477-4860 | Not Available  |
| National Car Rental | 1602 Northfield Rd    |           | Nanaimo              | BC       | V9S3A7      | (250) 758-3509 | (250) 758-9419 |
| National Car Rental | 17725 Yonge St        | 16        | Newmarket            | ON       | L3Y7C1      | (905) 895-6227 | (905) 895-9736 |
| National Car Rental | 1640 Electra Blvd     | 134       | North Saanich        | BC       | V8L5V4      | (250) 656-2541 | Not Available  |
| National Car Rental | 1291 1st Ave          |           | Prince George        | BC       | V2L2Y7      | (250) 564-4847 | (250) 564-4804 |
| National Car Rental | 4141 Airport Rd       |           | Prince George        | BC       | V2N4M6      | (250) 963-7473 | (250) 963-9179 |
| National Car Rental | 815 1st Ave W         | 106       | Prince Rupert        | BC       | V8J1B3      | (250) 624-5318 | (250) 624-5367 |
| National Car Rental | 37428 Range Road 273  | 172       | Red Deer County      | AB       | T4E0A1      | (403) 347-5811 | Not Available  |
| National Car Rental | 2627 Airport Rd       |           | Regina               | SK       | S4W1A4      | (306) 757-5757 | Not Available  |
|                     | Vancouver Internati   |           |                      |          |             |                |                |
|                     | Airport, 3211 Grant   |           |                      |          |             |                |                |
| National Car Rental | Mcconachie Way        |           | Richmond             | вс       | V7B0A4      | (604) 207-3730 | Not Available  |
| National Car Rental | 430 Ch Larocque       |           | Salaberry-Vlyfid     | QC       | J6T4C5      | (450) 373-9465 | (450) 371-6489 |
| National Car Rental | 1500 Airport Rd       |           | Sarnia               | ON       | N7W1B6      | (519) 542-5231 | Not Available  |
| National Car Rental | 2625 Airport Dr       |           | Saskatoon            | SK       | S7L7L1      | (306) 665-5665 | Not Available  |
| National Car Rental | 308 Great Northern Rd |           | Sault Ste. Marie     | ON       | P6B4Z7      | (705) 949-5121 | Not Available  |
| National Car Rental | 152 Airport Rd        |           | St. John's           | NL       | A1A4Y3      | (709) 722-4307 | Not Available  |
| National Car Rental | 4542 Lakelse Ave      |           | Terrace              | BC       | V8G1P8      | (250) 635-6855 | (250) 635-5665 |

|                                |                       | Car Renta | al Agencies in Cana | ada      | <u> </u>    |                |                |
|--------------------------------|-----------------------|-----------|---------------------|----------|-------------|----------------|----------------|
| · //                           |                       | Suite     |                     |          |             |                |                |
| Company Name                   | Address               | Number    | City                | Province | Postal Code | Phone Number   | Fax Number     |
| National Car Rental            | 450 Sifton Ave        |           | Thunder Bay         | ON       | P7E6M2      | (807) 577-1234 | (807) 457-3881 |
| National Car Rental            | 4599 Airport Rd       |           | Timmins             | ON       | P4N7C3      | (705) 264-2237 | Not Available  |
|                                | Airport, RR#2 Airport |           |                     |          |             | 1              |                |
| National Car Rental            | Road                  |           | Timmins             | ON       | P4N7C3      | (705) 474-5111 | Not Available  |
| National Car Rental            | 200 Wellington St W   |           | Toronto             | ON       | M5J1E6      | (416) 364-4191 | Not Available  |
| National Car Rental            | 16 Meadow Dr          |           | Truro               | NS       | B2N5V4      | (902) 895-5383 | (902) 893-3402 |
| National Car Rental            | Main 180 W Georgia    |           | Vancouver           | BC       | V6B4P4      | (604) 662-7803 | Not Available  |
| National Car Rental            | 2898 Douglas St       |           | Victoria            | BC       | V8T4M9      | (250) 386-1828 | (250) 386-1886 |
| National Car Rental            | 767 Douglas St        |           | Victoria            | BC       | V8W2B4      | (250) 386-1213 | (250) 381-1184 |
| National Car Rental            | 160 Eugenie St W      |           | Windsor             | ON       | N8X2X6      | (519) 966-7070 | (519) 966-2272 |
| National Car Rental            | 3200 County Rd 42     |           | Windsor             | ON       | N9A6J3      | (519) 966-9672 | (519) 966-9671 |
| National Car Rental            | 2000 Wellington Ave   |           | Winnipeg            | MB       | R3H1C1      | (204) 925-3529 | Not Available  |
| National Car Rental            | 323 Connell St        |           | Woodstock           | NB       | E7M5E7      | (506) 328-1967 | (506) 328-1824 |
| Nunavut Auto-Heavy             |                       |           |                     |          |             |                |                |
| Equipment                      | Bldg 2018             | В         | lqaluit             | NU       | ХОАОНО      | (867) 979-0922 | (867) 979-4326 |
| Oregan's Car & Truck Rentals   | 3224 Kempt Rd         |           | Halifax             | NS       | B3K4X1      | (902) 453-0444 | (902) 453-1988 |
| Payless Car Rental             | 17 Kelfield St        |           | Etobicoke           | ON       | M9W5A1      | (416) 245-8757 | Not Available  |
| Payless Car Rental             | 801 Dixon Rd          |           | Etobicoke           | ON       | M9W1J5      | (416) 674-6306 | Not Available  |
| Payless Car Rental             | 6465 Airport Rd       |           | Mississauga         | ON       | L4V1E4      | (416) 675-2000 | Not Available  |
| Payless Car Rental             | 677 Queen St E        |           | Toronto             | ON       | M4M1G6      | (416) 406-0908 | Not Available  |
| Payless Car Rental             | 1504 Regent Ave W     | -         | Winnipeg            | MB       | R2C3B4      | (204) 589-7111 | Not Available  |
| Payless Car Rental             | 167 Mcphillips St     |           | Winnipeg            | MB       | R3E2K1      | (204) 783-3111 | Not Available  |
| Peel Car & Truck Rentals       | 1102 Westport Cres    |           | Mississauga         | ON       | L5T1G1      | (905) 670-2442 | Not Available  |
| Phoenix Car & Truck Rentals    | 4808 50 St            | 407       | Red Deer            | AB       | T4N1X5      | (780) 460-8388 | Not Available  |
| Practi Car Car & Truck Rentals | 909 Topsail Rd        |           | Mount Pearl         | NL       | A1N3K1      | (709) 753-2277 | (709) 753-1108 |
| Practical New Car Rental       | 4494 Joyce Ave        |           | Powell River        | BC       | V8A3A6      | (604) 485-7929 | Not Available  |
| Practicar Car & Truck Rentals  | 302 Windmill Rd       | -         | Dartmouth           | NS       | B3A1H1      | (902) 469-0660 | Not Available  |

|                               |                      | Car Renta | al Agencies in Canad | la       |             |                |                |
|-------------------------------|----------------------|-----------|----------------------|----------|-------------|----------------|----------------|
|                               |                      | Suite     |                      |          |             | [              |                |
| Company Name                  | Address              | Number    | City                 | Province | Postal Code | Phone Number   | Fax Number     |
| Practicar Car & Truck Rentals | 3808 Bloor St W      |           | Etobicoke            | ON       | M9B6C2      | (416) 644-2022 | Not Available  |
| Practicar Car & Truck Rentals | 524 Nelson Ave       |           | Nelson               | вс       | V1L2N3      | (250) 352-5122 | Not Available  |
| Practicar Car & Truck Rentals | 8137 Old Waneta Rd   |           | Trail                | вс       | V1R4X1      | (250) 364-0211 | Not Available  |
| Rent-A-Wreck                  | 1480 Hunt Club Rd    |           | Ottawa               | ÓN       | K1T1M6      | (613) 521-1520 | Not Available  |
| Rent-A-Wreck                  | 610 South Dr         |           | Summerside           | PE       | C1N3Z7      | (902) 436-4757 | (902) 436-4944 |
| Rexdale Car & Truck Rental    | 305 Rexdale Blvd     |           | Etobicoke            | ON       | M9W1R8      | (416) 748-7283 | Not Available  |
| Riz Rent A Car & Truck Ltd    | 1361 Robson St       |           | Vancouver            | BC       | V6E1C6      | (604) 689-1231 | Not Available  |
| Ron Ridley Rentals Ltd        | 665 Oliver St        |           | Williams Lake        | BC       | V2G1M8      | (250) 392-6828 | (250) 392-7750 |
| Routes Car & Truck Rentals    | 3687 Nashua Dr       | 5         | Mississauga          | ON       | L4V1V5      | (416) 675-4445 | Not Available  |
| Runway Car Rentals            | 286 Attwell Dr       |           | Etobicoke            | ON       | M9W5B2      | (416) 213-8990 | Not Available  |
| Rustic Car Rentals            | 605 Oceanview Dr     |           | Queen Charlotte      | BC       | V0T1S0      | (250) 559-4641 | (250) 559-4557 |
| Sears Car & Truck Rental      | 451 Second St W      |           | Cornwall             | ON       | K6J1H3      | (613) 938-0454 | Not Available  |
| Sears Car & Truck Rental      | 10830 Rue Lajeunesse | 10        | Montreal             | QC       | H3L2E8      | (514) 482-4581 | Not Available  |
| Sears Rent-A-Car              | 271 Front St E       |           | Toronto              | ON       | M5A1G1      | (416) 368-0501 | Not Available  |
| Seven Car & Truck Rentals     | 435 Gregoire Dr      |           | Fort Mcmurray        | AB       | T9H4K7      | (780) 750-2710 | Not Available  |
| Thrifty Car Rental            | 2 Kennedy Rd South   | -<br>-    | Brampton             | ON       | L6W3E1      | (289) 632-3020 | Not Available  |
| Thrifty Car Rental            | 22 Kennedy Rd S      |           | Brampton             | ON       | L6W3E2      | (905) 454-5599 | (905) 874-9917 |
| Thrifty Car Rental            | 160 Charing Cross St |           | Brantford            | ON       | N3R2J4      | (519) 756-3779 | Not Available  |
| Thrifty Car Rental            | 720 Stewart Blvd     |           | Brockville           | ON       | K6V7H2      | (613) 498-1223 | Not Available  |
| Thrifty Car Rental            | 2005 55e Av          |           | Dorval               | QC       | H9P2Y6      | (514) 631-8000 | Not Available  |
| Thrifty Car Rental            | 1 Bell Blvd          |           | Enfield              | NS       | B2T1K2      | (902) 873-3527 | (902) 477-6164 |
| Thrifty Car Rental            | 639 Bishop Dr        |           | Fredericton          | NB       | E3C2M6      | (506) 457-5704 | Not Available  |
| Thrifty Car Rental            | 619 Bayfield Rd      |           | Goderich             | ON       | N7A3X8      | (519) 524-1795 | Not Available  |
| Thrifty Car Rental            | 70 Wyndham St S      |           | Guelph               | ON       | N1E5R4      | (519) 824-5300 | Not Available  |
| Thrifty Car Rental            | 1278 Rue Boreale     |           | Havre-St-Pierre      | QC       | G0G1P0      | (418) 538-2199 | (418) 538-3608 |
| Thrifty Car Rental            | 1790 Springfield Rd  | В         | Kelowna              | BC       | V1Y5V6      | (250) 868-2151 | Not Available  |
| Thrifty Car Rental            | 4470 Highway 7       |           | Markham              | ON       | L3R1M3      | (905) 475-1114 | Not Available  |
| Thrifty Car Rental            | 6 Harris Dr          | Α         | Marystown            | NL       | A0E2M0      | (709) 279-1171 | (709) 279-3385 |

~

|                         | ······································ | Car Renta                             | Agencies in Canac | la       |             |                |                |
|-------------------------|--|---------------------------------------|-------------------|----------|-------------|----------------|----------------|
|                         |  | Suite                                 |                   |          | 1           |                |                |
| Company Name            | Address                                | Number                                | City              | Province | Postal Code | Phone Number   | Fax Number     |
| Thrifty Car Rental      | 1675 The Chase                         | 6                                     | Mississauga       | ON       | L5M5Y7      | (905) 828-8181 | Not Available  |
| Thrifty Car Rental      | 1155 Rue Guy                           |                                       | Montreal          | QC       | H3H2K5      | (514) 989-7100 | Not Available  |
| Thrifty Car Rental      | 255 Westville Rd                       |                                       | New Glasgow       | NS       | B2H2J6      | (902) 752-5506 | (902) 755-2434 |
| Thrifty Car Rental      | 10 Maplewood Ave                       |                                       | North Bay         | ON       | P1B5H2      | (705) 498-6800 | Not Available  |
| Thrifty Car Rental      | 161 Av Laurier O                       |                                       | Ottawa            | ON       | K1P5J2      | (613) 233-8194 | (613) 233-0629 |
| Thrifty Car Rental      | 2055 16th Ave E                        |                                       | Owen Sound        | ON       | N4K5R7      | (519) 371-3381 | Not Available  |
| Thrifty Car Rental      | 1975 Broad St                          |                                       | Regina            | SK       | S4P1Y2      | (306) 525-1000 | Not Available  |
| Thrifty Car Rental      | 3910 Rochdale Blvd                     |                                       | Regina            | SK       | S4X4P7      | (306) 352-7000 | Not Available  |
| Thrifty Car Rental      | 3826 Mcdonald Rd                       |                                       | Richmond          | BC       | V7B1L8      | (604) 606-1695 | Not Available  |
| Thrifty Car Rental      | 5200 Miller Rd                         | 115                                   | Richmond          | BC       | V7B1K5      | (604) 276-0848 | Not Available  |
| Thrifty Car Rental      | 143 Robin Crescent                     |                                       | Saskatoon         | SK       | S7L6M3      | (306) 244-8000 | Not Available  |
| Thrifty Car Rental      | 295 Trunk Rd                           |                                       | Sault Ste. Marie  | ON       | P6A3S7      | (705) 575-2753 | Not Available  |
| Thrifty Car Rental      | 4142 Sheppard Ave E                    |                                       | Scarborough       | ON       | M1S1T3      | (416) 291-9761 | Not Available  |
| Thrifty Car Rental      | 1810 Boul Fiset                        |                                       | Sorel-Tracy       | QC       | J3P5K4      | (450) 742-8335 | Not Available  |
| Thrifty Car Rental      | 152 Scott St                           |                                       | St Catharines     | ON       | L2N1H1      | (905) 688-4460 | (905) 937-7300 |
| Thrifty Car Rental      | 4 Craig Dobbin's Way                   |                                       | St. John's        | NL       | A1A4Y3      | (709) 722-6000 | (709) 576-0953 |
| Thrifty Car Rental      | 1492 Paris St                          |                                       | Sudbury           | ON       | P3E3B8      | (705) 523-2137 | Not Available  |
| <br> Thrifty Car Rental | 7319 King George Blvd                  |                                       | Surrey            | вс       | V3W5A8      | (604) 635-3535 | Not Available  |
| Thrifty Car Rental      | 801 Cheadle St W                       |                                       | Swift Current     | SK       | S9H0C3      | (306) 971-4000 | Not Available  |
| Thrifty Car Rental      | 1141 Golf Links Rd                     | · · · · · · · · · · · · · · · · · · · | Thunder Bay       | ON       | P7B7A3      | (807) 345-7111 | (807) 346-7187 |
| Thrifty Car Rental      | 140 Carlton St                         | D                                     | Toronto           | ON       | M5A3W7      | (416) 515-7700 | (416) 921-7446 |
| Thrifty Car Rental      | 191 Parliament St                      |                                       | Toronto           | ON       | M5A2Z4      | (647) 497-7229 | Not Available  |
| Thrifty Car Rental      | 330 Front St W                         | <u> </u>                              | Toronto           | ON       | M5V3B7      | (416) 947-1385 | Not Available  |
| Thrifty Car Rental      | 330 Front St W                         |                                       | Toronto           | ON       | M5V3B7      | (647) 795-7244 | Not Available  |
|                         | 398 Steeles Avenue                     |                                       |                   |          |             |                |                |
| Thrifty Car Rental      | West                                   | 8                                     | Toronto           | ON       | L4J1A1      | (416) 291-9762 | Not Available  |
| Thrifty Car Rental      | 51 Highway 33                          |                                       | Trenton           | ON       | K8V5P6      | (613) 697-4271 | Not Available  |
| Thrifty Car Rental      | 413 Seymour St                         |                                       | Vancouver         | BC       | V6B3H5      | (604) 343-2433 | Not Available  |
| Thrifty Car Rental      | 1456 Mosley St                         | В                                     | Wasaga Beach      | ON       | L9Z2E7      | (705) 352-3500 | Not Available  |
| Thrifty Car Rental      | 285 Weber St N                         | 1A                                    | Waterloo          | ON       | N2J3H8      | (519) 742-4411 | (519) 742-1653 |

ς.

| Car Rental Agencies in Canada  |                    |                 |             |          |             |                |                |  |  |  |  |
|--------------------------------|--------------------|-----------------|-------------|----------|-------------|----------------|----------------|--|--|--|--|
| Company Name                   | Address            | Suite<br>Number | City        | Province | Postal Code | Phone Number   | Fax Number     |  |  |  |  |
| Thrifty Car Rental             | 212 Henderson Hwy  |                 | Winnipeg    | МВ       | R2L1L8      | (204) 989-2994 | Not Available  |  |  |  |  |
| Thrifty Car Rental             | 340 Old Airport Rd |                 | Yellowknife | NT       | X1A3T3      | (867) 669-9277 | Not Available  |  |  |  |  |
| Toronto Rentals                | 2035 Kennedy Rd    |                 | Scarborough | ON       | M1T3G2      | (416) 293-5510 | Not Available  |  |  |  |  |
| Trinity Car Rental & Leasing   | 55 Scarlett Rd     |                 | York        | ON       | M6N4J8      | (416) 243-0877 | (416) 243-1667 |  |  |  |  |
| Universal Sales Ltd Car Rental | 55 Watertower Rd   |                 | Saint John  | NB       | E2M7K2      | (506) 632-5809 | Not Available  |  |  |  |  |
| U-Save Car & Truck Rental      | 176 Bullock Dr     | 12              | Markham     | ON       | L3P7N1      | (905) 554-8800 | Not Available  |  |  |  |  |
| U-Save Car & Truck Rental      | 1415 Bonhill Rd    | 3               | Mississauga | ON       | L5T1R2      | (905) 256-3221 | (905) 256-3224 |  |  |  |  |
| Winnipeg U-Drive               | 167 Mcphillips St  |                 | Winnipeg    | MB       | R3E2K1      | (204) 779-3811 | Not Available  |  |  |  |  |
| Zoom Car Rentals               | 5835 Dixie Rd      |                 | Mississauga | ON       | L4W4V7      | (416) 694-7368 | (905) 565-6600 |  |  |  |  |
| Zoom Rent A Car                | 5825 Dixie Rd      |                 | Mississauga | ON       | L4W4V7      | (416) 232-2277 | (905) 565-6600 |  |  |  |  |
| Zoom Rent-A-Car                | 5835 Dixie Rd      | 3               | Mississauga | ON       | L4W4V7      | (416) 222-4473 | (905) 565-6600 |  |  |  |  |

٠

|                            | Taxi Con              | panies in Ca | nada with Annual Sal | es over \$500, | 000         |                |                |
|----------------------------|-----------------------|--------------|----------------------|----------------|-------------|----------------|----------------|
|                            |                       | Suite        | <u> </u>             |                |             |                |                |
| Company Name               | Address               | Number       | City                 | Province       | Postal Code | Phone Number   | Fax Number     |
| 1-United Taxi Group        | 33 8 Street           |              | Brandon              | MB             | R7A3W9      | (204) 571-1111 | Not Available  |
| 287 Taxi                   | 301 Dogwood St        | 204          | Campbell River       | BC             | V9W2Y1      | (250) 287-8294 | Not Available  |
| 4-Way Taxi                 | 1302 Rosser           |              | Brandon              | MB             | R7A4E9      | (204) 727-2500 | (204) 571-5199 |
| 5 0 Taxi Limo Div          | 16 Mitchell St        |              | St Catharines        | ON             | L2R3W4      | (905) 685-5464 | Not Available  |
| 5th Ave Taxi               | 91305 Alaska Hwy      | С            | Whitehorse           | YT             | Y1A6E4      | (867) 668-4111 | Not Available  |
| 5th Avenue Taxi            | Po Box 40227 Stn Main |              | Whitehorse           | ΥT             | Y1A6N3      | (867) 667-4111 | Not Available  |
| 8800 Taxi Beer & Liquor    | 60 Kilpatrick Dr      |              | Holland Landing      | ON             | L9N1H7      | (905) 853-8800 | (905) 830-4978 |
| A & A Deluxe Taxi          | 3472 Landmark Rd      |              | Burlington           | ON             | L7M1S8      | (905) 333-3000 | Not Available  |
| A 1 Designated Driver      | 10675 150 St          | 378          | Surrey               | BC             | V3R4C1      | (604) 581-0558 | Not Available  |
| A 1 Rose Taxi              | 8060 Lawson Rd        | 7            | Milton               | ON             | L9T5C5      | (905) 875-1900 | Not Available  |
| A Association Taxi         |                       |              |                      |                |             |                |                |
| Hochelaga                  | 6895 Rue Beaubien E   |              | Montreal             | QC             | H1M3B2      | (514) 256-9033 | (514) 256-8608 |
| A Aylmer Allo Taxi         | 203 Rue Principal     |              | Gatineau             | QC             | J9H6H4      | (819) 685-6666 | Not Available  |
| A Bowmanville Taxi &       |                       |              |                      |                |             |                |                |
| Limousine                  | 169 King St E         |              | Bowmanville          | ON             | L1C1N9      | (905) 623-4422 | Not Available  |
| A C Taxi & Delivery        | 537 115 Ave           |              | Dawson Creek         | BC             | V1G3B5      | (250) 782-2214 | Not Available  |
| A Dial A Cab               | 27 Haggert Ave S      | A            | Brampton             | ON             | L6Y2C2      | (905) 455-7777 | (905) 455-7000 |
| A Plus Taxi Ltd            |                       |              |                      |                |             |                |                |
| Administration             | 10th St N             | 416D         | Lethbridge           | AB ·           | T1H2C7      | (403) 394-9101 | Not Available  |
| A Premier Taxi             | 4 Stonebridge Dr      |              | Port Colborne        | ON             | L3K5V5      | (905) 835-8000 | (905) 835-1627 |
| A1 Niagara River Taxi      | 8166 Alpine Dr        |              | Niagara Falls        | ON             | L2H3C1      | (905) 658-3030 | (905) 228-0455 |
| A1 Taxi                    | 77 Mee Rd             |              | Kentville            | NS             | B4N1W8      | (902) 542-4414 | Not Available  |
| A-1 Taxi                   | 20 Regan Rd           | 5            | Brampton             | ON             | L7A1C3      | (905) 453-6666 | (905) 453-3131 |
| A-1 Taxi                   | 1024 A Division St    |              | Cobourg              | ON             | K9A5V1      | (905) 372-4449 | (905) 372-0430 |
| A-1 Taxi                   | 19 Prospect Ave       |              | Kirkland Lake        | ON             | P2N2T9      | (705) 568-4500 | (705) 568-5418 |
| A-1 Taxi                   | 189 Wilson Ave        |              | Timmins              | ON             | P4N2T2      | (705) 268-6868 | Not Available  |
| A-1 United Taxi            | 150 Victoria St N     |              | Kitchener            | ON             | N2H5C6      | (519) 743-6323 | (519) 743-5765 |
|                            | 785 Av Jean-Baptiste- |              |                      |                |             |                |                |
| A3 Taxi La Prairie Candiac | Varin                 |              | La Prairie           | QC             | J5R6P3      | (450) 444-2000 | Not Available  |
| A-A Mississauga Taxi       | 1105 Crestlawn Dr     | 16           | Mississauga          | ON             | L4W1A7      | (905) 277-7777 | Not Available  |
| AAA Best Choice Taxi       | 557 Dixon Rd          | 111          | Etobicoke            | ON             | M9W6K1      | (416) 241-4700 | Not Available  |

|                                | Taxi Companies in Canada with Annual Sales over \$500,000 |        |               |          |             |                |                |  |  |  |  |
|--------------------------------|---|--------|---------------|----------|-------------|----------------|----------------|--|--|--|--|
|                                |   | Suite  |               |          |             |                | 1              |  |  |  |  |
| Company Name                   | Address   | Number | City          | Province | Postal Code | Phone Number   | Fax Number     |  |  |  |  |
| AAA Best Choice Taxi Inc       | 34 Kelfield St  |        | Etobicoke     | ON       | M9W5A2      | (416) 748-7888 | Not Available  |  |  |  |  |
| AAA Black Cab                  | 410 Industrial Dr   | A      | Milton        | ON       | L9T5A6      | (905) 636-0555 | Not Available  |  |  |  |  |
| AAA Bolton Taxi                | 5 Cedargrove Rd   |        | Bolton        | ON       | L7E2L4      | (905) 857-4200 | Not Available  |  |  |  |  |
| AAA Loyal Taxi Enr             | 331 Rue Main  |        | Gatineau      | QC       | J8P5K5      | (819) 663-7177 | Not Available  |  |  |  |  |
| AAA Sherwood Park Taxi         | 52304 Range Road 233                                      | 2      | Sherwood Park | AB       | T8B1C9      | (780) 464-1333 | Not Available  |  |  |  |  |
| AAA Transpo                    | 3454 Petawawa Blvd  | D      | Petawawa      | ON       | K8H1X3      | (613) 687-1471 | Not Available  |  |  |  |  |
| AAA Viau 24/7 Taxi-Beer-       |   |        |               |          |             |                |                |  |  |  |  |
| Fast                           | 15 Collier St   |        | Barrie        | ON       | L4M1G5      | (705) 333-3333 | (705) 726-5109 |  |  |  |  |
| A-AAA Unicity Taxi             | 340 Hargrave Pl   |        | Winnipeg      | МВ       | R3C0X5      | (204) 925-3131 | (204) 956-5736 |  |  |  |  |
| Aaaaa United Discount Taxi     | 15 Collier St   |        | Barrie        | ON       | L4M1G5      | (705) 733-3333 | (705) 726-5109 |  |  |  |  |
| Aaaaa Viau Intl Taxi-Limo      | 518 Tiffin St   |        | Barrie        | ON       | L4N9W8      | (705) 720-2222 | Not Available  |  |  |  |  |
| Aaron Taxi                     | 11j Rayborn Cres  |        | St. Albert    | AB       | T8N4A9      | (780) 460-4444 | Not Available  |  |  |  |  |
| Aaron Taxi                     | 434 Elgin St  |        | Sudbury       | ON       | P3B1B2      | (705) 523-3333 | Not Available  |  |  |  |  |
| Abbotsford Taxi Ltd            | 31265 Wheel Ave   | 207    | Abbotsford    | BC       | V2T6H2      | (604) 855-1111 | Not Available  |  |  |  |  |
| ABC Airport & Out-Town         |   |        |               |          |             |                |                |  |  |  |  |
| Taxi                           | 2163 Lawrence Ave E                                       | 3      | Scarborough   | ON       | M1P2P5      | (416) 690-3200 | Not Available  |  |  |  |  |
| ABC Car Svc Ltd                | 880 Hanwell Rđ  | 301    | Fredericton   | NB       | E3B6A3      | (506) 455-5555 | Not Available  |  |  |  |  |
| ABC Taxi                       | 658 10th St   | С      | Brandon       | MB       | R7A4G5      | (204) 726-1600 | Not Available  |  |  |  |  |
| ABC Taxi                       | 36 Murphy St  |        | Trenton       | ON       | K8V4S7      | (613) 392-3525 | Not Available  |  |  |  |  |
| Able Atlantic Taxi Svc         | 2933 Kennedy Rd   |        | Scarborough   | ON       | M1V1S9      | (416) 298-1111 | Not Available  |  |  |  |  |
| Able Taxi                      | 17 James St W   |        | Orillia       | ON       | L3V8A6      | (705) 325-0632 | Not Available  |  |  |  |  |
| Able Taxi Ltd                  | 61 Summerwood Rd SE                                       |        | Airdrie       | AB       | T4B1W7      | (403) 948-5353 | Not Available  |  |  |  |  |
| Able Taxi Ltd                  | 2933 Kennedy Rd   |        | Scarborough   | ON       | M1V1S9      | (416) 291-7763 | (416) 291-7765 |  |  |  |  |
| Access Taxi                    | 8123 Fraser Ave   |        | Fort Mcmurray | AB       | T9H1W5      | (780) 799-3333 | Not Available  |  |  |  |  |
| Access Taxi                    | 5 Chambers St   |        | Smiths Falls  | ON       | K7A2Y2      | (613) 264-1212 | Not Available  |  |  |  |  |
| Access Taxi & Limousine<br>Svc | 8123 Fraser Ave   |        | Fort Mcmurray | AB       | T9H1W5      | (780) 799-1234 | Not Available  |  |  |  |  |

| <u> </u>                    | Taxi Com               |        | nada with Annual S | ales over \$500, | 000         | η <u>—</u>     | ·····          |
|-----------------------------|------------------------|--------|--------------------|------------------|-------------|----------------|----------------|
| <u>_</u>                    |                        | Suite  |                    |                  |             |                |                |
| Company Name                | Address                | Number | City               | Province         | Postal Code | Phone Number   | Fax Number     |
| Accessible Vehicle Svc-     |                        |        |                    |                  |             |                |                |
| Ontario                     | 4000 Steeles Ave W     | 206    | Woodbridge         | ON               | L4L4V9      | (416) 744-9991 | Not Available  |
| Ace Cabs                    | 19 Stewart Rd          |        | Collingwood        | ON               | L9Y4M7      | (705) 445-3300 | (705) 446-0524 |
| A-City Taxi Ltd             | 205 Church St          |        | Pembroke           | ON               | K8A4K8      | (613) 735-1044 | (613) 587-4613 |
| Act Taxi                    | 6 Peter St S           |        | Orillia            | ON               | L3V5A9      | (705) 327-3272 | (705) 326-2555 |
| Action Cab                  | 12 Adelaide St S       |        | Chatham            | ON               | N7M4P8      | (519) 354-8800 | (519) 351-1000 |
| Action Transportation       | 9 Douglas Rd           |        | Trenton            | ON               | K8V5R2      | (613) 392-9233 | (613) 394-2428 |
| Acton Taxi Svc              | 98 Mill St             |        | Georgetown         | ON               | L7G2C9      | (519) 853-1133 | Not Available  |
| Aeroport A-Cab              | 849 Westport Cres      |        | Mississauga        | ON               | L5T1E7      | (905) 629-2000 | (905) 908-5027 |
| Aeroport Taxi & Limousine   |                        |        |                    |                  |             | ]              |                |
| Svc                         | 849 Westport Cres      |        | Mississauga        | ON               | L5T1E7      | (416) 255-2211 | Not Available  |
| Aeroport Taxi Svc           | 849 W Port Cres        | -      | Mississauga        | ON               | L5R1K1      | (905) 624-0515 | Not Available  |
| Aimey's Taxi Ltd            | 795 Division St        |        | Kingston           | ON               | K7K4C2      | (613) 549-4444 | (613) 546-0424 |
| Air Cab Airport             |                        |        |                    |                  |             |                |                |
| Transportation              | 417 Hammonds Plains Rd |        | Bedford            | NS               | B4B1A5      | (902) 832-4297 | Not Available  |
| Air Cab Ltd                 | 189 Gordon St          |        | Moncton            | NB               | E1C1N3      | (506) 857-2000 | (506) 855-7193 |
| Air Flight Limousine Svc    | 3300 Steeles Ave W     | 202    | Concord            | ON               | L4K2Y4      | (416) 445-1999 | (905) 761-7210 |
| Air Taxi                    | 7294 Rue Lajeunesse    | Α      | Montreal           | QC               | H2R2H4      | (514) 840-9595 | (514) 840-0079 |
| Airline A Cabs              | 69 Wentworth St        |        | Sydney             | NS               | B1P6T5      | (902) 564-4445 | (902) 539-5321 |
| Airport Bus                 | 767 Audley St          |        | Victoria           | BC               | V8X2V4      | (250) 475-2010 | (250) 475-3010 |
| Airport Limo Toronto        | 849 Westport Cres      |        | Mississauga        | ON               | L5T 1E7     | (416) 366-0404 | Not Available  |
| Airport Sedan Svc           | 404 35 Ave NE          | _      | Calgary            | AB               | T2E2K7      | (403) 299-1123 | (403) 299-9559 |
|                             |                        |        |                    |                  |             |                |                |
| Airport Shuttle-Greg's Taxi | 768 Campbell St        |        | Sarnia             | ON               | N7T2J6      | (519) 337-2040 | Not Available  |
| Airport Silver Cab          | 71 Woodlot Cres        |        | Etobicoke          | ON               | M9W6T3      | (416) 674-6464 | (905) 677-7787 |
| Airport Taxi Svc            | 4608 101 St NW         |        | Edmonton           | AB               | T6E5G9      | (780) 890-7070 | Not Available  |
| Alberta Co-Op Taxi Line Ltd | 10148 105 St NW        |        | Edmonton           | AB               | T5J1C9      | (780) 822-6926 | Not Available  |
| Alberta Co-Op Taxi Line Ltd | 10538 114 St NW        |        | Edmonton           | AB               | т5нзј7      | (780) 425-2525 | Not Available  |

-

-

~

|                             | Taxi Con              | npanies in Ca | nada with Annual Sa | les over \$500, | ,000          |                |                |
|-----------------------------|-----------------------|---------------|---------------------|-----------------|---------------|----------------|----------------|
|                             |                       | Suite         |                     |                 | 1             |                |                |
| Company Name                | Address               | Number        | City                | Province        | Postal Code   | Phone Number   | Fax Number     |
| Alberta Cooperative Taxi    |                       |               |                     |                 | -             |                |                |
| Line                        | 10205 100 Ave NW      |               | Edmonton            | АВ              | T5J4B5        | (780) 414-5040 | Not Available  |
| Alberta Cooperative Taxi    |                       |               |                     |                 | · · · · · · · |                |                |
| Line                        | 10220 103 St NW       |               | Edmonton            | AB              | T5J0Y8        | (780) 414-5024 | Not Available  |
| Alberta Gold Taxi Ltd       | 4845 79 St            | 4             | Red Deer            | AB              | T4P2T4        | (403) 341-7777 | (403) 347-9052 |
| Aldergrove-Langley Taxi     | 5151 214a St          |               | Langley             | BC              | V3A8S7        | (604) 530-4444 | Not Available  |
| All Canadian Taxi           | 49 Colborne St E      |               | Orillia             | ON              | L3V1T5        | (705) 329-3333 | (705) 329-0095 |
| All Star Taxi               | 5159 Tomken Rd        |               | Mississauga         | ON              | L4W1P1        | (905) 602-0000 | Not Available  |
| Alliance Taxi               | Po Box 209            |               | Errington           | BC              | VOR1VO        | (250) 954-5567 | (250) 248-2475 |
| Allied Limousine            | 307 41 Ave NE         |               | Calgary             | AB              | T2E2N4        | (403) 299-9555 | (403) 299-9559 |
| Allied Limousine            | 404 35 Ave NE         |               | Calgary             | AB              | T2E2K7        | (403) 299-9555 | (403) 299-9559 |
| Alliston Cabs               | 33 Tottenham Rd       |               | Alliston            | ON              | L9R1J9        | (705) 435-2056 | (705) 434-9731 |
| Allo Taxi                   | 956 Rue Sainte-Helene |               | Longueuil           | QC              | J4K3R9        | (450) 646-6060 | Not Available  |
| Allstrathcona Taxi Ltd      | 10577 109 St NW       | 201           | Edmonton            | AB              | T5H3B1        | (780) 977-7666 | Not Available  |
|                             |                       |               |                     |                 |               |                |                |
| Amey's Greenwood Taxi Ltd   | 795 Division St       |               | Kingston            | ON              | К7К4С2        | (613) 546-1111 | (613) 546-0424 |
| Amherst Taxi                | 25 Church St          |               | Amherst             | NS              | B4H3A5        | (902) 667-8690 | Not Available  |
| Amil's Taxi                 | 1105 Stadacona St E   |               | Moose Jaw           | SK              | S6H0A1        | (306) 692-7878 | Not Available  |
| A-Milton Taxi               | 32 Bronte St S        |               | Milton              | ON              | L9T1Y8        | (905) 875-2994 | (905) 875-9224 |
| Ancaster Taxi Inc           | 160 John St S         |               | Hamilton            | ON              | L8N2C4        | (905) 648-8294 | (905) 525-6700 |
| Angus Taxi                  | 147 Mill St           | C             | Angus               | ON              | LOM1B2        | (705) 424-4444 | Not Available  |
| Apex Cabs                   | 215a Miskow Close     |               | Canmore             | AB              | T1W3G7        | (403) 609-0030 | Not Available  |
| A-Plus Taxi Ltd             | 416 10 St N           |               | Lethbridge          | AB              | T1H2C7        | (403) 317-7777 | (403) 394-9133 |
| Arch Transco Ltd            | 3405 Saskatchewan Dr  |               | Regina              | SK              | S4T1H7        | (306) 924-8734 | (306) 522-8830 |
| Arrow Cab Ltd               | 1 Credit Union Dr     |               | North York          | ON              | M4A2S6        | (416) 233-1111 | Not Available  |
|                             |                       |               |                     |                 |               |                |                |
| Assn Des Proprietaires-Taxi | 845 Rue Saint-Jacques |               | St-Jn-S-Richlieu    | QC              | J3B2N2        | (450) 346-6666 | (450) 357-9144 |
| Associated Cab Ltd          | 6812 52 Ave           |               | Red Deer            | AB              | T4N4L1        | (403) 343-3300 | (403) 342-6896 |
| Associated Cabs Ltd         | 307 41 Ave NE         |               | Calgary             | AB              | T2E2N4        | (403) 299-1111 | Not Available  |
| Astro Taxi Co-Op            | 2781 Highway 7        |               | Concord             | ON              | L4K1W1        | (905) 738-3311 | Not Available  |
| Astrol Taxi                 | 2016 Sherwood Dr      |               | Sherwood Park       | AB              | T8A3X3        | (780) 467-2222 | Not Available  |

a

|                            | Taxi Com                 | panies in Ca | nada with Annual Sa | les over \$500, | 000         | •                    |                |
|----------------------------|--------------------------|--------------|---------------------|-----------------|-------------|----------------------|----------------|
|                            |                          | Suite        |                     |                 |             |                      |                |
| Company Name               | Address                  | Number       | City                | Province        | Postal Code | Ph <u>one Number</u> | Fax Number     |
| Astrol Taxi                | 52304 Range Road 233     | 2            | Sherwood Park       | AB              | T8B1C9      | (780) 467-2222       | Not Available  |
| Atl Airport Car Svc        | 545 Caledonia Rd         |              | Moncton             | NB              | E1H2E8      | (506) 853-0001       | Not Available  |
| Atlas Taxi Inc             | 4999 Rue Ste-Catherine O | 302          | Westmount           | QC              | H3Z1T3      | (514) 485-8585       | (514) 485-0946 |
| Banff Taxi Tours &         |                          |              |                     |                 |             |                      |                |
| Limousine                  | 101 Owl St               |              | Banff               | AB              | T1L1C2      | (403) 762-4444       | (403) 762-2244 |
| Barands Regional           | 69 Wentworth St          |              | Sydney              | NS              | B1P6T5      | (902) 564-4444       | (902) 539-5321 |
| Barrel Taxi                | 10135 31 Ave NW          |              | Edmonton            | AB              | T6N1C2      | (780) 489-7777       | (780) 462-2722 |
| Barrie Innisfil Taxi       | 37 Saunders Rd           |              | Barrie              | ON              | L4N9A7      | (705) 436-8888       | Not Available  |
| Barrie Taxi                | 37 Saunders Rd           |              | Barrie              | ON              | L4N9A7      | (705) 721-7777       | Not Available  |
| Barry Taxi                 | 37 Saunders Rd           |              | Barrie              | ON              | L4N9A7      | (705) 727-9999       | Not Available  |
| Baseline Taxi              | 19024 53 Ave NW          |              | Edmonton            | AB              | T6M2L2      | (780) 444-4556       | Not Available  |
| Bayshore Taxi & Parcel Svc | 777 3rd Ave E            |              | Owen Sound          | ON              | N4K2K4      | (519) 371-5555       | (519) 371-0486 |
| Bby Elite Cabs             | 5525 Imperial St         |              | Burnaby             | BC              | V5J1E8      | (604) 434-8000       | (604) 435-6757 |
| Beaubien Radio Taxi Inc    | 6893 Rue Beaubien E      |              | Montreal            | QC              | H1M3B2      | (514) 376-4040       | Not Available  |
|                            | 110d Boul Saint-Jean-    |              |                     |                 |             |                      |                |
| Beauharnois Taxi           | Baptiste                 |              | Chateauguay         | QC              | J6K3A8      | (450) 225-3366       | Not Available  |
| Bee Line Taxi              | 2933 Kennedy Rd          |              | Scarborough         | ON              | M1V1S9      | (416) 412-1212       | (416) 291-7765 |
| Bee Line Taxi Ltd          | 1423 16th Ave            | D            | Campbell River      | BC              | V9W2E4      | (250) 287-8383       | (250) 286-6251 |
| Bel-Air Taxi               | 2121 Hartley Ave         |              | Coquitlam           | BC              | V3K6Z3      | (604) 524-1111       | Not Available  |
| Bell City Cabs             | 190 West St              |              | Brantford           | ON              | N3R3T9      | (519) 759-1300       | (519) 759-0122 |
| Bell Taxi                  | 109 Dundas St W          |              | Whitby              | ON              | L1N2M1      | (905) 668-8888       | (905) 668-1127 |
| Bellamare Taxi             | 2933 Av Beaudry Leman    |              | Shawinigan          | QC              | G9N3H8      | (819) 539-5598       | Not Available  |
| Bellemare Taxi             | 398 6e Av                |              | Grand-Mere          | QC              | G9T2P8      | (819) 538-3000       | Not Available  |
| Big Rock Taxi              | Po Box 1091 Stn Main     |              | Okotoks             | AB              | T1S1B2      | (403) 938-9501       | Not Available  |
| Black Top & Checker Cabs   | 777 Pacific St           |              | Vancouver           | BC              | V6Z2R7      | (604) 683-4567       | Not Available  |
| Black Top Taxi             | 2717 5 Ave NE            |              | Calgary             | AB              | T2A2L6      | (403) 735-3222       | Not Available  |
| Blacktop Cab Gp            | 11315 96 Av              |              | Grande Prairie      | AB              | T8V5M3      | (780) 539-3339       | (780) 539-4600 |

۰

۰,

|                          | Taxi Con              | npanies in Ca | nada with Annual Sal | es over \$500 | ,000        |                |                |
|--------------------------|-----------------------|---------------|----------------------|---------------|-------------|----------------|----------------|
| ······                   |                       | Suite         | 1                    |               | 1           |                |                |
| Company Name             | Address               | Number        | City                 | Province      | Postal Code | Phone Number   | Fax Number     |
| Blue & White Taxi        | 360 Revus Ave         | 18            | Mississauga          | ON            | L5G4S4      | (905) 274-4444 | (905) 274-8939 |
| Blue Bell Taxi           | 269 Wyse Rd           |               | Dartmouth            | NS            | B3A1N5      | (902) 465-5555 | (902) 469-9295 |
| Blue Bird Cabs Ltd       | 2162 Quadra St        | 200           | Victoria             | BC            | V8T4E4      | (250) 382-3611 | (250) 382-8931 |
| Blue Bird Cabs Ltd       | 2612 Quadra St        | 2             | Victoria             | BC            | V8T4E4      | (250) 382-4235 | Not Available  |
| Blue Line Taxi           | 160 John St S         | 200           | Hamilton             | ON            | L8N2C4      | (905) 525-2583 | Not Available  |
| Blue Line Taxi           | 470 Taunton Rd E      |               | Oshawa               | ON            | L1H7L6      | (905) 440-2000 | (905) 434-2571 |
| Blue Line Taxi           | 225 Avenue B S        |               | Saskatoon            | SK            | S7M1M3      | (306) 653-3333 | (306) 664-0076 |
| Blue Mountain Cabs       | 19 Stewart Rd         |               | Collingwood          | ON            | L9Y4M7      | (705) 445-0300 | (705) 445-3300 |
| Blue Star Taxi           | 1100 Ord Rd           |               | Kamloops             | BC            | V2B7V4      | (250) 554-1918 | (250) 554-7791 |
| Bluebird Taxi            | 50 Grenfell Hts       |               | Grand-Fls-Windsr     | NL            | A2A1W4      | (709) 489-5757 | (709) 489-8821 |
| Blueline Taxi            | 979 Brock Rd          | 16            | Pickering            | ON            | L1W3A4      | (905) 837-1111 | (905) 686-0467 |
| Bob Taxi                 | 756 Rue Saint-Louis   |               | Gatineau             | QC            | J8T2S4      | (819) 561-1512 | Not Available  |
| Bonny's Taxi Ltd         | 5525 Imperial St      |               | Burnaby              | BC            | V5J1E8      | (604) 451-1111 | Not Available  |
| Boomerang Taxi Ltd       | 77 Russell St         |               | Saint John           | NB            | E2J2H6      | (506) 672-7272 | Not Available  |
| Bp Airport Shuttle Ltd   | 162 Panatella Blvd NW |               | Calgary              | AB            | T3K6B5      | (403) 259-5959 | Not Available  |
| Bram City Taxi           | 27 Av Haggert S       |               | Brampton             | ON            | L6Y2C2      | (905) 455-1000 | Not Available  |
| Brampton Bramalea Kwik   |                       |               |                      |               |             |                |                |
| Kab                      | 160 Wilkinson Rd      | 37            | Brampton             | ON            | L6T4Z4      | (905) 451-8000 | Not Available  |
| Brampton Taxi            | 160 Wilkinson Rd      | 37            | Brampton             | ON            | L6T4Z4      | (905) 452-0100 | (905) 451-6732 |
| Brock-Q-Taxi             | 131 Read Rd           |               | St Catharines        | ON            | L2R7K6      | (905) 935-5000 | Not Available  |
| Bud's Taxi               | 11 Brookside St       |               | Glace Bay            | NS            | B1A1K2      | (902) 849-4055 | Not Available  |
| Bugden's Taxi Ltd        | 5 Lemarchant Rd       |               | St. John's           | NL            | A1C2G4      | (709) 726-4400 | Not Available  |
| Burlington Taxi          | 3472 Landmark Rd      |               | Burlington           | ON            | L7M1S8      | (905) 333-3333 | Not Available  |
| Burnaby Select Metrotown |                       |               |                      |               |             |                |                |
| Taxi                     | 5525 Imperial St      |               | Burnaby              | BC            | V5J1E8      | (604) 526-7000 | Not Available  |
| Burnaby Select Taxi Ltd  | 4049 Ledger Ave       |               | Burnaby              | BC            | V5G3T2      | (604) 759-1449 | Not Available  |
| Burnaby Taxi             | 5525 Imperial St      |               | Burnaby              | BC            | V5J1E8      | (604) 433-4466 | Not Available  |
| Busy Bee Taxi Inc        | 539 Bay St            |               | Midland              | ON            | L4R1L4      | (705) 526-2424 | Not Available  |
| C J Cabs                 | 74 Avalon Dr          |               | Labrador City        | NL            | A2V2Y2      | (709) 944-7757 | Not Available  |
| Cadillac Cabs Inc        | 218 William St        |               | Wallaceburg          | ON            | N8A4B4      | (519) 627-2221 | Not Available  |
| Call A Cab               | 2026 Bensfort Rd      |               | Peterborough         | ON            | К9Ј6Ү7      | (705) 745-2424 | Not Available  |

٠

|                            | Taxi Com                  | panies in Ca | nada with Annual Sa | les over \$500, | 000         | <b></b>        | <del></del>    |
|----------------------------|---------------------------|--------------|---------------------|-----------------|-------------|----------------|----------------|
| <u> </u>                   |                           | Suite        |                     |                 |             |                | 1              |
| Company Name               | Address                   | Number       | City                | Province        | Postal Code | Phone Number   | Fax Number     |
| Camille Taxi               | 145 Victoria St           |              | Edmundston          | NB              | E3V2H7      | (506) 739-8480 | Not Available  |
| Canadian Cab               | 88 Macdonell St           |              | Guelph              | ON              | N1H2Z6      | (519) 824-3110 | (519) 824-8434 |
| Canadian Checker Cab       | 506 Hanna St E            |              | Windsor             | ON              | N8X2N6      | (519) 254-7777 | (519) 253-6907 |
| Canyon Taxi                | 500 King St               |              | Норе                | BC              | V0X1L0      | (604) 869-5211 | (604) 869-2202 |
| Capital Cab D & G Taxi Ltd | 1320 St John St           |              | Regina              | SK              | S4R1S3      | (306) 791-2222 | (306) 791-2255 |
| Capital Cabs               | 1320 St John St           |              | Regina              | SK              | S4R1S3      | (306) 781-7777 | (306) 791-2255 |
| Capital Taxi               | 4248 93 St NW             |              | Edmonton            | AB              | T6E5P5      | (780) 423-2425 | (780) 468-2405 |
| Capitol Hill Taxi          | 5525 Imperial St          |              | Burnaby             | BC              | V5J1E8      | (604) 299-6262 | (604) 435-6757 |
| Capitol Taxi               | 728 Rye St                |              | Peterborough        | ON              | K9J6W9      | (705) 743-2111 | Not Available  |
| Capitol Taxi               | 135 Campbell Ave          | A            | St. John's          | NL              | A1E2Z5      | (709) 726-3333 | Not Available  |
| Capitol Taxi               | 2701 35 St                | A            | Vernon              | BC              | V1T6B4      | (250) 545-1355 | (250) 545-3830 |
| Care Cabs Ltd              | 232 Maple Ave SE          | -            | Medicine Hat        | AB              | T1A3A4      | (403) 529-2211 | (403) 529-1492 |
| <u> </u>                   | 4981 East River East Side |              | <u> </u>            |                 |             |                |                |
| Cargo Cabs                 | Rd                        |              | New Glasgow         | NS              | B2H5C5      | (902) 695-3030 | Not Available  |
| Cariboo Taxi               | 528 Marsh Rd              |              | Quesnel             | BC              | V2J6G8      | (250) 991-0007 | (250) 991-0380 |
| Casino Taxi                | 3558 Novalea Dr           |              | Halifax             | NS              | B3K3E8      | (902) 429-6666 | Not Available  |
| Casino Taxi                | 135 Campbell Ave          | A            | St. John's          | NL              | A1E2Z5      | (709) 579-5151 | Not Available  |
| Castlegar Taxi             | 306a 11th Ave             |              | Castlegar           | BC              | V1N1J3      | (250) 365-7222 | Not Available  |
| Catch A Cab                | 485 Dominion Ave          |              | Midland             | ON              | L4R1P5      | (705) 526-8585 | (705) 526-8088 |
| Central Cabs               | 31 Mccurdy Dr             |              | Gander              | NL              | A1V1A1      | (709) 256-7777 | Not Available  |
| Central Taxi               | 569 Dundas St E           |              | Belleville          | ON              | K8N1G6      | (613) 962-5353 | (613) 968-6466 |
| Central Taxi               | 7 Gale Cres               |              | St Catharines       | ON              | L2R7M8      | (905) 685-7343 | (905) 685-3404 |
| Central Taxi Ltd           | 1120 Athabasca St E       |              | Moose Jaw           | SK              | S6H0N3      | (306) 694-1922 | Not Available  |
| Central Valley Limousine   | 1643 Salton Rd            |              | Abbotsford          | вс              | V2S7P2      | (604) 870-1111 | (604) 859-1125 |
| Central Valley Taxi Ltd    | 1643 Salton Rd            |              | Abbotsford          | BC              | V2S7P2      | (604) 859-1111 | Not Available  |
| Centre City Cabs Ltd       | 331 1st Ave               |              | Prince George       | BC              | V2L2Y1      | (250) 564-4444 | Not Available  |
| Century Taxi               | 24 Taylor Ave             |              | Saint John          | NB              | E2K3E6      | (506) 696-6969 | (506) 652-3331 |
| Champion Cab               | 8045 Old Waneta Rd        |              | Trail               | BC              | V1R4X1      | (250) 364-3344 | Not Available  |
| Champlain Taxi Inc         | 5775 Rue Saint-Andre      |              | Montreal            | QC              | H2S2K2      | (514) 273-2435 | (514) 273-4445 |

|                     | Taxi Co            | mpanies in Ca | nada with Annual Sa | les over \$500 | .000        |                |                |
|---------------------|--------------------|---------------|---------------------|----------------|-------------|----------------|----------------|
|                     |                    | Suite         | 1                   |                |             |                |                |
| Company Name        | Address            | Number        | City                | Province       | Postal Code | Phone Number   | Fax Number     |
| Champ's Taxi        | 8 Lewis St         |               | Grafton             | NB             | E7N1N2      | (506) 325-9516 | Not Available  |
| Chatham Radio Cabs  | 733 Park Ave W     |               | Chatham             | ON             | N7M1X3      | (519) 351-1232 | Not Available  |
| Cheam Taxi Ltd      | 6942 Fraser Dr     |               | Agassiz             | BC             | V0M1A3      | (604) 796-2424 | Not Available  |
| Checker Cabs Inc    | 10135 31 Ave NW    |               | Edmonton            | AB             | T6N1C2      | (780) 484-8888 | (780) 462-2722 |
| Checker Limousine   | 573 Admiral Crt    |               | London              | ON             | N5V4L3      | (519) 659-0400 | (519) 455-4402 |
| Checker Taxi        | 86 10th St W       |               | Prince Albert       | SK             | S6V3A4      | (306) 763-3773 | (306) 763-4977 |
| Checker Yellow Cabs | 316 Meridian Rd SE |               | Calgary             | AB             | T2A1X2      | (403) 299-9999 | (403) 248-0584 |
| Cie Leduc Ltee      | 171 Rue Alexandre  |               | Salaberry-Vlyfld    | QC             | J6S3J1      | (450) 371-4743 | Not Available  |
| Circle Cab Co       | 73 University Ave  |               | Charlottetown       | PE             | C1A4L1      | (902) 892-6563 | Not Available  |
| Circle Taxi         | 128 Brock St N     |               | Whitby              | ON             | L1N4H4      | (905) 668-6666 | (905) 668-1817 |
| City Cabs           | 4 Caribou Rd       |               | Corner Brook        | NL             | A2H4W6      | (709) 634-6565 | Not Available  |
| City Cabs           | 268 Breithaupt St  |               | Kitchener           | ON             | N2H5H5      | (519) 747-7777 | (519) 742-0401 |
| City Cabs           | 230 Edwards Ave    |               | The Pas             | МВ             | R9A1L4      | (204) 623-2500 | Not Available  |
| City Cabs Inc       | 130 Downie St      |               | Stratford           | ON             | N5A1X1      | (519) 272-2222 | (519) 272-1425 |
| City Taxi           | 40 Dalhousie St    |               | Brantford           | ON             | N3T2H8      | (519) 759-7800 | (519) 759-7873 |
| City Taxi           | 223 King St W      |               | Brockville          | ON             | K6V3R7      | (613) 345-5888 | (613) 345-5881 |
| City Taxi           | 195 Kent St        |               | Charlottetown       | PE             | C1A1P1      | (902) 892-6567 | (902) 569-2331 |
| City Taxi           | 745 Lorne St       |               | Sudbury             | ON             | P3C4R5      | (705) 673-9999 | Not Available  |
| City Taxi Ltd       | 5 Woods St         |               | Kirkland Lake       | ON             | P2N3B5      | (705) 567-5383 | (705) 567-5224 |
| City Wide Taxi      | 5 Adelaide         |               | St. John's          | NL             | A1C5R6      | (709) 722-0003 | Not Available  |
| City Wide Taxi      | 53 Macaluays Ln    |               | Sydney              | NS             | B1P3A4      | (902) 564-6111 | Not Available  |
| City Wide Taxi      |                    |               |                     |                |             |                |                |
| Administration      | 190 Waterloo St    | 1A            | Oshawa              | ON             | L1H8A7      | (905) 571-1331 | (905) 571-5650 |
| City-Wide Taxi      | 190 Waterloo St    | 1A            | Oshawa              | ON             | L1H8A7      | (905) 579-9393 | (905) 571-5650 |
| Cj's Taxi Ltd       | 56 Hwy 358         |               | Port Williams       | NS             | BOP1TO      | (902) 670-6700 | Not Available  |
| Classic Cabs        | 2717 5 Ave NE      |               | Calgary             | AB             | T2A2L6      | (403) 275-8000 | (403) 274-4444 |
| Clover Cabs         | 17921 55 Ave       | 3             | Surrey              | BC             | V3S6C4      | (604) 574-5311 | (604) 576-2900 |
| Coastal Taxi        | 52 Mallard St      |               | Kitimat             | BC             | V8C1N1      | (250) 632-7250 | (250) 632-6268 |
| Coastal Taxi Ltd    | 40 Main St W       |               | Saint John          | NB             | E2M3N1      | (506) 635-1144 | (506) 633-8808 |
| Comfort Cab         | 1224 Ontario Ave   |               | Saskatoon           | SK             | S7K1S5      | (306) 664-6464 | (306) 664-6477 |
| Comox Taxi          | 1507 Mcphee Ave    |               | Courtenay           | BC             | V9N3A6      | (250) 339-7955 | (250) 334-2926 |

۰

o

|                                       | Taxi Com                 | panies in Ca | nada with Annual Sal | les over \$500, | ,000        |                |                |
|---------------------------------------|--------------------------|--------------|----------------------|-----------------|-------------|----------------|----------------|
|                                       |                          | Suite        |                      |                 |             |                |                |
| Company Name                          | Address                  | Number       | City                 | Province        | Postal Code | Phone Number   | Fax Number     |
| · · · · · · · · · · · · · · · · · · · | 6610 Boul Les Galeries   |              |                      |                 |             |                |                |
| Co-Op De L'Est Taxi                   | D'Anjou                  |              | Anjou                | QC              | H1M2T4      | (514) 352-6000 | (514) 352-7189 |
| Co-Op De Taxi De Montreal             | 3822 Rue Dandurand       |              | Montreal             | QC              | H1X1P2      | (514) 725-9885 | (514) 725-2658 |
| Co-Op Taxi                            | 570 Av Marshall          |              | Dorval               | QC              | H9P1C9      | (514) 636-6666 | (514) 636-4941 |
| Co-Op Taxi                            | 9515 116 St              | . <u></u>    | Grande Prairie       | AB              | T8V5W3      | (780) 402-8181 | (780) 831-0243 |
| <br>Co-Op Taxi                        | 32 Duncan Ave S          |              | Kirkland Lake        | ON              | P2N1X6      | (705) 567-5363 | (705) 567-4249 |
| Co-Op Taxi                            | 2614 6th Ave             | 200          | Regina               | SK              | S4T0N3      | (306) 525-2727 | (306) 791-9031 |
|                                       | 18 Rue Monseigneur-      | -            |                      |                 |             |                |                |
| Co-Op Taxi                            | Tessier O                |              | Rouyn-Noranda        | QC              | J9X2S4      | (819) 762-1733 | (819) 762-1947 |
| Co-Op Taxi                            | 11 Lemarchant Rd         |              | St. John's           | NL              | A1C2G4      | (709) 579-4197 | (709) 579-3025 |
| Co-Op Taxi Energie Inc                | 762 5e St                |              | Shawinigan           | QC              | G9N1E9      | (819) 536-2525 | Not Available  |
| Co-Op Taxi Laval                      | 4405 Boul Saint-Martin O |              | Laval                | QC              | H7T1C5      | (450) 688-8700 | Not Available  |
| Co-Op Taxi Line Ltd                   | 305 Allen St             |              | Charlottetown        | PE              | C1A2W1      | (902) 628-8200 | Not Available  |
| Cooperative Cabs                      | 130 Rivalda Rd           |              | North York           | ON              | M9M2M8      | (416) 504-2667 | (416) 504-0663 |
| Cooperative Taxi                      | 466 Boul Des Seigneurs   |              | Terrebonne           | QC              | J6W1T3      | (450) 471-4723 | (450) 471-4997 |
| Cooperative Taxi Line Ltd             | 297 Allen St             |              | Charlottetown        | PE              | C1A2W1      | (902) 628-6761 | Not Available  |
| Cooperative Taxi Line Ltu             | 2121 Hartley Ave         |              | Coquitlam            | BC              | V3K6Z3      | (604) 939-4641 | Not Available  |
| Coral Cabs Ltd                        | 11180 Voyageur Way       | 260          | Richmond             | BC              | V6X3N8      | (604) 278-9191 | Not Available  |
| Corner Taxi                           | 191 O'Connell Dr         | 200          | Corner Brook         | NL              | A2H5M9      | (709) 634-5662 | Not Available  |
|                                       |                          |              |                      |                 |             | <u> </u>       |                |
| Country Cabs Duncan Ltd               | 3420 Auchinachie Rd      | 310          | Duncan               | BC              | V9L4P2      | (250) 746-0009 | Not Available  |
| County Cab Inc                        | 123 Bridge St            |              | Carleton Place       | ON              | K7C2V4      | (613) 256-7979 | (613) 253-7841 |
| Courtesy Cab                          | 12 Adelaide St S         |              | Chatham              | ON              | N7M4P8      | (519) 352-2300 | (519) 351-0000 |
| Courtesy Cab Inc                      | 4 Peel St                |              | Simcoe               | ON              | N3Y1R9      | (519) 426-9911 | Not Available  |
| Courtesy Cabs Ltd                     | 2219 Government St       |              | Penticton            | BC              | V2A4W4      | (250) 492-7777 | Not Available  |
| Courtesy Cabs Ltd                     | 324 Carmi Ave            | 102          | Penticton            | BC              | V2A3G5      | (250) 492-7778 | Not Available  |
| Cox Couriers                          | 8534 Centennial Ave      |              | St Thomas            | ON              | N5P3S6      | (519) 631-1800 | Not Available  |
| Cronkie's Cab Co                      | 209 Tripp Rd             |              | Picton               | ON              | КОК2ТО      | (613) 476-7678 | Not Available  |

5

|                             | Taxi Cor              | npanies in Ca | nada with Annual Sal | les over \$500, | ,000        |                |                |
|-----------------------------|-----------------------|---------------|----------------------|-----------------|-------------|----------------|----------------|
| · · · · · ·                 |                       | Suite         |                      |                 |             | [              |                |
| Company Name                | Address               | Number        | City                 | Province        | Postal Code | Phone Number   | Fax Number     |
| Crown Cab                   | 1942 100th St         |               | North Battleford     | SK              | S9A1E6      | (306) 445-8155 | Not Available  |
| Crown Taxi                  | 165 Ch Proulx         |               | Gatineau             | QC              | J8R3B9      | (819) 777-1645 | Not Available  |
| D & J Taxi                  | 25 Church St          |               | Amherst              | NS              | B4H3A5      | (902) 667-8288 | Not Available  |
| D J's Cabs                  |                       |               | Drayton Valley       | AB              |             | (780) 542-7655 | Not Available  |
| Dad's Taxi                  | 852 Pembroke St W     |               | Pembroke             | ON              | K8A5P7      | (613) 732-3237 | Not Available  |
| Danny's Delivery            | 186 Victoria St       |               | Pembroke             | ON              | K8A4K3      | (613) 735-7007 | Not Available  |
| Dauphin Premier Taxi        | 620 Main St S         |               | Dauphin              | МВ              | R7N1L4      | (204) 638-3315 | Not Available  |
| Dd4u                        | 920 Mount Allan Ave   |               | Oshawa               | ON              | L1J8K3      | (905) 440-4469 | Not Available  |
| Delta Bayside Taxi          | 13119 84 Ave          | 107           | Surrey               | BC              | V3W1B3      | (604) 943-9800 | (604) 597-8002 |
| Delta Cab                   | 124 Mckenzie Rd       | 7             | Inuvik               | NT              | XOEOTO      | (867) 777-5100 | Not Available  |
| Delta Express Cab Ltd       | 7621 Vantage Way      |               | Delta                | BC              | V4G1A6      | (604) 946-7272 | Not Available  |
| Delta Sunshine Taxi         | 12837 76 Ave          | 203           | Surrey               | BC              | V3W2V3      | (604) 594-5444 | (604) 594-1785 |
| Deluxe Central Taxi         | 656 1/2 3 St SE       |               | Medicine Hat         | AB              | T1A0H5      | (403) 526-3333 | (403) 527-8344 |
| Deluxe Taxi                 | 9 Clapperton St       |               | Barrie               | ON              | L4M3E4      | (705) 728-4444 | Not Available  |
| Deluxe Taxi Barrie Ltd      | 9 Clapperton St       |               | Barrie               | ON              | L4M3E4      | (705) 728-5517 | (705) 728-9936 |
| Deluxe Taxi Central         | 656 1/2 3 St SE       |               | Medicine Hat         | AB              | T1A0H5      | (403) 527-7777 | (403) 527-8344 |
| Deluxe Taxi Ltd             | 98 Bayfield St        |               | Barrie               | ON              | L4M3A8      | (705) 728-4444 | Not Available  |
| Deluxe Taxi St Eustache Inc | 129 Boul Arthur-Sauve |               | Saint-Eustache       | QC              | J7P2A3      | (450) 473-3333 | (450) 473-4444 |
| Dg Executive Limousine      | Rr 2                  |               | Cameron              | ON              | K0M1G0      | (705) 341-4233 | Not Available  |
| Dial Tax Inc                | 1145 Hunt Club Rd     | 3             | Ottawa               | ON              | K1V0Y3      | (613) 738-7712 | (613) 738-8788 |
| Dial-A-Ride                 | 3472 Landmark Rd      |               | Burlington           | ON              | L7M1S8      | (905) 333-1113 | (905) 333-1688 |
| Diamond Lacey's Taxi        | 113 Leith St          |               | Thunder Bay          | ON              | P7C1M7      | (807) 622-6001 | Not Available  |
| Diamond Taxi                | 1105 Stadacona St E   |               | Moose Jaw            | SK              | S6H 0A1     | (306) 693-3999 | Not Available  |
| Diamond Taxi                | 251 Queen St E        |               | Toronto              | ON              | M5A1S6      | (416) 366-6600 | (416) 366-6753 |
| Diamond Taxi Oromocto       | 235 Restigouche Rd    |               | Oromocto             | NB              | E2V2H1      | (506) 446-5035 | Not Available  |
| Diamond Taxicab Dispatch    |                       |               |                      |                 |             |                |                |
| Svc                         | 251 Queen St E        |               | Toronto              | ON              | M5A1S6      | (416) 366-6868 | Not Available  |
| Dj's Taxi                   | 455 Coventry Rd       |               | Ottawa               | ON              | K1K2C5      | (613) 829-9900 | (613) 746-0814 |
| Dorset Taxi Co              | Hotle                 |               | Cape Dorset          | NU              | X0A0C0      | (867) 897-8005 | Not Available  |

\_

|                            | Taxi Con              | panies in Ca | nada with Annual Sales | over \$500, | 000         |                |                |
|----------------------------|-----------------------|--------------|------------------------|-------------|-------------|----------------|----------------|
|                            |                       | Suite        |                        |             | <b>1</b>    |                |                |
| Company Name               | Address               | Number       | City                   | Province    | Postal Code | Phone Number   | Fax Number     |
| Driftwood Nickel City Taxi | 725 Thompson Dr       |              | Thompson               | мв          | R8N0C7      | (204) 677-6000 | Not Available  |
| Driving Miss Daisy         | 11 Madison Ave        |              | St. Albert             | AB          | T8N1H2      | (780) 470-0123 | Not Available  |
| Duffy's Taxi               | 1100 Notre Dame Ave   |              | Winnipeg               | МВ          | R3E0N8      | (204) 925-0101 | (204) 774-7093 |
| Duncan Taxi Ltd            | 2910 Allenby Rd       |              | Duncan                 | BC          | V9L6V3      | (250) 746-4444 | (250) 746-4987 |
| Dundas Valley Taxi         | 15 Wentworth St N     |              | Hamilton               | ON          | L8L5T8      | (905) 310-8294 | (905) 628-1122 |
| Dupont Courier-Taxi        | 22 Dallyn Ave         | В            | Kapuskasing            | ON          | P5N1S6      | (705) 335-5491 | Not Available  |
| Durham Rapid Taxi Inc      | 979 Brock Rd          |              | Pickering              | ON          | L1W3A4      | (905) 831-2345 | Not Available  |
| E Barrie Taxi              | 80 Bradford St        | 808          | Barrie                 | ON          | L4N6S7      | (705) 726-8222 | Not Available  |
| Eagle Taxi                 | 5002 48 Av            | -            | Wetaskiwin             | AB          | T9A0M9      | (780) 352-7699 | Not Available  |
| Eastview Taxi              | 5205 51 Ave           |              | Camrose                | AB          | T4V3Y4      | (780) 679-6216 | Not Available  |
| Easy Ride Taxi Ltd         | 6105 46 St            | 2            | Leduc                  | AB          | T9E6T8      | (780) 980-5112 | (780) 980-5113 |
| Econo Airport Svc          | 1012 Main St          |              | Port Williams          | NS          | BOP1T0      | (902) 542-4495 | (902) 542-1698 |
| Economy Taxi               | 307 Third Ave         |              | Strathmore             | AB          | T1P1B4      | (403) 888-9058 | Not Available  |
| Edson Taxi Ltd             | 7 Ave                 |              | Yellowhead County      | AB          | T7E 3B2     | (780) 723-6633 | Not Available  |
| Emerald Taxi Ltd           | 966 5th Ave           |              | Prince George          | BC          | V2L3K8      | (250) 563-3333 | (250) 564-8294 |
| Esquimalt Taxi             | 817 Fisgard St        |              | Victoria               | BC          | V8W1R9      | (250) 386-7766 | (250) 386-7711 |
| Etobicoke Taxi             | 34 Kelfield St        |              | Etobicoke              | ON          | M9W5A2      | (416) 745-7777 | Not Available  |
| Evangeline Taxi            | 1090 Highbury Rd      |              | New Minas              | NS          | B4N3P7      | (902) 681-2222 | (902) 681-0477 |
| Executive Cab              | 2016 Sherwood Dr      |              | Sherwood Park          | AB          | T8A5V3      | (780) 464-1333 | (780) 757-4766 |
| Executive Cabs             | 3760 13 St NW         |              | Edmonton               | AB          | T6T0E5      | (780) 468-5411 | Not Available  |
| Executive Cabs             | 52304 Range Road 233  |              | Sherwood Park          | AB          | T8B1C9      | (780) 416-5555 | Not Available  |
| Executive Cabs Of Surrey   | 12975 84 Ave          | 103          | Surrey                 | вс          | V3W1B3      | (604) 588-3030 | (604) 507-6420 |
| Executive Taxi             | 28 Perth St           | 1            | Brockville             | ON          | K6V 5C2     | (613) 342-5655 | Not Available  |
| Executive Taxi             | 34 Jordan Vice        |              | St. John's             | NL_         | A1E2Z5      | (709) 726-2600 | (709) 747-1500 |
| Executive Taxi & Limo Svc  | 28 Perth St           |              | Brockville             | ON          | K6V5C2      | (613) 342-2000 | Not Available  |
| Express Taxi               | 161 Pennsylvannia Ave |              | Concord                | ON          | L4K1C3      | (905) 761-9655 | Not Available  |
| Fergus Elora Taxi Ltd      | 825 St George St E    |              | Fergus                 | ON          | N1M3N6      | (519) 787-7433 | Not Available  |
| First Class Cabs           | 370 Macalpine Cres    |              | Fort Mcmurray          | AB          | T9H4B1      | (780) 743-0090 | (780) 790-1166 |

|                            | Taxi Co           | ompanies in Ca | nada with Annual Sal | les over \$500, | ,000        |                |                |
|----------------------------|-------------------|----------------|----------------------|-----------------|-------------|----------------|----------------|
|                            |                   | Suite          |                      |                 |             | 1              |                |
| Company Name               | Address           | Number         | City                 | Province        | Postal Code | Phone Number   | Fax Number     |
| First Rate Taxi            | 240 Manitoba Ave  |                | Selkirk              | MB              | R1A0Y5      | (204) 785-8294 | Not Available  |
| Fort Langley Walnut Grove  |                   |                |                      |                 |             |                |                |
| Taxi                       | 17921 55 Ave      | 3              | Surrey               | BC              | V3S6C4      | (604) 882-2111 | (604) 576-2900 |
| Fort St John Cabs Ltd      | 8915 72 St        | 1              | Fort St. John        | BC              | V1J1B2      | (250) 785-3342 | (250) 787-9149 |
| Fort Taxi                  | 11201 88 Ave      |                | Ft Saskatchewan      | AB              | T8L2X4      | (780) 998-9446 | (780) 992-1229 |
| G P Cabs Inc               | 10001 101 Ave     | 205            | Grande Prairie       | AB              | T8V0X9      | (780) 814-6800 | Not Available  |
| Garden City Cabs           | 2633 Viking Way   | 148            | Richmond             | BC              | V6V3B6      | (604) 233-1111 | (604) 247-4002 |
| Georges Sky Cab            | 688 Prospect St   |                | Fredericton          | NB              | E3B6G9      | (506) 460-1111 | (506) 450-3848 |
| George's Taxi              | 163 Duke St       |                | Dryden               | ON              | P8N1G5      | (807) 223-6565 | (807) 223-7888 |
| Gerry's Taxi               | 506 Hanna St E    |                | Windsor              | ON              | N8X2N6      | (519) 966-3800 | Not Available  |
|                            |                   |                |                      |                 |             |                |                |
| Gervais Towing & Recovery  | 1485 Star Top Rd  | A              | Gloucester           | ON              | K1B3W5      | (613) 747-4666 | Not Available  |
| Glacier Cab Co             | 410 Innes St      |                | Nelson               | BC              | V1L5E7      | (250) 354-1111 | (250) 352-2222 |
| Golden Cabs Bc Ltd         | 10415 10 St       | 2              | Dawson Creek         | BC              | V1G3T8      | (250) 782-7714 | Not Available  |
| Good Taxi Svc              | 92 Capital Dr     |                | Charlottetown        | PE              | C1E1E7      | (902) 629-2000 | Not Available  |
|                            |                   |                |                      |                 |             |                |                |
| Good Taxi Svc Admin Ofc    | 152 Belvedere Ave |                | Charlottetown        | PE              | C1A2Z1      | (902) 370-2222 | Not Available  |
| Gord's Taxi                | 59 King           | 2              | Sioux Lookout        | ON              | P8T 1A1     | (807) 737-1087 | (807) 737-3024 |
| Grabba Cab                 | 161 Maypoint Rd   |                | Charlottetown        | PE              | C1E1X6      | (902) 892-6000 | Not Available  |
| Grace Taxi & Limousine Svc | 764 Campbell St   |                | Sarnia               | ON              | N7T2J6      | (519) 337-7516 | (519) 337-7994 |
| Grand Quest Taxi           | 1 Testa Rd        |                | Uxbridge             | ON              | L9P1Y9      | (905) 852-2222 | Not Available  |
| Green Taxi                 | 233 8th St E      |                | Owen Sound           | ON              | N4K1L2      | (519) 477-3535 | Not Available  |
| Grey Cab Prince Albert     | 341 16th St W     |                | Prince Albert        | SK              | S6V3V6      | (306) 764-4444 | (306) 922-4442 |
| Grumpy's Taxi & Limousine  |                   |                |                      |                 |             |                |                |
| Svc                        | 5545 Gelert Rd    |                | Minden               | ON              | ком2ко      | (705) 457-2276 | Not Available  |
| Gtown Taxi Inc             | 72 Miller Dr      |                | Georgetown           | ON              | L7G5T2      | (905) 873-2222 | Not Available  |
| Hamilton Cab               | 430 Cannon St E   |                | Hamilton             | ON              | L8L2C8      | (905) 777-7777 | Not Available  |
| Hay River Taxi             | Po Box 111        |                | Hay River            | NT              |             | (867) 874-2777 | Not Available  |
| Helping Hands              |                   |                |                      |                 |             |                |                |
| Transportation             | 433 Merritt Ave   | 1              | Chatham              | ON              | N7M3G3      | (519) 355-1654 | (519) 351-1234 |

|                           | Taxi Com              | panies in Ca | nada with Annual Sa | les over \$500, | ,000        |                |                |
|---------------------------|-----------------------|--------------|---------------------|-----------------|-------------|----------------|----------------|
|                           |                       | Suite        |                     |                 |             | !              |                |
| Company Name              | Address               | Number       | City                | Province        | Postal Code | Phone Number   | Fax Number     |
| Hemlock Taxi Inc          | 6385 Boul Monk        |              | Montreal            | QC              | H4E3H8      | (514) 766-7771 | (514) 767-0473 |
| Hms Transportation        | 260 Water St          |              | St Andrews          | NB              | E5B1B5      | (506) 529-3371 | (506) 529-4930 |
| Hms Transportation Tour   |                       |              |                     |                 |             |                |                |
| Svc                       | 260 Water St          |              | St Andrews          | NB              | E5B1B5      | (506) 529-4443 | (506) 529-4930 |
| Норе Тахі                 | 539 5th Ave           |              | Норе                | BC              | VOX 1L0     | (604) 869-5501 | Not Available  |
| Ikaluktutiak Co-Op Ltd    | 18 Omingmak St        |              | Cambridge Bay       | NU              | X0B0C0      | (867) 983-2201 | (867) 983-2085 |
| Imperial Taxi             | 40 Wynford Dr         | 106          | North York          | ON              | M3C1J5      | (416) 603-1600 | Not Available  |
| Independent Taxi          | 45 Cairns Cres        |              | Huntsville          | ON              | P1H1Y3      | (705) 788-9125 | Not Available  |
| Ingersoll Taxi & Delivery |                       |              |                     |                 |             |                |                |
| Svc                       | 55 Thames St S        |              | Ingersoli           | ON              | N5C2S8      | (519) 425-7878 | (519) 485-6071 |
| In-Home Care Inc          | 141 Victoria St       | 314          | Kamloops            | BC              | V2C1Z5      | (250) 851-0078 | Not Available  |
| International Limousine   | 1199 Aldersbrook Rd   |              | London              | ON              | N6G4T3      | (519) 474-0800 | Not Available  |
| J & D's Taxi Svc          | 37 King St            |              | Aylmer              | ON              | N5H1Z9      | (519) 773-9566 | Not Available  |
| Jiffy Cabs                | 322 Pennywell Rd      | A            | St. John's          | NL              | A1C2M4      | (709) 722-2222 | (709) 488-8888 |
| Jim's Taxi & Svc          |                       |              | Pangnirtung         | NU              | XOAORO      | (867) 473-4030 | Not Available  |
| Johnnie's Taxi            | 796 Lundell Rd        |              | Revelstoke          | BC              | VOE2SO      | (250) 837-3000 | (250) 837-4450 |
| Junash Taxi & Court Svc   | 800 Arrow Rd          |              | North York          | ON              | M9M2Z8      | (416) 748-1100 | (416) 748-9832 |
| Kalum Kabs Ltd            | 4449 Lakelse Ave      |              | Terrace             | BC              | V8G1P1      | (250) 635-7177 | (250) 635-2773 |
| Kamloops Transportation   |                       |              |                     |                 |             |                |                |
| Ltd                       | 1100 Ord Rd           | i            | Kamloops            | BC              | V2B7V4      | (250) 376-6666 | (250) 554-7791 |
| Kelowna Cabs Ltd          | 5-3312 Appalossa Road |              | Kelowna             | BC              | V1V 2W5     | (250) 762-2222 | Not Available  |
| Kings Taxi                | 43 Mississaga St E    |              | Orillia             | ON              | L3V1V4      | (705) 345-2100 | (705) 329-3369 |
|                           |                       |              |                     |                 |             |                |                |
| Kingston & Amherst Taxi   | 439 Montreal St       | 101          | Kingston            | ON              | к7КЗН4      | (613) 542-3333 | Not Available  |
| Kinngait Taxi Ltd         |                       |              | Cape Dorset         | NU              | X0A0C0      | (867) 897-7000 | Not Available  |
| Klein Cabs                | 116 2 St E            |              | Brooks              | AB              | T1R0G9      | (403) 362-8866 | Not Available  |
| Kootenay Taxi             | 1502 9th Ave          | B            | Fernie              | BC              | V0B1M0      | (250) 425-2121 | (250) 423-4413 |
| Krammer's Taxi            | 461 Campbell St       |              | Sarnia              | ON              | N7T2J1      | (519) 336-0000 | Not Available  |
| K-V Cab                   | 178 Pettingill Rd     |              | Quispamsis          | NB              | E2E2V6      | (506) 848-8294 | (506) 849-6185 |
| L A Family Taxi           | 119 Cuthand St        |              | La Ronge            | SK              | SOJ1LO      | (306) 425-3500 | (306) 425-5588 |
| Lachute Taxi Enr          | 544 Rue Lafleur       |              | Lachute             | QC              | J8H1R6      | (450) 562-2222 | (450) 562-2429 |

|                                | Taxi Com                 | panies in Ca | nada with Annual Sa | les over \$500 | ,000        |                |                |
|--------------------------------|--------------------------|--------------|---------------------|----------------|-------------|----------------|----------------|
|                                |                          | Suite        |                     |                |             |                | 1              |
| Company Name                   | Address                  | Number       | City                | Province       | Postal Code | Phone Number   | Fax Number     |
| Langley Cabs Ltd               | 17921 55 Ave             | 3            | Surrey              | BC             | V3S6C4      | (604) 533-3333 | Not Available  |
| Langley Taxi Cabs              | 17921 55 Ave             | 3            | Surrey              | BC             | V3S6C4      | (604) 534-5311 | (604) 576-2900 |
| Layton's Taxi                  | 42 Outram St             | , <b></b>    | Truro               | NS             | B2N4B9      | (902) 895-4471 | (902) 895-1456 |
| Leamington Taxi                | 268 Erie St S            |              | Leamington          | ON             | N8H3C5      | (519) 322-1444 | Not Available  |
| Lee's Taxi                     | 98 Centennial Dr         |              | Windsor             | NS             | BON2TO      | (902) 798-2062 | (902) 798-4391 |
| Legion Taxi                    | 101 Owl St               |              | Banff               | AB             | T1L1C4      | (403) 762-3353 | (403) 762-5811 |
| Len's Taxi                     | 1 Union St               |              | Campbellton         | NB             | E3N1B8      | (506) 789-0088 | Not Available  |
| Leo's Taxi                     | 336 Crammond Crt         |              | Miramichi           | NB             | E1V2B7      | (506) 624-7132 | Not Available  |
| Limo Vip Taxi                  |                          |              | lqaluit             | NU             | хоаоно      | (867) 979-3030 | Not Available  |
| Limousine Medaillon Inc        | 241 Rue Rivard           |              | Magog               | QC             | J1X4W4      | (819) 847-1989 | Not Available  |
| Livraison Clement              | 150 Boul Albiny-Paquette |              | Mont-Laurier        | QC             | J9L1J4      | (819) 623-3434 | (819) 623-1533 |
| Lloyd Taxi                     | 4210 44 St               |              | Lloydminster        | SK             | S9V1Z8      | (306) 825-3333 | (306) 825-0777 |
| Lockerby Taxi                  | 339 Harrison Dr          |              | Sudbury             | ON             | P3E5E1      | (705) 560-3210 | (705) 522-1654 |
| Lockterby Taxi                 | 339 Harrison Dr          |              | Sudbury             | ON             | P3E5E1      | (705) 522-2222 | Not Available  |
| London Taxi                    | 1540 Fanshawe Park Rd W  |              | London              | ON             | N6H5L8      | (519) 657-1111 | Not Available  |
| Loyal Taxi                     | 513 Union St             |              | Fredericton         | NB             | E3A3N3      | (506) 455-6789 | (506) 458-2711 |
| Loyal Taxi                     | 331 Rue Main             |              | Gatineau            | QC             | J8P5K5      | (819) 663-5252 | Not Available  |
| M & G Cab                      | 8799 Commercial St       |              | New Minas           | NS             | B4N3C4      | (902) 691-2500 | Not Available  |
| M & G Cab & Van Svc            | 4281 Scottsbay Rd        |              | Canning             | NS             | BOP1HO      | (902) 691-2500 | (902) 582-1103 |
| M & M'z Taxi                   | 55 Alberta Ave           | 4            | Spruce Grove        | AB             | T7X3A2      | (780) 962-3256 | (780) 962-3263 |
| M S Taxi                       | 456 Wellesly St          |              | Hawkesbury          | ON             | K6A2G1      | (613) 632-2141 | Not Available  |
| M Taxi Inc                     | 7575 Rte Transcanadienne |              | Saint-Laurent       | QC             | H4T1V6      | (514) 336-5000 | Not Available  |
| Mac Lure's Cabs Ltd            | 1510 3rd Ave W           |              | Vancouver           | BC             | V6J1J7      | (604) 683-6666 | (604) 731-9219 |
| Magog Orford Taxi Inc          | 45 Rue Edouard E         |              | Magog               | QC             | J1X1H3      | (819) 843-3377 | (819) 843-6401 |
| Maple Leaf Taxi                | 40 Wynford Dr            |              | North York          | ON             | M3C1J5      | (416) 465-5555 | Not Available  |
| Marsch Limousine & Taxi<br>Svc | 410 Pim St               |              | Sault Ste. Marie    | ON             | P6B2V1      | (705) 759-0159 | Not Available  |

|  | Taxi Com               |        | nada with Annual Sale | es over \$500, | ,000        |                | 1              |
|--|------------------------|--------|-----------------------|----------------|-------------|----------------|----------------|
|  |                        | Suite  |                       |                |             |                |                |
| Company Name                           | Address                | Number | City                  | Province       | Postal Code | Phone Number   | Fax Number     |
| Mary T Taxi                            |                        |        | Pangnirtung           | NU             | XOAORO      | (867) 473-4000 | Not Available  |
| Matsqui Taxi Ltd                       | 2625 Montrose Ave      |        | Abbotsford            | BC             | V2S3T5      | (604) 852-8600 | (604) 859-1129 |
| Mayfair Taxi Ltd                       | 7003 Farrell Rd SE     | 9      | Calgary               | AB             | T2H0T3      | (403) 255-6555 | (403) 255-3968 |
| Mc Kab Taxi                            | 55 Sinclair Ave        | 13     | Georgetown            | ON             | L7G4X4      | (905) 877-1234 | (905) 702-9900 |
| Meadow Ridge Taxi                      | 20542 Dewdney Trunk Rd |        | Maple Ridge           | BC             | V2X3E3      | (604) 465-5555 | Not Available  |
| Meadowlark Taxi                        | 124 South Ave          | В      | Spruce Grove          | AB             | T7X3A9      | (780) 962-8771 | (780) 960-2501 |
| Mercury Taxi                           | 113 Main St            |        | Penetanguishene       | ON             | L9M1S9      | (705) 549-8877 | (705) 549-9388 |
| ······································ | 3930 Rue Sainte-       |        |                       |                |             |                |                |
| Michael Siroky Taxi Reg'd              | Catherine E            | А      | Montreal              | QC             | H1W2G6      | (514) 524-5651 | (514) 527-3507 |
| Midtown Taxi                           |                        |        | Iqaluit               | NU             | X0A0H0      | (867) 979-1555 | Not Available  |
| Midway Taxi                            | 3366 Plummer Ave       |        | New Waterford         | NS             | B1H1Y7      | (902) 862-6800 | Not Available  |
| Mike's Taxi                            | 116 Cedar St           |        | Woodstock             | NB             | E7M1E9      | (506) 328-9581 | Not Available  |
| Mission Taxi Ltd                       | 31265 Wheel Ave        | 207    | Abbotsford            | BC             | V2T6H2      | (604) 864-2311 | (604) 855-9122 |
| Mj Taxi                                |                        |        | Cape Dorset           | NU             | X0A0C0      | (867) 897-7979 | Not Available  |
| Modern Taxi Cab Ltd                    | 665 Montreal St        |        | Kingston              | ON             | к7к3ј3      | (613) 546-2222 | (613) 546-0044 |
| Mountain Taxi & Tours Ltd              | 101 Owl St             |        | Banff                 | АВ             | T1L1C4      | (403) 762-3351 | (403) 762-5811 |
| Mt 7 Taxi                              | 801 10th Ave S         |        | Golden                | BC             | V0A1H0      | (250) 344-5237 | (250) 344-7190 |
| Naders Taxi                            | 36 Mill St W           |        | Leamington            | ON             | N8H1S8      | (519) 322-2262 | (519) 324-0830 |
| New York Taxi                          | 420 Main St            |        | Biggar                | SK             | SOKOMO      | (306) 948-4949 | (306) 948-4899 |
| Newfie Cab Inc                         | 10012 Franklin Ave     | 201    | Fort Mcmurray         | АВ             | T9H2K6      | (587) 276-2222 | Not Available  |
| Newfound Cabs                          | 322a Penny Rd          |        | St. John's            | NL             |             | (709) 726-5050 | Not Available  |
| Newmarket Taxi & Limo                  | 1211 Gorham St         | 13     | Newmarket             | ON             | L3Y8Y3      | (905) 895-0900 | (905) 868-9424 |
| Newton Whalley Hiway                   |                        |        |                       |                |             |                |                |
| Taxi Ltd                               | 13119 84 Ave           | 107    | Surrey                | BC             | V3W1B3      | (604) 581-1111 | (604) 597-8002 |
| Niagara Air Bus Taxi                   | 8626 Lundy's Lane      |        | Niagara Falls         | ON             | L2H1H4      | (905) 374-8111 | (905) 374-1633 |
| Niagara Falls Taxi Ltd                 | 4986 Victoria Ave      |        | Niagara Falls         | ON             | L2E4C8      | (905) 357-4000 | (905) 357-3322 |
| Nic Taxi                               | 956 Rue Sainte-Helene  |        | Longueuil             | QC             | J4K3R9      | (450) 659-9292 | Not Available  |
| Nice Taxi & Limousine Svc              | 537 Vellore Woods Blvd |        | Woodbridge            | ON             | L4H2W3      | (905) 482-1666 | Not Available  |

\*

٠

|                           | Taxi Co              | npanies in Ca | nada with Annual Sal | es over \$500 | ,000        |                |                |
|---------------------------|----------------------|---------------|----------------------|---------------|-------------|----------------|----------------|
|                           |                      | Suite         |                      |               |             | 1              |                |
| Company Name              | Address              | Number        | City                 | Province      | Postal Code | Phone Number   | Fax Number     |
| North Shore Taxi Ltd      | 264 Pemberton Ave    |               | North Vancouver      | BC            | V7P2R5      | (604) 922-2222 | Not Available  |
| North West Taxi           | 430 Topsail Rd       | 201           | St. John's           | NL            | A1E4N1      | (709) 364-1451 | (709) 364-3227 |
| Northern Frontier Svc Ltd |                      |               | lqaluit              | NU            | хоаоно      | (867) 979-2851 | Not Available  |
| Nunaga Taxi               |                      |               | Iqaluit              | NU            | ХОАОНО      | (867) 979-1999 | Not Available  |
| O K Taxi Svc              | Po Box 1091 Stn Main |               | Okotoks              | AB            | T1S1B2      | (403) 938-3828 | Not Available  |
| Old Town Taxi             | 89 Water St          |               | Summerside           | PE            | C1N1A6      | (902) 436-4947 | Not Available  |
| Olde Tyme Taxi            | 113 Mutual St N      |               | Ingersoll            | ON            | N5C1Z8      | (519) 425-0110 | Not Available  |
| On-Time Taxi Inc          | 100 Rexdale Blvd     | 102           | Etobicoke            | ON            | M9W1N7      | (416) 243-2323 | Not Available  |
| Orangeville Taxi          | 302 Broadway Ave     |               | Orangeville          | ON            | L9W1L3      | (519) 941-8294 | Not Available  |
| Orillia Taxi Svc          | 10 Kitchener St      | В             | Orillia              | ON            | L3V6Z9      | (705) 326-8294 | Not Available  |
| Oxford Taxi               | 255 Boul D'Anjou     |               | Chateauguay          | QC            | J6J2R4      | (450) 691-2020 | Not Available  |
| Pacific Cabs              | 17921 55 Ave         | 3             | Surrey               | BC            | V3S6C4      | (604) 596-6666 | Not Available  |
| Pai-Pa Taxi Garage        | 1087 Airport Rd      |               | Iqaluit              | NU            | ходоно      | (867) 979-2585 | (867) 979-2383 |
| Pai-Pa Taxi Ltd           | 1087 Airport Rd      |               | Iqaluit              | NU            | ХОАОНО      | (867) 979-5222 | Not Available  |
| Paris Taxi Transportation |                      |               |                      |               |             |                |                |
| Svc                       | 764 Governors Rd E   | В             | Paris                | ON            | N3L3E1      | (519) 442-7873 | (519) 442-7875 |
| Parker's Taxi             | 260 Foord St         |               | Stellarton           | NS            | BOK1SO      | (902) 755-5050 | Not Available  |
| Parkland Taxi             | 1430 Main St S       |               | Dauphin              | MB            | R7N1M8      | (204) 638-5860 | (204) 638-5946 |
| Paul's First Choice Taxi  | 381 Lake Dr N        |               | Keswick              | ON            | L4P3C8      | (905) 989-2222 | Not Available  |
| Peninsula Taxi            | 10326 Gabriola Pl    |               | Sidney               | BC            | V8L3J8      | (250) 656-1111 | Not Available  |
| Penticton Taxi            | 2319 Government St   |               | Penticton            | BC            | V2A4W5      | (250) 492-5555 | Not Available  |
| People's 24 Hour Taxi Svc | 1409 Sunset Dr       |               | Bathurst             | NB            | E2A4C7      | (506) 546-4435 | (506) 548-3712 |
| People's Taxi             | 711 Finley Ave       |               | Ajax                 | ON            | L1S3T1      | (905) 427-7770 | Not Available  |
| Plaza Taxi                | 11 Liberty St        |               | Sydney               | NS            | B1P7C7      | (902) 539-8770 | Not Available  |
| Pontiac Hemlock Taxi Cie  | 6385 Boul Monk       |               | Montreal             | QC            | H4E3H8      | (514) 322-6000 | Not Available  |
| Pontiac Vip Hemlock       | 6385 Boul Monk       | 1             | Montreal             | QC            | H4E3H8      | (514) 931-6666 | Not Available  |
| Port Coquitiam Taxi       | 2121 Hartley Ave     | 1             | Coquitlam            | BC            | V3K6Z3      | (604) 942-8080 | (604) 521-7790 |
| Port Hope Taxi            | 3 Maitland St        | 1             | Port Hope            | ON            | L1A4H6      | (905) 885-6369 | (905) 885-2464 |

|                            | Taxi Con              | npanies in Ca | nada with Annual Sale | s over \$500, | ,000        |                        |                |
|----------------------------|-----------------------|---------------|-----------------------|---------------|-------------|------------------------|----------------|
| <u> </u>                   |                       | Suite         |                       |               |             |                        |                |
| Company Name               | Address               | Number        | City                  | Province      | Postal Code | Phone Number           | Fax Number     |
| Port Hope/Cobourg Taxi     | 3 Maitland St         |               | Port Hope             | ON            | L1A4H6      | (905) 885-7264         | (905) 885-2464 |
| Port Moody Taxi Ltd        | 2121 Hartley Ave      |               | Coquitlam             | BC            | V3K6Z3      | (604) 942-7777         | (604) 521-7790 |
| Prairie Cabs               | 11315 96 Ave          |               | Grande Prairie        | AB            | T8V5M3      | (780) 532-1060         | (780) 539-4600 |
| Prestige Cabs              | 10135 31 Ave NW       |               | Edmonton              | AB            | T6N1C2      | (780) 462-4444         | (780) 462-2722 |
| Prince George Taxi Holding |                       |               |                       |               |             |                        |                |
| Со                         | 331 1st Ave           |               | Prince George         | ВС            | V2L2Y1      | (250) 564-4444         | Not Available  |
| R & E Transportation       | Po Box 309            |               | Burin Bay Arm         | NL            | A0E1G0      | (709) 891-1866         | Not Available  |
| R Taxi                     | 790 Lundell Rd        |               | Revelstoke            | BC            | V0E2S0      | (250) 837-4000         | (250) 837-4450 |
| Radio Cab                  | 197 Waterloo St S     |               | Stratford             | ON            | N5A4B6      | (519) 271-4242         | (519) 271-4245 |
| Radio Cabs                 | 71 Queen St           | A             | Marystown             | NL            | A0E2M0      | (709) 279-2222         | (709) 279-1067 |
| Radio Taxi Monaco Ltee     | 948 Rue Saint-Louis   |               | Joliette              | QC            | J6E3A4      | (450 <u>)</u> 753-7575 | (450) 753-7576 |
| Radio Taxi Union Ltee      | 1605 Rue Vercheres    |               | Longueuil             | QC            | J4K2Z6      | (450) 679-6262         | (450) 674-6263 |
| Rebel Taxi                 | 6105 46 St            | 2             | Leduc                 | AB            | T9E6T8      | (780) 980-0567         | Not Available  |
| Red Lake Taxi              | 2 Pugsley St          |               | Red Lake              | ON            | P0V2M0      | (807) 727-2100         | (807) 727-2127 |
| Red Line Taxi              | 1615 2nd Ave E        |               | Owen Sound            | ON            | N4K2J6      | (519) 376-5050         | (519) 373-0972 |
| Redcliff Taxi              | 216 Broadway Ave E    |               | Redcliff              | AB            | T0J2P0      | (403) 527-1969         | Not Available  |
| Redline Taxi               | 372 Fifth Ave N       | 4             | Yorkton               | SK            | S3N3G2      | (306) 783-1010         | (306) 783-8446 |
| Regal Taxi                 | 165 Rue Jean-Proulx   |               | Gatineau              | QC            | J8Z1T4      | (819) 777-5231         | (819) 777-9285 |
| Richmond & Coral Taxi      |                       |               |                       |               |             |                        |                |
| Cabs                       | 11180 Voyageur Way    | 260           | Richmond              | BC            | V6X3N8      | (604) 272-1111         | Not Available  |
| Richmond Taxi Co Ltd       | 5971 North Service Rd |               | Richmond              | BC            | V7B0A7      | (604) 303-3378         | Not Available  |
|                            |                       |               |                       |               |             |                        |                |
| Rick's Delivery & Taxi Svc | 215 7th St            |               | Weyburn               | SK            | S4H1C3      | (306) 842-6014         | (306) 842-2524 |
| Rita's Taxi                | 59 King St            | 2             | Sioux Lookout         | ON            | P8T1A5      | (807) 737-3636         | (807) 737-3204 |
| Riviera Taxi               | 756 Rue Saint-Louis   |               | Gatineau              | QC            | J8T2S4      | (819) 243-5316         | (819) 246-1057 |
| Roach's Yellow Cabs        | 216 Camelot St        |               | Thunder Bay           | ON            | P7A4B1      | (807) 345-7721         | (807) 346-9210 |
| Roach's Yellow Taxi        | 218 Camelot St        |               | Thunder Bay           | ON            | P7A4B1      | (807) 344-8481         | Not Available  |
| Royal City Taxi Ltd        | 436 Rousseau St       |               | New Westminster       | BC            | V3L3R3      | (604) 521-6666         | Not Available  |
| Royal Taxi                 | 200 Elizabeth St      |               | Toronto               | ON            | M5G2C4      | (416) 977-6869         | Not Available  |
| Royal Taxi                 | 2075 Bayview Ave      |               | Toronto               | ON            | M4N3M5      | (416) 481-1488         | Not Available  |
| Royal Taxi Ltd             | 238 12b St N          |               | Lethbridge            | AB            | T1H2K7      | (403) 328-5333         | Not Available  |

|                            | Taxi Con              | npanies in Ca | nada with Annual Sa | les over \$500, | ,000        | -              |                |
|----------------------------|-----------------------|---------------|---------------------|-----------------|-------------|----------------|----------------|
|                            |                       | Suite         |                     |                 | r           |                | 1              |
| Company Name               | Address               | Number        | City                | Province        | Postal Code | Phone Number   | Fax Number     |
| Royal Taxi/AAA Markham     |                       |               |                     |                 |             |                | 1              |
| Rchmn                      | 9078 Leslie St        | 1             | Richmond Hill       | ON              | L4B3L8      | (905) 895-0455 | (905) 889-6919 |
| Ruest Taxi                 | 19 Rue Dugal          |               | Edmundston          | NB              | E3V1X4      | (506) 735-3313 | Not Available  |
| Salmon Arm Taxi Ltd        | 875 Lakeshore Dr SW   |               | Salmon Arm          | BC              | V1E1E4      | (250) 832-2252 | (250) 832-4228 |
| Saskatoon Radio Cabs Ltd   | 50 23rd St E          |               | Saskatoon           | SK              | S7К0Н8      | (306) 242-1221 | Not Available  |
| Satellite Taxi             | 388 Cobequid Rd       |               | Lower Sackville     | NS              | B4C4C5      | (902) 865-4444 | (902) 864-3500 |
| Scarborough City Cab       | 1940 Ellesmere Rd     | 18            | Scarborough         | ON              | M1H2V7      | (416) 438-5151 | (416) 438-1733 |
| Sea To Sky Taxi Ltd        | 1080 Millar Creek Rd  | 201           | Whistler            | BC              | VON1B1      | (604) 932-3333 | Not Available  |
| Semiahmoo Taxi             | 13119 84 Ave          | 107           | Surrey              | BC              | V3W1B3      | (604) 531-5555 | (604) 597-8002 |
| Serge's Taxi Svc           | 175 Rexdale Bivd      |               | Etobicoke           | ON              | M9W1P7      | (416) 744-2224 | Not Available  |
| Shilo Taxi Knight-Line     |                       |               |                     |                 | <u></u>     |                |                |
| Express                    | 601 9th St            |               | Brandon             | мв              | R7A4B3      | (204) 571-6570 | Not Available  |
| Sidney Taxi                | 25 Frankford Cres     |               | Trenton             | ON              | K8V6H8      | (613) 394-1462 | (613) 394-3218 |
| South Coast Haldimand      |                       |               |                     |                 |             |                |                |
| Тахі                       | 881 Haldibrook Rd     |               | Caledonia           | ON              | N3W1N5      | (905) 765-5445 | (905) 765-4461 |
| South Shore Transportation | 25 Renaud Dr          |               | Amherstburg         | ON              | N9V4A9      | (519) 736-1761 | (519) 736-3986 |
| Sparky's Taxi              | 4104 Kepler St        |               | Whitecourt          | AB              | T7S0A3      | (780) 778-8994 | Not Available  |
| Spring Taxi                | 880 Logan Ave         |               | Winnipeg            | MB              | R3E1N8      | (204) 774-8294 | (204) 774-6232 |
| St John Taxi               | 125 Av Millidge       |               | Saint John          | NB              | E2K2M4      | (506) 693-0000 | Not Available  |
| St Marys Taxi Svc          | 728 Queen Street East |               | St Marys            | ON              | N4X 1G2     | (519) 284-3773 | Not Available  |
| Standard Taxi              | 15 Mackenzie Rd       |               | Fredericton         | NB              | E3B6B6      | (506) 450-4444 | (506) 457-2490 |
| Standard Taxi              | 245 Hilton Rd         | 24            | Fredericton         | NB              | E3B7B5      | (506) 459-1100 | Not Available  |
| Star Taxi                  | 1475e Theatre Rd      |               | Cranbrook           | BC              | V1C7G3      | (250) 426-3888 | Not Available  |
| Star Taxi Ltd              | 100 West St           |               | Corner Brook        | NL              | A2H2Z3      | (709) 634-4343 | (709) 634-9424 |
| Stony Plain Towing & Taxi  | 124 South Ave         | В             | Spruce Grove        | АВ              | T7X3A9      | (780) 963-1525 | (780) 962-0806 |
| Sun Parlor Cabs            | 9 Erie St S           |               | Leamington          | ON              | N8H3A6      | (519) 326-9063 | (519) 326-9064 |
| Sun Taxi                   | 140 Mackenzie King Rd |               | Fort Mcmurray       | AB              | T9H4L2      | (780) 743-5050 | Not Available  |
| Sun Taxi Inc               | 36 Riedel St          | 4             | Fort Mcmurray       | AB              | T9H3E1      | (780) 743-0045 | Not Available  |

|                           | Taxi Com                 | panies in Ca | nada with Annual Sale | es over \$500, | ,000        |                |                |
|---------------------------|--------------------------|--------------|-----------------------|----------------|-------------|----------------|----------------|
|                           |                          | Suite        |                       |                |             |                |                |
| Company Name              | Address                  | Number       | City                  | Province       | Postal Code | Phone Number   | Fax Number     |
| Sunny's Taxi Svc          | 220 Princess St          |              | Strathroy             | ON             | N7G2S7      | (519) 245-7777 | Not Available  |
| Sunrise Cabs              | 6534 Sparrow Dr          |              | Leduc                 | AB             | T9E6T9      | (780) 986-5400 | (780) 986-3035 |
| Sunset Taxi               | 759 1st St               | G            | Brandon               | MB             | R7A2X5      | (204) 729-9000 | Not Available  |
| Sunshine Coast Taxi Svc   | 761 O'Shea Rd            |              | Gibsons               | BC             | V0N1V9      | (604) 865-0466 | Not Available  |
| Super Taxi Inc            | 6893 Rue Beaubien E      |              | Montreal              | QC             | H1M3B2      | (514) 382-1010 | (514) 256-8608 |
| Superior Accessible Taxi  | 307 Simpson St           |              | Thunder Bay           | ON             | P7C3H8      | (807) 345-8595 | Not Available  |
| Surdell-Kennedy Taxi Ltd  | 12975 84 Ave             | 103          | Surrey                | BC             | V3W1B3      | (604) 588-8888 | Not Available  |
| Sushi Taxi                | 585 116 Rte              | 107          | Saint-Nicolas         | QC             | G7A2P6      | (418) 831-9393 | Not Available  |
| Swan Taxi                 | 9515 116 St              |              | Grande Prairie        | AB             | T8V5W3      | (780) 539-4000 | (780) 831-0243 |
| T Z Ctr Du Quebec         | 27 Rue Dorilla           |              | Victoriaville         | QC             | G6T1M7      | (819) 758-0780 | Not Available  |
| Taxi 2000                 | 116 South Cayuga St W    |              | Dunnville             | ON             | N1A1P5      | (905) 774-9229 | (905) 774-9229 |
| Taxi 2151                 | 1111 Av Du Port          |              | La Baie               | QC             | G7B1W2      | (418) 544-2151 | (418) 544-6084 |
| Taxi 24                   | 1329 4e Av               |              | Val-D'Or              | QC             | J9P1K5      | (819) 874-2424 | Not Available  |
| Taxi 6000                 | 352 Rue Principale       |              | La Sarre              | QC             | J9Z1Z5      | (819) 333-6000 | Not Available  |
| Taxi 800 Inc              | 55 Rue De L'Eveche E     |              | Rimouski              | QC             | G5L1X7      | (418) 723-3344 | (418) 723-3377 |
| Taxi A 52 Chicoutimi      | 1736 Boul St-Paul        |              | Chicoutimi            | QC             | G7J4N1      | (418) 549-5200 | Not Available  |
| Taxi Adapte Auger         | 880 Rue De L'Eglise      |              | Saint-Romuald         | QC             | G6W5M6      | (418) 833-3339 | (418) 833-2393 |
| Taxi Adapte Roland Deziel | 398 6e Av                |              | Grand-Mere            | QC             | G9T2P8      | (819) 539-5598 | (819) 538-8411 |
| Taxi Beaulieu             | 23 Ch Des Raymond        |              | Riviere-Du-Loup       | QC             | G5R2H9      | (418) 862-3111 | Not Available  |
| Taxi Bma Soucy            | 384 Av De Buckingham     |              | Gatineau              | QC             | J8L2G7      | (819) 986-3332 | Not Available  |
| Taxi Bruneau              | 35 Rue Lessard           |              | St-Jean-De-Matha      | QC             | JOK2SO      | (450) 886-2331 | Not Available  |
| Taxi Bus                  | 720 Trotter              |              | St-Jn-S-Richlieu      | QC             | J3B8T2      | (450) 348-8287 | Not Available  |
| Taxi Candare              | 7294 Rue Lajeunesse      |              | Montreal              | QC             | H2R2H4      | (514) 336-1313 | (514) 273-1866 |
| Taxi Chambly              | 250 Boul Frechette       | A            | Chambly               | QC             | J3L2Z5      | (450) 658-3000 | (450) 447-3223 |
| Taxi Clement              | 150 Boul Albiny-Paquette |              | Mont-Laurier          | QC             | J9L1J4      | (819) 623-3434 | (819) 623-1533 |
| Тахі Со-Ор                | 40 Rue Notre-Dame E      |              | Thetford Mines        | QC             | G6G1J2      | (418) 338-5115 | Not Available  |
|                           | 455 Boul Henri-Bourassa  |              |                       |                |             |                |                |
| Taxi Co-Op De L'Est       | 0                        |              | Montreal              | QC             | H3L1P3      | (514) 325-6000 | (514) 352-7189 |

|                             | Taxi Com               | panies in Ca | nada with Annual Sal | les over \$500 | ,000        |                |                |
|-----------------------------|------------------------|--------------|----------------------|----------------|-------------|----------------|----------------|
|                             |                        | Suite        |                      |                |             |                |                |
| Company Name                | Address                | Number       | City                 | Province       | Postal Code | Phone Number   | Fax Number     |
| Taxi Co-Op Loretteville     | 11000 Boul Valcartier  |              | Quebec               | QC             | G2A 2M3     | (418) 842-2724 | Not Available  |
| Taxi Co-Op Quebec           | 975 8e Av              |              | Quebec               | QC             | G1J3M8      | (418) 525-8123 | (418) 525-2004 |
|                             | 2639 Boul Du Versant-  |              |                      |                |             |                |                |
| Taxi Co-Op Ste-Foy Sillery  | Nord                   |              | Quebec               | QC             | G1V1A3      | (418) 653-7777 | Not Available  |
| Taxi Co-Op-Charlesbourg     | 111 58e Rue E          |              | Quebec               | QC             | G1H2E7      | (418) 626-5252 | Not Available  |
| Taxi Cooperative Quebec     | 496 2e Ave             |              | Quebec               | QC             | G1L3B1      | (418) 525-5191 | (418) 525-8661 |
| Taxi D'Amos Inc & Royal     | 122 10e Av O           |              | Amos                 | QC             | J9T1W8      | (819) 732-3355 | (819) 732-5356 |
| Taxi De Co-Op De Mont-Joli  | 20 Av De La Gare       |              | Mont-Joli            | QC             | G5H1N7      | (418) 775-4343 | (418) 775-4344 |
| Taxi De Hull                | 423 Rue Jean-Perrin    |              | Gatineau             | QC             | J8V2R7      | (819) 775-8969 | Not Available  |
| Taxi Des Pays D'En Haut Enr |                        |              | Sainte-Adele         | QC             | J8B2N1      | (450) 229-3535 | Not Available  |
|                             | 2475 Rue Saint-        |              |                      |                |             |                |                |
| Taxi Diamond                | Dominique              |              | Jonquiere            | QC             | G7X6K4      | (418) 547-2121 | (418) 547-2122 |
| Taxi Diamond                | 7294 Rue Lajeunesse    | A            | Montreal             | QC             | H2R2H4      | (514) 273-6331 | Not Available  |
| Taxi Dr Enr                 | 658 Rue Saint-Francois |              | Granby               | loc            | J2G3K5      | (450) 777-7766 | Not Available  |
| Taxi Du Lac Enr             | 177 Rue Saint-Joseph   |              | Alma                 | QC             | G8B3E7      | (418) 662-6010 | (418) 662-3737 |
| Taxi Du Pont                | 112 90 2e Av           |              | Saint-Georges        | QC             | G5Y1W2      | (418) 227-7777 | (418) 227-9801 |
| Taxi Guy                    | 19 Charlotte St        |              | Toronto              | ON             | M5V2H5      | (416) 597-0563 | Not Available  |
| Taxi Hudson                 | 66 Rue Lauzon          |              | Rigaud               | QC             | JOP1P0      | (450) 458-4722 | Not Available  |
| Taxi Lachute Inc            | 544 Rue Lafleur        |              | Lachute              | QC             | J8H1R6      | (450) 562-2429 | Not Available  |
| Taxi Latuque                | 530 Rue Saint-Louis    |              | La Tuque             | QC             | G9X2X4      | (819) 523-2525 | (819) 523-2265 |
| Taxi Martin St Lin Des      |                        |              |                      |                |             |                |                |
| Lrntds                      | 281 Rue Salaberry      |              | St-Lin-Laurentds     | QC             | J5M3A5      | (450) 439-2209 | Not Available  |
| Taxi Matane Inc             | 394 Av Saint-Jerome    |              | Matane               | QC             | G4W3B5      | (418) 562-2030 | (418) 562-2600 |
| Taxi National Inc           | 6385 Boul Monk         |              | Montreal             | QC             | H4E3H8      | (514) 762-1200 | (514) 761-3633 |
| Taxi Naudville Ltee         | 177 Rue Saint-Joseph   |              | Alma                 | QC             | G8B3E7      | (418) 662-2110 | (418) 662-3737 |
| Taxi Newman Lafleur Inc     | 6385 Boul Monk         |              | Montreal             | QC             | H4E3H8      | (514) 363-5111 | (514) 761-3633 |

۴

•ى

|                             | Taxi Comp                  | anies in Ca | nada with Annual Sale | s over \$500, | ,000        |                             |                |
|-----------------------------|----------------------------|-------------|-----------------------|---------------|-------------|-----------------------------|----------------|
|                             |                            | Suite       |                       | 1             | 1           |                             |                |
| Company Name                | Address                    | Number      | City                  | Province      | Postal Code | Phone Number                | Fax Number     |
| Taxi Porlier                | 24 Rue Harbour             |             | Gaspe                 | QC            | G4X1V1      | (418) 368-3131              | Not Available  |
| Taxi Porlier                | Cp 1119                    |             | Schefferville         | QC            | G0G2T0      | (418) 585-3453              | (418) 585-2611 |
| Taxi Raymond Dulac          | 237 Av Gilbert             |             | Sainte-Marie          | QC            | G6E2R7      | (418) 387-3333              | (418) 387-3833 |
| Taxi Rigaud                 | 66 Rue Lauzon              |             | Rigaud                | QC            | JOP1P0      | (450) 451-3000              | Not Available  |
| Taxi Rive-Nord              | 105 Rue L'Ecuyer           |             | Repentigny            | QC            | J6A8C5      | (450) 581-2220              | (450) 581-5481 |
| Taxi Rod                    | 1393 Rue Ouiatchouan       |             | Mashteuiatsh          | QC            | G0W2H0      | (418) 275-3832              | (418) 275-3832 |
| Taxi Rosemont Plus          | 5903 Rue Belanger          | -           | Saint-Leonard         | QC            | H1T1G8      | (514) 255-1313              | Not Available  |
| Taxi Rouyn-Noranda &        |                            |             |                       |               |             |                             |                |
| Aeroport                    | 179 Av Carter              |             | Rouyn-Noranda         | QC            | J9X1S5      | (819) 763-7800              | Not Available  |
| Taxi Royal                  | 502 3e Rue                 |             | Chibougamau           | QC            | G8P1N9      | (81 <del>9</del> ) 748-4050 | Not Available  |
| Taxi Royal                  | 7294 Rue Lajeunesse        |             | Montreal              | QC            | H2R2H4      | (514) 274-3333              | (514) 273-6298 |
|                             |                            |             |                       |               |             |                             |                |
| Taxi Ste Jean Chrysostome   | 912 Commerciale            |             | St-Jn-Chrysostme      | QC            | G6Z2E5      | (418) 832-6661              | Not Available  |
| Taxi Ste-Agathe Enr         | 118 Rue Principale E       | 106         | Ste-Agathe-Monts      | QC            | J8C1K1      | (819) 326-3300              | (819) 326-0624 |
| Taxi Taxi                   | 101 Owl St                 |             | Banff                 | AB            | T1L1C4      | (403) 762-2244              | Not Available  |
| Taxi Trois Mille Inc        | 12 Rue Du Centre           |             | Granby                | QC            | J2G5B3      | (450) 372-3000              | (450) 378-7021 |
|                             |                            | -           |                       |               |             |                             |                |
| Taxi Unis Baie-Comeau Inc   | 181 Boul La Salle          |             | Baie-Comeau           | QC            | G4Z1S7      | (418) 296-3333              | Not Available  |
| Taxi Venise                 | 171 Rue Alexandre          |             | Salaberry-Vlyfld      | QC            | J6S3J1      | (450) 373-2727              | (450) 371-9383 |
|                             |                            |             |                       |               |             |                             |                |
| Taxi Veterans Victoriaville | 91 Rue Saint-Jean-Baptiste |             | Victoriaville         | QC            | G6P4E7      | (819) 752-2222              | (819) 752-4787 |
|                             |                            |             |                       |               | 1           |                             |                |
| Taxi Windsor                | 1305 Rue Calixa-Lavallee   |             | Saint-Hyacinthe       | QC            | J2S3E7      | (450) 774-2222              | (450) 773-3499 |
| Taxibus                     | 250 Boul Poliquin          | · · · · · · | Sorel-Tracy           | QC            | J3P7Y9      | (450) 743-3336              | Not Available  |
|                             |                            | · · ·       |                       |               |             |                             |                |
| Taxibus                     | 91 Rue Saint-Jean-Baptiste |             | Victoriaville         | QC            | G6P4E7      | (819) 752-4549              | (819) 752-4787 |
| Taxibus Sept-Iles           | 652 Av De Quen             |             | Sept-Iles             | QC            | G4R2R5      | (418) 968-2877              | (418) 962-2871 |
|                             | 2631 Boul Du Versant-      |             |                       |               |             |                             |                |
| Taxis Cooperative           | Nord                       |             | Quebec                | QC            | G1V1A3      | (418) 833-3000              | (418) 658-2005 |
| Taxis Cooperative-La        |                            |             |                       |               |             |                             |                |
| Mauricie                    | 3055 Boul Des Forges       |             | Trois-Rivieres        | QC            | G8Z1V4      | (819) 378-5444              | (819) 378-6544 |

•

Ú.

|                               | Taxi Co              | mpanies in Ca | nada with Annual Sal | es over \$500, | .000        |                |                |
|-------------------------------|----------------------|---------------|----------------------|----------------|-------------|----------------|----------------|
|                               |                      | Suite         | 1                    |                |             |                |                |
| Company Name                  | Address              | Number        | City                 | Province       | Postal Code | Phone Number   | Fax Number     |
| Taxis De Sherbrooke Inc       | 1320 Rue Belvedere S |               | Sherbrooke           | QC             | J1H4E3      | (819) 562-4717 | Not Available  |
| Taxis Quebec                  | 975 8e Av            |               | Quebec               | QC             | G1J3M8      | (418) 522-2001 | (418) 525-2004 |
| Taxi-Taxi                     | 164 Bloor St E       |               | Oshawa               | ON             | L1H3M4      | (905) 571-1234 | (905) 725-4771 |
| Taxi-Tem Inc                  | 3 Ogima St           |               | Temiscaming          | QC             | J0Z3R0      | (819) 627-3331 | Not Available  |
| Team Taxi                     | 89 Water St          |               | Summerside           | PE             | C1N1A6      | (902) 436-4555 | Not Available  |
| Teco Taxi                     | 8915 72 St           |               | Fort St. John        | BC             | V1J0B4      | (250) 787-0641 | Not Available  |
| Temiskaming Shore Taxi        | 32 Wellington St S   |               | New Liskeard         | ON             | P0J1P0      | (705) 647-7111 | Not Available  |
| Thompson Cabs Ltd             | 79 Selkirk           |               | Thompson             | MB             | R8N0L5      | (204) 677-6262 | (204) 778-3884 |
| Todays Taxi                   | 580 Steven Crt       | 3             | Newmarket            | ON             | L3Y6Z2      | (905) 201-1111 | Not Available  |
| Top Flyte Taxi                | 37 Saunders Rd       |               | Barrie               | ON             | L4N9A7      | (705) 721-8666 | Not Available  |
|                               |                      |               |                      |                |             |                |                |
| Toronto Exceptional Limo      | 136 Dundas St W      | 40            | Toronto              | ON             | M5G1C3      | (416) 532-5466 | Not Available  |
| Town Taxi                     | 31 Holland St W      |               | Bradford             | ON             | L3Z2B9      | (905) 775-5656 | (905) 775-8784 |
| Town Taxi                     | 132 Mackenzie Ave S  |               | Williams Lake        | BC             | V2G1C5      | (250) 392-4151 | (250) 392-4356 |
|                               |                      |               |                      |                |             |                |                |
| Town Taxi Chartered Vans      | 132 Mackenzie Ave S  |               | Williams Lake        | BC             | V2G1C5      | (250) 398-5888 | (250) 392-4356 |
| Trans-Cab                     | 64 Vaughan Rd        |               | Welland              | ON             | L3B5Y1      | (905) 735-4500 | (905) 788-2253 |
| Transpo Taxi                  | 28 Rue Mance         |               | Gatineau             | QC             | J8X4A3      | (819) 360-2999 | Not Available  |
|                               |                      |               |                      |                |             |                |                |
| Trinity Taxi & Livery Svc Ltd |                      | _             | Lindsay              | ON             | K9V3H5      | (705) 878-0001 | (705) 878-4044 |
| Trinity Transportation        | 1380 Wyandotte St E  |               | Windsor              | ON             | N9A3K7      | (519) 791-2801 | Not Available  |
| Trius Dieppe Taxi Ltd         | 981 Main St          | 201           | Moncton              | NB             | E1C1G9      | (506) 858-0000 | (506) 856-6383 |
| Trius Taxi                    | 212 Hodgson Rd       |               | Fredericton          | NB             | E3C2G4      | (506) 454-4444 | (506) 472-2288 |
| Tsawwassen Taxi Ltd           | 12837 76th Ave       | 203           | Surrey               | BC             | V3W2V3      | (604) 594-1111 | Not Available  |
| U-Need-A Cab Ltd              | 729 Dundas St        | _             | London               | ON             | N5W2Z5      | (519) 438-2121 | (519) 438-9544 |
| U-Need-A-Taxi                 | 11 Willow Lane       |               | Truro                | NS             | B2N1B4      | (902) 893-2899 | Not Available  |
| Union Cab                     | 456 Second Line E    |               | Sault Ste. Marie     | ON             | P6B4K1      | (705) 946-1300 | (705) 946-0343 |
| Union Cab & Limousine's       | 188 Gore St          |               | Sault Ste. Marie     | ON             | P6A1M2      | (705) 946-1300 | Not Available  |
| Union Taxi                    | 151 Ontario Ave      |               | Elliot Lake          | ON             | P5A2T2      | (705) 848-7111 | (705) 848-5959 |
| Union Taxi                    | 2 Robert St E        |               | Penetanguishene      | ON             | L9M1L6      | (705) 549-7666 | (705) 526-3372 |

γ

|                               | Taxi Con            | panies in Ca | nada with Annual Sa | les over \$500, | 000         | <del></del>    |                |
|-------------------------------|---------------------|--------------|---------------------|-----------------|-------------|----------------|----------------|
|                               |                     | Suite        |                     |                 |             |                |                |
| Company Name                  | Address             | Number       | City                | Province        | Postal Code | Phone Number   | Fax Number     |
| Union Taxi Midland            | 511 Dominion Ave    | _            | Midland             |                 | L4R1P7      | (705) 526-5666 | (705) 526-3372 |
| United Cabs                   | 3556 4th Ave        |              | Port Alberni        | BC              | V9Y4H4      | (250) 723-2121 | (250) 723-2117 |
| United Cabs Ltd               | 225 Avenue B N      |              | Saskatoon           | SK              | S7L1E1      | (306) 652-2222 | Not Available  |
| United Class Cabs             | 8222 Fraser Ave     |              | Fort Mcmurray       | AB              | T9H1W8      | (780) 743-1234 | (780) 743-3011 |
| United Taxi                   | 8507 86 St          | 303          | Fort St. John       | BC              | V1J0E4      | (250) 262-3011 | Not Available  |
| United Taxi                   | 26 King Rd          |              | Inuvik              | NT              | XOEOTO      | (867) 777-4777 | Not Available  |
| United Taxi                   | 150 Victoria St N   |              | Kitchener           | ON              | N2H5C6      | (519) 888-9999 | (519) 743-5765 |
| United Taxi                   | 116 Vidal St N      |              | Sarnia              | ON              | N7T5X5      | (519) 336-1918 | (519) 344-4221 |
| United Taxi                   | 31 1st Ave NE       |              | Swift Current       | SK              | S9H2A9      | (306) 773-2084 | (306) 773-2086 |
| University Cab Inc            | 127 Prince Edward   |              | Saint John          | NB              | E2L1W1      | (506) 631-1111 | (506) 631-0010 |
| Ut Transnational Shuttle      |                     |              |                     |                 |             |                |                |
| Svc                           | 150 Skyway Ave      |              | Etobicoke           | ON              | M9W4Y9      | (416) 213-1131 | (416) 213-1132 |
| Valley Cab                    | 1090 Highbury Rd    |              | New Minas           | NS              | B4N3P7      | (902) 681-0129 | (902) 681-0477 |
| Valley Cabs Ltd               | 75 Barbour Dr       |              | Mount Pearl         | NL              | A1N3C7      | (709) 368-6001 | Not Available  |
| Valley Taxi                   | 29 Hutson St        |              | Rothesay            | NB              | E2S1A5      | (506) 849-9000 | (506) 847-2377 |
| Van Air Taxi                  | 14 King St E        | -            | Cobourg             | ON              | K9A1K7      | (905) 373-8850 | Not Available  |
| Van Air Taxi                  | 2 Strathy Rd        |              | Cobourg             | ON              | K9A5J7      | (905) 885-5995 | (905) 373-8806 |
| Vancouver Taxi Ltd            | 790 Clark Dr        |              | Vancouver           | BC              | V5L3J2      | (604) 871-1111 | (604) 664-0412 |
| Veteran Cab                   | 350 Tuscarora St    |              | Windsor             | ON              | N9A3L7      | (519) 256-2621 | Not Available  |
| Vet's Taxi Ltd                | 17 North Market     |              | Saint John          | NB              | E2L2E5      | (506) 658-2020 | Not Available  |
| Victoria Taxi                 | 2925 Douglas St     |              | Victoria            | BC              | V8T4M8      | (250) 383-7111 | (250) 380-7111 |
| Village Taxi Of Lakefield Ltd | 711 Rye St          | 4            | Peterborough        | ON              | К9Ј6Х1      | (705) 652-8294 | (705) 652-0138 |
| Vincent's Taxi & Limousine    | 61 Lochiel St N     |              | Renfrew             | ON              | K7V1W1      | (613) 432-2921 | (613) 432-8416 |
| Waterloo Taxi Ltd             | 55 King St N        | <u> </u>     | Waterloo            | ON              | N2J2W9      | (519) 886-1200 | Not Available  |
| Wells Gray Taxi               | 228e Yellowhead Hwy |              | Clearwater          | BC              | VOE1N0      | (250) 674-1542 | Not Available  |
| Westshore Taxi Ltd            | 2513 Mill Hill Rd   |              | Victoria            | BC              | V9B4X4      | (250) 478-7888 | Not Available  |
| Westside Cabs                 | 3312 Appaloosa Rd   | 5            | Kelowna             | BC              | V1V2W5      | (250) 768-1333 | (250) 491-9820 |
| Westside Taxi Inc             | 3312 Appaloosa Rd   | 5            | Kelowna             | BC              | V1V2W5      | (250) 878-1111 | (250) 491-9820 |
| West-Way Taxi                 | 11 Bentley Ave      |              | Nepean              | ON              | K2E6T7      | (613) 727-0166 | (613) 727-8757 |

а.

| Taxi Companies in Canada with Annual Sales over \$500,000 |                     |        |                |          |             |                |                |
|---|---------------------|--------|----------------|----------|-------------|----------------|----------------|
|   |                     | Suite  |                |          |             |                |                |
| Company Name  | Address             | Number | City           | Province | Postal Code | Phone Number   | Fax Number     |
| West-Way Taxi   | 380 Hunt Club Rd    |        | Ottawa         | ON       | K1V1C1      | (613) 248-0110 | Not Available  |
| Westwind Taxi   | 2925 Douglas St     |        | Victoria       | BC       | V8T4M8      | (250) 474-4747 | (250) 380-7111 |
| Whistler Resort Cabs                                      | 2063 Lake Placid Rd | 225    | Whistler       | BC       | V0N1B2      | (604) 938-1515 | (604) 938-1518 |
| White Cab Co Ltd  | 981 Main St         | 201    | Moncton        | NB       | E1C1G9      | (506) 857-3000 | Not Available  |
| White Rock South Surrey                                   |                     |        |                |          |             |                |                |
| Taxi  | 17921 55 Ave        | 3      | Surrey         | BC       | V3S6C4      | (604) 536-7666 | Not Available  |
| Whitehorse Central Taxi                                   |                     |        |                |          |             |                |                |
| Dsptch  | 44 Macdonald Rd     |        | Whitehorse     | YT       | Y1A4L2      | (867) 393-2227 | Not Available  |
| Windsor Service Swan River                                |                     |        |                |          |             |                |                |
| Ltd   | 113 4th Ave N       |        | Swan River     | мв       | ROL1ZO      | (204) 734-9383 | (204) 734-5674 |
| Woodstock Taxi  | 603 Dundas St       |        | Woodstock      | ON       | N4S1C9      | (519) 537-5500 | Not Available  |
| Woodstock Taxi  | 79 Montclair Dr     |        | Woodstock      | ON       | N4V1C5      | (519) 539-5181 | Not Available  |
| Wright Taxi   | 1113b 4 Ave         |        | Wainwright     | AB       | T9W1H1      | (780) 842-8882 | Not Available  |
| Yarmouth Town Taxi  | 133 Argyle St       |        | Yarmouth       | NS       | B5A3X1      | (902) 742-7801 | Not Available  |
| Yellow Cab  | 7 Perth St          |        | Brockville     | ON       | K6V5B9      | (613) 345-5911 | (613) 345-4039 |
| Yellow Cab  | 10118 100 Ave       | 202    | Grande Prairie | AB       | T8V0V5      | (780) 539-3366 | (780) 831-0243 |
| Yellow Cab  | 2756 Gladstone St   |        | Halifax        | NS       | B3K4W5      | (902) 865-6500 | Not Available  |
| Yellow Cab Co   | 1 Credit Union Dr   |        | North York     | ON       | M4A2S6      | (416) 504-4141 | Not Available  |
| Yellow Cab Of Victoria                                    | 817 Fisgard St      |        | Victoria       | BC       | V8W1R9      | (250) 381-2222 | (250) 381-2227 |
| Yellow Cabs   | 10118 100th Ave     |        | Grande Prairie | AB       | T8V0V5      | (780) 831-0244 | (780) 831-0243 |
| Yellow Cabs   | 1100 Ord Rd         |        | Kamloops       | BC       | V2B7V4      | (250) 374-3333 | (250) 554-7791 |
| Yellow Cabs Ltd   | 537 115 Ave         |        | Dawson Creek   | BC       | V1G3B5      | (250) 782-9191 | Not Available  |
| Yellow Taxi   | 61 Erie St S        |        | Leamington     | ON       | N8H3A8      | (519) 326-0555 | (519) 326-4055 |
| York City Taxi Limo Inc                                   | 34 Berczy St        | B4     | Aurora         | ON       | L4G1W9      | (905) 841-9696 | Not Available  |
| Your Cab  | 1 Erncliffe St      |        | Amherst        | NS       | B4H1A7      | (902) 667-5558 | Not Available  |

5

2

| Car Sharing Companies in Canada |   |           |            |               |                  |  |  |  |  |
|---------------------------------|---|-----------|------------|---------------|------------------|--|--|--|--|
|                                 |   |           | Province / | Postal Code / |                  |  |  |  |  |
| Company                         | Address                                 | City      | State      | ZIP           | Attention        |  |  |  |  |
|                                 | BCCA Head Office                        |           |            |               | EVO Legal        |  |  |  |  |
| EVO                             | 4567 Canada Way                         | Burnaby   | BC         | V5G 4T1       | Department       |  |  |  |  |
| Zipcar                          | 129 Spadina Avenue                      | Toronto   | ON         | M5V 2L3       |                  |  |  |  |  |
| Zipcar                          | 654 Seymour Street                      | Vancouver | BC         | V6B 3K4       |                  |  |  |  |  |
| Modo                            | Suite 200 - 470 Granville Street        | Vancouver | BC         | V6C 1V5       | Calvin Stokes    |  |  |  |  |
| Car2Go                          | 1717 W. 6th Street - Suite 425          | Austin    | TX         | 78703         | Legal Department |  |  |  |  |
| Community CarShare              | 290 King St. East, Unit 203             | Kitchener | ON         | N2G 2L3       |                  |  |  |  |  |
| Community CarShare              | 175 Longwood Rd. South, Unit 304A       | Hamilton  | ON         | L8P 0A1       |                  |  |  |  |  |
| Communauto                      | 335, rue St-Joseph Est, Suite 310       | Quebec    | QC         | G1K 3B4       |                  |  |  |  |  |
| Communauto                      | 117, rue Ste-Catherine Ouest Bureau 806 | Montreal  | QC         | H3B 1H9       |                  |  |  |  |  |
| Coast Car Co-Op                 | PO Box 103                              | Gibsons   | BC         | V0N 1V0       |                  |  |  |  |  |
| CarShare Atlantic               | 2-5553 Bloomfield Street                | Halifax   | NS         | B3K 1S7       |                  |  |  |  |  |
| Saskatoon CarShare Co-Operative | 28 Campus Drive                         | Saskatoon | SK         | S7N 0X1       |                  |  |  |  |  |
| VRTUCAR                         | 322 Waverley Street                     | Ottawa    | ON         | K2P 0W3       |                  |  |  |  |  |
| Peg City Car Co-op              | 400-460 Main Street                     | Winnipeg  | MB         | R3B 1B6       |                  |  |  |  |  |
| OGO Car Share Co-Op             | 304-1353 Ellis Street                   | Kelowna   | BC         | V1Y 1Z9       |                  |  |  |  |  |
| Enterprise CarShare             | 1646 N. Milwaukee                       | Chicago   | IL         | 60647         |                  |  |  |  |  |

7

SHERIDAN CHEVROLET et al v JTEKT CORPORATION et al

rj.

٣

Court File No. CV-14-506652-00CP

| Lawyers for the Plaintiffs | SOTOS LLP<br>Barristers & Solicitors<br>180 Dundas Street West<br>Suite 1200<br>Toronto, ON M5G 1Z8<br>David Sterns LSO #36274J<br>Jean-Marc Leclerc LSO #43974F<br>Tel: (416) 977-0007<br>Fax: (416) 977-0717   | )<br>- Electric Powere<br>- Mitsubishi Electric Notice A   | 0<br>SUPERIOR C<br>Proceeding or<br>Proceeding under the  |
|----------------------------|--|--|---|
|                            | SISKINDS LLP<br>Barristers & Solicitors<br>680 Waterloo Street<br>P.O. Box 2520<br>London, ON N6A 3V8<br>Charles M. Wright LSO #36599Q<br>Linda Visser LSO #52158I<br>Tel: (519) 672-2121<br>Fax: (519) 672-6065 | ORDER<br>- Electric Powered Steering Assemblies -<br>- Mitsubishi Electric Notice Approval and Consent Certification - | ONTARIO<br>SUPERIOR COURT OF JUSTICE<br>Proceeding commenced at Toronto<br>Proceeding under the Class Proceedings Act, 1992 |